In presenting this thesis or dissertation as a partial fulfillment of the requirements for an advanced degree from Emory University, I hereby grant to Emory University and its agents the non-exclusive license to archive, make accessible, and display my thesis or dissertation in whole or in part in all forms of media, now or hereafter known, including display on the world wide web. I understand that I may select some access restrictions as part of the online submission of this thesis or dissertation. I retain all ownership rights to the copyright of the thesis or dissertation. I also retain the right to use in future works (such as articles or books) all or part of this thesis or dissertation.

Signature:

Michael John Griffin

Date

Applications of Harmonic Maass Forms

Bу

Michael John Griffin Doctor of Philosophy

Mathematics

Ken Ono Advisor

David Borthwick Committee Member

David Zureick-Brown Committee Member

Accepted:

Lisa A. Tedesco, Ph.D. Dean of the James T. Laney School of Graduate Studies

Date

Applications of Harmonic Maass Forms

By

Michael John Griffin

Advisor: Ken Ono, Ph.D.

An abstract of A dissertation submitted to the Faculty of the James T. Laney School of Graduate Studies of Emory University in partial fulfillment of the requirements for the degree of Doctor of Philosophy in Mathematics 2015

#### Abstract

### Applications of Harmonic Maass Forms By Michael John Griffin

In this thesis, we prove various results in the theory of modular forms and harmonic Maass forms, representation theory, elliptic curves and differential geometry. In particular, we give a broad framework of Rogers-Ramanujan identities and algebraic values; we prove that Ramanujan's mock theta functions satisfy his original conjectured definition; and we show that certain harmonic Maass forms which arise naturally from the arithmetic of elliptic curves encode central L-values and L-derivatives involved in the Birch and Swinnerton-Dyer conjecture. We also prove a conjecture of Moore and Witten connecting the regularized *u*-plane integral on the complex projective plane with Donaldson invariants for the SU(2)-gauge theory. In our final two applications, we turn to moonshine phenomena. Monstrous Moonshine relates the Fourier coefficients of certain modular functions to values of the irreducible characters of the Monster group-the largest of the sporadic simple groups. We give the asymptotic distribution of these character values, answering a question of Witten with applications to mathematical physics. The Umbral Moonshine conjectures relate the the values of irreducible characters of prescribed finite groups with the Fourier coefficients of distinguished mock modular forms. Gannon has proved this for the special case involving the largest sporadic simple Mathieu group. We complete the proof in the remaining cases.

Applications of Harmonic Maass Forms

By

Michael John Griffin

Advisor: Ken Ono, Ph.D.

A dissertation submitted to the Faculty of the James T. Laney School of Graduate Studies of Emory University in partial fulfillment of the requirements for the degree of Doctor of Philosophy in Mathematics 2015

# Contents

1	Introduction			
<b>2</b>	Bac	kgroun	ıd	51
	2.1	Harmo	nic Maass forms	51
	2.2	Maass-	Poincaré series	55
	2.3	Holom	orphic projection	58
3	Rog	gers–Ra	amanujan identities	61
	3.1	The H	all–Littlewood polynomials	61
	3.2	Proof	of Theorems 1.2–1.5	66
		3.2.1	The Watson–Andrews approach $\ . \ . \ . \ . \ . \ .$ .	66
		3.2.2	Proof of Theorem 1.4 $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$	70
		3.2.3	Proof of Theorem 1.2 (1.8a) $\ldots \ldots \ldots \ldots \ldots \ldots$	74
		3.2.4	Proof of Theorem 1.2 (1.8b) $\ldots \ldots \ldots \ldots \ldots \ldots$	79
		3.2.5	Proof of Theorem 1.5	81
	3.3	Proof	of Theorem 1.7	86
	3.4	Siegel	Functions	88
		3.4.1	Basic Facts about Siegel functions	89
		3.4.2	Galois theory of singular values of Siegel functions	92
	3.5	Proofs	of Theorems 1.8 and $1.15 \dots \dots \dots \dots \dots \dots \dots$	94
		3.5.1	Reformulation of the $\Phi_*(m, n; \tau)$ series	94

		3.5.2 Proofs of Theorems 1.8 and $1.15 \ldots \ldots 38$
	3.6	Examples
<b>4</b>	Rar	nanujan's mock theta functions 104
	4.1	Proof of Theorem 1.11 $\ldots \ldots 104$
	4.2	Proof of Corollary 4.1
	4.3	The $f(q)$ example
5	Wei	ierstrass mock modular forms 108
	5.1	Weierstrass Theory and the proof of Theorems 1.14, 1.15 and
		1.16
		5.1.1 Basic facts about Weierstrass theory
		5.1.2 Proofs of Theorems 1.14 and $1.15$ 109
		5.1.3 Proof of Theorem 1.16 $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$ 111
	5.2	Vector valued harmonic Maass forms
		5.2.1 A lattice related to $\Gamma_0(\mathbf{N})$
		5.2.2 The Weil representation and vector-valued automor-
		phic forms $\ldots \ldots 116$
		5.2.3 Poincaré series and Whittaker functions $\ldots \ldots \ldots \ldots 118$
		5.2.4 Twisted theta series $\ldots \ldots 119$
	5.3	Theta lifts of harmonic Maass forms
		5.3.1 Fourier expansion of the holomorphic part $\ldots \ldots \ldots 127$
		5.3.2 Lift of Poincaré series and constants
	5.4	General version of Theorem 1.17 and its proof
	5.5	Examples
6	SU(	2)-Donaldson Invariants 142
	6.1	Some relevant classical functions
	6.2	$SU(2)$ -Donaldson invariants on $\mathbb{C}P^2$
		6.2.1 The work of Göttsche and his collaborators 145

		6.2.2	The $u$ -plane integral and the work of Moore and Witten 147			
		6.2.3	Criterion for proving Theorem 1.20			
	6.3	The p	roof of Theorem 1.20 $\ldots$ 154			
		6.3.1	q-series identities			
		6.3.2	Work of Zwegers			
		6.3.3	Modularity Properties of $K_0(\tau)$			
		6.3.4	The proof of Theorem 1.20			
		6.3.5	Examples			
7	Moonshine 164					
	7.1	Vertex	x operators and the proof of classical moonshine $\ .$ 164			
	7.2	Modu	larity $\ldots \ldots 170$			
	7.3	Raden	nacher Sums $\ldots \ldots 173$			
	7.4	Quant	um Gravity			
	7.5	Moons	shine Tower			
	7.6	Monstrous Moonshine's Distributions				
		7.6.1	The modular groups in monstrous moonshine $\ldots$ 189			
		7.6.2	Exact formulas for $T_g^{(-m)}$			
		7.6.3	Exact formulas for $U_g$ up to a theta function			
		7.6.4	Proof of Theorem 7.10 $\ldots$ 195			
		7.6.5	Examples of the exact formulas			
8	Umbral Moonshine 199					
	8.1	Proof	of Theorem 1.32			
	8.2	Proof	of Theorem 1.30			
$\mathbf{A}_{]}$	ppen	dix	208			
	A.1	Monst	rous Groups			
	A.2	The U	mbral Groups			
		A.2.1	Construction			

	A.2.2	Character Tables	13
	A.2.3	Twisted Euler Characters	21
A.3	The U	mbral McKay-Thompson Series 2	29
	A.3.1	Special Functions	29
	A.3.2	Shadows	32
	A.3.3	Rademacher Sums	33
	A.3.4	Explicit Prescriptions	35
	_		
bliog	raphy	2	64

### Bibliography

## Chapter 1

### Introduction

In 1913, The mathematician G. H. Hardy received a letter from a young and unknown Indian clerk named Srinivas Ramanujan. "Dear Sir," the letter began, "I beg to introduce myself to you as a clerk in the Accounts Department of the Port Trust Office at Madras... I am now about 23 years of age... I have not trodden through the conventional regular course which is followed in a University course, but I am striking out a new path for myself. I have made a special investigation of divergent series in general and the results I get are termed by the local mathematicians as 'startling'." [25] Near the end of the letter, Ramanujan claimed that

$$\frac{1}{1 + \frac{e^{-2\pi}}{1 + \frac{e^{-4\pi}}{1 + \frac{e^{-6\pi}}{1 + \frac{e^{-6\pi}}{1 - \frac{e^{-\pi}}{1 - \frac{e^{-3\pi}}{1 - \frac{e^{-3\pi}}{$$

and that the function

$$\frac{\frac{1}{1 + \frac{e^{-\pi\sqrt{n}}}{1 + \frac{e^{-2\pi\sqrt{n}}}{1 + \frac{e^{-3\pi\sqrt{n}}}{1 + \frac{e^{-3\pi\sqrt{n}}}{1 + \frac{e^{-3\pi\sqrt{n}}}{1 + \frac{e^{-3\pi\sqrt{n}}}{1 + \frac{e^{-3\pi\sqrt{n}}}{1 + \frac{e^{-3\pi\sqrt{n}}{1 + \frac{e^{-3\pi\sqrt{n}}}}}{1 + \frac{e^{-3\pi\sqrt{n}}{1 + \frac{e^{-3\pi\sqrt{n}}{1 + \frac{e^{-3\pi\sqrt{n}}{1 + \frac{e^{-3\pi\sqrt{n}}}{1 + \frac{e^{-3\pi\sqrt{n}}}{1 + \frac{e^{-3\pi\sqrt{n}}{1 + \frac{e^{1$$

"can be exactly found if n be any positive rational."

Speaking of these particular formulas, Hardy later said "They defeated me completely. I had never seen anything in the least like this before... they could only be written down by a mathematician of the highest class. They must be true because no one would have the imagination to invent them."

The heart of Ramanujan's evaluations of 1.1 are the Rogers-Ramanujan identities which show that this is essentially a *modular function*. Modular functions and more general *modular forms* have been studied since the early nineteenth century in connection with elliptic curves. They have since been shown to connect broadly across mathematics from fields such as number theory, class field theory, and representation theory, to combinatorics, mathematical physics and more.

Many modern advances in the study of modular forms are rooted in another letter from Ramanujan to Hardy-his last one. After returning to India, Ramanujan wrote to Hardy and introduced an enigmatic collection of functions which he called the *mock theta functions*. These functions would eventually lead to the theory of *harmonic Maass forms*, special functions which generalize the notion of a modular form and provide a foundational framework which has lead to a deeper understanding of these important mathematical objects. Through the course of this thesis, we will examine several new results and applications of the mathematics rooted in Ramanujan's first and last letters.

A framework of Rogers-Ramanujan identities (Joint work with K. Ono and S. O. Warnaar)

Returning to Ramanujan's first letter, the Rogers-Ramanujan (RR) identities [263] are given by

$$G(q) := \sum_{n=0}^{\infty} \frac{q^{n^2}}{(1-q)\cdots(1-q^n)} = \prod_{n=0}^{\infty} \frac{1}{(1-q^{5n+1})(1-q^{5n+4})}$$
(1.2)

and

$$H(q) := \sum_{n=0}^{\infty} \frac{q^{n^2+n}}{(1-q)\cdots(1-q^n)} = \prod_{n=0}^{\infty} \frac{1}{(1-q^{5n+2})(1-q^{5n+3})}.$$
 (1.3)

These are essentially modular functions, and their ratio H(q)/G(q) is the Rogers–Ramanujan q-continued fraction

$$\frac{H(q)}{G(q)} = \frac{1}{1 + \frac{q}{1 + \frac{q^2}{1 + \frac{q^3}{\cdot \cdot \cdot}}}}.$$
(1.4)

Note that when  $q = e^{-\pi\sqrt{n}}$ , this is (1.1).

The golden ratio  $\phi$  satisfies  $H(1)/G(1) = 1/\phi = (-1+\sqrt{5})/2$ . Ramanujan computed further values such as

$$e^{-\frac{2\pi}{5}} \cdot \frac{H(e^{-2\pi})}{G(e^{-2\pi})} = \sqrt{\frac{5+\sqrt{5}}{2}} - \frac{\sqrt{5}+1}{2}.$$
 (1.5)

The minimal polynomial of this value is

$$x^4 + 2x^3 - 6x^2 - 2x + 1,$$

which shows that it is an algebraic integral unit. All of Ramanujan's evaluations are such units.

Ramanujan's evaluations inspired early work by Watson [234, 296, 297] and Ramanathan [257]. Then in 1996, Berndt, Chan and Zhang [24]<sup>1</sup> finally obtained general theorems concerning such values. The theory pertains to values at  $q := e^{2\pi i \tau}$ , where the  $\tau$  are quadratic irrational points in the upperhalf of the complex plane. We refer to such a point  $\tau$  as a *CM point* with

<sup>&</sup>lt;sup>1</sup>Cais and Conrad [60] and Duke [102] later revisited these results from the perspective of arithmetic geometry and the symmetries of the regular icosahedron respectively.

discriminant -D < 0, where -D is the discriminant of the minimal polynomial of  $\tau$ . The corresponding evaluation is known as a singular value. Berndt, Chan and Zhang proved that the singular values  $q^{-1/60}G(q)$  and  $q^{11/60}H(q)$ are algebraic numbers in abelian extensions of  $\mathbb{Q}(\tau)$  which enjoy the surprising property (see [24, Theorem 6.2]) that their ratio  $q^{1/5}H(q)/G(q)$  is an algebraic integral unit which generates specific abelian extensions of  $\mathbb{Q}(\tau)$ . *Remark.* The individual values of  $q^{-1/60}G(q)$  and  $q^{11/60}H(q)$  generically are

*Remark.* The individual values of  $q^{-1/60}G(q)$  and  $q^{11/60}H(q)$  generically are not algebraic integers. For example, in (1.5) we have  $\tau = i$ , and the numerator and denominator

$$q^{-\frac{1}{60}}G(q) = -\sqrt[4]{\frac{1+3\sqrt{5}-2\sqrt{10+2\sqrt{5}}}{10}} \quad \text{and} \quad q^{\frac{11}{60}}H(q) = -\sqrt[4]{\frac{1+3\sqrt{5}+2\sqrt{10+2\sqrt{5}}}{10}}$$

share the minimal polynomial  $625x^{16} - 250x^{12} - 1025x^8 - 90x^4 + 1$ .

In addition to the deep algebraic properties described above, (1.2) and (1.3) have been related to a large number of different areas of mathematics. They were were first recognized by MacMahon and Schur as identities for integer partitions [220,266], but have since been linked to algebraic geometry [59, 140], K-theory [111], conformal field theory [22, 186, 209], group theory [128], Kac-Moody, Virasoro, vertex and double affine Hecke algebras [76,120, 206, 207, 211–214], knot theory [159, 160], modular forms [38–40, 43, 46, 250], orthogonal polynomials [14,35,138], statistical mechanics [10,20], probability [129] and transcendental number theory [262].

In 1974 Andrews [7] extended (1.2) and (1.3) to an infinite family of Rogers-Ramanujan-type identities by proving that

$$\sum_{r_1 \ge \dots \ge r_m \ge 0} \frac{q^{r_1^2 + \dots + r_m^2 + r_i + \dots + r_m}}{(q)_{r_1 - r_2} \cdots (q)_{r_{m-1} - r_m}(q)_{r_m}} = \frac{(q^{2m+3}; q^{2m+3})_\infty}{(q)_\infty} \cdot \theta(q^i; q^{2m+3}),$$
(1.6)

where  $1 \le i \le m+1$ . As usual, here we have that

$$(a)_k = (a;q)_k := \begin{cases} (1-a)(1-aq)\cdots(1-aq^{k-1}) & \text{if } k \ge 0, \\ \prod_{j=0}^{\infty} (1-aq^j) & \text{if } k = \infty \end{cases}$$

and

$$\theta(a;q) := (a;q)_{\infty}(q/a;q)_{\infty}$$

is a modified theta function. The identities (1.6), which can be viewed as the analytic counterpart of Gordon's partition theorem [139], are now commonly referred to as the Andrews–Gordon (AG) identities.

Remark. The specializations of  $\theta(a; q)$  in (1.6) are (up to powers of q) modular functions, where  $q := e^{2\pi i \tau}$  and  $\tau$  is any complex point with  $\text{Im}(\tau) > 0$ . It should be noted that this differs from our use of q and  $\tau$  above where we required  $\tau$  to be a quadratic irrational point. Such infinite product modular functions were studied extensively by Klein and Siegel.

There are numerous algebraic interpretations of the Rogers-Ramanujan and Andrews-Gordon identities. For example, the above-cited papers by Milne, Lepowsky and Wilson show that they arise as principally specialized characters of integrable highest-weight modules of the affine Kac-Moody algebra  $\mathbb{A}_1^{(1)}$ . Similarly, Feigin and Frenkel proved the Rogers-Ramanujan and Andrews-Gordon identities by considering certain irreducible minimal representations of the Virasoro algebra [120]. We should also mention the much larger program by Lepowsky and others on combinatorial and algebraic extensions of Rogers-Ramanujan-type identities, leading to the introduction of Z-algebras for all affine Lie algebras, vertex-operator-theoretic proofs of the Rogers-Ramanujan identities, and Rogers-Ramanujan identities for arbitrary affine Lie algebras in which the sum side is replaced by a combinatorial sum, see e.g., [126, 205, 236] and references therein. Here we have a similar but distinct aim, namely to find a concrete framework of Rogers-Ramanujan type identities in the q-series sense of "Infinite sum = Infinite product", where the infinite products arise as specialized characters of appropriately chosen affine Lie algebras  $X_N^{(r)}$  for arbitrary N. Such a general framework would give new connections between Lie algebras and the theory of modular functions.

In [11] (see also [119, 292]) some partial results concerning the above question were obtained, resulting in Rogers-Ramanujan-type identities for  $\mathbb{A}_2^{(1)}$ . Unfortunately, the approach of [11] does not in any obvious manner extend to  $\mathbb{A}_n^{(1)}$  for all n, and this paper aims to give a more complete answer. By using a level-m Rogers-Selberg identity for the root system  $\mathbb{C}_n$  as recently obtained by Bartlett and Warnaar [19], we show that the Rogers-Ramanujan and Andrews-Gordon identities are special cases of a doubly-infinite family of q-identities arising from the Kac-Moody algebra  $\mathbb{A}_{2n}^{(2)}$  for arbitrary n. In their most compact form, the "sum-sides" are expressed in terms of Hall-Littlewood polynomials  $P_{\lambda}(x;q)$  evaluated at infinite geometric progressions (see Section 3.1 for definitions and further details), and the "product-sides" are essentially products of modular theta functions. We shall present four pairs (a, b) such that for all  $m, n \geq 1$  we have an identity of the form

$$\sum_{\substack{\lambda\\\lambda_1\leq m}} q^{a|\lambda|} P_{2\lambda}(1,q,q^2,\ldots;q^{2n+b}) = \text{``Infinite product modular function''}.$$

To make this precise, we fix notation for *integer partitions*, nonincreasing sequences of nonnegative integers with at most finitely many nonzero terms. For a partition  $\lambda = (\lambda_1, \lambda_2, ...)$ , we let  $|\lambda| := \lambda_1 + \lambda_2 + \cdots$ , and we let  $2\lambda := (2\lambda_1, 2\lambda_2, ...)$ . We also require  $\lambda'$ , the *conjugate* of  $\lambda$ , the partition which is obtained by transposing the Ferrers-Young diagram of  $\lambda$ . Finally, for convenience we let

$$\theta(a_1, \dots, a_k; q) := \theta(a_1; q) \cdots \theta(a_k; q).$$
(1.7)

*Example* 1.1. If  $\lambda = (5, 3, 3, 1)$ , then we have that  $|\lambda| = 12, 2\lambda = (10, 6, 6, 2)$  and  $\lambda' = (4, 3, 3, 1, 1)$ .

Using this notation, we have the following pair of doubly-infinite Rogers– Ramanujan type identities which correspond to specialized characters of  $\mathbb{A}_{2n}^{(2)}$ .

**Theorem 1.2** ( $\mathbb{A}_{2n}^{(2)}$  RR and AG identities). If m and n are positive integers and  $\kappa := 2m + 2n + 1$ , then we have that

$$\sum_{\substack{\lambda\\\lambda_1 \le m}} q^{|\lambda|} P_{2\lambda} \left( 1, q, q^2, \dots; q^{2n-1} \right)$$

$$= \frac{(q^{\kappa}; q^{\kappa})_{\infty}^n}{(q)_{\infty}^n} \cdot \prod_{i=1}^n \theta \left( q^{i+m}; q^{\kappa} \right) \prod_{1 \le i < j \le n} \theta \left( q^{j-i}, q^{i+j-1}; q^{\kappa} \right)$$
(1.8a)

 $= \frac{(q^{\kappa}; q^{\kappa})_{\infty}^m}{(q)_{\infty}^m} \cdot \prod_{i=1}^m \theta\left(q^{i+1}; q^{\kappa}\right) \prod_{1 \le i < j \le m} \theta\left(q^{j-i}, q^{i+j+1}; q^{\kappa}\right),$ 

and

$$\sum_{\substack{\lambda\\\lambda_{1}\leq m}} q^{2|\lambda|} P_{2\lambda} \left( 1, q, q^{2}, \dots; q^{2n-1} \right)$$

$$= \frac{(q^{\kappa}; q^{\kappa})_{\infty}^{n}}{(q)_{\infty}^{n}} \cdot \prod_{i=1}^{n} \theta \left( q^{i}; q^{\kappa} \right) \prod_{1\leq i< j\leq n} \theta \left( q^{j-i}, q^{i+j}; q^{\kappa} \right)$$

$$= \frac{(q^{\kappa}; q^{\kappa})_{\infty}^{m}}{(q)_{\infty}^{m}} \cdot \prod_{i=1}^{m} \theta \left( q^{i}; q^{\kappa} \right) \prod_{1\leq i< j\leq m} \theta \left( q^{j-i}, q^{i+j}; q^{\kappa} \right).$$

$$(1.8b)$$

Four remarks.

(1) When m = n = 1, Theorem 1.2 gives the Rogers-Ramanujan identities (1.2) and (1.3). The summation defining the series is over the empty partition,  $\lambda = 0$ , and partitions consisting of *n* copies of 1, i.e.,  $\lambda = (1^n)$ . Since

$$q^{(\sigma+1)|(1^n)|}P_{(2^n)}(1,q,q^2,\ldots;q) = \frac{q^{n(n+\sigma)}}{(1-q)\cdots(1-q^n)},$$

identities (1.2) and (1.3) thus follow from Theorem 1.2 by letting  $\sigma = 0, 1$ . (2) When n = 1, Theorem 1.2 gives the i = 1 and the i = m + 1 instances of the Andrews–Gordon identities in a representation due to Stembridge [278] (see also Fulman [128]). The equivalence with (1.6) follows from the specialization formula [219, p. 213]

$$q^{(\sigma+1)|\lambda|} P_{2\lambda}(1,q,q^2,\ldots;q) = \prod_{i\geq 1} \frac{q^{r_i(r_i+\sigma)}}{(q)_{r_i-r_{i+1}}},$$

where  $r_i := \lambda'_i$ . Note that  $\lambda_1 \leq m$  implies that  $\lambda'_i = r_i = 0$  for i > m.

(3) We note the beautiful level-rank duality exhibited by the products on the right-hand sides of the expressions in Theorem 1.2 (especially those of (1.8b)).

(4) In the next section we shall show that the more general series

$$\sum_{\substack{\lambda\\\lambda_1 \le m}} q^{(\sigma+1)|\lambda|} P_{2\lambda}(1, q, q^2, \dots; q^n)$$
(1.9)

are also expressible in terms of q-shifted factorials, allowing for a formulation of Theorem 1.2 (see Lemma 3.3) which is independent of Hall-Littlewood polynomials.

Example 1.3. Here we illustrate Theorem 1.2 when m = n = 2. Then (1.8a) is

$$\sum_{\substack{\lambda \\ \lambda_1 \le 2}} q^{|\lambda|} P_{2\lambda} (1, q, q^2, \dots; q^3) = \prod_{n=1}^{\infty} \frac{(1-q^{9n})}{(1-q^n)},$$

giving another expression for the q-series in Dyson's favorite identity, as recalled in his "A walk through Ramanujan's Garden" [112]:

"The end of the war was not in sight. In the evenings of that winter I kept sane by wandering in Ramanujan's garden. ... I found a lot of identities of the sort that Ramanujan would have enjoyed. My favorite one was this one:

$$\sum_{n=0}^{\infty} x^{n^2+n} \cdot \frac{(1+x+x^2)(1+x^2+x^4)\cdots(1+x^n+x^{2n})}{(1-x)(1-x^2)\cdots(1-x^{2n+1})} = \prod_{n=1}^{\infty} \frac{(1-x^{9n})}{(1-x^n)}.$$

In the cold dark evenings, while I was scribbling these beautiful identities amid the death and destruction of 1944, I felt close to Ramanujan. He had been scribbling even more beautiful identities amid the death and destruction of 1917."

The series in (1.8b) is

$$\sum_{\substack{\lambda\\\lambda_1\leq 2}} q^{2|\lambda|} P_{2\lambda}(1,q,q^2,\ldots;q^3) = \prod_{n=1}^{\infty} \frac{(1-q^{9n})(1-q^{9n-1})(1-q^{9n-8})}{(1-q^n)(1-q^{9n-4})(1-q^{9n-5})}.$$

We also have an even modulus analog of Theorem 1.2. Surprisingly, the a = 1 and a = 2 cases correspond to dual affine Lie algebras, namely  $\mathbb{C}_n^{(1)}$  and  $\mathbf{D}_{n+1}^{(2)}$ .

**Theorem 1.4** ( $\mathbb{C}_n^{(1)}$  RR and AG identities). If m and n are positive integers and  $\kappa := 2m + 2n + 2$ , then we have that

$$\sum_{\substack{\lambda\\\lambda_{1} \leq m}} q^{|\lambda|} P_{2\lambda} (1, q, q^{2}, \dots; q^{2n})$$

$$= \frac{(q^{2}; q^{2})_{\infty} (q^{\kappa/2}; q^{\kappa/2})_{\infty} (q^{\kappa}; q^{\kappa})_{\infty}^{n-1}}{(q)_{\infty}^{n+1}} \cdot \prod_{i=1}^{n} \theta(q^{i}; q^{\kappa/2}) \prod_{1 \leq i < j \leq n} \theta(q^{j-i}, q^{i+j}; q^{\kappa})$$

$$= \frac{(q^{\kappa}; q^{\kappa})_{\infty}^{m}}{(q)_{\infty}^{m}} \cdot \prod_{i=1}^{m} \theta(q^{i+1}; q^{\kappa}) \prod_{1 \leq i < j \leq m} \theta(q^{j-i}, q^{i+j+1}; q^{\kappa}).$$
(1.10)

**Theorem 1.5** ( $\mathbf{D}_{n+1}^{(2)}$  RR and AG identities). If m and n are positive integers such that  $n \geq 2$ , and  $\kappa := 2m + 2n$ , then we have that

$$\sum_{\substack{\lambda\\\lambda_{1} \leq m}} q^{2|\lambda|} P_{2\lambda} (1, q, q^{2}, \dots; q^{2n-2})$$

$$= \frac{(q^{\kappa}; q^{\kappa})_{\infty}^{n}}{(q^{2}; q^{2})_{\infty}(q)_{\infty}^{n-1}} \cdot \prod_{1 \leq i < j \leq n} \theta (q^{j-i}, q^{i+j-1}; q^{\kappa})$$

$$= \frac{(q^{\kappa}; q^{\kappa})_{\infty}^{m}}{(q)_{\infty}^{m}} \cdot \prod_{i=1}^{m} \theta (q^{i}; q^{\kappa}) \prod_{1 \leq i < j \leq m} \theta (q^{j-i}, q^{i+j}; q^{\kappa}).$$
(1.11)

### Two remarks.

(1) The (m, n) = (1, 2) case of (1.11) is equivalent to Milne's modulus 6 Rogers-Ramanujan identity [238, Theorem 3.26].

(2) If we take m = 1 in (1.10) (with  $n \mapsto n - 1$ ) and (1.11), and apply formula (3.7) below (with  $\delta = 0$ ), we obtain the i = 1, 2 cases of Bressoud's even modulus identities [36]

$$\sum_{r_1 \ge \dots \ge r_n \ge 0} \frac{q^{r_1^2 + \dots + r_n^2 + r_i + \dots + r_n}}{(q)_{r_1 - r_2} \cdots (q)_{r_{n-1} - r_n} (q^2; q^2)_{r_n}} = \frac{(q^{2n+2}; q^{2n+2})_{\infty}}{(q)_{\infty}} \cdot \theta(q^i; q^{2n+2}).$$
(1.12)

By combining (1.8)–(1.11), we obtain an identity of "mixed" type.

**Corollary 1.6.** If m and n are positive integers and  $\kappa := 2m + n + 2$ , then for  $\sigma = 0, 1$  we have that

$$\sum_{\substack{\lambda\\\lambda_1 \le m}} q^{(\sigma+1)|\lambda|} P_{2\lambda}(1,q,q^2,\ldots;q^n)$$
(1.13)

$$= \frac{(q^{\kappa}; q^{\kappa})_{\infty}^{m}}{(q)_{\infty}^{m}} \cdot \prod_{i=1}^{m} \theta\left(q^{i-\sigma+1}; q^{\kappa}\right) \prod_{1 \le i < j \le m} \theta\left(q^{j-i}, q^{i+j-\sigma+1}; q^{\kappa}\right)$$
(1.14)

Identities for  $\mathbb{A}_{n-1}^{(1)}$  also exist, although their formulation is perhaps slightly less satisfactory. We have the following "limiting" Rogers–Ramanujan type identities.

**Theorem 1.7** ( $\mathbb{A}_{n-1}^{(1)}$  RR and AG identities). If m and n are positive integers and  $\kappa := m + n$ , then we have that

$$\lim_{r \to \infty} q^{-m\binom{r}{2}} P_{(m^r)}(1, q, q^2, \dots; q^n) = \frac{(q^{\kappa}; q^{\kappa})_{\infty}^{n-1}}{(q)_{\infty}^n} \cdot \prod_{1 \le i < j \le n} \theta(q^{j-i}; q^{\kappa})$$
$$= \frac{(q^{\kappa}; q^{\kappa})_{\infty}^{m-1}}{(q)_{\infty}^m} \cdot \prod_{1 \le i < j \le m} \theta(q^{j-i}; q^{\kappa}).$$

Now we turn to the question of whether the new q-series appearing in these theorems, which arise so simply from the Hall–Littlewood polynomials, enjoy the same deep algebraic properties as (1.2), (1.3), and the Rogers–Ramanujan continued fraction. As it turns out they do: their singular values are algebraic numbers. Moreover, we can characterize those ratios which simplify to algebraic integral units.

To make this precise, we recall that  $q = e^{2\pi i \tau}$  for  $\text{Im}(\tau) > 0$ , and that m and n are arbitrary positive integers. The auxiliary parameter  $\kappa = \kappa_*(m, n)$  in Theorems 1.2, 1.4 and 1.5 is defined as follows:

$$\kappa = \begin{cases} \kappa_1(m,n) := 2m + 2n + 1 & \text{for } \mathbb{A}_{2n}^{(2)} \\ \kappa_2(m,n) := 2m + 2n + 2 & \text{for } \mathbb{C}_n^{(1)} \\ \kappa_3(m,n) := 2m + 2n & \text{for } \mathbf{D}_{n+1}^{(2)}. \end{cases}$$
(1.15)

*Remark.* The parameter  $\kappa$  has a representation theoretic interpretation arising from the corresponding affine Lie algebra  $X_N^{(r)}$  (see Section 3.2). It turns out that

$$\kappa_*(m,n) = \frac{2}{r} \big( \operatorname{lev}(\Lambda) + h^{\vee} \big),$$

where  $lev(\Lambda)$  is the level of the corresponding representation,  $h^{\vee}$  is the dual Coxeter number and r is the tier number.

To obtain algebraic values, we require certain normalizations of these series. The subscripts below correspond to the labelling in the theorems. In particular,  $\Phi_{1a}$  and  $\Phi_{1b}$  appear in Theorem 1.2,  $\Phi_2$  is in Theorem 1.4, and  $\Phi_3$  is in Theorem 1.5. Using this notation, the series are

$$\Phi_{1a}(m,n;\tau) := q \frac{mn(4mn-4m+2n-3)}{12\kappa} \sum_{\lambda: \lambda_1 \le m} q^{|\lambda|} P_{2\lambda}(1,q,q^2,\dots;q^{2n-1}) \quad (1.16a)$$

$$\Phi_{1b}(m,n;\tau) := q \frac{mn(4mn+2m+2n+3)}{12\kappa} \sum_{\lambda: \lambda_1 \le m} q^{2|\lambda|} P_{2\lambda}(1,q,q^2,\ldots;q^{2n-1}) \quad (1.16b)$$

$$\Phi_2(m,n;\tau) := q^{\frac{m(2n+1)(2mn-m+n-1)}{12\kappa}} \sum_{\lambda:\,\lambda_1 \le m} q^{|\lambda|} P_{2\lambda}(1,q,q^2,\ldots;q^{2n}) \quad (1.16c)$$

$$\Phi_3(m,n;\tau) := q^{\frac{m(2n-1)(2mn+n+1)}{12\kappa}} \sum_{\lambda:\,\lambda_1 \le m} q^{2|\lambda|} P_{2\lambda}(1,q,q^2,\ldots;q^{2n-2}).$$
(1.16d)

#### Two remarks.

(1) We note that  $\Phi_3(m, n; \tau)$  is not well defined when n = 1.

(2) We note that the  $\kappa_*(m, n)$  are odd in the  $\mathbb{A}_{2n}^{(2)}$  cases, and are even for the  $\mathbb{C}_n^{(1)}$  and  $\mathbf{D}_{n+1}^{(2)}$  cases. This dichotomy will be important when seeking pairs of  $\Phi_*$  whose singular values have ratios that are algebraic integral units.

Our first result concerns the algebraicity of these values and their Galois theoretic properties. We show that these values are in specific abelian extensions of imaginary quadratic fields (see [33,85] for background on the explicit class field theory of imaginary quadratic fields). For convenience, if -D < 0 is a discriminant, then we define

$$D_0 := \begin{cases} \frac{D}{4} & \text{if } D \equiv 0 \pmod{4}, \\ \frac{-D-1}{4} & \text{if } -D \equiv 1 \pmod{4} \end{cases}$$

**Theorem 1.8.** Assume the notation above, and let  $\kappa := \kappa_*(m, n)$ . If  $\kappa \tau$  is a CM point with discriminant -D < 0, then the following are true:

- 1. The singular value  $\Phi_*(m,n;\tau)$  is an algebraic number.
- 2. The multiset

$$\left\{\Phi_*(m,n,\tau_Q/\kappa)^{12\kappa}_{(\gamma\cdot\delta_Q(\tau))}: (\gamma,Q)\in W_{\kappa,\tau}\times\mathcal{Q}_D\right\}$$

(see Section 3.4 for definitions) consists of multiple copies of a Galois orbit over  $\mathbb{Q}$ .

3. If  $\kappa > 10$ ,  $|-D| > \kappa^4/2$ , and  $gcd(D_0, \kappa) = 1$ , then the multiset in (2) is a Galois orbit over  $\mathbb{Q}$ .

Four remarks.

(1) For each pair of positive integers m and n, the inequality in Theorem 1.8(3) holds for all but finitely many discriminants.

(2) In Section 3.4 we will show that the values  $\Phi_*(m, n; \tau)^{12\kappa}$  are in a distinguished class field over the ring class field  $\mathbb{Q}(j(\kappa^2 \tau))$ , where  $j(\tau)$  is the usual Klein *j*-function.

(3) The  $\Phi_*$  singular values do not in general contain full sets of Galois conjugates. In particular, the singular values in the multiset in Theorem 1.8 (2) generally require q-series which are not among the four families  $\Phi_*$ . For instance, only the i = 1 and i = m + 1 cases of the Andrews–Gordon identities arise from specializations of  $\Phi_{1a}$  and  $\Phi_{1b}$  respectively. However, the values associated to the other AG identities arise as Galois conjugates of these specializations. One then naturally wonders whether there are even further families of identities, perhaps those which can be uncovered by the theory of complex multiplication.

(4) Although Theorem 1.8 (3) indicates that the multiset in (2) is generically a single orbit of Galois conjugates, it turns out that there are indeed situations where the set is more than a single copy of such an orbit. Indeed, the two examples in Section 3.6 will be such accidents.

We now address the question of singular values and algebraic integral units. Although the singular values of  $q^{-1/60}G(q)$  and  $q^{11/60}H(q)$  are not generally algebraic integers, their denominators can be determined exactly, and their ratios always are algebraic integral units. The series  $\Phi_*$  exhibit similar behavior. The following theorem determines the integrality properties of the singular values. Moreover, it gives algebraic integral unit ratios in the case of the  $\mathbb{A}_{2n}^{(2)}$  identities, generalizing the case of the Rogers-Ramanujan continued fraction.

**Theorem 1.9.** Assume the notation and hypotheses in Theorem 1.8. Then the following are true:

- 1. The singular value  $1/\Phi_*(m,n;\tau)$  is an algebraic integer.
- 2. The singular value  $\Phi_*(m, n; \tau)$  is a unit over  $\mathbb{Z}[1/\kappa]$ .
- 3. The ratio  $\Phi_{1a}(m,n;\tau)/\Phi_{1b}(m,n;\tau)$  is an algebraic integral unit.

#### Two remarks.

(1) We have that  $\Phi_{1a}(1,1;\tau) = q^{-1/60}G(q)$  and  $\Phi_{1b}(1,1;\tau) = q^{11/60}H(q)$ . Therefore, Theorem 1.15 (3) implies the theorem of Berndt, Chan, and Zhang that the ratios of these singular values—the singular values of the Rogers–Ramanujan continued fraction—are algebraic integral units.

(2) It is natural to ask whether Theorem 1.15 (3) is a special property enjoyed only by the  $\mathbb{A}_{2n}^{(2)}$  identities. More precisely, are ratios of singular values of further pairs of  $\Phi_*$  series algebraic integral units? By Theorem 1.15 (2), it is natural to restrict attention to cases where the  $\kappa_*(m,n)$  integers agree. Indeed, in these cases the singular values are already integral over the common ring  $\mathbb{Z}[1/\kappa]$ . Due to the parity of the  $\kappa_*(m,n)$ , the only other cases to consider are pairs involving  $\Phi_2$  and  $\Phi_3$ . In Section 3.6 we give an example illustrating that such ratios for  $\Phi_2$  and  $\Phi_3$  are not generically units.

*Example* 1.10. In Section 3.6 we shall consider the q-series  $\Phi_{1a}(2,2;\tau)$  and  $\Phi_{1b}(2,2;\tau)$ . For  $\tau = i/3$ , the first 100 coefficients of the q-series respectively

give the numerical approximations

$$\Phi_{1a}(2,2;i/3) = 0.577350 \cdots \stackrel{?}{=} \frac{1}{\sqrt{3}}$$
$$\Phi_{1b}(2,2;i/3) = 0.217095 \dots$$

Here we have that  $\kappa_1(2,2) = 9$ . Indeed, these values are not algebraic integers. Respectively, they are roots of

$$3x^2 - 1$$
  
19683 $x^{18} - 80919x^{12} - 39366x^9 + 11016x^6 - 486x^3 - 1$ 

However, Theorem 1.15 (2) applies, and we find that  $\sqrt{3}\Phi_{1a}(2,2;i/3)$  and  $\sqrt{3}\Phi_{1b}(2,2;i/3)$  are units. Respectively, they are roots of

$$x - 1$$
  
$$x^{18} + 6x^{15} - 93x^{12} - 304x^9 + 420x^6 - 102x^3 + 1.$$

Lastly, Theorem 1.15(3) applies, and so their ratio

$$\frac{\Phi_{1a}(2,2;i/3)}{\Phi_{1b}(2,2;i/3)} = 4.60627\dots$$

is a unit. Indeed, it is a root of

$$x^{18} - 102x^{15} + 420x^{12} - 304x^9 - 93x^6 + 6x^3 + 1.$$

**Ramanujan's mock theta functions** (Joint work with K. Ono and L. Rolen)

Ramanujan's collaborations with Hardy were fruitful, but were cut short. After falling deathly ill, he returned to India in 1919. Near the end of his life, Ramanujan wrote a final letter [25] to Hardy which included a list of 17 enigmatic functions which he referred to *mock theta functions*. Thanks to Zwegers [311, 312], it is now known that these functions are essentially the holomorphic parts of weight 1/2 harmonic Maass forms (see section 2.1) whose nonholomorphic parts are period integrals of weight 3/2 unary theta functions. This realization has many applications (e.g. [250,306]).

Here we revisit Ramanujan's original definition from his deathbed letter [25]. After a discussion of the asymptotics of certain modular forms which are given as *Eulerian series*, he writes:

"...Suppose there is a function in the Eulerian form and suppose that all or an infinity of points  $q = e^{2i\pi m/n}$  are exponential singularities and also suppose that at these points the asymptotic form of the function closes as neatly as in the cases of (A) and (B). The question is: - is the function taken the sum of two functions one of which is an ordinary theta function and the other a (trivial) function which is O(1) at all the points  $e^{2i\pi m/n}$ ? The answer is it is not necessarily so. When it is not so I call the function Mock  $\vartheta$ -function. I have not proved rigorously that it is not necessarily so. But I have constructed a number of examples in which it is inconceivable to construct a  $\vartheta$ -function to cut out the singularities of the original function."

Remark. By ordinary theta function, Ramanujan meant a meromorphic modular form with  $k \in \frac{1}{2}\mathbb{Z}$  on some  $\Gamma_1(N)$  (see [249] for background), whose poles (if any) are supported at cusps. We refer to such forms as weakly holomorphic modular forms.

Little attention has been given to Ramanujan's original definition, prompting Berndt to remark [23] that "it has not been proved that any of Ramanujan's mock theta functions are really mock theta functions according to his definition." The following fact fills in this gap.

**Theorem 1.11.** Suppose that  $f(z) = f^{-}(z) + f^{+}(z)$  is a harmonic Maass form of weight  $k \in \frac{1}{2}\mathbb{Z}$  on  $\Gamma_1(N)$ , where  $f^{-}(z)$  (resp.  $f^{+}(z)$ ) is the nonholomorphic (resp. holomorphic) part of f(z). If  $f^{-}(z)$  is nonzero and g(z) is a weight k weakly holomorphic modular form on any  $\Gamma_1(N')$ , then  $f^{+}(z) - g(z)$  has exponential singularities as q approaches infinitely many roots of unity  $\zeta$ .

As a corollary, we obtain the following fitting conclusion to Ramanujan's enigmatic question by proving that his alleged examples indeed satisfy his original definition (Note. Throughout, we let  $q := e^{2\pi i z}$ ). More precisely, we prove the following.

**Corollary 1.12.** Suppose that M(z) is one of Ramanujan's mock theta functions, and let  $\gamma$  and  $\delta$  be integers for which  $q^{\gamma}M(\delta z)$  is the holomorphic part of a weight 1/2 harmonic Maass form. Then there does not exist a weakly holomorphic modular form g(z) of any weight  $k \in \frac{1}{2}\mathbb{Z}$  on any congruence subgroup  $\Gamma_1(N')$  such that for every root of unity  $\zeta$  we have

$$\lim_{q \to \zeta} \left( q^{\gamma} M(\delta z) - g(z) \right) = O(1).$$

*Remark.* The limits in Corollary 4.1 are *radial* limits taken from within the unit disk.

*Example* 1.13. Although a weakly holomorphic modular and a mock theta function cannot cut out each other's singularities, Ramanujan discusses a *near miss*. He considers his mock theta function

$$f(q) := 1 + \frac{q}{(1+q)^2} + \frac{q^4}{(1+q)^2(1+q^2)^2} + \dots,$$
(1.17)

and he compares it to a q-series b(q) which is essentially a weight 1/2 weakly holomorphic modular form. He then conjectures, as q approaches an even order 2k primitive root of unity  $\zeta$ , that

$$\lim_{q \to \zeta} \left( f(q) - (-1)^k b(q) \right) = O(1).$$

Watson confirmed this in [294], and Folsom, Ono, and Rhoades went further by deriving formulas for the O(1) numbers as explicit numbers in  $\mathbb{Z}[\zeta]$ . Weierstrass mock modular forms (Joint work C. Alfes, K. Ono, and L. Rolen)

The theory of *mock modular forms* provides the underlying theoretical framework for Ramanujan's enigmatic mock theta functions [31,44,309,311]. Here we consider mock modular forms and the arithmetic of elliptic curves.

There is a canonical weight 0 harmonic Maass form which arises from the analytic realization of an elliptic curve  $E/\mathbb{Q}$ . This was first observed by Guerzhoy [151, 152]. To define it we recall that  $E \cong \mathbb{C}/\Lambda_E$ , where  $\Lambda_E$  is a 2-dimensional lattice in  $\mathbb{C}$ . The parameterization of E is given by  $\mathfrak{z} \mapsto P_{\mathfrak{z}} = (\wp(\Lambda_E; \mathfrak{z}), \wp'(\Lambda_E; \mathfrak{z}))$ , where

$$\wp(\Lambda_E;\mathfrak{z}) := \frac{1}{\mathfrak{z}^2} + \sum_{w \in \Lambda_E \setminus \{0\}} \left( \frac{1}{(\mathfrak{z} - w)^2} - \frac{1}{w^2} \right)$$

is the usual Weierstrass  $\wp$ -function for  $\Lambda_E$ . Here E is given by the Weierstrass equation

$$E: y^2 = 4x^3 - 60G_4(\Lambda_E)x - 140G_6(\Lambda_E),$$

where  $G_{2k}(\Lambda_E) := \sum_{w \in \Lambda_E \setminus \{0\}} w^{-2k}$  is the classical weight 2k Eisenstein series. The canonical harmonic Maass form arises from the Weierstrass zeta-function

$$\zeta(\Lambda_E;\mathfrak{z}) := \frac{1}{\mathfrak{z}} + \sum_{w \in \Lambda_E \setminus \{0\}} \left( \frac{1}{\mathfrak{z} - w} + \frac{1}{w} + \frac{z}{w^2} \right) = \frac{1}{\mathfrak{z}} - \sum_{k=1}^{\infty} G_{2k+2}(\Lambda_E)\mathfrak{z}^{2k+1}.$$
(1.18)

This function already plays important roles in the theory of elliptic curves. The first role follows from the well-known "addition law"

$$\zeta(\Lambda_E;\mathfrak{z}_1+\mathfrak{z}_2) = \zeta(\Lambda_E;\mathfrak{z}_1) + \zeta(\Lambda_E;\mathfrak{z}_2) + \frac{1}{2} \frac{\wp'(\Lambda_E;\mathfrak{z}_1) - \wp'(\Lambda_E;\mathfrak{z}_2)}{\wp(\Lambda_E;\mathfrak{z}_1) - \wp(\Lambda_E;\mathfrak{z}_2)}, \quad (1.19)$$

which can be interpreted in terms of the "group law" of E.

To obtain the canonical forms from  $\zeta(\Lambda_E; \mathfrak{z})$ , we make use of the modularity of elliptic curves over  $\mathbb{Q}$ , which gives the modular parameterization

$$\phi_E: X_0(N_E) \to \mathbb{C}/\Lambda_E \cong E,$$

where  $N_E$  is the conductor of E. For convenience, we suppose throughout that E is a strong Weil curve. Let  $F_E(z) = \sum_{n=1}^{\infty} a_E(n)q^n \in S_2(\Gamma_0(N_E))$  be the associated newform, and let  $\mathcal{E}_E(z)$  be its *Eichler integral* 

$$\mathcal{E}_E(z) := -2\pi i \int_z^{i\infty} F_E(\tau) d\tau = \sum_{n=1}^\infty \frac{a_E(n)}{n} \cdot q^n.$$
(1.20)

Using an observation of Eisenstein, we define the function  $\mathcal{Z}_E^+(\mathfrak{z})$  by

$$\mathcal{Z}_{E}^{+}(\mathfrak{z}) := \zeta(\Lambda_{E}; \mathfrak{z}) - S(\Lambda_{E})\mathfrak{z}, \qquad (1.21)$$

where

$$S(\Lambda_E) := \lim_{s \to 0^+} \sum_{w \in \Lambda_E \setminus \{0\}} \frac{1}{w^2 |w|^{2s}}.$$
 (1.22)

We define the nonholomorphic function  $\mathcal{Z}_E(\mathfrak{z})$  by

$$\mathcal{Z}_E(\mathfrak{z}) := \mathcal{Z}_E^+(\mathfrak{z}) - \frac{\deg(\phi_E)}{4\pi ||F_E||^2} \cdot \overline{\mathfrak{z}}, \qquad (1.23)$$

where  $||F_E||$  is the Petersson norm of  $F_E$ . Finally, we define the nonholomorphic function  $\widehat{\mathfrak{Z}}_E(z)$  on  $\mathbb{H}$  by the specialization of this function at  $\mathfrak{z} = \mathcal{E}_E(z)$  given by

$$\widehat{\mathfrak{Z}}_E(z) = \widehat{\mathfrak{Z}}_E^+(z) + \widehat{\mathfrak{Z}}_E^-(z) := \mathcal{Z}_E(\mathcal{E}_E(z)).$$
(1.24)

In particular, the holomorphic part of  $\widehat{\mathfrak{Z}}_{E}(z)$  is  $\widehat{\mathfrak{Z}}_{E}^{+}(z) = \mathcal{Z}_{E}^{+}(\mathcal{E}_{E}(z)).$ 

**Theorem 1.14.** Assume the notation and hypotheses above. The following are true:

(1) The poles of  $\widehat{\mathfrak{Z}}_{E}^{+}(z)$  are precisely those points z for which  $\mathcal{E}_{E}(z) \in \Lambda_{E}$ . (2) If  $\widehat{\mathfrak{Z}}_{E}^{+}(z)$  has poles in  $\mathbb{H}$ , then there is a canonical modular function  $M_{E}(z)$  with algebraic coefficients on  $\Gamma_{0}(N_{E})$  for which  $\widehat{\mathfrak{Z}}_{E}^{+}(z) - M_{E}(z)$  is holomorphic on  $\mathbb{H}$ .

(3) We have that  $\widehat{\mathfrak{Z}}_{E}(z) - M_{E}(z)$  is a weight 0 harmonic Maass form on  $\Gamma_{0}(N_{E})$ . In particular,  $\widehat{\mathfrak{Z}}_{E}^{+}(z)$  is a weight 0 mock modular form.

*Remark.* Guerzhoy [152] has used such harmonic Maass forms in his work on the Kaneko-Zagier hypergeometric differential equation, and in [151] he studies their p-adic properties.

*Remark.* We refer to  $\widehat{\mathfrak{Z}}_{E}^{+}(z)$  as the *Weierstrass mock modular form* for *E*. It is a simple task to compute this mock modular form. Using the two Eisenstein numbers  $G_4(\Lambda_E)$  and  $G_6(\Lambda_E)$ , one then computes the remaining Eisenstein numbers using the recursion

$$G_{2n}(\Lambda_E) := \sum_{j=2}^{n-2} \frac{3(2j-1)(2n-2j-1)}{(2n+1)(2n-1)(n-3)} \cdot G_{2j}(\Lambda_E) G_{2n-2j}(\Lambda_E)$$

Armed with the Fourier expansion of  $F_E(z)$  and  $S(\Lambda_E)$ , one then simply applies (1.20)-(1.24).

Remark. The number  $\deg(\phi_E)$ , which appears in (1.23), gives information about modular form congruences. The congruence number for E is the largest integer, say  $r_E$ , with the property that there is a  $g \in S_2(\Gamma_0(N_E)) \cap \mathbb{Z}[[q]]$ , which is orthogonal to  $F_E$  with respect to the Petersson inner product, which also satisfies  $F_E \equiv g \pmod{r_E}$ . A theorem of Ribet asserts that  $\deg(\phi_E) \mid r_E$ (see Theorem 2.2 of [2]).

Many applications require the explicit Fourier expansions of harmonic Maass forms at cusps. The following theorem gives such expansions for the forms  $\widehat{\mathfrak{Z}}_E(z)$  in Theorem 1.14 at certain cusps. These expansions follow from the fact that these forms transform nicely under  $\Gamma_0^*(N_E)$ , the extension of  $\Gamma_0(N_E)$  by the Atkin-Lehner involutions. For each positive integer  $q|N_E$  we have a determinant  $q^{\alpha}$  matrix

$$W_q := \begin{pmatrix} q^{\alpha}a & b\\ N_Ec & q^{\alpha}d \end{pmatrix}, \qquad (1.25)$$

where  $q^{\alpha}||N_E$ . By Atkin-Lehner Theory, there is a  $\lambda_q \in \{\pm 1\}$  for which  $F_E|_2W_q = \lambda_q F_E$ . The following result uses these involutions to give the

Fourier expansions of  $\widehat{\mathfrak{Z}}_E(z)$  at cusps. When the level N is squarefree, the next theorem gives the expansion at all cusps of  $\Gamma_0(N)$ , which can be explicitly computed using (1.19).

**Theorem 1.15.** If  $q|N_E$ , then

$$\widehat{\mathfrak{Z}}_E(z)|_0 W_q = \mathcal{Z}_E^+(\lambda_q(\mathcal{E}_E(z) - \Omega_q(F_E))) - \frac{\deg(\phi_E)}{4\pi ||F_E||^2} \cdot \overline{\lambda_q(\mathcal{E}_E(z) - \Omega_q(F_E))},$$

where we have

$$\Omega_q(F_E) := -2\pi i \int_{W_q^{-1}i\infty}^{i\infty} F_E(z) dz.$$

Remark. In particular, we have  $\Omega_{N_E}(F_E) = L(F_E, 1)$ . By the modular parameterization, we have that  $\wp(\Lambda_E; \mathcal{E}_E(z))$  is a modular function on  $\Gamma_0(N_E)$ . We then have for each  $q|N_E$  that  $\Omega_q(F_E) \in r\Lambda_E$ , where r is a rational number. This can be seen by considering the constant term of  $\wp(\Lambda_E; \mathcal{E}_E(z))$  at cusps. The constant term of  $\wp(\Lambda_E; \mathcal{E}_E(z))$  is  $\wp(\Lambda_E; \Omega_q(F_E))$  (see Section 5.1.2 for more details). More generally, if  $N_E$  is square free, then  $\Omega_q(F_E)$  maps to a rational torsion point of E.

As these facts illustrate, the harmonic Maass form  $\widehat{\mathfrak{Z}}_{E}(z)$  and the mock modular form  $\widehat{\mathfrak{Z}}_{E}^{+}(z)$  encode the degree of the modular parameterization  $\phi_{E}$ , which in turns gives information about the congruence number  $r_{E}$ , and it encodes information about Q-rational torsion.

By the work of Bruinier, Ono, and Rhoades [50] and Candelori [61], the coefficients of  $\widehat{\mathfrak{Z}}_{E}^{+}(z)$  are  $\mathbb{Q}$ -rational when E has complex multiplication. For example, consider the elliptic curve  $E: y^{2} + y = x^{3} - 38x + 90$  of conductor 361 with CM in the field  $K = \mathbb{Q}(\sqrt{-19})$ . We find

$$F_E(z) = q - 2q^4 - q^5 + 3q^7 - 3q^9 - 5q^{11} + 4q^{16} - 7q^{17} + \dots$$

and

$$\zeta(\Lambda_E; \mathcal{E}_E(z)) = q^{-1} + \frac{1}{2}q^2 - \frac{7}{3}q^3 + \frac{12}{5}q^5 + 4q^6 - \frac{6}{7}q^7 - \frac{27}{4}q^8 - \frac{13}{3}q^9 + \frac{17}{2}q^{10} + \dots$$

As an illustration of this Q-rationality, we find that  $S(\Lambda_E) = -2$ , which in turns gives

$$\widehat{\mathfrak{Z}}_{E}^{+}(z) = q^{-1} + 2q + \frac{1}{2}q^{2} - \frac{7}{3}q^{3} - q^{4} + 2q^{5} + 4q^{6} - \frac{27}{4}q^{8} - 5q^{9} + \frac{17}{2}q^{10} + 14q^{11} - \dots$$

This power series enjoys some deep p-adic properties with respect to Hecke operators. For example, it turns out that

$$\lim_{n \to +\infty} \frac{\left[q \frac{d}{dq} \zeta(\Lambda_E; \mathcal{E}_E(z))\right] | T(5^n)}{a_E(5^n)} = -2F_E(z)$$

as a 5-adic limit. To illustrate this phenomenon we offer:

$$\frac{\left[q\frac{d}{dq}\zeta(\Lambda_E;\mathcal{E}_E(z))\right]|T(5)}{a_E(5)} + 2F_E(z) = 5q^{-5} - 20q - 85q^2 - 430q^3 - \dots \equiv 0 \pmod{5}$$

$$\frac{\left[q\frac{d}{dq}\zeta(\Lambda_E;\mathcal{E}_E(z))\right]|T(5^2)}{a_E(5^2)} + 2F_E(z) = \frac{25}{4}q^{-25} - \frac{9525}{4}q - 2031975q^2 - \dots \equiv 0 \pmod{5^2}$$

$$\frac{\left[q\frac{d}{dq}\zeta(\Lambda_E;\mathcal{E}_E(z))\right]|T(5^3)}{a_E(5^3)} + 2F_E(z) = -\frac{125}{9}q^{-125} - 89698470642375q + \dots \equiv 0 \pmod{5^3}$$

Our next result explains this phenomenon. There are such *p*-adic formulas for every *E* provided that  $p \nmid N_E$  has the property that  $p \nmid a_E(p)$  (i.e. *p* is ordinary). In analogy with recent work of Guerzhoy, Kent, and Ono [153], we obtain the following formulas.

**Theorem 1.16.** If  $p \nmid N_E$  is ordinary, then there is a constant  $\mathfrak{S}_E(p)$  for which

$$\lim_{n \to +\infty} \frac{\left[q \frac{d}{dq} \zeta(\Lambda_E; \mathcal{E}_E(z))\right] | T(p^n)}{a_E(p^n)} = \mathfrak{S}_E(p) F_E(z).$$

*Remark.* If E has CM in Theorem 1.16, then  $\mathfrak{S}_E(p) = S(\Lambda_E)$  as rational numbers. In other cases  $S(\Lambda_E)$  is expected to be transcendental, and one can interpret  $\mathfrak{S}_E(p)$  as its p-adic expansion.

The harmonic Maass forms  $\widehat{\mathfrak{Z}}_E(z)$  also encode much information about Hasse-Weil *L*-functions. The seminal works by Birch and Swinnerton-Dyer [26, 27] give an indication of this role in the case of CM elliptic curves. They obtained beautiful formulas for L(E, 1), for certain CM elliptic curves, as finite sums of numbers involving special values of  $\zeta(\Lambda_E, s)$ . Such formulas have been generalized by many authors for CM elliptic curves (for example, see the famous papers by Damerell [89,90]), and these generalizations have played a central role in the study of the arithmetic of CM elliptic curves.

Here we obtain results which show that the arithmetic of Weierstrass zeta-functions gives rise to deep information which hold for all elliptic curves  $E/\mathbb{Q}$ , not just those with CM. We prove that the canonical harmonic Maass forms  $\widehat{\mathfrak{Z}}_E(z)$  "encode" the vanishing and nonvanishing of the central values  $L(E_D, 1)$  and central derivatives  $L'(E_D, 1)$  for the quadratic twist elliptic curves  $E_D$  of all modular elliptic curves.

The connection between these values and the theory of harmonic Maass forms was first made by Bruinier and Ono [47]. Their work proved that there are weight 1/2 harmonic Maass forms whose coefficients give exact formulas for  $L(E_D, 1)$ , and which also encode the vanishing of  $L'(E_D, 1)$ . For central *L*-values their work relied on deep previous results of Shimura and Waldspurger. In the case of central derivatives, they made use of the theory of generalized Borcherds products and the Gross-Zagier Theorem. Bruinier [52] has recently refined this work by obtaining exact formulas involving periods of algebraic differentials.

The task of computing these weight 1/2 harmonic Maass forms has been nontrivial. Natural difficulties arise (see [51]). These weight 1/2 forms are preimages under  $\xi_{1/2}$  of certain weight 3/2 cusp forms, and as mentioned earlier, there are infinitely many such preimages. Secondly, the methods implemented to date for constructing such forms have relied on the theory of Poincaré series, forms whose coefficients are described as infinite sums of Kloosterman sums weighted by Bessel functions. Establishing the convergence of these expressions can already pose difficulties. Moreover, there are infinitely many linear relations among Poincaré series.

Here we circumvent these issues. We construct canonical weight 1/2 harmonic Maass forms by making use of the canonical weight 0 harmonic Maass form  $\hat{\mathfrak{Z}}_E(z)$ . More precisely, we define a twisted theta lift using the usual Siegel theta function modified by a simple polynomial. This function was studied by Hövel [166] in his Ph.D. thesis. The twisted lift  $\mathcal{I}_{\Delta,r}(\bullet; z)$  (see Section 5.3) then maps weight 0 harmonic Maass forms to weight 1/2 harmonic Maass forms. Here  $\Delta$  is a fundamental discriminant and r is an integer satisfying  $r^2 \equiv \Delta \pmod{4N_E}$ . For simplicity, we drop the dependence on  $\Delta$  and r in the introduction. The canonical weight 1/2 harmonic Maass form we define is

$$f_E(z) := \mathcal{I}\left(\widehat{\mathfrak{Z}}_E^*(z) - M_E^*(z); z\right), \qquad (1.26)$$

where  $\widehat{\mathfrak{Z}}_{E}^{*}(z)$  and  $M_{E}^{*}(z)$  denote a suitable normalization of  $\widehat{\mathfrak{Z}}_{E}(z)$  and  $M_{E}(z)$ (see Section 5.4). The normalization originates from the fact that we need the rationality of the principal part of  $f_{E}$  and we need to substract constant terms from the input. Following (2.1), we let

$$f_E(z) = f_E^+(z) + f_E^-(z) = \sum_{n \gg -\infty} c_E^+(n) q^n + \sum_{n < 0} c_E^-(n) \Gamma\left(\frac{1}{2}, 4\pi |n|y\right) q^n.$$
(1.27)

Although we treat the general case in this paper (see Theorem 5.13), to simplify exposition, in the remainder of the introduction we shall assume that  $N_E = p$  is prime, and we shall assume that the sign of the functional equation of L(E, s) is  $\epsilon(E) = -1$ . Therefore, we have that L(E, 1) = 0. The coefficients of  $f_E$  then satisfy the following theorem.

**Theorem 1.17.** Suppose that  $N_E = p$  is prime and that  $\epsilon(E) = -1$ . Then we have that  $f_E(z)$  is a weight 1/2 harmonic Maass form on  $\Gamma_0(4p)$ . Moreover, the following are true:

(1) If d < 0 is a fundamental discriminant for which \$\begin{pmatrix} d \ p \end{pmatrix}\$ = 1, then \$L(E\_d, 1) = 0\$ if and only if \$c\_E^-(d) = 0\$.</li>
(2) If d > 0 is a fundamental discriminant for which \$\begin{pmatrix} d \ p \end{pmatrix}\$ = 1, then \$L'(E\_d, 1) = 0\$ if and only if \$c\_E^+(d)\$ is in \$\mathbb{Q}\$.

*Remark.* Assume that E is as in Theorem 1.17. By work of Kolyvagin [195] and Gross and Zagier [149] on the Birch and Swinnerton-Dyer Conjecture, we then have the following for fundamental discriminants d:

- 1. If d < 0,  $\left(\frac{d}{p}\right) = 1$ , and  $c_E(d) \neq 0$ , then the rank of  $E_d(\mathbb{Q})$  is 0.
- 2. If d > 0,  $\left(\frac{d}{p}\right) = 1$ , and  $c_E^+(d)$  is transcendental, then the rank of  $E_d(\mathbb{Q})$  is 1.

Criterion (1) is analogous to Tunnell's [289] work on the *Congruent Number Problem*.

*Remark.* Theorem 1.17 follows from exact formulas. In particular, Theorem 1.17 (1) follows from the exact formula

$$L(E_d, 1) = 8\pi^2 ||F_E||^2 \cdot ||g_E||^2 \cdot \sqrt{\frac{|d|}{p}} \cdot c_E^{-}(d)^2.$$

Here  $g_E$  is the weight 3/2 cusp form which is the image of  $f_E(z)$  under the differential operator  $\xi_{\frac{1}{2}}$  (see (5.9)). More precisely, we require that  $\xi_{1/2}(f_E) = ||g_E||^{-2}g_E$  (resp.  $\xi_{1/2}(f_E) \in \mathbb{R} \cdot g_E$ ). Theorem 1.17 (2) is also related to exact formulas, ones involving periods of algebraic differentials. Recent work by Bruinier [52] establishes that

$$c_E^+(d) = \frac{\Re \int_{C_{F_E}} \zeta_d(f_E)}{\sqrt{d} \int_{C_{F_E}} \omega_{F_E}},$$

where  $\zeta_d(f_E)$  is the normalized differential of the third kind for a certain divisor associated to  $f_E$  and  $\omega_{F_E} = 2\pi i F_E(z) dz$ . Here  $C_{F_E}$  is a generator of the  $F_E$ -isotypical component of the first homology of X. The interested reader should consult [52] for further details.

Theorem 1.17 follows from a general result on the theta lift  $\mathcal{I}(\bullet, z)$  we define in Section 5.3. Earlier work of Bruinier and Funke [55], Alfes and Ehlen [6], and more recent work of Alfes [4] and of Bruinier and Ono [57], consider similar theta lifts which implement the Kudla-Millson theta function as the kernel function. Those works give lifts which map weight -2k forms to weight 3/2 + k forms when k is even. For odd k, these lifts map to weight 1/2 - k forms. The new theta lift here makes use of the usual Siegel theta kernel which is modified with a simple polynomial. Using this weight 1/2 function Hövel [166] defined a theta lift going in the direction "opposite" to ours, i.e. from forms for the symplectic group to forms for the orthogonal group.

We prove that the lift we consider maps weight 0 forms to weight 1/2 forms. Moreover, it satisfies Hecke equivariant commutative diagrams, involving  $\xi_0, \xi_{1/2}$  and the Shintani lift, of the form:

Here  $g_E$  is the weight 3/2 cusp form in Remark 8.

Remark. It turns out that the coefficients  $c_E^+(n)$  of  $f_E(\tau)$  are "twisted traces" of the singular moduli for the weight 0 harmonic Maass form  $\widehat{\mathfrak{Z}}_E^*(z) - M_E^*(z)$ . This is Theorem 5.10. This phenomenon is not new. Seminal works by Zagier [308] and Katok and Sarnak [185], followed by subsequent works by Bringmann, Bruinier, Duke, Funke, Imamoğlu, Jenkins, Miller, Pixton, and Tóth [42, 55, 56, 102–105, 237], among many others, give situations where Fourier coefficients are such traces. In particular, we obtain (vector valued versions of) the generating functions for the twisted traces of the *j*-invariant that Zagier called  $f_d$ , where *d* is a fundamental discriminant, in [308]. We explain this in more detail in Example 5.16.

Example 1.18. In Section 5.5 we shall consider the conductor 37 elliptic curve

$$E: y^2 - y = x^3 - x.$$

The sign of the functional equation of L(E, s) is -1, and  $E(\mathbb{Q})$  has rank 1.

d	$c^+(d)$	$L'(E_d, 1)$	$\operatorname{rk}(E_d(\mathbb{Q}))$
1	$-0.2817617849\dots$	$0.3059997738\ldots$	1
12	$-0.4885272382\dots$	4.2986147986	1
21	$-0.1727392572\ldots$	9.0023868003	1
28	$-0.6781939953\dots$	4.3272602496	1
33	$0.5663023201\ldots$	$3.6219567911\ldots$	1
:	÷	÷	:
1489	9	0	3
:	÷	:	•
4393	66	0	3

The table below illustrates Theorem 1.17, and its implications for ranks of elliptic curves.

For the *d* in the table we have that the sign of the functional equation of  $L(E_d, s)$  is -1. Therefore, if  $L'(E_d, 1) \neq 0$ , then we have that  $\operatorname{ord}_{s=1}(L(E_d, s)) = 1$ , which then implies that  $\operatorname{rk}(E_d(\mathbb{Q})) = 1$  by Kolyvagin's Theorem. For such *d*, Theorem 1.17 asserts that  $L'(E_d, 1) = 0$  if and only if  $c_E^+(d) \in \mathbb{Q}$ . Therefore, for these *d* the Birch and Swinnerton-Dyer Conjecture implies that  $\operatorname{rk}(E_d(\mathbb{Q})) \geq 3$  is odd if and only if  $c_E^+(d) \in \mathbb{Q}$ . We
note that for  $d \in \{1489, 4393\}$ , we find<sup>2</sup> that the curves have rank 3.

### SU(2)-Donaldson invariants (Joint work with A. Malmendier and K. Ono)

Maass forms have also found deep applications to the study of differential topology. Donaldson invariants of smooth simply connected four-dimensional manifolds [97] are diffeomorphism class invariants which play a central role in differential topology and mathematical physics. There are two families of Donaldson invariants, corresponding to the SU(2)-gauge theory and the SO(3)-gauge theory with non-trivial Stiefel-Whitney class. In each case, the invariants are graded homogeneous polynomials on the homology  $H_0(\mathbb{CP}^2) \oplus$  $H_2(\mathbb{CP}^2)$ , where  $H_i(\mathbb{CP}^2)$  is considered to have degree (4-i)/2, defined using the fundamental homology classes of the corresponding moduli spaces of antiselfdual instantons arising in gauge theory. These invariants are typically very difficult to calculate.

Here we consider the simplest manifold to which Donaldson's definition applies, the complex projective plane  $\mathbb{CP}^2$  with the Fubini-Study metric. In earlier work, Göttsche and Zagier [144] gave a formula for the Donaldson invariants of rational surfaces in terms of theta functions of indefinite lattices. As an application, Göttsche [141] derived closed expressions for the two families of the Donaldson invariants of  $\mathbb{CP}^2$  assuming the truth of the Kotschick-Morgan conjecture. Recently, Göttsche, Nakajima, Hiraku, and Yoshioka [143] have unconditionally proved these formulas.

Deep conjectures exist which relate these formulas to constructions in theoretical physics. From the viewpoint of theoretical physics [301], these two families of Donaldson invariants and the related Seiberg-Witten invariants are the correlation functions of a supersymmetric topological gauge theory with gauge group SU(2) and SO(3). Witten [300] argued that one should be

<sup>&</sup>lt;sup>2</sup>These computations were done using Sage [277] by Bruinier and Strömberg in [51]. Stephan Ehlen obtained the same numbers using our results (also using Sage).

able to compute these correlation functions in a so called *low energy effec*tive field theory. This theory has the advantage of being an abelian  $\mathcal{N}=2$ supersymmetric topological gauge theory, and the data required to define the theory only involves line bundles of even (resp. odd) first Chern class on  $\mathbb{CP}^2$  if the gauge group is SU(2) (resp. SO(3)). The vacua of the low energy effective field theory are parametrized by the *u*-plane which Seiberg and Witten [268] describe in terms of the classical modular curve  $\mathbb{H}/\Gamma_0(4)$ , together with a meromorphic one-form. Finally, Moore and Witten [241] obtained the correlation functions as regularized integrals over the u-plane, where the integrands are modular functions which are determined by the gauge group. These regularized *u*-plane integrals define a way of extracting certain contributions for each boundary component near the cusps at  $\tau = 0, 2, \infty$  of the modular curve, and Moore and Witten observed [241] that the cuspidal contributions at  $\tau = 0, 2$  vanish trivially. This vanishing corresponds to the mathematical statement that the Seiberg-Witten invariants on  $\mathbb{C}P^2$  vanish due to the presence of a Fubini-Study metric of positive scalar curvature [302].

Concerning the contribution from the cusp  $\tau = \infty$ , Moore and Witten made the following deep conjecture which relates the *u*-plane integral to Donaldson invariants.

Conjecture 1.19 (Moore and Witten [241]). The contribution at  $\tau = \infty$  to the regularized *u*-plane integral is the generating function for the Donaldson invariants of  $\mathbb{C}P^2$ .

As evidence for this conjecture, in the case of the gauge group SU(2), Moore and Witten [241] computed the first 40 invariants and found them to be in agreement with the results of Ellingsrud and Göttsche [115]. In recent work, Ono and Malmendier [227] proved this conjecture in the SO(3) case. Here we complete the proof of the conjecture by confirming the claim for the gauge group SU(2). We prove the following theorem. **Theorem 1.20.** The conjecture of Moore and Witten in the case of the SU(2)-gauge theory on  $\mathbb{CP}^2$  is true.

Moonshine (Joint work with J. F. Duncan and K. Ono)

Our final application of the theory of harmonic Maass forms is to the theory of monstrous moonshine which once again connects the theory of modular and mock modular forms with representation theory, and in recent years, with important applications to mathematical physics.

The classification of finite simple groups [13] distinguishes twenty-six examples above the others; namely, the *sporadic* simple groups, which are those that belong to none of the naturally occurring infinite families: cyclic groups, alternating groups, or finite groups of Lie type. Distinguished amongst the sporadic simple groups is the *Fischer-Griess monster*  $\mathbb{M}$ , on account of its size, which is

$$|\mathbb{M}| = 2^{46} \cdot 3^{20} \cdot 5^9 \cdot 7^6 \cdot 11^2 \cdot 13^3 \cdot 17 \cdot 19 \cdot 23 \cdot 29 \cdot 31 \cdot 41 \cdot 47 \cdot 59 \cdot 71 \quad (1.28)$$

(cf. [145]). Note that the margin is not small, for the order of the monster is

$$2^{5} \cdot 3^{7} \cdot 5^{3} \cdot 7^{4} \cdot 11 \cdot 13^{2} \cdot 29 \cdot 41 \cdot 59 \cdot 71 \tag{1.29}$$

times that of the next largest sporadic simple group, the baby monster (cf. [203]).

Fischer and Griess independently produced evidence for the monster group in 1973 (cf. [145]). Well before it was proven to exist, Tits gave a lecture on its conjectural properties at the Collège de France in 1975. In particular, he described its order (1.28). Around this time, Ogg had been considering the automorphism groups of certain algebraic curves, and had arrived at the set of primes

$$\{2, 3, 5, 7, 11, 13, 17, 19, 23, 29, 31, 41, 47, 59, 71\}$$
(1.30)

in a purely geometric way (cf. the Corollaire of [248]). Making what may now be identified as the first observation of monstrous moonshine, Ogg offered a bottle of Jack Daniels <sup>3</sup> for an explanation of this coincidence (cf. Remarque 1 of [248]).

Ogg's observation would ultimately be recognized as reflecting another respect in which the monster is distinguished amongst finite simple groups: as demonstrated by the pioneering construction of Frenkel–Lepowsky–Meurman [125–127], following the astonishing work of Griess [146, 147], the "most natural" representation of the monster, is infinite-dimensional.

The explanation of this statement takes us back to McKay's famous observation, that

$$196884 = 1 + 196883 \tag{1.31}$$

(cf. [81, 284]), and the generalizations of this observed by Thompson [284], including

$$21493760 = 1 + 196883 + 21296876,$$
  

$$864299970 = 2 \times 1 + 2 \times 196883 + 21296876 + 842609326,$$
  

$$20245856256 = 3 \times 1 + 3 \times 196883 + 21296876 + 2 \times 842609326 + 18538750076$$
  
(1.32)

Of course the left hand sides of (1.31) and (1.32) are familiar to number theorists and algebraic geometers, as coefficients in the Fourier coefficients of the *normalized elliptic modular invariant* 

$$J(\tau) := \frac{1728g_2(\tau)^3}{g_2(\tau)^3 - 27g_3(\tau)^2} - 744$$
  
=  $q^{-1} + 196884q + 21493760q^2 + 864299970q^3 + 20245856256q^4 + \dots$   
(1.33)

 $^{3}$ We refer the reader to [109] for a recent analysis of the Jack Daniels problem.

Here  $q := e^{2\pi i \tau}$ , and we set  $g_2(\tau) := 60G_4(\tau)$  and  $g_3(\tau) := 140G_6(\tau)$ , where  $G_{2k}(\tau)$  denotes the Eisenstein series of weight 2k,

$$G_{2k}(\tau) := \sum_{(m,n) \neq (0,0)} (m+n\tau)^{-2k}, \qquad (1.34)$$

for  $k \geq 2$ . The functions  $g_2$  and  $g_3$  serve to translate between the two most common parameterizations of a complex elliptic curve: as a quotient  $C/(\mathbb{Z} + \mathbb{Z}\tau)$  for  $\tau$  in the upper-half plane,  $\mathbb{H} := \{\tau \in C \mid \Im(\tau) > 0\}$ , and as the locus of a Weierstrass equation,  $y^2 = 4x^3 - g_2x - g_3$ .

The fundamental property of  $J(\tau)$ , from both the number theoretic and algebro-geometric points of view, is that it is a modular function for  $SL_2(\mathbb{Z})$ . In fact, and importantly for the monster's natural infinite-dimensional representation,  $J(\tau)$  is a generator for the field of  $SL_2(\mathbb{Z})$ -invariant holomorphic functions on  $\mathbb{H}$  that have at most exponential growth as  $\Im(\tau) \to \infty$ .

The right hand sides of (1.31) and (1.32) are familiar to finite group theorists, as simple sums of dimensions of irreducible representations of the monster M. In fact, the irreducible representations appearing in (1.31) and (1.32) are just the first five, of a total of 194, in the character table of M (cf. [80]), when ordered by size. We have that

$$\chi_1(e) = 1$$
  

$$\chi_2(e) = 196883$$
  

$$\chi_3(e) = 21296876$$
  

$$\chi_4(e) = 842609326$$
  

$$\chi_5(e) = 18538750076$$
  

$$\vdots$$
  

$$\chi_{194}(e) = 258823477531055064045234375.$$
  
(1.35)

Here e denotes the identity element of  $\mathbb{M}$ , so  $\chi_i(e)$  is just the dimension of the irreducible representation of  $\mathbb{M}$  with character  $\chi_i$ .

The coincidences (1.31) and (1.32) led Thompson to make the following conjecture [284] which realizes the natural representation of the monster alluded to above.

Conjecture 1.21 (Thompson). There is a naturally defined graded infinitedimensional monster module, denoted  $V^{\natural} = \bigoplus_{n=-1}^{\infty} V_n^{\natural}$ , which satisfies

$$\dim(V_n^{\natural}) = c(n) \tag{1.36}$$

for  $n \ge -1$  (Cf. (1.33)), such that the decompositions into irreducible representations of the monster satisfy (1.31) and (1.32) for n = 1, 2, 3 and 4 (and a similar condition for n = 5).

At the time that Thompson's conjecture was made, the monster had not yet been proven to exist, but Griess [145], and Conway–Norton [81], had independently conjectured the existence of a faithful representation of dimension 196883, and Fischer–Livingstone–Thorne had constructed the character table of M, by assuming the validity of this claim (cf. [81]) together with conjectural statements (cf. [145]) about the structure of M.

Thompson also suggested [283] to investigate the properties of the gradedtrace functions

$$T_g(\tau) := \sum_{n=-1}^{\infty} \operatorname{tr}(g|V_n^{\natural}) q^n, \qquad (1.37)$$

for  $g \in \mathbb{M}$ , now called the monstrous McKay-Thompson series, that would arise from the conjectural monster module  $V^{\natural}$ . Using the character table constructed by Fischer-Livingstone-Thorne, it was observed [81, 283] that the functions  $T_g$  are in many cases directly similar to J in the following respect: the first few coefficients of each  $T_g$  coincide with those of a generator for the function field of a discrete group<sup>4</sup>  $\Gamma_g < SL_2(\mathbb{R})$ , commensurable with

<sup>&</sup>lt;sup>4</sup>The relevant groups  $\Gamma_g$  shall be discussed in detail in Section 7.6.1.

 $SL_2(\mathbb{Z})$ , containing -I, and having width one at infinity, meaning that the subgroup of upper-triangular matrices in  $\Gamma_g$  coincides with

$$\Gamma_{\infty} := \left\{ \pm \begin{pmatrix} 1 & n \\ 0 & 1 \end{pmatrix} \mid n \in \mathbb{Z} \right\}, \tag{1.38}$$

for all  $g \in \mathbb{M}$ .

This observation was refined and developed by Conway–Norton [81], leading to their famous monstrous moonshine conjectures:.

Conjecture 1.22 (Monstrous Moonshine: Conway–Norton). For each  $g \in \mathbb{M}$ there is a specific group  $\Gamma_g < SL_2(\mathbb{R})$  such that  $T_g$  is the unique normalized principal modulus<sup>5</sup> for  $\Gamma_g$ .

This means that each  $T_g$  is the unique  $\Gamma_g$ -invariant holomorphic function on  $\mathbb{H}$  which satisfies

$$T_g(\tau) = q^{-1} + O(q), \tag{1.39}$$

as  $\Im(\tau) \to \infty$ , and remains bounded as  $\tau$  approaches any non-infinite cusp of  $\Gamma_g$ . We refer to this feature of the  $T_g$  as the *principal modulus property* of monstrous moonshine.

The hypothesis that  $T_g$  is  $\Gamma_g$ -invariant, satisfying (1.39) near the infinite cusp of  $\Gamma_g$  but having no other poles, implies that  $T_g$  generates the field of  $\Gamma_g$ -invariant holomorphic functions on  $\mathbb{H}$  that have at most exponential growth at cusps, in direct analogy with J. In particular, the natural Riemann surface structure on  $\Gamma_g \setminus \mathbb{H}$  (cf. e.g. [272]) must be that of the Riemann sphere  $\widehat{\mathbb{C}} = \mathbb{C} \cup \{\infty\}$  with finitely many points removed, and for this reason the groups  $\Gamma_g$  are said to have *genus zero*, and the principal modulus property is often referred to as the *genus zero property* of monstrous moonshine.

The reader will note the astonishing predictive power that the principal modulus property of monstrous moonshine bestows: the fact that a normalized principal modulus for a genus zero group  $\Gamma_q$  is unique, means that we

<sup>&</sup>lt;sup>5</sup>A principal modulus is also referred to as a *Hauptmodul*.

can compute the trace of an element  $g \in \mathbb{M}$ , on any homogeneous subspace of the monster's natural infinite-dimensional representation  $V^{\natural}$ , without any information about the monster, as soon as we can guess correctly the discrete group  $\Gamma_g$ . The analysis of Conway–Norton in [81] establishes very strong guidelines for the determination of  $\Gamma_g$ , and once  $\Gamma_g$  has been chosen, the "theory of replicability" (cf. [3, 81, 245]) allows for efficient computation of the coefficients of the normalized principal modulus  $T_g$ , given the knowledge of just a few of them (cf. [122], or (7.6)).

It was verified by Atkin–Fong–Smith [275], using results of Thompson [283] (cf. also [261]), that a graded (possibly virtual) infinite-dimensional monster module  $V^{\natural}$ , such that the functions  $T_g$  of (1.37) are exactly those predicted by Conway–Norton in [81], exists.

**Theorem 1.23** (Atkin–Fong–Smith). There exists a (possibly virtual) graded  $\mathbb{M}$ -module  $V^{\natural} = \bigoplus_{n=-1}^{\infty} V_n^{\natural}$  such that if  $T_g$  is defined by (1.37), then  $T_g$  is the Fourier expansion of the unique  $\Gamma_g$ -invariant holomorphic function on  $\mathbb{H}$  that satisfies  $T_g(\tau) = q^{-1} + O(q)$  as  $\tau$  approaches the infinite cusp, and has no poles at any non-infinite cusps of  $\Gamma_g$ , where  $\Gamma_g$  is the discrete subgroup of  $SL_2(\mathbb{R})$  specified by Conway–Norton in [81].

Thus Thompson's conjecture was confirmed, albeit indirectly. By this point in time, Griess, in an astonishing tour de force, had constructed the monster explicitly, by hand, by realizing it as the automorphism group of a commutative but non-associative algebra of dimension 196884 [146,147]. (See also [79, 285].) Inspired by Griess' construction, and by the representation theory of affine Lie algebras, which also involves graded infinite-dimensional vector spaces whose graded dimensions enjoy good modular properties (cf. e.g. [178–180, 184]), Frenkel–Lepowsky–Meurman established Thompson's conjecture in a strong sense.

**Theorem 1.24** (Frenkel–Lepowsky–Meurman). Thompson's Conjecture is

true. In particular, the moonshine module  $V^{\natural}$  is constructed in [125, 127].

Frenkel-Lepowsky-Meurman generalized the homogeneous realization of the basic representation of an affine Lie algebra  $\hat{\mathbf{g}}$  due, independently, to Frenkel-Kac [124] and Segal [267], in such a way that *Leech's lattice*  $\Lambda$  [201, 202]—the unique [78] even self-dual positive-definite lattice of rank 24 with no roots—could take on the role played by the root lattice of  $\mathbf{g}$  in the Lie algebra case. In particular, their construction came equipped with rich algebraic structure, furnished by vertex operators, which had appeared first in the physics literature in the late 1960's.

We refer to [124], and also the introduction to [126] for accounts of the role played by vertex operators in physics (up to 1988) along with a detailed description of their application to the representation theory of affine Lie algebras. The first application of vertex operators to affine Lie algebra representations was obtained by Lepowsky–Wilson in [210].

Borcherds described a powerful axiomatic formalism for vertex operators in [29]. In particular, he introduced the notion of a vertex algebra, which can be regarded as similar to a commutative associative algebra, except that multiplications depend upon formal variables  $z_i$ , and can be singular, in a certain formal sense, along the canonical divisors  $\{z_i = 0\}, \{z_i = z_j\}$  (cf. [32,123]). This lead eventually to Borcherds' proof of monstrous moonshine.

**Theorem 1.25** (Borcherds). Let  $V^{\natural}$  be the moonshine module vertex operator algebra constructed by Frenkel-Lepowsky-Meurman, whose automorphism group is  $\mathbb{M}$ . If  $T_g$  is defined by (1.37) for  $g \in \mathbb{M}$ , and if  $\Gamma_g$  is the discrete subgroup of  $SL_2(\mathbb{R})$  specified by Conway-Norton in [81], then  $T_g$  is the unique normalized principal modulus for  $\Gamma_g$ .

For more on vertex operator algebras and the proof of Frenkel–Lepowsky– Meurman theorem and Borcherd's theorem, see section 7.1. Witten was the first to predict a role for the monster in quantum gravity. In [304] Witten considered *pure quantum gravity* in three dimensions with negative cosmological constant, and presented evidence that the moonshine module  $V^{\natural}$  is a chiral half of the conformal field theory dual to such a quantum gravity theory, at the most negative possible value of the cosmological constant.

To explain some of the content of this statement, note that the action in pure three-dimensional quantum gravity is given explicitly by

$$I_{\rm EG} := \frac{1}{16\pi G} \int d^3x \sqrt{-g} (R - 2\Lambda), \qquad (1.40)$$

where G is the Newton or gravitational constant, R denotes the Ricci scalar, and the cosmological constant is the scalar denoted by  $\Lambda$ .

The case that the cosmological constant  $\Lambda$  is negative is distinguished, since then there exist black hole solutions to the action (1.40), as was discovered by Bañados-Teitelboim-Zanelli [18]. These black hole solutions—the *BTZ black holes*—are locally isomorphic to three-dimensional anti-de Sitter space [17], which is a Lorentzian analogue of hyperbolic space, and may be realized explicitly as the universal cover of a hyperboloid

$$-X_{-1}^2 - X_0^2 + X_1^2 + X_2^2 + X_3^3 = -\ell^2$$
(1.41)

in  $\mathbb{R}^{2,3}$  (cf. e.g. [222]). The parameter  $\ell$  in (1.41) is called the *radius of curvature*. For a locally anti-de Sitter (AdS) solution to (1.40), the radius of curvature is determined by the cosmological constant, according to

$$\ell^2 = -1/\Lambda. \tag{1.42}$$

In what has become the most cited<sup>6</sup> paper in the history of high energy physics, Maldacena opened the door on a new, and powerful approach to

<sup>&</sup>lt;sup>6</sup>Maldacena's groundbreaking paper [223] on the gauge/gravity duality has over 10,000 citations at the time of writing, according to inspirehep.net.

quantum gravity in [223], by presenting evidence for a gauge/gravity duality, in which gauge theories serve as duals to gravity theories in one dimension higher. (See [222] for a recent review.) In the simplest examples, the gauge theories are conformal field theories, and the gravity theories involve locally AdS spacetimes. The gauge/gravity duality for these cases is now known as the AdS/CFT correspondence.

Maldacena's duality furnishes a concrete realization of the *holographic* principle, introduced by 't Hooft [12], and elaborated on by Susskind [282]. For following refinements to Maldacena's proposal due to Gubser-Klebanov-Polyakov [150], and Witten [303], it is expected that gravity theories with (d + 1)-dimensional locally AdS spacetimes can be understood through the analysis of d-dimensional conformal field theories defined on the boundaries of these AdS spaces. Thus in the case of AdS solutions to three-dimensional quantum gravity, a governing role may be played by two-dimensional conformal theories, which can be accessed mathematically via vertex operator algebras (as we have mentioned previously).

The conjecture of [304] is that the two-dimensional conformal field theory corresponding to a tensor product of two copies of the moonshine module  $V^{\ddagger}$ (one "left-moving," the other "right-moving") is the holographic dual to pure three-dimensional quantum gravity with  $\ell = 16G$ , and therefore

$$\Lambda = -\frac{1}{256G^2}.\tag{1.43}$$

It is also argued that the only physically consistent values of  $\ell$  are  $\ell = 16Gm$ , for m a positive integer, so that (1.43) is the most negative possible value for  $\Lambda$ , by force of (1.42).

Shortly after this conjecture was formulated, problems with the quantum mechanical interpretation were identified by Maloney–Witten in [229]. Moreover, Gaiotto [134] and Höhn [163] cast doubt on the relevance of the monster to gravity by demonstrating that it cannot act on a holographically dual conformal field theory corresponding to  $\ell = 32G$  (i.e. m = 2), at least under the hypotheses (namely, an extremality condition, and holomorphic factorization) presented in [304].

Interestingly, the physical problems with the analysis of [304] seem to disappear in the context of *chiral three-dimensional gravity*, which was introduced and discussed in detail by Li–Song–Strominger in [215] (cf. also [228, 279]). This is the gravity theory which motivates much of the discussion in §7 of [110].

In order to define chiral three-dimensional gravity, we first describe *topologically massive gravity*, which was introduced in 1982 by Deser-Jackiw-Templeton [92, 265]. (See also [91].) The action for topologically massive gravity is given by

$$I_{\rm TMG} := I_{\rm EG} + I_{\rm CSG},\tag{1.44}$$

where  $I_{\text{EG}}$  is the Einstein-Hilbert action (cf. (1.40)) of pure quantum gravity, and  $I_{\text{CSG}}$  denotes the gravitational Chern-Simons term

$$I_{\rm CSG} := \frac{1}{32\pi G\mu} \int d^3x \sqrt{-g} \epsilon^{\lambda\mu\nu} \Gamma^{\rho}_{\lambda\sigma} \left( \partial_{\mu} \Gamma^{\sigma}_{\rho\nu} + \frac{2}{3} \Gamma^{\sigma}_{\mu\tau} \Gamma^{\tau}_{\nu\rho} \right).$$
(1.45)

The  $\Gamma^*_{**}$  are Christoffel symbols, and the parameter  $\mu$  is called the *Chern–Simons coupling constant*.

Chiral three-dimensional gravity is the special case of topologically massive gravity in which the Chern–Simons coupling constant is set to  $\mu = 1/\ell = \sqrt{-\Lambda}$ . It is shown in [215] that at this special value of  $\mu$ , the leftmoving central charges of the boundary conformal field theories vanish, and the right-moving central charges are

$$c = \frac{3\ell}{2G} = 24m,\tag{1.46}$$

for m a positive integer,  $\ell = 16Gm$ .

Much of the analysis of [304] still applies in this setting, and the natural analogue of the conjecture mentioned above states that  $V^{\ddagger}$  is holographically dual to chiral three-dimensional quantum gravity at  $\ell = 16G$ , i.e. m =1. However, as argued in detail in [228], the problem of quantizing chiral three-dimensional gravity may be regarded as equivalent to the problem of constructing a sequence of extremal chiral two-dimensional conformal field theories (i.e. vertex operator algebras), one for each central charge c = 24m, for m a positive integer. Here, a vertex operator algebra  $V = \bigoplus_n V_n$  with central charge c = 24m is called *extremal*, if its graded<sup>7</sup> dimension function satisfies

$$\sum_{n \in \mathbb{Z}} \dim(V_n) q^n = q^{-m} \frac{1}{\prod_{n>1} (1-q^n)} + O(q).$$
(1.47)

The moonshine module is the natural candidate for m = 1 (indeed, it is the only candidate if we assume the uniqueness conjecture of [126]), as the right hand side of (1.47) reduces to  $q^{-1} + O(q)$  in this case, but the analysis of [134, 163] also applies here, indicating that the monster cannot act nontrivially on any candidate<sup>8</sup> for m = 2. Thus the role of the monster in quantum gravity is still unclear, even in the more physically promising chiral gravity setting.

Nonetheless, the moonshine module  $V^{\natural}$  may still serve as the holographic dual to chiral three-dimensional quantum gravity at  $\ell = 16G$ , m = 1. In this interpretation, the graded dimension, or genus one partition function for  $V^{\natural}$ —namely, the elliptic modular invariant J—serves as the exact spectrum of physical states of chiral three-dimensional gravity at  $\mu = \sqrt{-\Lambda} = 1/16G$ , in spacetime asymptotic to the three-dimensional anti-de Sitter space (cf. (1.41)).

<sup>&</sup>lt;sup>7</sup>We regard all vertex operator algebras as graded by L(0) - c/24. Cf. (7.1).

<sup>&</sup>lt;sup>8</sup>The existence of extremal vertex operator algebras with central charge c = 24m for m > 1 remains an open question. We refer to [133, 135, 163, 305] for analyses of this problem.

Recall that if V is a representation of the Virasoro algebra  $\mathcal{V}$  (cf. (7.1)), then  $v \in V$  is called a Virasoro highest weight vector with highest weight  $h \in C$  if  $L(m)v = h\delta_{m,0}v$  whenever  $m \geq 0$ . A Virasoro descendant is a vector of the form

$$L(m_1)\cdots L(m_k)v, \tag{1.48}$$

where v is a Virasoro highest weight vector, and  $m_1 \leq \cdots \leq m_k \leq -1$ .

Assuming that  $V^{\natural}$  is dual to chiral three-dimensional gravity at m = 1, the Virasoro highest weight vectors in  $V^{\natural}$  define operators that create black holes, and the Virasoro descendants of a highest weight vector describe black holes embellished by boundary excitations. In particular, the 196883-dimensional representation of the monster which is contained in the 196884-dimensional homogenous subspace  $V_1^{\natural} < V^{\natural}$  (cf. (1.31) and (1.36)), represents an 196883dimensional space of black hole states in the chiral gravity theory.

More generally, the black hole states in the theory are classified, by the monster, into 194 different kinds, according to which monster irreducible representation they belong to.

Question 1.26. Assuming that the moonshine module  $V^{\natural}$  serves as the holographic dual to chiral three-dimensional quantum gravity at m = 1, how are the 194 different kinds of black hole states distributed amongst the homogeneous subspaces  $V_n^{\natural} < V^{\natural}$ . Are some kinds of black holes more or less common than others?

This question will be answered precisely by Corollary 1.28. Monstrous moonshine implies that the McKay–Thompson series can be written as

$$T_g(\tau) = q^{-1} + \sum_{n=1}^{\infty} \sum_{i=1}^{194} \mathbf{m}_i(n) \chi_i(g) q^n$$

where the  $\chi_i$  range over the 194 irreducible characters for the monster. In section 7.6, we find exact formulas for the multiplicities  $\mathbf{m}_i(n)$ , which lead to the following asymptotics.

**Theorem 1.27.** If  $1 \le i \le 194$ , then as  $n \to +\infty$  we have

$$\mathbf{m}_{i}(n) \sim \frac{\dim(\chi_{i})|m|^{1/4}}{\sqrt{2}|n|^{3/4}|\mathbb{M}|} \cdot e^{4\pi\sqrt{|mn|}}$$

These asymptotics immediately imply that the following limits are welldefined

$$\delta\left(\mathbf{m}_{i}\right) := \lim_{n \to +\infty} \frac{\mathbf{m}_{i}(n)}{\sum_{i=1}^{194} \mathbf{m}_{i}(n)}$$
(1.49)

Corollary 1.28. In particular, we have that

$$\delta(\mathbf{m}_i) = \frac{\dim(\chi_i)}{\sum_{j=1}^{194} \dim(\chi_j)} = \frac{\dim(\chi_i)}{5844076785304502808013602136}$$

*Remark.* Theorem 1.27 and Corollary 1.28 are the m = 1 cases of Theorem 7.10 and Corollary 7.11 respectively.

We illustrate these asymptotics explicitly, for  $\chi_1$ ,  $\chi_2$ , and  $\chi_{194}$  in Table 1.1. The precise values given in the bottom row of Table 1.1 admit the following decimal approximations:

$$\delta\left(\mathbf{m}_{1}\right) = \frac{1}{5844076785304502808013602136} \approx 1.711 \dots \times 10^{-28}$$
196883

$$\delta\left(\mathbf{m}_{2}\right) = \frac{196883}{5844076785304502808013602136} \approx 3.368\dots \times 10^{-23} \quad (1.50)$$

$$\delta\left(\mathbf{m}_{194}\right) = \frac{258823477531055064045234375}{5844076785304502808013602136} \approx 4.428\ldots \times 10^{-2}$$

Umbral Moonshine (Joint work with J. F. Duncan and K. Ono)

In 2010, Eguchi, Ooguri, and Tachikawa reignited moonshine with their observation [114] that dimensions of some representations of  $M_{24}$ , the largest sporadic simple Mathieu group (cf. e.g. [80, 83]), are multiplicities of superconformal algebra characters in the K3 elliptic genus. This observation

•

u	$\delta\left(\mathbf{m}_{1}(-1,n)\right)$	$\delta\left(\mathbf{m}_{2}(-1,n)\right)$	:	$\delta\left(\mathbf{m}_{194}(-1,n) ight)$
		0	:	0
0	0	0	:	0
<del>,</del> _	1/2	1/2	•	0
5	1/3	1/3	:	0
•••			•••	
40	$4.011\ldots  imes 10^{-4}$	$2.514\ldots  imes 10^{-3}$	•	0.00891
00	$2.699\ldots  imes 10^{-9}$	$2.732\ldots  imes 10^{-8}$	:	0.04419
80	$4.809\ldots \times 10^{-14}$	$7.537\ldots  imes 10^{-13}$	:	0.04428
100	$4.427\ldots  imes 10^{-18}$	$1.077\ldots  imes 10^{-16}$	:	0.04428
120	$1.377 \ldots \times 10^{-21}$	$5.501\ldots  imes 10^{-20}$	:	0.04428
140	$1.156\ldots  imes 10^{-24}$	$1.260\ldots  imes 10^{-22}$	:	0.04428
160	$2.621\ldots  imes 10^{-27}$	$3.443  imes 10^{-23}$	:	0.04428
180	$1.877 \dots \times 10^{-28}$	$3.371\ldots  imes 10^{-23}$	:	0.04428
200	$1.715  imes 10^{-28}$	$3.369\ldots  imes 10^{-23}$	•	0.04428
220	$1.711\ldots  imes 10^{-28}$	$3.368\ldots  imes 10^{-23}$	:	0.04428
240	$1.711\ldots  imes 10^{-28}$	$3.368\ldots  imes 10^{-23}$	:	0.04428
•••				
8	$\frac{1}{5844076785304502808013602136}$	$\frac{196883}{5844076785304502808013602136}$	•	$\frac{258823477531055064045234375}{5844076785304502808013602136}$

Table 1.1

suggested a manifestation of moonshine for  $M_{24}$ : Namely, there should be an infinite-dimensional graded  $M_{24}$ -module whose McKay-Thompson series are holomorphic parts of *harmonic Maass forms*, or mock modular forms.

Following the work of Cheng [70], Eguchi and Hikami [113], and Gaberdiel, Hohenegger, and Volpato [132,217], Gannon established the existence of this infinite-dimensional graded  $M_{24}$ -module in [136].

It is natural to seek a general mathematical and physical setting for these results. Here we consider the mathematical setting, which develops from the close relationship between the monster group  $\mathbb{M}$  and the Leech lattice  $\Lambda_{24}$ . Recall (cf. e.g. [83]) that the Leech lattice is even, unimodular, and positivedefinite of rank 24. It turns out that  $M_{24}$  is closely related to another such lattice. Such observations led Cheng, Duncan and Harvey to further instances of moonshine within the setting of even unimodular positive-definite lattices of rank 24. In this way they arrived at the *Umbral Moonshine Conjectures* (cf. §5 of [75], §6 of [68], and §2 of [69]), predicting the existence of 22 further, graded infinite-dimensional modules, relating certain finite groups to distinguished mock modular forms.

To explain this prediction in more detail we recall Niemeier's result [244] that there are 24 (up to isomorphism) even unimodular positive-definite lattices of rank 24. The Leech lattice is the unique one with no root vectors (i.e. lattice vectors with norm-square 2), while the other 23 have root systems with full rank, 24. These *Niemeier root systems* are unions of simple simply-laced root systems with the same Coxeter numbers, and are given explicitly as

$$A_{1}^{24}, A_{2}^{12}, A_{3}^{8}, A_{4}^{6}, A_{6}^{2}, A_{12}^{2}, A_{5}^{4}D_{4}, A_{7}^{2}D_{5}^{2}, A_{8}^{3}, A_{9}^{2}D_{6}, A_{11}D_{7}E_{6}, A_{15}D_{9}, A_{17}E_{7}, A_{24},$$
(1.51)  
$$D_{4}^{6}, D_{6}^{4}, D_{8}^{3}, D_{10}E_{7}^{2}, D_{12}^{2}, D_{16}E_{8}, D_{24}, E_{6}^{4}, E_{8}^{3},$$

in terms of the standard ADE notation. (Cf. e.g. [83] or [167] for more on

root systems.)

For each Niemeier root system X let  $N^X$  denote the corresponding unimodular lattice, let  $W^X$  denote the (normal) subgroup of  $\operatorname{Aut}(N^X)$  generated by reflections in roots, and define the *umbral group* of X by setting

$$G^X := \operatorname{Aut}(N^X)/W^X. \tag{1.52}$$

(See §A.2.1 for explicit descriptions of the groups  $G^X$ .)

Let  $m^X$  denote the Coxeter number of any simple component of X. An association of distinguished  $2m^X$ -vector-valued mock modular forms  $H_g^X(\tau) = (H_{g,r}^X(\tau))$ —called *umbral McKay-Thompson series*—to elements  $g \in G^X$  is described and analyzed in [68,69,75].

For  $X = A_1^{24}$  we have  $G^X \simeq M_{24}$  and  $m^X = 2$ , and the functions  $H_{g,1}^X(\tau)$ are precisely the mock modular forms assigned to elements  $g \in M_{24}$  in the works [70, 113, 132, 217] mentioned above. Generalizing the  $M_{24}$  moonshine initiated by Eguchi, Ooguri and Tachikawa, we have the following conjecture of Cheng, Duncan and Harvey (cf. §2 of [69] or §9.3 of [106]).

Conjecture 1.29 (Umbral Moonshine Modules). Let X be a Niemeier root system X and set  $m := m^X$ . There is a naturally defined bi-graded infinite-dimensional  $G^X$ -module

$$\check{K}^X = \bigoplus_{r \in I^X} \bigoplus_{\substack{D \in \mathbb{Z}, \ D \le 0, \\ D = r^2 \pmod{4m}}} \check{K}^X_{r, -D/4m}$$
(1.53)

such that the vector-valued mock modular form  $H_g^X = (H_{g,r}^X)$  is related<sup>9</sup> to

<sup>&</sup>lt;sup>9</sup>In the statement of Conjecture 6.1 of [68] the function  $H_{g,r}^X$  in (1.54) is replaced with  $3H_{g,r}^X$  in the case that  $X = A_8^3$ . This is now known to be an error, arising from a misspecification of some of the functions  $H_g^X$  for  $X = A_8^3$ . Our treatment of the case  $X = A_8^3$  in this work reflects the corrected specification of the corresponding  $H_g^X$  which is described and discussed in detail in [69].

the graded trace of g on  $\check{K}^X$  by

$$H_{g,r}^{X}(\tau) = -2q^{-1/4m}\delta_{r,1} + \sum_{\substack{D \in \mathbb{Z}, \ D \le 0, \\ D = r^{2}(\text{mod }4m)}} \operatorname{tr}(g|\check{K}_{r,-D/4m}^{X})q^{-D/4m}$$
(1.54)

for  $r \in I^X$ .

In (1.53) and (1.54) the set  $I^X \subset \mathbb{Z}/2m\mathbb{Z}$  is defined in the following way. If X has an A-type component then  $I^X := \{1, 2, 3, \ldots, m-1\}$ . If X has no A-type component but does have a D-type component then  $m = 2 \mod 4$ , and  $I^X := \{1, 3, 5, \ldots, m/2\}$ . The remaining cases are  $X = E_6^4$  and  $X = E_8^3$ . In the former of these,  $I^X := \{1, 4, 5\}$ , and in the latter case  $I^X := \{1, 7\}$ .

The functions  $H_q^X(\tau)$  are described explicitly in §A.3.4.

Here we prove the following theorem.

### **Theorem 1.30.** The umbral moonshine modules exist.

Two remarks.

1) Theorem 1.30 for  $X = A_1^{24}$  is the main result of Gannon's work [136].

2) The vector-valued mock modular forms  $H^X = (H_{g,r}^X)$  have "minimal" *principal parts*. This minimality is analogous to the fact that the original McKay-Thompson series  $T_g(\tau)$  for the Monster are hauptmoduln, and plays an important role in our proof.

Example 1.31. Many of Ramanujan's mock theta functions [258] are components of the vector-valued umbral McKay-Thompson series  $H_g^X = (H_{g,r}^X)$ . For example, consider the root system  $X = A_2^{12}$ , whose umbral group is a double cover  $2.M_{12}$  of the sporadic simple Mathieu group  $M_{12}$ . In terms of Ramanujan's 3rd order mock theta functions

$$f(q) = 1 + \sum_{n=1}^{\infty} \frac{q^{n^2}}{(1+q)^2(1+q^2)^2 \cdots (1+q^n)^2},$$
  

$$\phi(q) = 1 + \sum_{n=1}^{\infty} \frac{q^{n^2}}{(1+q^2)(1+q^4) \cdots (1+q^{2n})},$$
  

$$\chi(q) = 1 + \sum_{n=1}^{\infty} \frac{q^{n^2}}{(1-q+q^2)(1-q^2+q^4) \cdots (1-q^n+q^{2n})},$$
  

$$\omega(q) = \sum_{n=0}^{\infty} \frac{q^{2n(n+1)}}{(1-q)^2(1-q^3)^2 \cdots (1-q^{2n+1})^2},$$
  

$$\rho(q) = \sum_{n=0}^{\infty} \frac{q^{2n(n+1)}}{(1+q+q^2)(1+q^3+q^6) \cdots (1+q^{2n+1}+q^{4n+2})},$$

we have that

$$\begin{split} H^X_{2B,1}(\tau) &= H^X_{2C,1}(\tau) = H^X_{4C,1}(\tau) = -2q^{-\frac{1}{12}} \cdot f(q^2), \\ H^X_{6C,1}(\tau) &= H^X_{6D,1}(\tau) = -2q^{-\frac{1}{12}} \cdot \chi(q^2), \\ H^X_{8C,1}(\tau) &= H^X_{8D,1}(\tau) = -2q^{-\frac{1}{12}} \cdot \phi(-q^2), \\ H^X_{2B,2}(\tau) &= -H^X_{2C,2}(\tau) = -4q^{\frac{2}{3}} \cdot \omega(-q), \\ H^X_{6C,2}(\tau) &= -H^X_{6D,2}(\tau) = 2q^{\frac{2}{3}} \cdot \rho(-q). \end{split}$$

See §5.4 of [68] for more coincidences between umbral McKay-Thompson series and mock theta functions identified by Ramanujan almost a hundred years ago.

Our proof of Theorem 1.30 involves the explicit determination of each  $G^X$ module  $\check{K}^X$  by computing the multiplicity of each irreducible component for each homogeneous subspace. It guarantees the existence and uniqueness of a  $\check{K}^X$  which is compatible with the representation theory of  $G^X$  and the Fourier expansions of the vector-valued mock modular forms  $H_g^X(\tau) = (H_{g,r}^X(\tau))$ .

At first glance our methods do not appear to shed light on any deeper algebraic properties of the  $\check{K}^X$ , such as might correspond to the vertex operator algebra structure on  $V^{\natural}$ , or the monster Lie algebra introduced by Borcherds in [30]. However, we do determine, and utilize, specific recursion relations for the coefficients of the umbral McKay-Thompson series which are analogous to the replicability properties of monstrous moonshine formulated by Conway and Norton in §8 of [81] (cf. also [3]). More specifically, we use recent work [168] of Imamoğlu, Raum and Richter, as generalized [235] by Mertens, to obtain such recursions. These results are based on the process of *holomorphic projection*.

**Theorem 1.32.** For each  $g \in G^X$  and 0 < r < m, the McKay-Thompson series  $H_{g,r}^X(\tau)$  is replicable in the mock modular sense.

A key step in Borcherds' proof [30] of the monstrous moonshine conjecture is the reformulation of replicability in Lie theoretic terms. We may speculate that the *mock modular replicability* utilized in this work will ultimately admit an analogous algebraic interpretation. Such a result remains an important goal for future work.

In the statement of Theorem 1.32, replicable means that there are explicit recursion relations for the coefficients of the vector-valued mock modular form in question. For example, we recall the recurrence formula for Ramanujan's third order mock theta function  $f(q) = \sum_{n=0}^{\infty} c_f(n)q^n$  that was obtained recently by Imamoğlu, Raum and Richter [168]. If  $n \in \mathbb{Q}$ , then let

$$\sigma_1(n) := \begin{cases} \sum_{d|n} d & \text{if } n \in \mathbb{Z}, \\ 0 & \text{otherwise} \end{cases}$$

$$\operatorname{sgn}^+(n) := \begin{cases} \operatorname{sgn}(n) & \text{if } n \neq 0, \\ 1 & \text{if } n = 0, \end{cases}$$

and then define

$$d(N, \widetilde{N}, t, \widetilde{t}) := \operatorname{sgn}^+(N) \cdot \operatorname{sgn}^+(\widetilde{N}) \cdot \left( |N + t| - |\widetilde{N} + \widetilde{t}| \right).$$

Then for positive integers n, we have that

$$\sum_{\substack{m \in \mathbb{Z} \\ 3m^2 + m \le 2n}} \left( m + \frac{1}{6} \right) c_f \left( n - \frac{3}{2}m^2 - \frac{1}{2}m \right)$$
$$= \frac{4}{3}\sigma(n) - \frac{16}{3}\sigma\left(\frac{n}{2}\right) - 2\sum_{\substack{a,b \in \mathbb{Z} \\ 2n = ab}} d\left( N, \widetilde{N}, \frac{1}{6}, \frac{1}{6} \right),$$

where  $N := \frac{1}{6}(-3a+b-1)$  and  $\widetilde{N} := \frac{1}{6}(3a+b-1)$ , and the sum is over integers a, b for which  $N, \widetilde{N} \in \mathbb{Z}$ . This is easily seen to be a recurrence relation for the coefficients  $c_f(n)$ . The replicability formulas for all of the  $H_{g,r}^X(\tau)$  are similar (although some of these relations are slightly more complicated and involve the coefficients of weight 2 cusp forms).

It is important to emphasize that, despite the progress which is represented by our main results, Theorems 1.30 and 1.32, the following important question remains open in general.

Question 1.33. Is there a "natural" construction of  $\check{K}^X$ ? Is  $\check{K}^X$  equipped with a deeper algebra structure as in the case of the monster module  $V^{\natural}$  of Frenkel, Lepowsky and Meurman?

We remark that this question has been answered positively, recently, in one special case: A vertex operator algebra structure underlying the umbral moonshine module  $\check{K}^X$  for  $X = E_8^3$  has been described explicitly in [107]. See also [74,108], where the problem of constructing algebraic structures that illuminate the umbral moonshine observations is addressed from a different point of view.

The proof of Theorem 1.30 is not difficult. It is essentially a collection of tedious calculations. We use the theory of mock modular forms and the character table for each  $G^X$  (cf. §A.2.2) to solve for the multiplicities of the irreducible  $G^X$ -module constituents of each homogeneous subspace in the alleged  $G^X$ -module  $\check{K}^X$ . To prove Theorem 1.30 it suffices to prove that these multiplicities are non-negative integers. To prove Theorem 1.32 we apply recent work [235] of Mertens on the holomorphic projection of weight  $\frac{1}{2}$  mock modular forms, which generalizes earlier work [168] of Imamoğlu, Raum and Richter.

In §1.24 we recall the facts about mock modular forms that we require, and we prove Theorem 1.32. We prove Theorem 1.30 in §8.2. The appendices furnish all the data that our method requires. In particular, the umbral groups  $G^X$  are described in detail in §A.2, and the umbral McKay-Thompson series  $H_g^X(\tau)$  are given explicitly in §A.3.

# Chapter 2

## Background

## 2.1 Harmonic Maass forms

We begin by briefly recalling the definition of a harmonic Maass form of weight  $k \in \frac{1}{2}\mathbb{Z}$  and multiplier  $\nu$  (a generalization of the notion of a Nebentypus). If  $\tau = x + iy$  with x and y real, we define the weight k hyperbolic Laplacian by

$$\Delta_k := -y^2 \left( \frac{\partial^2}{\partial x^2} + \frac{\partial^2}{\partial y^2} \right) + iky \left( \frac{\partial}{\partial x} + i \frac{\partial}{\partial y} \right), \qquad (2.1)$$

and if  $\gamma = \begin{pmatrix} a & b \\ c & d \end{pmatrix} \in SL_2(\mathbb{Z})$ , define

$$(\gamma:\tau) := (c\tau + d).$$

Suppose  $\Gamma$  is a subgroup of finite index in  $SL_2(\mathbb{Z})$  and  $\frac{3}{2} \leq k \in \frac{1}{2}\mathbb{Z}$ . Then a real analytic function  $F(\tau)$  is a harmonic Maass form of weight k on  $\Gamma$  with multiplier  $\nu$  if:

(a) The function  $F(\tau)$  satisfies the modular transformation with respect to the weight k slash operation,

$$F(\tau)|_k \gamma := (\gamma : \tau)^{-k} F(\gamma \tau) = \nu(\gamma) F(\tau)$$

for every matrix  $\gamma \in \Gamma$ , where if  $k \in \mathbb{Z} + \frac{1}{2}$ , the square root is taken to be the principal branch. In particular, if  $\nu$  is trivial, then F is invariant under the action of the slash operator.

- (b) We have that  $\Delta_k F(\tau) = 0$ ,
- (c) The singularities of F (if any) are supported at the cusps of  $\Gamma$ , and for each cusp  $\rho$  there is a polynomial  $P_{F,\rho}(q^{-1}) \in \mathbb{C}[q^{-1/t_{\rho}}]$  and a constant c > 0 such that  $F(\tau) - P_{F,\rho}(e^{-2\pi i \tau}) = O(e^{-cy})$  as  $\tau \to \rho$  from inside a fundamental domain. Here  $t_{\rho}$  is the width of the cusp  $\rho$ . If  $\rho$  is not specified, we assume  $\rho = \infty$ .

Remark. The polynomial  $P_{F,\rho}$  above is referred to as the principal part of F at  $\rho$ . In certain applications, condition (c) of the definition may be relaxed to admit larger classes of harmonic Maass forms. However, for our purposes we will only be interested in those satisfying the given definition, having a holomorphic principal part.

We denote the complex vector space of such functions by  $H_k(\Gamma, \nu)$ , and note that in order for  $H_k(\Gamma, \nu)$  to be nonzero,  $\nu$  must satisfy

$$(\gamma:\delta\tau)^k(\delta:\tau)^k\nu(\gamma)\nu(\delta) = (\gamma\delta:\tau)^k\nu(\gamma\delta)$$

for every  $\gamma, \delta \in \Gamma$ .

Let  $\mathcal{S}(\Gamma)$  denote some fixed complete set of inequivalent representatives of the cusps of  $\Gamma$ . For each representative  $\rho = \frac{\alpha}{\gamma}$  with  $(\alpha, \gamma) = 1$ , fix a matrix

$$L_{\rho} = \begin{pmatrix} -\delta & \beta \\ \gamma & -\alpha \end{pmatrix} \in SL_2(\mathbb{Z})$$

so that  $\rho = L_{\rho}^{-1}\infty$ . Following Rankin [259], let  $t_{\rho}$  be the cusp width and let  $\kappa_{\rho}$  be the cusp parameter, defined as the least nonnegative integer so that  $\nu(L_{\rho}T^{t_{\rho}}L_{\rho}^{-1}) = e^{2\pi i\kappa_{\rho}}$ , where  $T := \begin{pmatrix} 1 & 1 \\ 0 & 1 \end{pmatrix}$ . The stabilizer of  $\rho$  in  $\Gamma$  is given

by  $\Gamma_{\rho} := \langle \pm T^{t_{\rho}} \rangle$ , so for example  $\Gamma_{\infty} = \langle \pm T \rangle$ . Given  $F(\tau) \in H_{2-k}(\Gamma, \nu)$ , we refer to  $F_{\rho}(\tau) := F(\tau)|_{2-k}L_{\rho}$  as the expansion of F at the cusp  $\rho$ . We note that this expansion depends on the choice of  $L_{\rho}$ . These facts imply that the expansion of  $F_{\rho}$  can be given as a Fourier series of the form

$$F_{\rho}(\tau) = \sum_{n} a(n, y) e^{2\pi i x(n+\kappa_{\rho})/t_{\rho}}$$

More precisely, we have the following. The Fourier expansion of harmonic Maass forms F at a cusp  $\rho$  (see Proposition 3.2 of [54]) splits into two components. As before, we let  $q := e^{2\pi i \tau}$ .

**Lemma 2.1.** If  $F(\tau)$  is a harmonic Maass form of weight 2 - k for  $\Gamma$  where  $\frac{3}{2} \leq k \in \frac{1}{2}\mathbb{Z}$ , and if  $\rho$  is a cusp of  $\Gamma$ , then

$$F_{\rho}(\tau) = F_{\rho}^{+}(\tau) + F_{\rho}^{-}(\tau)$$

where  $F_{\rho}^{+}$  is the holomorphic part of  $F_{\rho}$ , given by

$$F_{\rho}^{+}(\tau) := \sum_{n \gg -\infty} c_{F,\rho}^{+}(n) q^{(n+\kappa_{\rho})/t_{\rho}},$$

and  $F_{\rho}^{-}$  is the nonholomorphic part, given by

$$F_{\rho}^{-}(\tau) + \sum_{n < 0} c_{F\rho}^{-}(n) \Gamma(k-1, 4\pi y | (n+\kappa_{\rho})/t_{\rho} |) q^{(n+\kappa_{\rho})/t_{\rho}}$$

The holomorphic part of a harmonic Maass form is referred to as a *mock modular form*. By inspection, we see that weakly holomorphic modular forms are themselves harmonic Maass forms. In fact, under the given definition, all harmonic Maass forms of positive weight are weakly holomorphic.

The  $\xi$ -operator is a differential operator on harmonic Maass forms which is useful for understanding the nonholomorphic part. It is defined as  $\xi_k := 2iy^k \cdot \overline{\frac{\partial}{\partial z}}$ .

Zagier calls the image under  $\xi$  of f the shadow of f, and we note that the shadow is nonzero if and only if f has a nonzero non-holomorphic part. We

further note that for our definition of a harmonic Maass form, the shadow is always a cusp form of weight 2 - k.

Suppose  $F(\tau)$  is a harmonic Maass form of weight 2-k as in Lemma 2.1 with shadow  $h(\tau) := \xi_{2-k}F(\tau)$ . Then if  $\rho$  is a cusp, we have that

$$F_{\rho}^{-}(\tau) = \int_{-\overline{\tau}}^{i\infty} h_{\rho}(z) (-i(\tau+z))^{k-2} \mathrm{d}z.$$
 (2.2)

Bruinier and Funke used the  $\xi$  operator to define a bilinear pairing  $\{\cdot, \cdot\}$ :  $M_k \times H_{2-k} \to \mathbb{C}$  by

$$\{g, f\} := (g, \xi_{2-k}f)_k, \qquad (2.3)$$

where  $(\cdot, \cdot)_k$  is the regularized Petersson scalar product. Proposition (3.5) of [53] gives this pairing in terms of the Fourier coefficients of g and the holomorphic part of f. In particular, suppose at a cusp h, g has an expansion  $\sum_n a(h,n)q^n$  and f has an expansion with holomorphic part  $\sum_n b(h,n)q^n$ . They then show that

$$\{g, f\} = \sum_{h} \sum_{n \le 0} a(-n)b(n).$$
(2.4)

The first sum here is over the components of a vector-valued form. In their notation, all Maass forms are level 1, and higher level forms may be viewed as level 1 vector-valued forms if we sum over the cusps.

This pairing is important to us primarily because of the following observation:  $\{\xi_k f, f\} \neq 0$ . This follows from the properties of the Peterson scalar product. However, since  $\xi_k f$  is a cusp form, (2.4) gives us the following theorem:

**Theorem 2.2** (Bruinier, Funke). If f(z) is a harmonic weak Maass form with a nonzero non-holomorphic part, then f must have a non-zero principal part at least one cusp.

### 2.2 Maass-Poincaré series

The Maass–Poincaré series define a basis for a space of harmonic Maass forms and provide exact formulae for their coefficients. The following construction of the Maass–Poincaré series follows the method and notation of Bringmann and Ono [45] which builds on the early work of Rademacher, followed by more contemporary work of Fay, Niebur, among many others [117, 242, 243]. The Poincaré series we construct in this section are modular for congruence subgroups  $\Gamma_0(N)$ .

For  $s \in \mathbb{C}$ ,  $w \in \mathbb{R} \setminus \{0\}$ , and  $k \ge 3/2$ ,  $k \in \frac{1}{2}\mathbb{Z}$ , let

$$\mathcal{M}_{s}(w) := |y|^{\frac{k}{2}-1} M_{\operatorname{sign}(w)(1-k/2), s-\frac{1}{2}}(|w|), \qquad (2.5)$$

where  $M_{\nu,\mu}(z)$  is the *M*-Whittaker function which is a solution to the differential equation

$$\frac{\partial^2 u}{\partial z^2} + \left( -\frac{1}{4} + \frac{\nu}{z} + \frac{\frac{1}{4} - \mu^2}{z^2} \right) u = 0,$$

and (here and throughout this paper)  $\tau = x + iy$ . Using this function, let

$$\phi_s(\tau) := \mathcal{M}_s(4\pi y)e^{2\pi ix}.$$
(2.6)

Given a positive integer m and a cusp  $\rho$ , Maass–Poincaré series provide a form with principal part equal to  $q^{(-m+\kappa_{\rho})/t_{\rho}}$  plus a constant at the cusp  $\rho$ , and constant at all other cusps, thereby forming a basis for  $H_{2-k}(\Gamma, \nu)$ .

Suppose m > 0 and  $L \in SL_2(\mathbb{Z})$  with  $\rho = L^{-1}\infty$ . Then we have the Maass-Poincaré series

$$\mathcal{P}_L(\tau, m, \Gamma, 2-k, s, \nu) := \sum_{M \in \Gamma_\rho \setminus \Gamma} \frac{\phi_s\left(\frac{-m+\kappa_\rho}{t_\rho} \cdot L^{-1}M\tau\right)}{(L^{-1}:M\tau)^{2-k}(M:\tau)^{2-k}\nu(M)}.$$
 (2.7)

It is easy to check that  $\phi_s(\tau)$  is an eigenfunction of  $\Delta_{2-k}$  with eigenvalue

$$s(1-s) + \frac{k^2 - 2k}{4}$$

The right hand side of (2.7) converges absolutely for  $\Re(s) > 1$ , however Bringmann and Ono establish conditional convergence when  $s \ge 3/4$  [45], giving Theorem 2.3 below. The theorem is stated for the specific case  $\Gamma = \Gamma_0(N)$  for some N and  $k \ge 3/2$ , in which case we modify the notation slightly and define

$$\mathcal{P}_{L}(\tau, m, N, 2-k, \nu) := \frac{1}{\Gamma(k)} \mathcal{P}_{L}(\tau, m, \Gamma_{0}(N), 2-k, \frac{k}{2}, \nu)$$
(2.8)

In the statement of the theorem below,  $K_c$  is a modified Kloosterman sum given by

$$K_{c}(2-k,L,\nu,m,n) := \sum_{\substack{0 \le a < ct_{\rho} \\ a \equiv -\frac{c \cdot (\alpha,N)}{\alpha\gamma} (\text{mod } \frac{N}{(\gamma,N)})}} \frac{(S:\tau)^{2-k} \exp\left(2\pi i \left(\frac{a \cdot \frac{(-m+\kappa_{\rho})}{t_{\rho}} + d \cdot \frac{(n+\kappa_{\infty})}{t_{\infty}}}{c}\right)\right)}{(L^{-1}:LS\tau)^{2-k}(LS:\tau)^{2-k}\nu(LS)}, \quad (2.9)$$

where  $S = \begin{pmatrix} a & b \\ c & d \end{pmatrix} \in SL_2(\mathbb{Z})$ . If  $\nu$  is trivial, we omit it from the notation. We also have that  $\delta_{L,S}(m)$  is an indicator function for the cusps  $\rho = L^{-1}\infty$ and  $\mu = S^{-1}\infty$  given by

$$\delta_{L,S}(m) := \begin{cases} \nu(M)^{-1} e^{2\pi i r \frac{-m + \kappa_{\rho}}{t_{\rho}}} & \text{if } M = LT^r S^{-1} \in \Gamma_0(N), \\ 0 & \text{if } \mu \not\sim \rho \text{ in } \Gamma_0(N). \end{cases}$$

Using this notation, we have the following theorem which gives exact formulae for the coefficients and principal part of  $\mathcal{P}_L(\tau, m, N, 2 - k, \nu)$ , which is a generalization of Theorem 3.2 of [45].

**Theorem 2.3.** Suppose that  $\frac{3}{2} \leq k \in \frac{1}{2}\mathbb{Z}$ , and suppose  $\rho = L^{-1}\infty$  is a cusp of  $\Gamma_0(N)$ . If m is a positive integer, then  $\mathcal{P}_L(\tau, m, N, 2 - k, \nu)$  is in  $H_{2-k}(\Gamma_0(N), \nu)$ . Moreover, the following are true:

1. We have

$$\mathcal{P}_L^+(\tau, m, N, 2-k, \nu) = \delta_{\rho, I}(m) \cdot q^{-m+\kappa_\infty} + \sum_{n \ge 0} a^+(n)q^n.$$

Moreover, if n > 0, then  $a^+(n)$  is given by

$$-i^{k}2\pi \left| \frac{-m+\kappa_{\rho}}{t_{\rho}(n+\kappa_{\infty})} \right|^{\frac{k-1}{2}} \sum_{\substack{c>0\\(c,N)=(\gamma,N)}} \frac{K_{c}(2-k,L,\nu,-m,n)}{c}$$
$$I_{k-1}\left(\frac{4\pi}{c}\sqrt{\frac{\left|-m+\kappa_{\rho}\right|\left|n+\kappa_{\infty}\right|}{t_{\rho}}}\right),$$

where  $I_k$  is the usual I-Bessel function.

2. If  $S \in SL_2(\mathbb{Z})$ , then there is some  $c \in \mathbb{C}$  so that the principal part of  $\mathcal{P}_L(\tau, m, N, 2-k)$  at the cusp  $\mu = S^{-1}\infty$  is given by

$$\delta_{L,S}(m)q^{\frac{-m+\kappa_{\rho}}{t_{\rho}}}+c$$

Sketch of the proof. Writing  $\rho = \frac{\alpha}{\gamma}$ , Bringmann and Ono prove this theorem for the case that  $\gamma \mid N$  and  $(\alpha, N) = 1$ , along with the assumption that  $\mu$ and  $\rho$  are in a fixed complete set of inequivalent cusps, so that  $\delta_{\mu,\rho} = 1$  or 0. This general form is useful to us particularly since it works equally well for the cusps  $\infty$  with L taken to be the identity, and for 0 with L taken to be  $\begin{pmatrix} 0 & -1 \\ 1 & 0 \end{pmatrix}$ .

Here and throughout, we let  $S_N$  denote any complete set of inequivalent cusps of  $\Gamma_0(N)$ , and for each  $\rho \in S_N$ , we fix some  $L_\rho$  with  $\rho = L_\rho^{-1}\infty$ . Rankin notes [259] (Proof of Theorem 4.1.1(iii)) that given some choice of  $S_N$ , each right coset of  $\Gamma_0(N) \setminus SL_2(\mathbb{Z})$  is in  $\Gamma_0(N) \cdot L_\rho T^r$  for some unique  $\rho \in S_N$ . Moreover, the *r* in the statement is unique modulo  $t_\rho$ , so the function  $\delta_{L,S}(m)$ given above is well-defined on all matrices in  $SL_q(\mathbb{Z})$ . In the proof given by Bringmann and Ono, the sum of Kloosterman sums  $\sum_{\substack{c>0\\(c,N)=(\gamma,N)}} \frac{K_c(2-k,L,\nu,-m,n)}{c} \dots$  is written as a sum over representatives

of the double coset  $\Gamma_{\rho} \setminus L^{-1}\Gamma_0(N)/\Gamma_{\infty}$  (omitting the identity if present). Following similar arguments, but without the assumptions on  $\alpha$  and  $\gamma$ , we find the indices of summation given in (2.9) and Theorem 2.3. As in their case, we find that the principal part of  $\mathcal{P}_{L_{\rho}}(\tau, m, N, 2 - k)$  at a cusp  $\mu$  is constant if  $\mu \not\sim \rho$  and is  $\delta_{L_{\rho},L_{\mu}}q^{\frac{-m}{t_{\rho}}} + c$  for some constant if  $\mu = \rho$ . Therefore, if  $\mu$  is a cusp with  $L_{\mu} = M^{-1}L_{\rho}T^r$  for some  $M \in \Gamma_0(N, )$ , then the principal part of  $\mathcal{P}_{L_{\rho}}(\tau, m, N, 2 - k, \nu)$  at  $\mu$  is clearly  $\nu(M)^{-1}e^{2\pi i r \frac{-m+\kappa_{\rho}}{t_{\rho}}}q^{\frac{-m+\kappa_{\rho}}{t_{\rho}}} + c$ .  $\Box$ 

Since harmonic Maass forms with a nonholomorphic part have a nonconstant principal part at some cusp, we have the following theorem.

**Theorem 2.4.** [45, Theorem 1.1] Assuming the notation above, if  $\frac{3}{2} \leq k \in \frac{1}{2}\mathbb{Z}$ , and  $F(\tau) \in H_{2-k}(\Gamma_0(N), \nu)$  has principal part

$$P_{\rho}(\tau) = \sum_{m \ge 0} a_{\rho}(-m) q^{\frac{-m+\kappa_{\rho}}{t_{\rho}}}$$

for each cusp  $\rho \in \mathcal{S}_N$ , then

$$F(\tau) = \sum_{\rho \in S_N} \sum_{m>0} a_{\rho}(-m) \mathcal{P}_{\rho}(\tau, m, N, 2 - k, \nu) + g(\tau),$$

where  $g(\tau)$  is a holomorphic modular form. Moreover, we have that c = 0whenever k > 2, and is a constant when k = 2.

## 2.3 Holomorphic projection

The theory of holomorphic projections provides a way to explicitly relate the coefficients of harmonic Maass forms and other nonholomorphic modular forms to classical modular or quasimodular forms<sup>1</sup>. For instance, given a weight 1/2 harmonic Maass form, we may multiply by its shadow to obtain a weight 2 nonholomorphic modular form with simple transformation properties. The holomorphic projection provides a simple explicit correction term to convert this nonholomorphic modular form into a weight 2 quasimodular form with the same transformation properties. In this way, we may essentially reduce many questions about the coefficients of weight  $\frac{1}{2}$  mock modular forms to questions about weight 2 holomorphic modular forms.

Although holomorphic projections may be applied more generally, we will restrict our attention in this section to weight 1/2 Harmonic Maass forms multiplied by weight 3/2 theta series. The following theorem is a special case of a more general theorem due to Mertens (cf. Theorem 6.3 of [235]).

**Theorem 2.5** (Mertens). Suppose  $g(\tau)$  and  $h(\tau)$  are both theta functions of weight  $\frac{3}{2}$  contained in  $S_{\frac{3}{2}}(\Gamma, \nu_g)$  and  $S_{\frac{3}{2}}(\Gamma, \nu_h)$  respectively, with Fourier expansions

$$g(\tau) := \sum_{i=1}^{s} \sum_{n \in \mathbb{Z}} n\chi_i(n) q^{n^2},$$
$$h(\tau) := \sum_{j=1}^{t} \sum_{n \in \mathbb{Z}} n\psi_j(n) q^{n^2},$$

where each  $\chi_i$  and  $\psi_i$  is a Dirichlet character. Moreover, suppose  $h(\tau)$  is the shadow of a weight  $\frac{1}{2}$  harmonic Maass form  $f(\tau) \in H_{\frac{1}{2}}(\Gamma, \overline{\nu_h})$ . Define the function

$$D^{f,g}(\tau) := 2 \sum_{r=1}^{\infty} \sum_{\chi_i,\psi_j} \sum_{\substack{m,n \in \mathbb{Z}^+ \\ m^2 - n^2 = r}} \chi_i(m) \overline{\psi_j(n)}(m-n) q^r.$$

If  $f(\tau)g(\tau)$  has no singularity at any cusp, then  $f^+(\tau)g(\tau) + D^{f,g}(\tau)$  is a weight 2 quasimodular form. In other words, it lies in the space  $\mathbb{C}E_2(\tau) \oplus M_2(\Gamma, \nu_g \overline{\nu_h})$ .

<sup>&</sup>lt;sup>1</sup>By quasimodular forms here, we mean spaces of the form  $\mathbb{C}E_2(\tau) \oplus M_2(\Gamma, \nu)$ .

### Two Remarks.

1) These identities give recurrence relations for the weight  $\frac{1}{2}$  mock modular form  $f^+$  in terms of the weight 2 quasimodular form which equals  $f^+(\tau)g(\tau) + D^{f,g}(\tau)$ . The example after Theorem 1.32 for Ramanujan's third order mock theta function f is an explicit example of such a relation.

2) Theorem 2.5 extends to vector-valued mock modular forms in the natural way by acting on components.

# Chapter 3

# **Rogers–Ramanujan** identities

### 3.1 The Hall–Littlewood polynomials

Let  $\lambda = (\lambda_1, \lambda_2, ...)$  be an integer partition [9], a nonincreasing sequence of nonnegative integers  $\lambda_1 \geq \lambda_2 \geq ...$  with only finitely nonzero terms. The positive  $\lambda_i$  are called the *parts* of  $\lambda$ , and the number of parts, denoted  $l(\lambda)$ , is the *length* of  $\lambda$ . The *size*  $|\lambda|$  of  $\lambda$  is the sum of its parts. The Ferrers– Young diagram of  $\lambda$  consists of  $l(\lambda)$  left-aligned rows of squares such that the *i*th row contains  $\lambda_i$  squares. For example, the Ferrers–Young diagram of  $\nu = (6, 4, 4, 2)$  of length 4 and size 16 is



The conjugate partition  $\lambda'$  corresponds to the transpose of the Ferrers-Young diagram of  $\lambda$ . For example, we have  $\nu' = (4, 4, 3, 3, 1, 1)$ . We define nonnegative integers  $m_i = m_i(\lambda)$ , for  $i \ge 1$ , to be the multiplicities of parts of size i, so that  $|\lambda| = \sum_i i m_i$ . It is easy to see that  $m_i = \lambda'_i - \lambda'_{i+1}$ . We say that a partition is even if its parts are all even. Note that  $\lambda'$  is even if all multiplicities  $m_i(\lambda)$  are even. The partition  $\nu$  above is an even partition. Given two partitions  $\lambda, \mu$  we write  $\mu \subseteq \lambda$  if the diagram of  $\mu$  is contained in the diagram of  $\lambda$ , or, equivalently, if  $\mu_i \leq \lambda_i$  for all *i*. To conclude our discussion of partitions, we define the generalized q-shifted factorial

$$b_{\lambda}(q) := \prod_{i \ge 1} (q)_{m_i} = \prod_{i \ge 1} (q)_{\lambda'_i - \lambda'_{i+1}}.$$
(3.1)

Hence, for  $\nu$  as above we have  $b_{\nu}(q) = (q)_1^2(q)_2$ .

For a fixed positive integer n, let  $x = (x_1, \ldots, x_n)$ . Given a partition  $\lambda$  such that  $l(\lambda) \leq n$ , write  $x^{\lambda}$  for the monomial  $x_1^{\lambda_1} \ldots x_n^{\lambda_n}$ , and define

$$v_{\lambda}(q) = \prod_{i=0}^{n} \frac{(q)_{m_i}}{(1-q)^{m_i}},$$
(3.2)

where  $m_0 := n - l(\lambda)$ . The Hall-Littlewood polynomial  $P_{\lambda}(x;q)$  is defined as the symmetric function [219]

$$P_{\lambda}(x;q) = \frac{1}{v_{\lambda}(q)} \sum_{w \in \mathfrak{S}_n} w \left( x^{\lambda} \prod_{i < j} \frac{x_i - qx_j}{x_i - x_j} \right), \tag{3.3}$$

where the symmetric group  $\mathfrak{S}_n$  acts on x by permuting the  $x_i$ . It follows from the definition that  $P_{\lambda}(x;q)$  is a homogeneous polynomial of degree  $|\lambda|$ , a fact used repeatedly in the rest of this paper.  $P_{\lambda}(x;q)$  is defined to be identically 0 if  $l(\lambda) > n$ . The Hall-Littlewood polynomials may be extended in the usual way to symmetric functions in countably-many variables, see [219].

Here we make this precise when x is specialized to an infinite geometric progression. For  $x = (x_1, x_2, ...)$  not necessarily finite, let  $p_r$  be the r-th power sum symmetric function

$$p_r(x) = x_1^r + x_2^r + \cdots,$$

and  $p_{\lambda} = \prod_{i \ge 1} p_{\lambda_i}$ . The power sums  $\{p_{\lambda}(x_1, \ldots, x_n)\}_{l(\lambda) \le n}$  form a Q-basis of the ring of symmetric functions in *n* variables. If  $\phi_q$  denotes the ring homomorphism  $\phi_q(p_r) = p_r/(1-q^r)$ , then the modified Hall-Littlewood polynomials  $P'_{\lambda}(x;q)$  are defined as the image of the  $P_{\lambda}(x;q)$  under  $\phi_q$ :

$$P_{\lambda}' = \phi_q (P_{\lambda}).$$

We note that as stated, this homomorphism is well defined only if nis at least the degree of  $P_{\lambda}$  (i.e.  $|\lambda|$ ). Otherwise nontrivial relations exist among products of the  $p_r(x)$  with degree greater than n. However if  $n \geq |\lambda|$ , then  $P_{\lambda}$  can be expressed uniquely as terms of the  $p_r$ . Moreover this expression is otherwise independent of n. Therefore for the purposes of this homomorphism, we identify  $P_{\lambda}$  with this expansion.

We also require the Hall–Littlewood polynomials  $Q_{\lambda}$  and  $Q'_{\lambda}$  defined by

$$Q_{\lambda}(x;q) := b_{\lambda}(q)P_{\lambda}(x;q) \quad \text{and} \quad Q_{\lambda}'(x;q) := b_{\lambda}(q)P_{\lambda}'(x;q). \tag{3.4}$$

Clearly,  $Q'_{\lambda} = \phi_q(Q_{\lambda}).$ 

Up to the point where the x-variables are specialized, our proof of Theorems 1.2–1.5 will make use of the modified Hall–Littlewood polynomials, rather than the ordinary Hall–Littlewood polynomials. Through specialization, we arrive at  $P_{\lambda}$  evaluated at a geometric progression thanks to

$$P_{\lambda}(1,q,q^2,\ldots;q^n) = P'_{\lambda}(1,q,\ldots,q^{n-1};q^n),$$
(3.5)

which readily follows from

$$\phi_{q^n}(p_r(1,q,\ldots,q^{n-1})) = \frac{1-q^{nr}}{1-q^r} \cdot \frac{1}{1-q^{nr}} = p_r(1,q,q^2,\ldots).$$

Example 3.1. Let  $\lambda = (2)$ . If we take n = 1, then  $P_{\lambda}(x;q) = x_1^2$ , whereas if n = 2, then  $P_{\lambda}(x;q) = x_1^2 + (1-q) \cdot x_1 x_2 + x_2^2$ , from which we can find

$$P_{\lambda}(x;q) = \frac{1-q}{2}p_1(x)^2 + \left(1 - \frac{1-q}{2}\right)p_2(x).$$

In order to find  $P_{\lambda}(1, q, q^2, ...; q)$  we replace  $p_1(x)$  in the expression above with  $\frac{1}{1-q}$ , and  $p_2(x)$  with  $\frac{1}{1-q^2}$  to get

$$P_{\lambda}(1,q,q^2,\ldots;q) = \frac{1-q}{2}\frac{1}{(1-q)^2} + \left(1-\frac{1-q}{2}\right)\frac{1}{1-q^2} = \frac{1}{1-q}.$$
Example 3.2. Let  $\lambda = (2, 2)$ . The table below gives values of  $P_{\lambda}(1, q, q^2, \dots, q^{n-1}; q)$  for various choices of n.

The limiting value  $P_{\lambda}(1, q, q^2, ...; q)$  can be found exactly as described above. We find that

$$P_{\lambda}(x;q) = \frac{2 - 3q + q^3}{24} p_1(x)^4 + \frac{q - q^3}{4} p_2(x) p_1(x)^2 + \frac{2 + q + q^3}{8} p_2(x)^2 - \frac{1 - q^3}{3} p_3(x) p_1(x) - \frac{q + q^3}{4} p_4(x),$$

which implies

$$P_{\lambda}(1,q,q^2,\ldots;q) = \frac{q^2}{(1-q)(1-q^2)}.$$

From [187, 293] we may infer the following combinatorial formula for the modified Hall–Littlewood polynomials:

$$Q_{\lambda}'(x;q) = \sum \prod_{i=1}^{\lambda_1} \prod_{a=1}^n x_a^{\mu_i^{(a-1)} - \mu_i^{(a)}} q^{\binom{\mu_i^{(a-1)} - \mu_i^{(a)}}{2}} \begin{bmatrix} \mu_i^{(a-1)} - \mu_{i+1}^{(a)} \\ \mu_i^{(a-1)} - \mu_i^{(a)} \end{bmatrix}_q,$$

where the sum is over partitions  $0 = \mu^{(n)} \subseteq \cdots \subseteq \mu^{(1)} \subseteq \mu^{(0)} = \lambda'$  and

$$\begin{bmatrix} n \\ m \end{bmatrix}_q = \begin{cases} \frac{(q)_n}{(q)_m(q)_{n-m}} & \text{if } m \in \{0, 1, \dots, n\} \\ 0 & \text{otherwise} \end{cases}$$

is the usual q-binomial coefficient. Therefore, by (3.1)-(3.5), we have obtained the following combinatorial description of the q-series we have assembled from the Hall-Littlewood polynomials.

**Lemma 3.3.** If m and n are positive integers, then

$$\sum_{\substack{\lambda\\\lambda_1 \le m}} q^{(\sigma+1)|\lambda|} P_{2\lambda}(1,q,q^2,\dots;q^n) \\ = \sum \prod_{i=1}^{\lambda} \left\{ \frac{q^{\frac{1}{2}(\sigma+1)\mu_i^{(0)}}}{(q^n;q^n)_{\mu_i^{(0)}-\mu_{i+1}^{(0)}}} \prod_{a=1}^n q^{\mu_i^{(a)}+n\binom{\mu_i^{(a-1)}-\mu_i^{(a)}}{2}} \left[ \frac{\mu_i^{(a-1)}-\mu_{i+1}^{(a)}}{\mu_i^{(a-1)}-\mu_i^{(a)}} \right]_{q^n} \right\}, \quad (3.6)$$

where the sum on the right is over partitions  $0 = \mu^{(n)} \subseteq \cdots \subseteq \mu^{(1)} \subseteq \mu^{(0)}$ such that  $(\mu^{(0)})'$  is even and  $l(\mu^{(0)}) \leq 2m$ .

Lemma 3.3 may be used to express the sum sides of (1.8)-(1.13) combinatorially. Moreover, we have that (3.6) generalizes the sums in (1.2), (1.3), and (1.6). To see this, we note that the above simplifies for n = 1 to

$$\sum_{\substack{\lambda \\ \lambda_1 \le m}} q^{(\sigma+1)|\lambda|} P_{2\lambda}(1, q, q^2, \dots; q) = \sum \prod_{i=1}^{2m} \frac{q^{\frac{1}{2}\mu_i(\mu_i + \sigma)}}{(q)_{\mu_i - \mu_{i+1}}}$$

summed on the right over partitions  $\mu$  of length at most 2m whose conjugates are even. Such partitions are characterized by the restriction  $\mu_{2i} = \mu_{2i-1} =: r_i$ so that we get

$$\sum_{\substack{\lambda \\ \lambda_1 \le m}} q^{(\sigma+1)|\lambda|} P_{2\lambda}(1, q, q^2, \dots; q) = \sum_{r_1 \ge \dots \ge r_m \ge 0} \prod_{i=1}^m \frac{q^{r_i(r_i+\sigma)}}{(q)_{r_i-r_{i+1}}}$$

in accordance with (1.6).

If instead we consider m = 1 and replace  $\mu^{(j)}$  by  $(r_j, s_j)$  for  $j \ge 0$ , we find

$$\sum_{r=0}^{\infty} q^{(\sigma+1)r} P_{(2^r)}(1, q, q^2, \dots; q^n)$$

$$= \sum_{r=0}^{\infty} \frac{q^{(\sigma+1)r_0}}{(q^n; q^n)_{r_0}} \prod_{j=1}^n q^{r_j + s_j + n\binom{r_{j-1} - r_j}{2} + n\binom{s_{j-1} - s_j}{2}} {r_{j-1} - r_j} {r_{j-1} - r_j}_{q^n} {s_{j-1} \choose s_j}_{q^n}$$

$$= \frac{(q^{n+4}; q^{n+4})_{\infty}}{(q)_{\infty}} \cdot \theta(q^{2-\sigma}; q^{n+4}),$$

where the second sum is over  $r_0, s_0, \ldots, r_{n-1}, s_{n-1}$  such that  $r_0 = s_0$ , and  $r_n = s_n := 0$ .

We conclude this section with a remark about Theorem 1.7. Due to the occurrence of the limit, the left-hand side does not take the form of the usual sum-side of a Rogers-Ramanujan-type identity. For special cases it is, however, possible to eliminate the limit. For example, for partitions of the form  $(2^r)$  we found that

$$P_{(2^{r})}(1,q,q^{2},\ldots;q^{2n+\delta}) = \sum_{r \ge r_{1} \ge \cdots \ge r_{n} \ge 0} \frac{q^{r^{2}-r+r_{1}^{2}+\cdots+r_{n}^{2}+r_{1}+\cdots+r_{n}}}{(q)_{r-r_{1}}(q)_{r_{1}-r_{2}}\cdots(q)_{r_{n-1}-r_{n}}(q^{2-\delta};q^{2-\delta})_{r_{n}}}$$
(3.7)

for  $\delta = 0, 1$ . This turns the m = 2 case of Theorem 1.7 into

$$\sum_{r_1 \ge \dots \ge r_n \ge 0} \frac{q^{r_1^2 + \dots + r_n^2 + r_1 + \dots + r_n}}{(q)_{r_1 - r_2} \cdots (q)_{r_{n-1} - r_n} (q^{2-\delta}; q^{2-\delta})_{r_n}} = \frac{(q^{2n+2+\delta}; q^{2n+2+\delta})_{\infty}}{(q)_{\infty}} \cdot \theta(q; q^{2n+2+\delta})$$

For  $\delta = 1$  this is the i = 1 case of the Andrews–Gordon identity (1.6) (with m replaced by n). For  $\delta = 0$  it corresponds to the i = 1 case of (1.12). We do not know how to generalize (3.7) to arbitrary rectangular shapes.

## 3.2 Proof of Theorems 1.2–1.5

Here we prove Theorems 1.2–1.5. We begin by recalling key aspects of the classical works of Andrews and Watson which give hints of the generalizations we obtain.

#### 3.2.1 The Watson–Andrews approach

In 1929 Watson proved the Rogers–Ramanujan identities (1.2) and (1.3) by first proving a new basic hypergeometric series transformation between a terminating balanced  $_4\phi_3$  series and a terminating very-well-poised  $_8\phi_7$  series [295]

$$\frac{(aq, aq/bc)_N}{(aq/b, aq/c)_N} \sum_{r=0}^N \frac{(b, c, aq/de, q^{-N})_r}{(q, aq/d, aq/e, bcq^{-N}/a)_r} q^r 
= \sum_{r=0}^N \frac{1 - aq^{2r}}{1 - a} \cdot \frac{(a, b, c, d, e, q^{-N})_r}{(q, aq/b, aq/c, aq/d, aq/e)_r} \cdot \left(\frac{a^2 q^{N+2}}{bcde}\right)^r. \quad (3.8)$$

Here a, b, c, d, e are indeterminates, N is a nonnegative integer and

$$(a_1, \ldots, a_m)_k := (a_1, \ldots, q_m; q) = (a_1; q)_k \cdots (a_m; q)_k$$

By letting b, c, d, e tend to infinity and taking the nonterminating limit  $N \rightarrow \infty$ , Watson arrived at what is known as the Rogers–Selberg identity [264, 269]<sup>1</sup>

$$\sum_{r=0}^{\infty} \frac{a^r q^{r^2}}{(q)_r} = \frac{1}{(aq)_{\infty}} \sum_{r=0}^{\infty} \frac{1 - aq^{2r}}{1 - a} \cdot \frac{(a)_r}{(q)_r} \cdot (-1)^r a^{2r} q^{5\binom{r}{2} + 2r}.$$
 (3.9)

For a = 1 or a = q the sum on the right can be expressed in product-form by the Jacobi triple-product identity

$$\sum_{r=-\infty}^{\infty} (-1)^r x^r q^{\binom{r}{2}} = (q)_{\infty} \cdot \theta(x;q),$$

resulting in (1.2) and (1.3).

Almost 50 years after Watson's work, Andrews showed that the Andrews– Gordon identities (1.6) for i = 1 and i = m + 1 follow in a similar way from a multiple series generalization of (3.8) in which the  $_8\phi_7$  series on the right is replaced by a terminating very-well-poised  $_{2m+6}\phi_{2m+5}$  series depending on

<sup>&</sup>lt;sup>1</sup>Here and elsewhere in the paper we ignore questions of convergence. From an analytic point of view, the transition from (3.8) to (3.9) requires the use of the dominated convergence theorem, imposing the restriction |q| < 1 on the Rogers–Selberg identity. We however choose to view this identity as an identity between formal power series in q, in line with the combinatorial and representation-theoretic interpretations of Rogers–Ramanujan-type identities.

2m + 2 parameters instead of b, c, d, e [8]. Again the key steps are to let all these parameters tend to infinity, to take the nonterminating limit, and to then express the a = 1 or a = q instances of the resulting sum as a product by the Jacobi triple-product identity.

Recently, Bartlett and Warnaar obtained an analog of Andrews' multiple series transformation for the  $C_n$  root system [19, Theorem 4.2]. Apart from the variables  $(x_1, \ldots, x_n)$ —which play the role of a in (3.8), and are related to the underlying root system—the  $C_n$  Andrews transformation again contains 2m + 2 parameters. Unfortunately, simply following the Andrews–Watson procedure is no longer sufficient. In [238] Milne already obtained the  $C_n$ analogue of the Rogers–Selberg identity (3.9) (the m = 1 case of (3.10) below) and considered specializations along the lines of Andrews and Watson. Only for  $C_2$  did this result in a Rogers–Ramanujan-type identity: the modulus 6 case of (1.11) mentioned previously.

The first two steps towards a proof of (1.8)-(1.13), however, are the same as those of Watson and Andrews: we let all 2m + 2 parameters in the  $C_n$ Andrews transformation tend to infinity and take the nonterminating limit. Then, as shown in [19], the right-hand side can be expressed in terms of modified Hall-Littlewood polynomials, resulting in the level- $m C_n$  Rogers-Selberg identity

$$\sum_{\substack{\lambda \\ \lambda_1 \le m}} q^{|\lambda|} P'_{2\lambda}(x;q) = L_m^{(0)}(x;q),$$
(3.10)

where

$$L_m^{(0)}(x;q) := \sum_{r \in \mathbb{Z}_+^n} \frac{\Delta_{\mathcal{C}}(xq^r)}{\Delta_{\mathcal{C}}(x)} \prod_{i=1}^n x_i^{2(m+1)r_i} q^{(m+1)r_i^2 + n\binom{r_i}{2}} \cdot \prod_{i,j=1}^n \left(-\frac{x_i}{x_j}\right)^{r_i} \frac{(x_i x_j)_{r_i}}{(qx_i/x_j)_{r_i}}$$

Here we have that

$$\Delta_{\mathcal{C}}(x) := \prod_{i=1}^{n} (1 - x_i^2) \prod_{1 \le i < j \le n} (x_i - x_j) (x_i x_j - 1)$$

is the C<sub>n</sub> Vandermonde product, and  $f(xq^r)$  is shorthand for  $f(x_1q^{r_1}, \ldots, x_nq^{r_n})$ .

*Remark.* As mentioned previously, (3.10) for m = 1 is Milne's  $C_n$  Rogers–Selberg formula [238, Corollary 2.21].

The strategy for the proofs of Theorems 1.2-1.5 is now simple to describe. By comparing the left-hand side of (3.10) with that of (1.8)-(1.11), it follows that we should make the simultaneous substitutions

$$q \mapsto q^n, \qquad x_i \mapsto q^{(n+\sigma+1)/2-i} \quad (1 \le i \le n).$$
 (3.11)

Then, by the homogeneity and symmetry of the (modified) Hall–Littlewood polynomials and (3.5), we have

$$\sum_{\substack{\lambda\\\lambda_1 \le m}} q^{|\lambda|} P'_{2\lambda}(x;q) \longmapsto \sum_{\substack{\lambda\\\lambda_1 \le m}} q^{(\sigma+1)|\lambda|} P_{2\lambda}(1,q,q^2,\ldots;q^n).$$

Therefore, we wish to carry out these maneuvers and prove that the resulting right-hand side can be described as a product of modified theta functions in the four families in the theorems. The problem we face is that making the substitutions (3.11) in the right-hand side of (3.10) and then writing the resulting q-series in product form is very difficult.

To get around this problem, we take a rather different route and (up to a small constant) first double the rank of the underlying  $C_n$  root system and then take a limit in which products of pairs of x-variables tend to one. To do so we require another result from [19].

First we extend our earlier definition of the q-shifted factorial to

$$(a)_k = (a)_{\infty} / (aq^k)_{\infty}.$$
 (3.12)

Importantly, we note that  $1/(q)_k = 0$  for k a negative integer. Then, for  $x = (x_1, \ldots, x_n)$ , p an integer such that  $0 \le p \le n$  and  $r \in \mathbb{Z}^n$ , we have

$$L_m^{(p)}(x;q) := \sum_{r \in \mathbb{Z}^n} \frac{\Delta_{\mathcal{C}}(xq^r)}{\Delta_{\mathcal{C}}(x)} \prod_{i=1}^n x_i^{2(m+p+1)r_i} q^{(m+1)r_i^2 + (n+p)\binom{r_i}{2}} \times \prod_{i=1}^n \prod_{j=p+1}^n \left(-\frac{x_i}{x_j}\right)^{r_i} \frac{(x_i x_j)_{r_i}}{(qx_i/x_j)_{r_i}}.$$
 (3.13)

Note that the summand of  $L_m^{(p)}(x;q)$  vanishes if one of  $r_{p+1},\ldots,r_n<0$ .

The following key lemma will be crucial for our strategy to work.

**Lemma 3.4** ([19, Lemma A.1]). For  $1 \le p \le n - 1$ ,

$$\lim_{x_{p+1}\to x_p^{-1}} L_m^{(p-1)}(x;q) = L_m^{(p)}(x_1,\dots,x_{p-1},x_{p+1},\dots,x_n;q).$$
(3.14)

This will be the key to the proof of all four generalized Rogers-Ramanujan identities, although the level of difficulty varies considerably from case to case. We begin with the simplest proof, that of Theorem 1.4 (i.e.equation (1.10)).

## 3.2.2 Proof of Theorem 1.4

Here we carry out the strategy described in the previous section by making use of the  $C_n$  and  $B_n$  Weyl denominator formulas, and the  $\mathbf{D}_{n+1}^{(2)}$  Macdonald identity.

Proof of Theorem 1.4. By iterating (3.14), we have

$$\lim_{y_1 \to x_1^{-1}} \dots \lim_{y_n \to x_n^{-1}} L_m^{(0)}(x_1, y_1, \dots, x_n, y_n) = L_m^{(n)}(x_1, \dots, x_n).$$

Hence, after replacing  $x \mapsto (x_1, y_1, \ldots, x_n, y_n)$  in (3.10) (which corresponds to the doubling of the rank mentioned previously) and taking the  $y_i \to x_i^{-1}$ limit for  $1 \le i \le n$ , we find

$$\sum_{\substack{\lambda\\\lambda_1 \le m}} q^{|\lambda|} P'_{2\lambda}(x^{\pm};q) = \frac{1}{(q)_{\infty}^n \prod_{i=1}^n \theta(x_i^2;q) \prod_{1 \le i < j \le n} \theta(x_i/x_j, x_i x_j;q)} \times \sum_{r \in \mathbb{Z}^n} \Delta_{\mathcal{C}}(xq^r) \prod_{i=1}^n x_i^{\kappa r_i - i + 1} q^{\frac{1}{2}\kappa r_i^2 - nr_i}, \quad (3.15)$$

where  $\kappa = 2m + 2n + 2$  and  $f(x^{\pm}) = f(x_1, x_1^{-1}, \dots, x_n, x_n^{-1})$ . Next we make the simultaneous substitutions

$$q \mapsto q^{2n}, \qquad x_i \mapsto q^{n-i+1/2} =: \hat{x}_i \ (1 \le i \le n),$$
 (3.16)

which corresponds to (3.11) with  $(n, \sigma) \mapsto (2n, 0)$ . By the identity

$$(q^{2n};q^{2n})_{\infty}^{n} \cdot \prod_{i=1}^{n} \theta(q^{2n-2i+1};q^{2n}) \cdot \prod_{1 \le i < j \le n} \theta(q^{j-i},q^{2n-i-j+1};q^{2n}) = \frac{(q)_{\infty}^{n+1}}{(q^2;q^2)_{\infty}},$$

and

$$\begin{aligned} q^{2n|\lambda|} P'_{2\lambda}(q^{n-1/2}, q^{1/2-n}, \dots, q^{1/2}, q^{-1/2}; q^{2n}) \\ &= q^{2n|\lambda|} P'_{2\lambda}(q^{1/2-n}, q^{3/2-n}, \dots, q^{n-1/2}; q^{2n}) & \text{by symmetry} \\ &= q^{|\lambda|} P'_{2\lambda}(1, q, \dots, q^{2n-1}; q^{2n}) & \text{by homogeneity} \\ &= q^{|\lambda|} P_{2\lambda}(1, q, q^2, \dots; q^{2n}) & \text{by } (3.5), \end{aligned}$$

we obtain

$$\sum_{\substack{\lambda\\\lambda_1 \le m}} q^{|\lambda|} P_{2\lambda} (1, q, q^2, \dots; q^{2n}) = \frac{(q^2; q^2)_{\infty}}{(q)_{\infty}^{n+1}} \mathcal{M},$$
(3.17)

where

$$\mathcal{M} := \sum_{r \in \mathbb{Z}^n} \Delta_{\mathcal{C}}(\hat{x}q^{2nr}) \prod_{i=1}^n \hat{x}_i^{\kappa r_i - i + 1} q^{nr_i^2 - 2n^2 r_i}.$$

We must express  $\mathcal{M}$  in product form. As a first step, we use the  $C_n$  Weyl denominator formula [198, Lemma 2]

$$\Delta_{\mathcal{C}}(x) = \det_{1 \le i, j \le n} \left( x_i^{j-1} - x_i^{2n-j+1} \right), \tag{3.18}$$

as well as multilinearity, to write  $\mathcal{M}$  as

$$\mathcal{M} = \det_{1 \le i,j \le n} \left( \sum_{r \in \mathbb{Z}} \hat{x}_i^{\kappa r - i + 1} q^{n \kappa r^2 - 2n^2 r} \left( (\hat{x}_i q^{2nr})^{j - 1} - (\hat{x}_i q^{2nr})^{2n - j + 1} \right) \right).$$
(3.19)

We now replace  $(i, j) \mapsto (n - j + 1, n - i + 1)$  and, viewing the resulting determinant as being of the form det  $(\sum_{r} u_{ij;r} - \sum_{r} v_{ij;r})$ , we change the summation index  $r \mapsto -r - 1$  in the sum over  $v_{ij;r}$ . Then we find that

$$\mathcal{M} = \det_{1 \le i,j \le n} \left( q^{a_{ij}} \sum_{r \in \mathbb{Z}} y_i^{2nr-i+1} q^{2n\kappa \binom{r}{2} + \frac{1}{2}\kappa r} \left( (y_i q^{\kappa r})^{j-1} - (y_i q^{\kappa r})^{2n-j} \right) \right), \quad (3.20)$$

where  $y_i = q^{\kappa/2-i}$  and  $a_{ij} = j^2 - i^2 + (i-j)(\kappa+1)/2$ . Since the factor  $q^{a_{ij}}$  does not contribute to the determinant, we can apply the  $B_n$  Weyl denominator formula [198]

$$\det_{1 \le i,j \le n} \left( x_i^{j-1} - x_i^{2n-j} \right) = \prod_{i=1}^n (1 - x_i) \prod_{1 \le i < j \le n} (x_i - x_j) (x_i x_j - 1) =: \Delta_{\mathcal{B}}(x) \quad (3.21)$$

to obtain

$$\mathcal{M} = \sum_{r \in \mathbb{Z}^n} \Delta_{\mathrm{B}}(yq^{\kappa r}) \prod_{i=1}^n y_i^{2nr_i - i + 1} q^{2n\kappa\binom{r_i}{2} + \frac{1}{2}\kappa r_i}.$$

By the  $\mathbf{D}_{n+1}^{(2)}$  Macdonald identity [218]

$$\sum_{r \in \mathbb{Z}^n} \Delta_{\mathrm{B}}(xq^r) \prod_{i=1}^n x_i^{2nr_i - i + 1} q^{2n\binom{r_i}{2} + \frac{1}{2}r_i} = (q^{1/2}; q^{1/2})_{\infty} (q)_{\infty}^{n-1} \prod_{i=1}^n \theta(x_i; q^{1/2})_{\infty} \prod_{1 \le i < j \le n} \theta(x_i/x_j, x_i x_j; q)$$

with  $(q, x) \mapsto (q^{\kappa}, y)$ , we obtain

$$\mathcal{M} = (q^{\kappa/2}; q^{\kappa/2})_{\infty} (q^{\kappa}; q^{\kappa})_{\infty}^{n-1} \prod_{i=1}^{n} \theta(q^{i}; q^{\kappa/2}) \prod_{1 \le i < j \le n} \theta(q^{j-i}, q^{i+j}; q^{\kappa}), \quad (3.22)$$

where we have also used the simple symmetry  $\theta(q^{a-b}; q^a) = \theta(q^b; q^a)$ . Substituting (3.22) into (3.17) proves the first equality of (1.10).

Establishing the second equality is a straightforward exercise in manipulating infinite products, and we omit the details.  $\Box$ 

There is a somewhat different approach to (1.10) based on the representation theory of the affine Kac–Moody algebra  $C_n^{(1)}$  [183]. Let  $I = \{0, 1, \ldots, n\}$ , and  $\alpha_i, \alpha_i^{\vee}$  and  $\Lambda_i$  for  $i \in I$  the simple roots, simple coroots and fundamental weights of  $C_n^{(1)}$ . Let  $\langle \cdot, \cdot \rangle$  denote the usual pairing between the Cartan subalgebra  $\mathfrak{h}$  and its dual  $\mathfrak{h}^*$ , so that  $\langle \Lambda_i, \alpha_j^{\vee} \rangle = \delta_{ij}$ . Finally, let  $V(\Lambda)$  be the integrable highest-weight module of  $C_n^{(1)}$  of highest weight  $\Lambda$  with character ch  $V(\Lambda)$ .

The homomorphism

$$F_{\mathscr{W}}: \ \mathbb{C}[[\mathrm{e}^{-\alpha_0}, \dots, \mathrm{e}^{-\alpha_n}]] \to \mathbb{C}[[q]], \qquad F_{\mathscr{W}}(\mathrm{e}^{-\alpha_i}) = q \quad \text{for all } i \in I \quad (3.23)$$

is known as principal specialization [204]. Subject to this specialization, ch  $V(\Lambda)$  admits a simple product form as follows. Let  $\rho$  be the Weyl vector (that is  $\langle \rho, \alpha_i^{\vee} \rangle = 1$  for  $i \in I$ ) and mult( $\alpha$ ) the multiplicity of  $\alpha$ . Then [179, 205] we have

$$F_{\mathscr{F}}\left(\mathrm{e}^{-\Lambda}\operatorname{ch}V(\Lambda)\right) = \prod_{\alpha\in\Delta_{+}^{\vee}} \left(\frac{1-q^{\langle\Lambda+\rho,\alpha\rangle}}{1-q^{\langle\rho,\alpha\rangle}}\right)^{\mathrm{mult}(\alpha)},\tag{3.24}$$

where  $\Delta^{\vee}_{+}$  is the set of positive coroots. This result, which is valid for all types  $X_N^{(r)}$ , can be rewritten in terms of theta functions. Assuming  $C_n^{(1)}$  and setting

$$\Lambda = (\lambda_0 - \lambda_1)\Lambda_0 + (\lambda_1 - \lambda_2)\Lambda_1 + \dots + (\lambda_{n-1} - \lambda_n)\Lambda_{n-1} + \lambda_n\Lambda_n,$$

for  $\lambda = (\lambda_0, \lambda_1, \dots, \lambda_n)$  a partition, this rewriting takes the form

$$F_{\mathbb{H}}\left(\mathrm{e}^{-\Lambda}\operatorname{ch}V(\Lambda)\right) = \frac{(q^{2};q^{2})_{\infty}(q^{\kappa/2};q^{\kappa/2})_{\infty}(q^{\kappa};q^{\kappa})_{\infty}^{n-1}}{(q;q)_{\infty}^{n+1}}$$
$$\times \prod_{i=1}^{n} \theta\left(q^{\lambda_{i}+n-i+1};q^{\kappa/2}\right) \prod_{1 \leq i < j \leq n} \theta\left(q^{\lambda_{i}-\lambda_{j}-i+j},q^{\lambda_{i}+\lambda_{j}+2n+2-i-j};q^{\kappa}\right), \quad (3.25)$$

where  $\kappa = 2n + 2\lambda_0 + 2$ .

The earlier product form now arises by recognizing (see e.g., [19, Lemma 2.1]) the right-hand side of (3.15) as

$$e^{-m\Lambda_0} \operatorname{ch} V(m\Lambda_0)$$

upon the identification

$$q = e^{-\alpha_0 - 2\alpha_1 - \dots - 2\alpha_{n-1} - \alpha_n}$$
 and  $x_i = e^{-\alpha_i - \dots - \alpha_{n-1} - \alpha_n/2}$   $(1 \le i \le n).$ 

Since (3.16) corresponds exactly to the principal specialization (3.23), it follows from (3.25) with  $\lambda = (m, 0^n)$ , that

$$F_{\mathbb{H}}(e^{-m\Lambda_{0}} \operatorname{ch} V(m\Lambda_{0})) = \frac{(q^{2};q^{2})_{\infty}(q^{\kappa/2};q^{\kappa/2})_{\infty}(q^{\kappa};q^{\kappa})_{\infty}^{n-1}}{(q;q)_{\infty}^{n+1}} \times \prod_{i=1}^{n} \theta(q^{n-i+1};q^{\kappa/2}) \prod_{1 \le i < j \le n} \theta(q^{j-i},q^{i+j};q^{\kappa}).$$

We should remark that this representation-theoretic approach is not essentially different from our earlier q-series proof. Indeed, the principal specialization formula (3.25) itself is an immediate consequence of the  $\mathbf{D}_{n+1}^{(2)}$  Macdonald identity, and if, instead of the right-hand side of (3.15), we consider the more general

$$e^{-\Lambda} \operatorname{ch} V(\Lambda) = \frac{1}{(q)_{\infty}^{n} \prod_{i=1}^{n} \theta(x_{i}^{2}; q) \prod_{1 \leq i < j \leq n} \theta(x_{i}/x_{j}, x_{i}x_{j}; q)}$$
$$\times \sum_{r \in \mathbb{Z}^{n}} \det_{1 \leq i, j \leq n} \left( (x_{i}q^{r_{i}})^{j-\lambda_{i}-1} - (x_{i}q^{r_{i}})^{2n-j+\lambda_{i}+1} \right) \prod_{i=1}^{n} x_{i}^{\kappa r_{i}+\lambda_{i}-i+1} q^{\frac{1}{2}\kappa r_{i}^{2}-nr_{i}}$$

for  $\kappa = 2n + 2\lambda_0 + 2$ , then all of the steps carried out between (3.15) and (3.22) carry over to this more general setting. The only notable changes are that (3.19) generalizes to

$$\mathcal{M} = \det_{1 \le i,j \le n} \left( \sum_{r \in \mathbb{Z}} \hat{x}_i^{\kappa r + \lambda_i - i + 1} q^{n \kappa r^2 - 2n^2 r} \cdot \left( (\hat{x}_i q^{2nr})^{j - \lambda_i - 1} - (\hat{x}_i q^{2nr})^{2n - j + \lambda_i + 1} \right) \right),$$

and that in (3.20) we have to redefine  $y_i$  as  $q^{\kappa/2-\lambda_{n-i+1}-i}$ , and  $a_{ij}$  as

$$j^{2} - i^{2} + (i - j)(\kappa + 1)/2 + (j - 1/2)\lambda_{n-j+1} - (i - 1/2)\lambda_{n-i+1}$$

## **3.2.3** Proof of Theorem 1.2 (1.8a)

Here we prove (1.8a) by making use of the  $B_n^{(1)}$  Macdonald identity.

Proof of Theorem 1.2(1.8a). Again we iterate (3.14), but this time the variable  $x_n$ , remains unpaired:

$$\lim_{y_1 \to x_1^{-1}} \dots \lim_{y_{n-1} \to x_{n-1}^{-1}} L_m^{(0)}(x_1, y_1, \dots, x_{n-1}, y_{n-1}, x_n) = L_m^{(n-1)}(x_1, \dots, x_n).$$

Therefore, if we replace  $x \mapsto (x_1, y_1, \ldots, x_{n-1}, y_{n-1}, x_n)$  in (3.10) (changing the rank from n to 2n-1) and take the  $y_i \to x_i^{-1}$  limit for  $1 \le i \le n-1$ , we obtain

$$\sum_{\substack{\lambda \\ \lambda_1 \le m}} q^{|\lambda|} P'_{2\lambda} \left( x_1^{\pm}, \dots, x_{n-1}^{\pm}, x_n; q \right)$$

$$= \frac{1}{(q)_{\infty}^{n-1} (q x_n^2)_{\infty} \prod_{i=1}^{n-1} (q x_i^{\pm} x_n, q x_i^{\pm 2})_{\infty} \prod_{1 \le i < j \le n-1} (q x_i^{\pm} x_j^{\pm})_{\infty}}$$

$$\times \sum_{r \in \mathbb{Z}^n} \frac{\Delta_{\mathcal{C}}(xq^r)}{\Delta_{\mathcal{C}}(x)} \prod_{i=1}^n \left( -\frac{x_i^{\kappa}}{x_n} \right)^{r_i} q^{\frac{1}{2}\kappa r_i^2 - \frac{1}{2}(2n-1)r_i} \frac{(x_i x_n)_{r_i}}{(q x_i/x_n)_{r_i}},$$
(3.26)

where  $\kappa = 2m + 2n + 1$ ,  $(ax_i^{\pm})_{\infty} := (ax_i)_{\infty} (ax_i^{-1})_{\infty}$  and

$$(ax_i^{\pm}x_j^{\pm})_{\infty} := (ax_ix_j)_{\infty}(ax_i^{-1}x_j)_{\infty}(ax_ix_j^{-1})_{\infty}(ax_i^{-1}x_j^{-1})_{\infty}.$$

Recalling the comment immediately after (3.13), the summand of (3.26) vanishes unless  $r_n \ge 0$ .

Let  $\hat{x} := (-x_1, \dots, -x_{n-1}, -1)$  and

$$\phi_r = \begin{cases} 1 & \text{if } r = 0 \\ 2 & \text{if } r = 1, 2, \dots \end{cases}$$
(3.27)

Letting  $x_n$  tend to 1 in (3.26), and using

$$\lim_{x_n \to 1} \frac{\Delta_{\mathcal{C}}(xq^r)}{\Delta_{\mathcal{C}}(x)} \prod_{i=1}^n \frac{(x_i x_n)_{r_i}}{(q x_i / x_n)_{r_i}} = \phi_{r_n} \frac{\Delta_{\mathcal{B}}(\hat{x}q^r)}{\Delta_{\mathcal{B}}(\hat{x})},$$

we find that

$$\sum_{\substack{\lambda \\ \lambda_1 \le m}} q^{|\lambda|} P'_{2\lambda} \left( x_1^{\pm}, \dots, x_{n-1}^{\pm}, 1; q \right)$$

$$= \frac{1}{(q)_{\infty}^n \prod_{i=1}^{n-1} (q x_i^{\pm}, q x_i^{\pm 2})_{\infty} \prod_{1 \le i < j \le n-1} (q x_i^{\pm} x_j^{\pm})_{\infty}}$$

$$\times \sum_{r_1, \dots, r_{n-1} = -\infty}^{\infty} \sum_{r_n = 0}^{\infty} \phi_{r_n} \frac{\Delta_{\mathrm{B}}(\hat{x}q^r)}{\Delta_{\mathrm{B}}(\hat{x})} \prod_{i=1}^n \hat{x}_i^{\kappa r_i} q^{\frac{1}{2}\kappa r_i^2 - \frac{1}{2}(2n-1)r_i}.$$

It is easily checked that the summand on the right (without the factor  $\phi_{r_n}$ ) is invariant under the variable change  $r_n \mapsto -r_n$ . Using the elementary relations

$$\theta(-1;q) = 2(-q)_{\infty}^{2}, \quad (-q)_{\infty}(q;q^{2})_{\infty} = 1, \quad \theta(z,-z;q)\theta(qz^{2};q^{2}) = \theta(z^{2}),$$
(3.28)

we can then simplify the above to obtain

$$\sum_{\substack{\lambda \\ \lambda_{1} \leq m}} q^{|\lambda|} P'_{2\lambda} \left( x_{1}^{\pm}, \dots, x_{n-1}^{\pm}, 1; q \right)$$

$$= \frac{1}{(q)_{\infty}^{n} \prod_{i=1}^{n} \theta(\hat{x}_{i}; q) \theta(q \hat{x}_{i}^{2}; q^{2}) \prod_{1 \leq i < j \leq n} \theta(\hat{x}_{i} / \hat{x}_{j}, \hat{x}_{i} \hat{x}_{j}; q)}$$

$$\times \sum_{r \in \mathbb{Z}^{n}} \Delta_{\mathrm{B}}(\hat{x}q^{r}) \prod_{i=1}^{n} \hat{x}_{i}^{\kappa r_{i} - i + 1} q^{\frac{1}{2} \kappa r_{i}^{2} - \frac{1}{2}(2n - 1)r_{i}}.$$
(3.29)

The remainder of the proof is similar to that of (1.10). We make the simultaneous substitutions

$$q \mapsto q^{2n-1}, \qquad x_i \mapsto q^{n-i} \ (1 \le i \le n),$$

$$(3.30)$$

so that from here on  $\hat{x}_i := -q^{n-i}$ . By the identity

$$(q^{2n-1}; q^{2n-1})_{\infty}^{n} \prod_{i=1}^{n} \theta(-q^{n-i}; q^{2n-1}) \theta(q^{2n-2i+1}; q^{4n-2}) \\ \times \prod_{1 \le i < j \le n} \theta(q^{j-i}, q^{2n-i-j}; q^{2n-1}) = 2(q)_{\infty}^{n}$$

and (3.5), we find that

$$\sum_{\substack{\lambda\\\lambda_1\leq m}} q^{|\lambda|} P_{2\lambda}(1,q,q^2,\ldots;q^{2n-1}) = \frac{\mathcal{M}}{2(q)_{\infty}^n},$$

where we have that

$$\mathcal{M} := \sum_{r \in \mathbb{Z}^n} \Delta_{\mathrm{B}} \left( \hat{x} q^{(2n-1)r} \right) \prod_{i=1}^n \hat{x}_i^{\kappa r_i - i + 1} q^{\frac{1}{2}(2n-1)\kappa r_i^2 - \frac{1}{2}(2n-1)^2 r_i}.$$

By (3.21) and multilinearity,  $\mathcal{M}$  can be rewritten in the form

$$\mathcal{M} = \det_{1 \le i,j \le n} \left( \sum_{r \in \mathbb{Z}} \hat{x}_i^{\kappa r - i + 1} q^{\frac{1}{2}(2n-1)\kappa r^2 - \frac{1}{2}(2n-1)^2 r} \times \left( \left( \hat{x}_i q^{(2n-1)r} \right)^{j-1} - \left( \hat{x}_i q^{(2n-1)r} \right)^{2n-j} \right) \right).$$

Following the same steps that led from (3.19) to (3.20), we obtain

$$\mathcal{M} = \det_{1 \le i,j \le n} \left( (-1)^{i-j} q^{b_{ij}} \sum_{r \in \mathbb{Z}} (-1)^r y_i^{(2n-1)r-i+1} q^{(2n-1)\kappa\binom{r}{2}} \times \left( (y_i q^{\kappa r})^{j-1} - (y_i q^{\kappa r})^{2n-j} \right) \right), \quad (3.31)$$

where

$$y_i = q^{\frac{1}{2}(\kappa+1)-i}$$
 and  $b_{ij} := j^2 - i^2 + \frac{1}{2}(i-j)(\kappa+3).$  (3.32)

Again, the factor  $(-1)^{i-j}q^{b_{ij}}$  does not contribute, and so (3.21) then gives

$$\mathcal{M} = \sum_{r \in \mathbb{Z}^n} \Delta_{\mathrm{B}}(y_i q^{\kappa r}) \prod_{i=1}^n (-1)^{r_i} y_i^{(2n-1)r_i - i + 1} q^{(2n-1)\kappa \binom{r_i}{2}}.$$

To complete the proof, we apply the following variant of the  $\mathbf{B}_n^{(1)}$  Macdonald

 $identity^2$ 

$$\sum_{r \in \mathbb{Z}^n} \Delta_{\mathrm{B}}(xq^r) \prod_{i=1}^n (-1)^{r_i} x_i^{(2n-1)r_i - i + 1} q^{(2n-1)\binom{r_i}{2}}$$
$$= 2(q)_{\infty}^n \prod_{i=1}^n \theta(x_i; q) \prod_{1 \le i < j \le n} \theta(x_i/x_j, x_i x_j; q), \quad (3.33)$$

with  $(q, x) \mapsto (q^{\kappa}, y)$ .

Identity (1.8a) can be understood representation-theoretically, but this time the relevant Kac–Moody algebra is  $\mathbb{A}_{2n}^{(2)}$ . According to [19, Lemma 2.3] the right-hand side of (3.29), with  $\hat{x}$  interpreted as

$$\hat{x}_i = \mathrm{e}^{-\alpha_0 - \dots - \alpha_{n-i}} \quad (1 \le i \le n)$$

and q as

$$q = e^{-2\alpha_0 - \dots - 2\alpha_{n-1} - \alpha_n}, \qquad (3.34)$$

is the  $\mathbb{A}_{2n}^{(2)}$  character

$$e^{-m\Lambda_n} \operatorname{ch} V(m\Lambda_n).$$

The substitution (3.30) corresponds to

$$e^{-\alpha_0} \mapsto -1$$
 and  $e^{-\alpha_i} \mapsto q \ (1 \le i \le n).$  (3.35)

Denoting this by F, we have the general specialization formula

$$F\left(\mathrm{e}^{-\Lambda}\operatorname{ch} V(\Lambda)\right) = \frac{\left(q^{\kappa}; q^{\kappa}\right)_{\infty}^{n}}{\left(q\right)_{\infty}^{n}} \prod_{i=1}^{n} \theta\left(q^{\lambda_{0}-\lambda_{i}+i}; q^{\kappa}\right) \prod_{1 \leq i < j \leq n} \theta\left(q^{\lambda_{i}-\lambda_{j}-i+j}, q^{\lambda_{i}+\lambda_{j}-i-j+2n+1}; q^{\kappa}\right),$$

$$(3.36)$$

<sup>&</sup>lt;sup>2</sup>The actual  $B_n^{(1)}$  Macdonald identity has the restriction  $|r| \equiv 0 \pmod{2}$  in the sum over  $r \in \mathbb{Z}^n$ , which eliminates the factor 2 on the right. To prove the form used here it suffices to take the  $a_1, \ldots, a_{2n-1} \to 0$  and  $a_{2n} \to -1$  limit in Gustafson's multiple  $_6\psi_6$  summation for the affine root system  $\mathbb{A}_{2n-1}^{(2)}$ , see [154].

where  $\kappa = 2n + 2\lambda_0 + 1$  and

$$\Lambda = 2\lambda_n\Lambda_0 + (\lambda_{n-1} - \lambda_n)\Lambda_1 + \dots + (\lambda_1 - \lambda_2)\Lambda_{n-1} + (\lambda_0 - \lambda_1)\Lambda_n$$

for  $\lambda = (\lambda_0, \lambda_1, \dots, \lambda_n)$  a partition. If we let  $\lambda = (m, 0^n)$  (so that  $\Lambda = m\Lambda_n$ ), then this is in accordance with (1.8a).

## **3.2.4 Proof of Theorem 1.2** (1.8b)

Here we prove the companion result to (1.8a).

Proof of Theorem 1.2 (1.8b). In (3.26) we set  $x_n = q^{1/2}$  so that

$$\sum_{\substack{\lambda \\ \lambda_1 \le m}} q^{|\lambda|} P'_{2\lambda} \left( x_1^{\pm}, \dots, x_{n-1}^{\pm}, q^{1/2}; q \right)$$

$$= \frac{1}{(q)_{\infty}^{n-1} (q^2)_{\infty} \prod_{i=1}^{n-1} (q^{3/2} x_i^{\pm}, q x_i^{\pm 2})_{\infty} \prod_{1 \le i < j \le n-1} (q x_i^{\pm} x_j^{\pm})_{\infty}}}{\times \sum_{r_1, \dots, r_{n-1} = -\infty}^{\infty} \sum_{r_n = 0}^{\infty} \frac{\Delta_{\mathcal{C}}(\hat{x}q^r)}{\Delta_{\mathcal{C}}(\hat{x})} \prod_{i=1}^{n} (-1)^{r_i} \hat{x}_i^{\kappa r_i} q^{\frac{1}{2} \kappa r_i^2 - nr_i},}$$

where  $\kappa = 2m + 2n + 1$  and  $\hat{x} = (x_1, \dots, x_{n-1}, q^{1/2})$ . The  $r_n$ -dependent part of the summand is

$$(-1)^{r_n} q^{\kappa \binom{r_n+1}{2} - nr_n} \frac{1 - q^{2r_n+1}}{1 - q} \prod_{i=1}^{n-1} \frac{x_i q^{r_i} - q^{r_n+1/2}}{x_i - q^{1/2}} \cdot \frac{x_i q^{r_n+r_i+1/2} - 1}{x_i q^{1/2} - 1},$$

which is readily checked to be invariant under the substitution  $r_n \mapsto -r_n - 1$ . Hence

$$\sum_{\substack{\lambda \\ \lambda_1 \le m}} q^{|\lambda|} P'_{2\lambda} (x_1^{\pm}, \dots, x_{n-1}^{\pm}, q^{1/2}; q)$$
  
= 
$$\frac{1}{2(q)_{\infty}^n \prod_{i=1}^{n-1} (-1)\theta(q^{1/2}x_i, x_i^2; q) \prod_{1 \le i < j \le n-1} \theta(x_i/x_j, x_i x_j; q)}$$
  
× 
$$\sum_{r \in \mathbb{Z}^n} \Delta_{\mathcal{C}} (\hat{x}q^r) \prod_{i=1}^n (-1)^{r_i} \hat{x}_i^{\kappa r_i - i} q^{\frac{1}{2}\kappa r_i^2 - nr_i + \frac{1}{2}}.$$

Our next step is to replace  $x_i \mapsto x_{n-i+1}$  and  $r_i \mapsto r_{n-i+1}$ . By  $\theta(x;q) = -x\theta(x^{-1};q)$  and (3.28), this leads to

$$\sum_{\substack{\lambda\\\lambda_{1} \leq m}} q^{|\lambda|} P_{2\lambda}'(q^{1/2}, x_{2}^{\pm}, \dots, x_{n}^{\pm}; q)$$

$$= \frac{1}{(q)_{\infty}^{n} \prod_{i=1}^{n} \theta(-q^{1/2} \hat{x}_{i}; q) \theta(\hat{x}_{i}^{2}; q^{2}) \prod_{1 \leq i < j \leq n} \theta(\hat{x}_{i}/\hat{x}_{j}, \hat{x}_{i} \hat{x}_{j}; q)}$$

$$\times \sum_{r \in \mathbb{Z}^{n}} \Delta_{C}(\hat{x}q^{r}) \prod_{i=1}^{n} (-1)^{r_{i}} \hat{x}_{i}^{\kappa r_{i} - i + 1} q^{\frac{1}{2} \kappa r_{i}^{2} - nr_{i}},$$
(3.37)

where now  $\hat{x} = (q^{1/2}, x_2, \dots, x_n)$ . Again we are at the point where we can specialize, letting

$$q \mapsto q^{2n-1}, \qquad x_i \mapsto q^{n-i+1/2} =: \hat{x}_i \ (1 \le i \le n).$$
 (3.38)

This is consistent, since  $x_1 = q^{1/2} \mapsto q^{n-1/2}$ . By the identity

$$\begin{split} (q^{2n-1};q^{2n-1})_{\infty}^n \prod_{i=1}^n \theta(-q^{2n-i};q^{2n-1})\theta(q^{2n-2i+1};q^{4n-2}) \\ \times \prod_{1 \le i < j \le n} \theta(q^{j-i},q^{2n-i-j+1};q^{2n-1}) = 2(q)_{\infty}^n, \end{split}$$

we obtain

$$\sum_{\substack{\lambda\\\lambda_1\leq m}} q^{2|\lambda|} P_{2\lambda}(1,q,q^2,\ldots;q^{2n-1}) = \frac{\mathcal{M}}{2(q)_{\infty}^n},$$

where

$$\mathcal{M} := \sum_{r \in \mathbb{Z}^n} \Delta_{\mathcal{C}}(\hat{x}q^{(2n-1)r}) \prod_{i=1}^n (-1)^{r_i} \hat{x}_i^{\kappa r_i - i + 1} q^{\frac{1}{2}(2n-1)\kappa r_i^2 - (2n-1)nr_i}.$$

Expressing  $\mathcal{M}$  in determinantal form using (3.18) yields

$$\mathcal{M} = \det_{1 \le i,j \le n} \left( \sum_{r \in \mathbb{Z}} (-1)^r \hat{x}_i^{\kappa r - i + 1} q^{\frac{1}{2}(2n - 1)\kappa r^2 - (2n - 1)nr} \times \left( (\hat{x}_i q^{(2n - 1)r})^{j - 1} - (\hat{x}_i q^{(2n - 1)r})^{2n - j + 1} \right) \right).$$

We now replace  $(i, j) \mapsto (j, i)$  and, viewing the resulting determinant as of the form det  $(\sum_{r} u_{ij;r} - \sum_{r} v_{ij;r})$ , we change the summation index  $r \mapsto -r$ in the sum over  $u_{ij;r}$ . The expression for  $\mathcal{M}$  we obtain is exactly (3.31) except that  $(-1)^{i-j}q^{b_{ij}}$  is replaced by  $q^{c_{ij}}$  and  $y_i$  is given by  $q^{n-i+1}$  instead of  $q^{(\kappa+1)/2-i}$ . Following the previous proof results in (1.8b).

To interpret (1.8b) in terms of  $\mathbb{A}_{2n}^{(2)}$ , we note that by [19, Lemma 2.2] the right-hand side of (3.37) in which  $\hat{x}$  is interpreted as

$$\hat{x}_i = -q^{1/2} e^{\alpha_0 + \dots + \alpha_{i-1}} \quad (1 \le i \le n)$$

(and q again as (3.34)) corresponds to the  $\mathbb{A}_{2n}^{(2)}$  character

$$e^{-2m\Lambda_0} \operatorname{ch} V(2m\Lambda_0).$$

The specialization (3.38) is then again consistent with (3.35). From (3.36) with  $\lambda = (m^{n+1})$ , the first product-form on the right of (1.8b) immediately follows. By level-rank duality, we can also identify (1.8b) as a specialization of the  $\mathbb{A}_{2m}^{(2)}$  character  $e^{-2n\Lambda_0} \operatorname{ch} V(2n\Lambda_0)$ .

### 3.2.5 Proof of Theorem 1.5

This proof, which uses the  $\mathbf{D}_n^{(1)}$  Macdonald identity, is the most complicated of the four.

Proof of Theorem 1.5. Once again we iterate (3.14), but now both  $x_{n-1}$  and  $x_n$  remain unpaired:

$$\lim_{y_1 \to x_1^{-1}} \dots \lim_{y_{n-2} \to x_{n-2}^{-1}} L_m^{(0)}(x_1, y_1, \dots, x_{n-2}, y_{n-2}, x_{n-1}, x_n) = L_m^{(n-2)}(x_1, \dots, x_n).$$

Accordingly, if we replace  $x \mapsto (x_1, y_1, \ldots, x_{n-2}, y_{n-2}, x_{n-1}, x_n)$  in (3.10) (thereby changing the rank from n to 2n - 2) and take the  $y_i \to x_i^{-1}$  limit, for  $1 \le i \le n - 2$ , we obtain

$$\begin{split} \sum_{\substack{\lambda \\ \lambda_1 \le m}} q^{|\lambda|} P'_{2\lambda} \left( x_1^{\pm}, \dots, x_{n-2}^{\pm}, x_{n-1}, x_n; q \right) \\ &= \frac{1}{(q)_{\infty}^{n-2} (q x_{n-1}^2, q x_{n-1} x_n, q x_n^2)_{\infty}} \\ &\times \frac{1}{\prod_{i=1}^{n-2} (q x_i^{\pm 2}, q x_i^{\pm} x_{n-1}, q x_i^{\pm} x_n)_{\infty} \prod_{1 \le i < j \le n-2} (q x_i^{\pm} x_j^{\pm})_{\infty}} \\ &\times \sum_{r \in \mathbb{Z}^n} \frac{\Delta_{\mathcal{C}} (x q^r)}{\Delta_{\mathcal{C}} (x)} \prod_{i=1}^n \left( \frac{x_i^{\kappa}}{x_{n-1} x_n} \right)^{r_i} q^{\frac{1}{2} \kappa r_i^2 - (n-1)r_i} \frac{(x_i x_{n-1}, x_i x_n)_{r_i}}{(q x_i / x_{n-1}, q x_i / x_n)_{r_i}}, \end{split}$$

where  $\kappa = 2m + 2n$ . It is important to note that the summand vanishes unless  $r_{n-1}$  and  $r_n$  are both nonnegative. Next we let  $(x_{n-1}, x_n)$  tend to  $(q^{1/2}, 1)$  using

$$\lim_{(x_{n-1},x_n)\to(q^{1/2},1)}\frac{\Delta_{\mathcal{C}}(xq^r)}{\Delta_{\mathcal{C}}(x)}\prod_{i=1}^n\frac{(x_ix_{n-1},x_ix_n)_{r_i}}{(qx_i/x_{n-1},qx_i/x_n)_{r_i}}=\phi_{r_n}\frac{\Delta_{\mathcal{B}}(\hat{x}q^r)}{\Delta_{\mathcal{B}}(\hat{x})},$$

with  $\phi_r$  as in (3.27) and  $\hat{x} := (-x_1, \dots, -x_{n-2}, -q^{1/2}, -1)$ . Hence we find that

$$\begin{split} \sum_{\substack{\lambda\\\lambda_1 \le m}} q^{|\lambda|} P_{2\lambda}'(x_1^{\pm}, \dots, x_{n-2}^{\pm}, q^{1/2}, 1; q) \\ &= \frac{1}{(q)_{\infty}^{n-1}(q^{3/2}; q^{1/2})_{\infty} \prod_{i=1}^{n-2} (qx_i^{\pm}; q^{1/2})_{\infty} (qx_i^{\pm 2})_{\infty} \prod_{1 \le i < j \le n-2} (qx_i^{\pm}x_j^{\pm})_{\infty}} \\ &\times \sum_{r_1, \dots, r_{n-2} = -\infty}^{\infty} \sum_{r_{n-1}, r_n = 0}^{\infty} \phi_{r_n} \frac{\Delta_{\mathcal{B}}(\hat{x}q^r)}{\Delta_{\mathcal{B}}(\hat{x})} \prod_{i=1}^n \hat{x}_i^{\kappa r_i} q^{\frac{1}{2}\kappa r_i^2 - \frac{1}{2}(2n-1)r_i}. \end{split}$$

Since the summand (without the factor  $\phi_{r_n}$ ) is invariant under the variable change  $r_n \mapsto -r_n$ , as well as the change  $r_{n-1} \mapsto -r_{n-1} - 1$ , we can rewrite

this as

$$\begin{split} \sum_{\substack{\lambda \\ \lambda_1 \leq m}} q^{|\lambda|} P'_{2\lambda} \big( x_1^{\pm}, \dots, x_{n-2}^{\pm}, q^{1/2}, 1; q \big) \\ = & \frac{1}{(q)_{\infty}^{n-1} (q^{1/2}; q^{1/2})_{\infty} \prod_{i=1}^n \theta(\hat{x}_i; q^{1/2}) \prod_{1 \leq i < j \leq n} \theta(\hat{x}_i / \hat{x}_j, \hat{x}_i \hat{x}_j)} \\ & \times \sum_{r \in \mathbb{Z}^n} \Delta_{\mathcal{B}}(\hat{x}q^r) \prod_{i=1}^n \hat{x}_i^{\kappa r_i - i + 1} q^{\frac{1}{2} \kappa r_i^2 - \frac{1}{2} (2n-1) r_i}, \end{split}$$

where, once again, we have used (3.28) to clean up the infinite products. Before we can carry out the usual specialization, we need to relabel  $x_1, \ldots, x_{n-2}$ as  $x_2, \ldots, x_{n-1}$  and, accordingly, we redefine  $\hat{x}$  as

$$(-q^{1/2}, -x_2, \ldots, -x_{n-1}, -1).$$

For  $n \geq 2$ , we then find that

$$\sum_{\substack{\lambda\\\lambda_{1} \leq m}} q^{|\lambda|} P_{2\lambda}'(q^{1/2}, x_{2}^{\pm}, \dots, x_{n-1}^{\pm}, 1; q)$$

$$= \frac{1}{(q)_{\infty}^{n-1}(q^{1/2}; q^{1/2})_{\infty} \prod_{i=1}^{n} \theta(\hat{x}_{i}; q^{1/2}) \prod_{1 \leq i < j \leq n} \theta(\hat{x}_{i}/\hat{x}_{j}, \hat{x}_{i}\hat{x}_{j})}$$

$$\times \sum_{r \in \mathbb{Z}^{n}} \Delta_{\mathrm{B}}(\hat{x}q^{r}) \prod_{i=1}^{n} \hat{x}_{i}^{\kappa r_{i} - i + 1} q^{\frac{1}{2}\kappa r_{i}^{2} - \frac{1}{2}(2n-1)r_{i}}.$$
(3.39)

We are now ready to make the substitutions

$$q \mapsto q^{2n-2}, \qquad x_i \mapsto q^{n-i} \ (2 \le i \le n-1),$$
 (3.40)

so that  $\hat{x}_i := -q^{n-i}$  for  $1 \le i \le n$ . By the identity

$$(q^{2n-2}; q^{2n-2})_{\infty}^{n-1}(q^{n-1}; q^{n-1})_{\infty} \prod_{i=1}^{n} \theta(-q^{n-i}; q^{n-1}) \\ \times \prod_{1 \le i < j \le n} \theta(q^{j-i}, q^{2n-i-j}; q^{2n-2}) = 4(q^2; q^2)_{\infty}(q)_{\infty}^{n-1}$$

and (3.5), we obtain

$$\sum_{\substack{\lambda \\ \lambda_1 \le m}} q^{2|\lambda|} P_{2\lambda} (1, q, q^2, \dots; q^{2n-3}) = \frac{\mathcal{M}}{4(q^2; q^2)_{\infty}(q)_{\infty}^{n-1}},$$

where  $\mathcal{M}$  is given by

$$\mathcal{M} := \sum_{r \in \mathbb{Z}^n} \Delta_{\mathrm{B}}(\hat{x}q^{2(n-1)r}) \prod_{i=1}^n \hat{x}_i^{\kappa r_i - i + 1} q^{(n-1)\kappa r_i^2 - (n-1)(2n-1)r_i}.$$

By the  $B_n$  determinant (3.21), we find that

$$\mathcal{M} = \det_{1 \le i,j \le n} \left( \sum_{r \in \mathbb{Z}} \hat{x}_i^{\kappa r - i + 1} q^{(n-1)\kappa r^2 - (n-1)(2n-1)r} \cdot \left( \left( \hat{x}_i q^{2(n-1)r} \right)^{j-1} - \left( \hat{x}_i q^{2(n-1)r} \right)^{2n-j} \right) \right).$$

By the same substitutions that transformed (3.19) into (3.20), we obtain

$$\mathcal{M} = \det_{1 \le i,j \le n} \left( (-1)^{i-j} q^{b_{ij}} \sum_{r \in \mathbb{Z}} y_i^{2(n-1)r-i+1} q^{2(n-1)\kappa\binom{r}{2}} \cdot \left( \left( y_i q^{\kappa r} \right)^{j-1} + \left( y_i q^{\kappa r} \right)^{2n-j-1} \right) \right),$$

where  $y_i$  and  $b_{ij}$  are as in (3.32). Recalling the Weyl denominator formula for  $\mathbf{D}_n$  [198]

$$\frac{1}{2} \det_{1 \le i,j \le n} \left( x_i^{j-1} + x_i^{2n-j-1} \right) = \prod_{1 \le i < j \le n} (x_i - x_j) (x_i x_j - 1) =: \Delta_{\mathbf{D}}(x)$$

we can rewrite  $\mathcal{M}$  in the form

$$\mathcal{M} = 2 \sum_{r \in \mathbb{Z}^n} \Delta_{\mathbf{D}}(xq^r) \prod_{i=1}^n y_i^{2(n-1)r_i - i + 1} q^{2(n-1)\kappa\binom{r_i}{2}}.$$

Taking the  $a_1, \ldots, a_{2n-2} \to 0$ ,  $a_{2n-1} \to 1$  and  $a_{2n} \to -1$  limit in Gustafson's multiple  $_6\psi_6$  summation for the affine root system  $\mathbb{A}_{2n-1}^{(2)}$  [154] leads to the

following variant of the  $\mathbf{D}_n^{(1)}$  Macdonald identity<sup>3</sup>

$$\sum_{r \in \mathbb{Z}^n} \Delta_{\mathbf{D}}(xq^r) \prod_{i=1}^n x_i^{2(n-1)r_i - i + 1} q^{2(n-1)\binom{r_i}{2}} = 2(q)_{\infty}^n \prod_{1 \le i < j \le n} \theta(x_i/x_j, x_i x_j; q).$$

This implies the claimed product form for  $\mathcal{M}$  and completes our proof.  $\Box$ 

Identity (1.11) has a representation-theoretic interpretation. By [19, Lemma 2.4], the right-hand side of (3.39) in which  $\hat{x}$  is interpreted as

$$\hat{x}_i = \mathrm{e}^{-\alpha_i - \dots - \alpha_n} \ (1 \le i \le n)$$

and q as

$$q = e^{-2\alpha_0 - \dots - 2\alpha_n}$$

yields the  $\mathbf{D}_{n+1}^{(2)}$  character

$$e^{-2m\Lambda_0} \operatorname{ch} V(2m\Lambda_0).$$

The specialization (3.40) then corresponds to

$$e^{-\alpha_0}, e^{-\alpha_n} \mapsto -1$$
 and  $e^{-\alpha_i} \mapsto q \ (2 \le i \le n-1).$ 

Denoting this by F, we have

$$F\left(\mathrm{e}^{-\Lambda}\operatorname{ch} V(\Lambda)\right) = \frac{(q^{\kappa}; q^{\kappa})_{\infty}^{n}}{(q^{2}; q^{2})_{\infty}(q)_{\infty}^{n-1}} \prod_{1 \le i < j \le n} \theta\left(q^{\lambda_{i} - \lambda_{j} - i + j}, q^{\lambda_{i} + \lambda_{j} - i - j + 2n + 1}; q^{\kappa}\right),$$

where  $\kappa = 2n + 2\lambda_0$  and

$$\Lambda = 2(\lambda_0 - \lambda_1)\Lambda_0 + (\lambda_1 - \lambda_2)\Lambda_1 + \dots + (\lambda_{n-1} - \lambda_n)\Lambda_{n-1} + 2\lambda_n\Lambda_n,$$

for  $\lambda = (\lambda_0, \lambda_1, \dots, \lambda_n)$  a partition or half-partition (i.e., all  $\lambda_i \in \mathbb{Z} + 1/2$ ). For  $\lambda = (m, 0^n)$  this agrees with (1.11).

<sup>&</sup>lt;sup>3</sup>As in the  $B_n^{(1)}$  case, the actual  $\mathbf{D}_n^{(1)}$  Macdonald identity contains the restriction  $|r| \equiv 0 \pmod{2}$  on the sum over r.

# 3.3 Proof of Theorem 1.7

For integers k and m, where  $0 \le k \le m$ , we denote the the nearly-rectangular partition  $(\underbrace{m, \ldots, m}_{r \text{ times}}, k)$  as  $(m^r, k)$ . Using these partitions, we have the following "limiting" Rogers-Ramanujan-type identities, which imply Theorem 1.7 when k = 0 or k = m.

**Theorem 3.5** ( $\mathbb{A}_{n-1}^{(1)}$  RR and AG identities). If m and n are positive integers and  $0 \leq k \leq m$ , then we have

$$\lim_{r \to \infty} q^{-m\binom{r}{2}-kr} Q_{(m^r,k)}(1,q,q^2,\dots;q^n) = \frac{(q^n;q^n)_{\infty}(q^\kappa;q^\kappa)_{\infty}^{n-1}}{(q)_{\infty}^n} \cdot \prod_{i=1}^{n-1} \theta(q^{i+k};q^\kappa) \cdot \prod_{1 \le i < j \le n-1} \theta(q^{j-i};q^\kappa), \quad (3.41)$$

where  $\kappa = m + n$ .

*Remark.* A similar calculation when  $k \ge m$  gives

$$\lim_{r \to \infty} q^{-m\binom{r+1}{2}} Q_{(k,m^r)}(1,q,q^2,\dots;q^n) = {\binom{k-m+n-1}{n-1}}_q \frac{(q^n;q^n)_{\infty}(q^\kappa;q^\kappa)_{\infty}^{n-1}}{(q)_{\infty}^n} \prod_{1 \le i < j \le n} \theta(q^{j-i};q^\kappa)$$

Proof of Theorem 3.5. It suffices to prove the identity for  $0 \le k < m$ , and below assume that k satisfies this inequality.

The following identity for the modified Hall–Littlewood polynomials indexed by near-rectangular partitions is a special case of [19, Corollary 3.2]:

$$Q'_{(m^r,k)}(x;q) = (q)_r(q)_1 \sum_{\substack{u \in \mathbb{Z}_+^n \\ |u|=r+1}} \sum_{\substack{v \in \mathbb{Z}_+^n \\ |v|=r}} \prod_{i=1}^n x_i^{ku_i + (m-k)v_i} q^{k\binom{u_i}{2} + (m-k)\binom{v_i}{2}} \times \prod_{i,j=1}^n \frac{(qx_i/x_j)_{u_i-u_j}}{(qx_i/x_j)_{u_i-v_j}} \cdot \frac{(qx_i/x_j)_{v_i-v_j}}{(qx_i/x_j)_{v_i}}$$

It is enough to compute the limit on the left-hand side of (3.41) for r a multiple of n. Hence we replace r by nr in the above expression, and then shift  $u_i \mapsto u_i + r$  and  $v_i \mapsto v_i + r$ , for all  $1 \le i \le n$ , to obtain

$$Q'_{(m^{nr},k)}(x;q) = (x_1 \cdots x_n)^{mr} q^{mn\binom{r}{2} + kr} (q)_{nr} (q)_1$$

$$\times \sum_{\substack{u \in \mathbb{Z}^n \\ |u|=1}} \sum_{\substack{v \in \mathbb{Z}^n \\ |v|=0}} \prod_{i=1}^n x_i^{ku_i + (m-k)v_i} q^{k\binom{u_i}{2} + (m-k)\binom{v_i}{2}} \prod_{i,j=1}^n \frac{(qx_i/x_j)_{u_i-u_j}}{(qx_i/x_j)_{u_i-v_j}} \cdot \frac{(qx_i/x_j)_{v_i-v_j}}{(qx_i/x_j)_{r+v_i}}$$

Since the summand vanishes unless  $u_i \ge v_i$  for all i and |u| = |v| + 1, it follows that  $u = v + \epsilon_{\ell}$ , for some  $\ell = 1, \ldots, n$ , where  $(\epsilon_{\ell})_i = \delta_{\ell i}$ . Hence we find that

$$Q'_{(m^{nr},k)}(x;q) = (x_1 \cdots x_n)^{mr} q^{mn\binom{r}{2} + kr} (q)_{nr}$$

$$\times \sum_{\substack{v \in \mathbb{Z}^n \\ |v|=0}} \prod_{i=1}^n x_i^{mv_i} q^{m\binom{v_i}{2}} \prod_{i,j=1}^n \frac{(qx_i/x_j)_{v_i-v_j}}{(qx_i/x_j)_{r+v_i}} \sum_{\ell=1}^n (x_\ell q^{v_\ell})^k \prod_{\substack{i=1 \\ i \neq k}}^n \frac{1}{1 - q^{v_i - v_\ell} x_i/x_\ell}$$

Next we use

$$\prod_{i,j=1}^{n} (qx_i/x_j)_{v_i-v_j} = \frac{\Delta(xq^v)}{\Delta(x)} (-1)^{(n-1)|v|} q^{-\binom{|v|}{2}} \prod_{i=1}^{n} x_i^{nv_i-|v|} q^{\binom{v_i}{2}+(i-1)v_i},$$

where  $\Delta(x) := \prod_{1 \le i < j \le n} (1 - x_i/x_j)$ , and

$$\sum_{\ell=1}^{n} x_{\ell}^{k} \prod_{\substack{i=1\\i\neq k}}^{n} \frac{1}{1 - x_{i}/x_{\ell}} = \sum_{1 \le i_{1} \le i_{2} \le \dots \le i_{k} \le n} x_{i_{1}} x_{i_{2}} \cdots x_{i_{k}} = h_{k}(x) = s_{(k)}(x),$$

where  $h_k$  and  $s_\lambda$  are the complete symmetric and Schur function, respectively. Thus we have

$$Q'_{(m^{nr},k)}(x;q) = (x_1 \cdots x_n)^{mr} q^{mn\binom{r}{2} + kr}(q)_{nr} \\ \times \sum_{\substack{v \in \mathbb{Z}^n \\ |v|=0}} s_{(k)}(xq^v) \frac{\Delta(xq^v)}{\Delta(x)} \prod_{i=1}^n x_i^{\kappa v_i} q^{\frac{1}{2}\kappa v_i^2 + iv_i} \prod_{i,j=1}^n \frac{1}{(qx_i/x_j)_{r+v_i}},$$

where  $\kappa := m + n$ . Note that the summand vanishes unless  $v_i \ge -r$  for all *i*. This implies the limit

$$\lim_{r \to \infty} q^{-mn\binom{r}{2} - kr} \frac{Q'_{(m^{nr},k)}(x;q)}{(x_1 \cdots x_n)^{mr}} = \frac{1}{(q)_{\infty}^{n-1} \prod_{1 \le i < j \le n} \theta(x_i/x_j;q)} \sum_{\substack{v \in \mathbb{Z}^n \\ |v| = 0}} s_{(k)}(xq^v) \Delta(xq^v) \prod_{i=1}^n x_i^{\kappa v_i} q^{\frac{1}{2}\kappa v_i^2 + iv_i}.$$

The expression on the right is exactly the Weyl–Kac formula for the level- $M_{n-1}^{(1)}$  character [183]

$$e^{-\Lambda} \operatorname{ch} V(\Lambda), \quad \Lambda = (m-k)\Lambda_0 + k\Lambda_1,$$

provided we identify

$$q = e^{-\alpha_0 - \alpha_1 - \dots - \alpha_{n-1}}$$
 and  $x_i / x_{i+1} = e^{-\alpha_i}$   $(1 \le i \le n-1).$ 

Hence

$$\lim_{r \to \infty} q^{-mn\binom{r}{2}-kr} \frac{Q'_{(m^{nr},k)}(x;q)}{(x_1 \cdots x_n)^{mr}} = \mathrm{e}^{-\Lambda} \mathrm{ch} \, V(\Lambda),$$

with  $\Lambda$  as above. For m = 1 and k = 0 this was obtained in [187] by more elementary means. The simultaneous substitutions  $q \mapsto q^n$  and  $x_i \mapsto q^{n-i}$ correspond to the principal specialization (3.23). From (3.24) we can then read off the product form claimed in (3.41).

# 3.4 Siegel Functions

The normalizations for the series  $\Phi_*$  were chosen so that the resulting q-series are modular functions on the congruence subgroups  $\Gamma(N)$ , where

$$\Gamma(N) := \left\{ \left(\begin{smallmatrix} a & b \\ c & d \end{smallmatrix}\right) \in SL_2(\mathbb{Z}) : a \equiv d \equiv 1 \pmod{N}, b \equiv c \equiv 0 \pmod{N} \right\}.$$

These groups act on  $\mathbb{H}$ , the upper-half of the complex plane, by  $\gamma \tau := \frac{a\tau+b}{c\tau+d}$ , where  $\gamma = \begin{pmatrix} a & b \\ c & d \end{pmatrix}$ . If f is a meromorphic function on  $\mathbb{H}$  and  $\gamma \in SL_2(Z)$ , then we define

$$(f|_k\gamma)(\tau) := (c\tau + d)^{-k} f(\gamma\tau).$$

Modular functions are meromorphic functions which are invariant with respect to this action. More precisely, a meromorphic function f on  $\mathbb{H}$  is a modular function on  $\Gamma(N)$  if for every  $\gamma \in \Gamma(N)$  we have

$$f(\gamma \tau) = (f|_0 \gamma)(\tau) = f(\tau).$$

The set of such functions form a field. We let  $\mathcal{F}_N$  denote the canonical subfield of those modular functions on  $\Gamma(N)$  whose Fourier expansions are defined over  $\mathbb{Q}(\zeta_N)$ , where  $\zeta_N := e^{2\pi i/N}$ .

The important work of Kubert and Lang [199] plays a central role in the study of these modular function fields. Their work, which is built around the Siegel  $g_a$  functions and the Klein  $\mathfrak{t}_a$  functions, allows us to understand the fields  $\mathcal{F}_N$ , as well as the Galois theoretic properties of the extensions  $\mathcal{F}_N/\mathcal{F}_1$ . These results will be fundamental tools in the proofs of Theorems 1.8 and 1.15.

#### 3.4.1 Basic Facts about Siegel functions

We begin by recalling the definitions of the Siegel and Klein functions. Let  $\mathbf{B}_2(x) := x^2 - x + \frac{1}{6}$  be the second Bernoulli polynomial and  $\mathbf{e}(x) := \mathbf{e}^{2\pi i x}$ . If  $a = (a_1, a_2) \in \mathbb{Q}^2$ , then Siegel function  $g_a$  is defined as

$$g_{a}(\tau) := -q^{\frac{1}{2}\mathbf{B}_{2}(a_{1})} e\left(a_{2}(a_{1}-1)/2\right) \prod_{n=1}^{\infty} \left(1 - q^{n-1+a_{1}}e(a_{2})\right) \left(1 - q^{n-a_{1}}e(-a_{2})\right),$$
(3.42)

and the *Klein function*  $\mathfrak{t}_a$  is defined as

$$\mathfrak{t}_a(\tau) := \frac{g_a(\tau)}{\eta(\tau)^2} \tag{3.43}$$

where  $\eta(\tau) := q^{1/24} \prod_{n=1}^{\infty} (1-q^n)$  is the Dedekind  $\eta$ -function.

Neither  $g_a$  nor  $\mathfrak{t}_a$  are modular on  $\Gamma(N)$ , however if  $N \cdot a \in \mathbb{Z}^2$ , then  $\mathfrak{t}_a^{2N}$  is on  $\Gamma(N)$  (or  $\mathfrak{t}_a^N$  if N is odd). Therefore if  $g_a^{\operatorname{lcm}(12,2N)} \in \mathcal{F}_N$ , and if  $N \cdot a' \in \mathbb{Z}^2$ , then  $\left(\frac{g_a(\tau)}{g_{a'}(\tau)}\right)^{2N} \in \mathcal{F}_N$  if N is even and  $\left(\frac{g_a(\tau)}{g_{a'}(\tau)}\right)^N \in \mathcal{F}_N$  if N is odd. Given  $a \in \mathbb{Q}^2$ , we denote the smallest  $N \in \mathbb{N}$  such that  $N \cdot a \in \mathbb{Z}^2$  by  $\operatorname{Den}(a)$ .

**Theorem 3.6** ([199, Ch. 2 of K1 and K2]). Assuming the notation above, the following are true:

1. If 
$$\gamma \in SL_2(\mathbb{Z})$$
, then

$$(\mathfrak{t}_a|_{-1}\gamma)(\tau) = \mathfrak{t}_{a\gamma}(\tau)$$

2. If  $b = (b_1, b_2)$ , then

$$\mathbf{t}_{a+b}(\tau) = e\big((b_1b_2 + b_1 + b_2 - b_1a_2 + b_2a_1)\big)\mathbf{t}_{a\gamma}(\tau).$$

These properties for  $\mathfrak{t}_a$ , (3.42), and the fact that  $\eta(\tau)^{24} = \Delta(\tau)$  is modular on  $SL_2(\mathbb{Z})$ , lead to the following properties for  $g_a$ .

**Theorem 3.7** ([199, Ch. 2, Thm 1.2]). If  $a \in \mathbb{Z}^2/N$  and Den(a) = N, then the following are true:

1. If  $\gamma \in SL_2(\mathbb{Z})$ , then

$$(g_a^{12}|_0\gamma)(\tau) = g_{a\gamma}^{12}(\tau).$$

2. If  $b = (b_1, b_2) \in \mathbb{Z}^2$ , then

$$g_{a+b}(\tau) = e \big( 1/2 \cdot (b_1 b_2 + b_1 + b_2 - b_1 a_2 + b_2 a_1) \big) g_a(\tau).$$

- 3. We have that  $g_{-a}(\tau) = -g_a(\tau)$ .
- 4. The  $g_a(\tau)^{12N}$  are modular functions on  $\Gamma(N)$ . Moreover, if  $\gamma \in SL_2(\mathbb{Z})$ , then we have

$$(g_a^{12}|_0\gamma)(\tau) = g_{a\gamma}^{12}(\tau).$$

The following theorem addresses the modularity properties of products and quotients of Siegel functions.

**Theorem 3.8** ([199, Ch. 3, Lemma 5.2, Thm 5.3]). Let  $N \ge 2$  be an integer, and let  $\{m(a)\}_{r\in \frac{1}{N}\mathbb{Z}^2/\mathbb{Z}^2}$  be a set of integers. Then the product of Siegel functions

$$\prod_{\substack{\in \frac{1}{N}\mathbb{Z}^2/\mathbb{Z}^2}} g_a^{m(a)}(\tau)$$

belongs to  $\mathcal{F}_N$  if  $\{m(a)\}$  satisfies the following:

- 1. We have that  $\sum_{a} m(a)(Na_1)^2 \equiv \sum_{a} m(a)(Na_2)^2 \equiv 0 \pmod{\gcd(2, N) \cdot N}$ .
- 2. We have that  $\sum_{a} m(a)(Na_1)(Na_2) \equiv 0 \pmod{N}$ .
- 3. We have that  $gcd(12, N) \cdot \sum_{a} m(a) \equiv 0 \pmod{12}$ .

Additionally, we have the following important results about the algebraicity of the singular values of the Siegel functions in relation to the singular values of the  $SL_2(\mathbb{Z})$  modular function

$$j(\tau) := \frac{\left(1 + 240 \sum_{n=1}^{\infty} \sum_{d|n} d^3 q^n\right)^3}{q \prod_{n=1}^{\infty} (1 - q^n)^{24}}$$
$$= \frac{\eta(\tau)^{24}}{\eta(2\tau)^{24}} + 3 \cdot 2^8 + 3 \cdot 2^{16} \frac{\eta(2\tau)^{24}}{\eta(\tau)^{24}} + 2^{24} \frac{\eta(2\tau)^{48}}{\eta(\tau)^{48}}$$
$$= q^{-1} + 744 + 196884q + \cdots,$$

which are well known to be algebraic by the theory of complex multiplication (for example, see [33,85]).

**Theorem 3.9** ([199, Ch. 1, Thm. 2.2]). If  $\tau$  is a CM point and N = Den(a), then the following are true:

1. We have that  $g_a(\tau)$  is an algebraic integer.

- 2. If N has at least two prime factors, then  $g_a(\tau)$  is a unit over  $\mathbb{Z}[j(\tau)]$ .
- 3. If  $N = p^r$  is a prime power, then  $g_a(\tau)$  is a unit over  $\mathbb{Z}[1/p][j(\tau)]$ .
- 4. If  $c \in \mathbb{Z}$  and (c, N) = 1, then  $(g_{ca}/g_a)$  is a unit over  $\mathbb{Z}[j(\tau)]$ .

### 3.4.2 Galois theory of singular values of Siegel functions

We now recall the Galois-theoretic properties of extensions of modular function fields, and we then relate these properties to the Siegel and Klein functions.

The Galois group  $\operatorname{Gal}(\mathcal{F}_N/\mathcal{F}_1)$  is isomorphic to  $GL_2(N)/\{\pm I\} = GL_2(\mathbb{Z}/N\mathbb{Z})/\{\pm I\}$ (see [199, Ch. 3, Lemma 2.1]), where I is the identity matrix. This group factors naturally as

$$\left\{ \begin{pmatrix} 1 & 0 \\ 0 & d \end{pmatrix} : d \in (\mathbb{Z}/N\mathbb{Z})^{\times} \right\} \times SL_2(N)/\{\pm I\},$$

where an element  $\begin{pmatrix} 1 & 0 \\ 0 & d \end{pmatrix}$  acts on the Fourier coefficients by sending  $\zeta_N \to \zeta_N^d$ , and a matrix  $\gamma \in SL_2(\mathbb{Z})$  acts by the standard fractional linear transformation on  $\tau$ . If  $f(\tau) \in \mathcal{F}_N$  and  $\gamma \in GL_2(N)$ , then we use the notation  $f(\tau)_{(\gamma)} := (\gamma \circ f)(\tau)$ . Applying these facts to the Siegel functions, we obtain the following.

**Proposition 3.10.** If  $a \in \mathbb{Q}^2$ , and Den(a) divides N, then the multiset

$$\left\{g_a^{12N}(\tau)(\gamma) := g_{a\gamma}^{12N}(\tau) : \ \gamma \in GL_2(N)\right\}$$

is a union of Galois orbits for  $g_a^{12N}(\tau)$  over  $\mathcal{F}_1$ .

If  $\theta$  is a CM point of discriminant -D, we define the field

$$K_{(N)}(\theta) := \mathbb{Q}(\theta) (f(\theta) : f \in \mathcal{F}_N \text{ s.t. } f \text{ is defined and finite at } \theta),$$

and  $\mathcal{H} := \mathbb{Q}(\theta, j(\theta))$  be the Hilbert class field over  $\mathbb{Q}(\theta)$ . The Galois group  $K_{(N)}(\theta)/\mathcal{H}$  is isomorphic to the matrix group  $W_{N,\theta}$  (see [274]) defined by

$$W_{N,\theta} = \left\{ \begin{pmatrix} t - sB & -sC \\ sA & t \end{pmatrix} \in GL_2(\mathbb{Z}/N\mathbb{Z}) \right\} / \left\{ \pm \begin{pmatrix} 1 & 0 \\ 0 & 1 \end{pmatrix} \right\},$$

where  $Ax^2 + Bx + C$  is a minimal polynomial for  $\theta$  over  $\mathbb{Z}$ . The Galois group  $\operatorname{Gal}(\mathcal{H}/\mathbb{Q})$  is isomorphic to the group  $\mathcal{Q}_D$  of primitive reduced positivedefinite integer binary quadratic forms of negative discriminant -D. For each  $Q = ax^2 + bxy + cy^2 \in \mathcal{Q}_D$ , we define the corresponding CM point  $\tau_Q = \frac{-b+\sqrt{-D}}{2a}$ . In order to define the action of this group, we must also define corresponding matrices  $\beta_Q \in GL_2(\mathbb{Z}/N\mathbb{Z})$  which we may build up by way of the Chinese Remainder Theorem and the following congruences. For each prime p dividing N, we have the following congruences which hold  $(\operatorname{mod} p^{\operatorname{ord}_p(N)})$ :

$$\beta_Q \equiv \begin{cases} \begin{pmatrix} a & \frac{b}{2} \\ 0 & 1 \end{pmatrix} & \text{if } p \nmid a \\ \begin{pmatrix} -\frac{b}{2} & -c \\ 1 & 0 \end{pmatrix} & \text{if } p | a, \text{ and } p \nmid c \\ \begin{pmatrix} -\frac{b}{2} - a & -\frac{b}{2} - c \\ 1 & -1 \end{pmatrix} & \text{if } p | a, \text{ and } p | c \end{cases}$$

if  $-D \equiv 0 \pmod{4}$ , and

$$\beta_Q \equiv \begin{cases} \begin{pmatrix} a & \frac{b-1}{2} \\ 0 & 1 \end{pmatrix} & \text{if } p \nmid a \\ \begin{pmatrix} -\frac{b+1}{2} & -c \\ 1 & 0 \end{pmatrix} & \text{if } p | a, \text{ and } p \nmid c \\ \begin{pmatrix} -\frac{b+1}{2} - a & \frac{1-b}{2} - c \\ 1 & -1 \end{pmatrix} & \text{if } p | a, \text{ and } p | c \end{cases}$$

if  $-D \equiv 1 \pmod{4}$ . Then given  $\theta = \tau_{Q'}$  for some  $Q' \in \mathcal{Q}_D$ , define  $\delta_Q(\theta) := \beta_{Q'}^{-1}\beta_Q$ . The Galois group  $\operatorname{Gal}(\mathcal{H}/\mathbb{Q})$  can be extended into  $\operatorname{Gal}(K_{(N)}(\theta)/\mathbb{Q})$  by taking the action of a quadratic form Q on the element  $f(\theta) \in K_{(N)}(\theta)$  to be given by

$$Q \circ f(\theta) = f(\tau_Q)_{(\delta_Q(\theta))}.$$

We combine these facts into the following theorem.

**Theorem 3.11.** Let  $F(\tau)$  be in  $\mathcal{F}_N$  and let  $\theta$  be a CM point of discriminant -D < 0. Then the multiset

$$\left\{F(\tau_Q)_{(\gamma\cdot\delta_Q(\theta))}:\ (\gamma,Q)\in W_{\kappa,\tau}\times\mathcal{Q}_D\right\}$$

is a union of the Galois orbits of  $F(\theta)$  over  $\mathbb{Q}$ .

## 3.5 Proofs of Theorems 1.8 and 1.15

Here we prove Theorems 1.8 and 1.15. We shall prove these theorems using the results of the previous section.

## **3.5.1** Reformulation of the $\Phi_*(m, n; \tau)$ series

To ease the proofs of Theorems 1.8 and 1.15, we begin by reformulating each of the  $\Phi_*(m, n; \tau)$  series, as well as

$$\frac{\Phi_{1a}(m,n;\tau)}{\Phi_{1b}(m,n;\tau)},$$

as pure products of modified theta functions. These factorizations will be more useful for our purposes. In order to ease notation, for a fixed  $\kappa$ , if  $1 \leq j < \kappa/2$ , then we let

$$\theta_{j,\kappa} := \theta(q^j; q^\kappa),$$

If  $\kappa$  is even, then we let

$$\theta_{\kappa/2,\kappa} := (q^{\kappa/2}; q^{\kappa})_{\infty} = \theta(q^{\kappa/2}; q^{2\kappa}),$$

which is a square root of  $\theta(q^{\kappa/2};q^{\kappa})$ .

The reformulations below follow directly from (1.16) by making use of the fact that

$$\frac{(q^{\kappa};q^{\kappa})_{\infty}}{(q)_{\infty}} = \prod_{j=1}^{\lfloor \kappa/2 \rfloor} \theta_{j,\kappa}.$$

**Lemma 3.12.** Let m and n be positive integers and  $\kappa_* = \kappa_*(m, n)$  as in (1.15). Then the following are true:

(1a) With  $\kappa = \kappa_1$ ,

$$\Phi_{1a}(m,n;\tau) = q^{\frac{mn(4mn-4m+2n-3)}{12\kappa}} \prod_{j=1}^{m} \theta_{j,\kappa}^{-1} \prod_{j=1}^{m+n} \theta_{j,\kappa}^{-\min(m,n-1,\lceil j/2\rceil-1)}.$$

(1a) With 
$$\kappa = \kappa_1$$
,

$$\Phi_{1b}(m,n;\tau) = q^{\frac{mn(4mn+2m+2n+3)}{12\kappa}} \prod_{j=1}^{m+n} \theta_{j,\kappa}^{-\min(m,n,\lfloor j/2 \rfloor)}.$$

(2) With 
$$\kappa = \kappa_2$$
,

$$\Phi_2(m,n;\tau) = q^{\frac{m(2n+1)(2mn-m+n-1)}{12\kappa}} \prod_{j=1}^m \theta_{j,\kappa}^{-1} \prod_{j=1}^{m+n+1} \theta_{j,\kappa}^{-\min(m,n-1,\lceil j/2\rceil-1)} \prod_{j=n}^{\lfloor (m+n)/2 \rfloor} \theta_{2j+1,\kappa}^{-1}.$$

(3) For  $n \geq 2$  and  $\kappa = \kappa_3$ ,

$$\Phi_3(m,n;\tau) = q^{\frac{m(2n-1)(2mn+n+1)}{12\kappa}} \prod_{j=1}^{m+n} \theta_{j,\kappa}^{-\min(m,n-1,\lfloor j/2 \rfloor)} \prod_{j=n}^{\lfloor (m+n)/2 \rfloor} \theta_{2j,\kappa}^{-1}.$$

Moreover, with  $\kappa = \kappa_1(m, n)$ ,

$$\Psi_1(m,n;\tau) := \frac{\Phi_{1a}(m,n;\tau)}{\Phi_{1b}(m,n;\tau)} = q^{-\frac{mn(m+1)}{2\kappa}} \prod_{j=1}^m \frac{\theta(q^{2j};q^\kappa)}{\theta(q^j;q^\kappa)},$$

and with  $\kappa = \kappa_2(m, n) = \kappa_3(m, n+1)$ ,

$$\Psi_2(m,n;\tau) := \frac{\Phi_2(m,n;\tau)}{\Phi_3(m,n;\tau)} = q^{-\frac{m(m+1)(2n+1)}{4\kappa}} \prod_{j=1}^m \frac{\theta(q^{2j};q^\kappa)}{\theta(q^j;q^\kappa)}.$$

*Proof.* Since the proofs of the four cases are essentially the same, we only prove Lemma 3.12 (1a).

Let  $\varphi = mn(4mn - 4m + 2n - 3)/(12\kappa_1)$ . By Theorem 1.2, we have that

$$\Phi_{1a}(m,n;\tau) = q^{\varphi} \cdot \frac{(q^{\kappa};q^{\kappa})_{\infty}^{n}}{(q)_{\infty}^{n}} \prod_{i=1}^{n} \theta(q^{i+m};q^{\kappa}) \prod_{1 \le i < j \le n} \theta(q^{j-i},q^{i+j-1};q^{\kappa})$$
$$= q^{\varphi} \cdot \frac{(q^{\kappa};q^{\kappa})_{\infty}^{m}}{(q)_{\infty}^{m}} \prod_{i=1}^{m} \theta(q^{i+1};q^{\kappa}) \prod_{1 \le i < j \le m} \theta(q^{j-i},q^{i+j+1};q^{\kappa}).$$

Using the simple identity

$$\frac{(q^{\kappa};q^{\kappa})_{\infty}}{(q)_{\infty}} = \prod_{j=1}^{m+n} \theta(q^j;q^{\kappa})^{-1},$$

we can rewrite these two forms as

$$\Phi_{1a}(m,n;\tau) = q^{\varphi} \cdot \frac{\prod_{i=1}^{n} \theta(q^{i+m};q^{\kappa})}{\prod_{i=1}^{m+n} \theta(q^{j};q^{\kappa})} \cdot \prod_{j=2}^{n} \frac{\prod_{i=1}^{j-1} \theta(q^{j-i},q^{i+j-1};q^{\kappa})}{\prod_{i=1}^{m+n} \theta(q^{i},q^{\kappa})} \\ = q^{\varphi} \cdot \frac{\prod_{i=1}^{m} \theta(q^{i+1};q^{\kappa})}{\prod_{i=1}^{m+n} \theta(q^{j};q^{\kappa})} \cdot \prod_{j=2}^{m} \frac{\prod_{i=1}^{j-1} \theta(q^{j-i},q^{i+j+1};q^{\kappa})}{\prod_{i=1}^{m+n} \theta(q^{i};q^{\kappa})}.$$

If  $m \ge n-1$  then the first identity reduces to

$$\Phi_{1a}(m,n;\tau) = q^{\varphi} \cdot \left(\prod_{j=1}^{m} \theta(q^{j};q^{\kappa})\right)^{-1} \prod_{j=2}^{n} \left(\prod_{i=2j-1}^{m+n} \theta(q^{i};q^{\kappa})\right)^{-1}$$
$$= q^{\varphi} \cdot \left(\prod_{j=1}^{m} \theta(q^{j};q^{\kappa})\right)^{-1} \prod_{j=1}^{n-1} \left(\prod_{i=2j+1}^{m+n} \theta(q^{i};q^{\kappa})\right)^{-1}.$$

If  $m \leq n-1$  then the second identity reduces to

$$\Phi_{1a}(m,n;\tau) = q^{\varphi} \cdot \left(\theta(q;q^{\kappa}) \prod_{j=m+2}^{m+n} \theta(q^j;q^{\kappa})\right)^{-1} \\ \times \prod_{j=2}^m \left(\theta(q^j,q^{j+1};q^{\kappa}) \prod_{i=2j+1}^{m+n} \theta(q^i;q^{\kappa})\right)^{-1} \\ = q^{\varphi} \cdot \left(\prod_{j=1}^m \theta(q^j;q^{\kappa})\right)^{-1} \prod_{j=1}^m \left(\prod_{i=2j+1}^{m+n} \theta(q^i;q^{\kappa})\right)^{-1}.$$

Together these imply Lemma 3.12 (1a).

Since the modified  $\theta$ -functions  $\theta(q^{\ell}; q^{\kappa})$  are essentially Siegel functions (up to powers of q), we can immediately rewrite Lemma 3.12 in terms of modular functions. We shall omit the proofs for brevity.

**Lemma 3.13.** Let m and n be positive integers and  $\kappa_* = \kappa_*(m, n)$  as in (1.15). Then the following are true:

(1a) With  $\kappa = \kappa_1$ ,

$$\Phi_{1a}(m,n;\tau) = \prod_{j=1}^{m} g_{j/\kappa,0}(\kappa\tau)^{-1} \prod_{j=1}^{m+n} g_{j/\kappa,0}(\kappa\tau)^{-\min(m,n-1,\lceil j/2\rceil-1)}.$$

(1b) With  $\kappa = \kappa_1$ ,

$$\Phi_{1b}(m,n;\tau) = \prod_{j=1}^{m+n} g_{j/\kappa,0}(\kappa\tau)^{-\min(m,n,\lfloor j/2 \rfloor)}.$$

(2) With  $\kappa = \kappa_2$ ,

$$\Phi_{2}(m,n;\tau) = g_{\frac{1}{4},0}(2\kappa\tau)^{-\min(m,n-1)-\delta_{1}} \prod_{j=1}^{m} g_{\frac{j}{\kappa},0}(\kappa\tau)^{-1} \\ \times \prod_{j=1}^{m+n} g_{\frac{j}{\kappa},0}(\kappa\tau)^{-\min(m,n-1,\lceil j/2\rceil-1)} \prod_{j=n}^{\lfloor (m+n-1)/2 \rfloor} g_{\frac{(2j+1)}{\kappa},0}(\kappa\tau)^{-1}.$$

(3) For  $n \geq 2$  and  $\kappa = \kappa_3$ ,

$$\Phi_{3}(m,n;\tau) = g_{\frac{1}{4},0}(2\kappa\tau)^{-\min(m,n-1)-\delta_{2}} \times \prod_{j=1}^{m+n-1} g_{\frac{j}{\kappa},0}(\kappa\tau)^{-\min(m,n-1,\lfloor j/2\rfloor)} \prod_{j=n}^{\lfloor (m+n-1)/2\rfloor} g_{\frac{2j}{\kappa},0}(\kappa\tau)^{-1}.$$

(4) With  $\kappa = \kappa_1$ ,

$$\Psi_1(m,n;\tau) := \frac{\Phi_{1a}(m,n;\tau)}{\Phi_{1b}(m,n;\tau)} = \prod_{j=1}^m \frac{g_{\frac{2j}{\kappa},0}(\kappa\tau)}{g_{\frac{j}{\kappa},0}(\kappa\tau)}.$$

(5) With  $\kappa = \kappa_2(m, n) = \kappa_3(m, n+1)$ ,

$$\Psi_{2}(m,n;\tau) := \frac{\Phi_{2}(m,n;\tau)}{\Phi_{3}(m,n;\tau)} = \prod_{j=1}^{m} \frac{g_{\frac{2j}{\kappa},0}(\kappa\tau)}{g_{\frac{j}{\kappa},0}(\kappa\tau)}$$

## 3.5.2 Proofs of Theorems 1.8 and 1.15

We now apply the results in Section 3.4 to prove Theorems 1.8 and 1.15.

Proof of Theorem 1.8 (1) and (2). Lemma 3.13 shows that each of the  $\Phi_*(m, n; \tau)$  is exactly a pure product of Siegel functions. Therefore, we may apply Theorem 3.7 directly to each of the Siegel function factors, and as a consequence to each  $\Phi_*(m, n; \tau)$ .

Since by Theorem 3.7 (4),  $g_a(\tau)^{12N}$  is in  $\mathcal{F}_N$  if N = Den(a), we may take  $N = \kappa_*(m, n)$ , and so we have that  $\Phi_*(m, n; \tau)^{12\kappa} \in \mathcal{F}_{\kappa_*(m,n)}$ . We now apply Theorem 3.11 to obtain Theorem 1.8 (1) and (2).

Sketch of the Proof of Theorem 1.8 (3). By Theorem 1.8 (2), we have that this multiset consists of multiple copies of a single Galois orbit of conjugates over  $\mathbb{Q}$ . Therefore to complete the proof, it suffices to show that the given conditions imply that there are singular values which are not repeated. To this end, we focus on those CM points with maximal imaginary parts. Indeed, because each  $\Phi_*(m, n; \tau)$  begins with a negative power of q, one generically expects that these corresponding singular values will be the one with maximal complex absolute value.

To make this argument precise requires some cumbersome but unenlightening details (which we omit)<sup>4</sup>. One begins by observing why the given conditions are necessary. For small  $\kappa$  it can happen that the matrices in  $W_{\kappa,\tau}$ permute the Siegel functions in the factorizations of  $\Phi_*(m, n; \tau)$  obtained in Lemma 3.13. However, if  $\kappa > 9$ , then this does not happen. The condition that  $gcd(D_0, \kappa) = 1$  is required for a similar reason. More precisely, the group does not act faithfully. However, under these conditions, the only obstruction to the conclusion would be a nontrivial identity between the evaluations of two different modular functions. In particular, under the given assumptions, we may view these functions as a product of distinct Siegel functions. Therefore, the proof follows by studying the asymptotic properties of the CM values of individual Siegel functions, and then considering the  $\Phi_*$  functions as a product of these values.

The relevant asymptotics arise by considering, for each -D, a canonical CM point with discriminant -D. Namely, we let

$$\tau_* := \begin{cases} \frac{\sqrt{-D}}{2} & \text{if } -D \equiv 0 \pmod{4}, \\ \frac{1+\sqrt{-D}}{2} & \text{if } -D \equiv 1 \pmod{4}. \end{cases}$$

By the theory of reduced binary quadratic forms, these points are the CM points with maximal imaginary parts corresponding to reduced forms with discriminant -D. Moreover, every other CM point with discriminant -D has imaginary part less than  $|\sqrt{-D}|/3$ . Now the singular values of each Siegel function then essentially arise from the values of the second Bernoulli polynomial. The point is that one can uniformly estimate the infinite product portion of each singular value, and it turns out that they are exponentially

<sup>&</sup>lt;sup>4</sup>A similar analysis is carried out in detail by Jung, Koo, and Shin in [172, Sec. 4].
close to the number 1. By assembling these estimates carefully, one obtains the result.  $\hfill \Box$ 

Proof of Theorem 1.15. Lemma 3.13 reformulates each  $\Phi_*$  function in terms of products of negative powers of Siegel functions of the form  $g_{j/\kappa,0}(\kappa\tau)$ , where  $1 \leq j \leq \kappa/2$ , and  $g_{1/4,0}(2\kappa\tau)$ , when  $\kappa$  is even. Theorem 3.9 (1) then implies Theorem 1.15 (1).

Since  $\text{Den}(j/\kappa, 0)$  may be any divisor of  $\kappa$ , and since  $j(\tau)$  is an algebraic integer [33,85], Theorem 3.9 (2) and (3) imply Theorem 1.15 (2).

Using Theorem 3.13(5), we have that

$$\frac{\Phi_{1a}(m,n;\tau)}{\Phi_{1b}(m,n;\tau)} = \prod_{j=1}^m \frac{g_{\frac{2j}{\kappa},0}(\kappa\tau)}{g_{\frac{j}{\kappa},0}(\kappa\tau)},$$

where  $\kappa = \kappa_1 = 2m + 2n + 1$ . Since  $\kappa$  is odd, Theorem 3.9 (4) implies that each term

$$\frac{g_{\frac{2j}{\kappa},0}(\kappa\tau)}{g_{\underline{i},0}(\kappa\tau)}$$

in the product is a unit. Therefore, Theorem 1.15 (3) follows.  $\Box$ 

### 3.6 Examples

Here we give two examples of the main results in this paper.

*Example* 3.14. This is a detailed discussion of the example in the introduction.

Consider the q-series

$$\Phi_{1a}(2,2;\tau) = q^{1/3} \prod_{n=1}^{\infty} \frac{(1-q^{9n})}{(1-q^n)}$$
  
=  $q^{1/3} + q^{4/3} + 2q^{7/3} + 3q^{10/3} + 5q^{13/3} + 7q^{16/3} + \cdots$ 

and

$$\Phi_{1b}(2,2;\tau) = q \prod_{n=1}^{\infty} \frac{(1-q^{9n})(1-q^{9n-1})(1-q^{9n-8})}{(1-q^n)(1-q^{9n-4})(1-q^{9n-5})}$$
$$= q + q^3 + q^4 + 3q^5 + 3q^6 + 5q^7 + 6q^8 + \cdots$$

For  $\tau = i/3$ , the first 100 coefficients of the q-series respectively give the numerical approximations

$$\Phi_{1a}(2,2;i/3) = 0.577350 \cdots \stackrel{?}{=} \frac{1}{\sqrt{3}}$$
$$\Phi_{1b}(2,2;i/3) = 0.217095 \dots$$

Here we have that  $\kappa_1(2,2) = 9$ . Theorem 3.8 tells us that  $\Phi_{1a}(2,2;\tau)^3$  and  $\Phi_{1b}(2,2;\tau)^3$  are in  $\mathcal{F}_9$ , so we may use Theorem 3.11 to find the conjugates of the values of the functions at  $\tau = i/3$ . We have  $\kappa_1(2,2) \cdot i/3 = 3i$  and

$$W_{9,3i} = \left\{ \begin{pmatrix} t & 0 \\ s & t \end{pmatrix} \in GL_2(\mathbb{Z}/9\mathbb{Z}) \right\} / \left\{ \pm \begin{pmatrix} 1 & 0 \\ 0 & 1 \end{pmatrix} \right\},$$

which has 27 elements. However each of these acts like the identity on  $\Phi_{1a}(2,2;\tau)$ , and the group has an orbit of size three when acting on  $\Phi_{1b}(2,2;\tau)$ . The set  $Q_{36}$  has two elements  $Q_1 = x^2 + 9y^2$  and  $Q_2 = 2x^2 + 2xy + 5y^2$ . These give us  $\beta_{Q_1}$  which is the identity, and  $\beta_{Q_2} = \begin{pmatrix} 2 & 1 \\ 0 & 1 \end{pmatrix}$ . Therefore  $\Phi_{1a}(2,2;i/3)^3$  has only one other conjugate,

$$\left(g_{2/9,1/9}\left(\frac{-1+3\mathrm{i}}{2}\right)g_{4/9,2/9}\left(\frac{-1+3\mathrm{i}}{2}\right)g_{6/9,3/9}\left(\frac{-1+3\mathrm{i}}{2}\right)g_{8/9,4/9}\left(\frac{-1+3\mathrm{i}}{2}\right)\right)^{-3},$$

although the multiset described in Theorem 1.8 (2) contains 27 copies of these two numbers. On the other hand,  $\Phi_{1b}(2,2;i/3)^3$  has an orbit of six conjugates, and the multiset from Theorem 1.8 (2) contains nine copies of this orbit. Theorem 1.15 (2) tells us that  $\Phi_{1a}(2,2;i/3)$  and  $\Phi_{1b}(2,2;i/3)$  may have denominators which are powers of three, whereas Theorem 1.15 (1) tells us that their inverses are algebraic integers. Therefore, we find the minimal polynomials for the inverses and then invert the polynomials. In this way, we find that  $\Phi_{1a}(2,2;i/3)$  and  $\Phi_{1b}(2,2;i/3)$  are roots of the irreducible polynomials

$$3x^2 - 1$$
  
19683 $x^{18} - 80919x^{12} - 39366x^9 + 11016x^6 - 486x^3 - 1.$ 

The full polynomials whose roots are the elements of the multisets corresponding to  $\Phi_{1a}(2,2;i/3)^3$  and  $\Phi_{1b}(2,2;i/3)^3$ , counting multiplicity are

$$(27x^2 - 1)^{27}$$
$$(19683x^6 - 80919x^4 - 39366x^3 + 11016x^2 - 486x^2 - 1)^9$$

Applying Theorem 1.15(2), we find that  $\sqrt{3}\Phi_{1a}(2,2;i/3)$  and  $\sqrt{3}\Phi_{1b}(2,2;i/3)$  are units and roots of the polynomials

$$\begin{aligned} x - 1 \\ x^{18} + 6x^{15} - 93x^{12} - 304x^9 + 420x^6 - 102x^3 + 1. \end{aligned}$$

Lastly, Theorem 1.15 (3) applies, and we know that the ratio

$$\frac{\Phi_{1a}(2,2;\tau)}{\Phi_{1b}(2,2;\tau)} = q^{-2/3} \prod_{n=1}^{\infty} \frac{(1-q^{9n-4})(1-q^{9n-5})}{(1-q^{9n-1})(1-q^{9n-8})}$$
$$= q^{-2/3} (1+q+q^2+q^3-q^5-q^6-q^7+\cdots)$$

evaluates to a unit at  $\tau = i/3$ . In fact we find that

$$\frac{\Phi_{1a}(2,2;i/3)}{\Phi_{1b}(2,2;i/3)} = 4.60627\dots$$

is a unit. Indeed, it is a root of

$$x^{18} - 102x^{15} + 420x^{12} - 304x^9 - 93x^6 + 6x^3 + 1.$$

Example 3.15. Here we give an example which illustrates the second remark after Theorem 1.15. This is the discussion concerning ratios of singular values of  $\Phi_2$  and  $\Phi_3$  with the same  $\kappa_*$ . Here we show that these ratios are not generically algebraic integral units as Theorem 1.15(3) guarantees for the  $\mathbb{A}_{2n}^{(2)}$  cases.

We consider  $\Phi_2(1,1;\tau)$  and  $\Phi_3(1,2;\tau)$ , with  $\tau = \sqrt{-1/3}$ . For these example we have  $\kappa_2(1,1) = \kappa_3(1,2) = 6$ . A short computation by way of the *q*-series shows that

$$\Phi_2(1,1;\sqrt{-1/3}) = 0.883210\ldots,$$

and

$$\Phi_3(1,2;\sqrt{-1/3}) = 0.347384\dots$$

Since  $\Phi_2(1,1;\tau)^{24}$  and  $\Phi_3(1,2;\tau)^{24}$  are in  $\mathcal{F}_{12}$ , we find that

$$\Phi_2(1,1;\sqrt{-1/3})^{24}$$
 and  $\Phi_3(1,2;\sqrt{-1/3})^{24}$ 

each have one other conjugate, namely

$$\left(g_{1/2,1/3}\left(\sqrt{-4/3}\right) \cdot g_{1/4,0}\left(2\sqrt{-4/3}\right)\right)^{-24}$$
 and  $\left(g_{0,1/3}\left(\sqrt{-4/3}\right) \cdot g_{1/2,0}\left(2\sqrt{-4/3}\right)\right)^{-24}$   
respectively, and the corresponding multisets described in Theorem 1.8 (2)  
each contain six copies of the respective orbits. In this way we find that  
 $\Phi_2(1,1;\sqrt{-1/3})$  is a root of

$$2^{20} x^{48} - 2^{12} \cdot 13x^{24} + 1$$

and  $\Phi_3(1,2;\sqrt{-1/3})$  is a root of

$$2^{20}3^{12}x^{48} - 12^6 \cdot 35113x^{24} + 1.$$

Therefore, their ratio

$$\frac{\Phi_2(1,1;\sqrt{-1/3})}{\Phi_3(1,2;\sqrt{-1/3})} = 2.542459\dots$$

is not a unit. Its minimal polynomial is

$$x^4 - 6x^2 - 3.$$

# Chapter 4

# Ramanujan's mock theta functions

## 4.1 Proof of Theorem 1.11

Proof. Suppose that g is such weakly holomorphic modular form on  $\Gamma_1(m)$  which cuts out the singularities of a mock modular form  $F^+$  on  $\Gamma_1(n)$  which has a non-zero shadow. Moreover, suppose that both are of weight  $k \in \frac{1}{2}\mathbb{Z}$ . Then the difference F - g is also a Maass form of weight k on  $\Gamma_1(\operatorname{GCD}(m, n))$  with non-zero shadow. Due to the growth condition required in by the definition of a harmonic Maass form given in 2.1, the non-holomorphic part of  $F^-$  exhibits exponential decay as  $\tau \to \infty$ . Suppose  $h(\tau) = \xi_k F$ , and pick  $\gamma = \begin{pmatrix} a & b \\ c & d \end{pmatrix}$  with  $c \neq 0$ , and let  $\zeta = e(a/c)$  and  $\rho$  be the cusp d/c. The growth of  $F^-(\tau)$  as  $q \to \zeta$  is bounded, as can be seen by taking the following limit.

$$\begin{split} \lim_{\tau \to +i\infty} (F^{-}) (\gamma \tau) &= \lim_{\tau \to +i\infty} F^{-} (\gamma \tau) - F^{-}_{\rho} (\tau) \\ &= \lim_{\tau \to +i\infty} \int_{-\overline{\gamma\tau}}^{i\infty} h(z) (-i(\gamma \tau + z))^{k-2} \mathrm{d}z - F^{-}_{\rho} (\tau) \\ &= \lim_{\tau \to +i\infty} \int_{-\overline{\tau}}^{\frac{d}{c}} h(\gamma^{*}z) (-i(\gamma \tau + \gamma^{*}z))^{k-2} \mathrm{d}(\gamma^{*}z) - F^{-}_{\rho} (\tau) \\ &= \lim_{\tau \to +i\infty} - \int_{\frac{d}{c}}^{i\infty} h_{\rho}(z) \left(\frac{-i(\tau + z)}{c\tau + d}\right)^{k-2} \mathrm{d}z. \end{split}$$

Here  $\gamma^* = \begin{pmatrix} a & -b \\ -c & d \end{pmatrix}$ . Thus  $F^+ - g$  is O(1) if and only if F - g is as well.

Suppose that F - g has a non-trivial principal part at the expansion at infinity (a similar arguments works if we have a non-trivial principal part at any other cusp). Then take  $\gamma = \begin{pmatrix} a & b \\ c & d \end{pmatrix} \in \Gamma_1(\operatorname{GCD}(m, n))$ , with  $\zeta$  as above. The radial limit  $\lim_{q \to \zeta} F - g$  may be computed by taking

$$\lim_{z \to +i\infty} (F-g) \left(\frac{az+b}{cz+d}\right) = (cz+d)^k (F-g)(z).$$

Since the principal part is nontrivial, we observe exponential growth on  $(cz + d)^k (F^+ - g)(z)$ , while  $(cz + d)^k F^-(z)$  must approach 0. Thus, neither F - g nor  $F^+ - g$  is O(1) as  $q \to \zeta$ .

Therefore, if g cuts out the singularities at roots of unity, it must also cut out the principal part of F at all cusps. Then F - g is a weak Maass form with no principal part at any cusp, but has a non-zero non-holomorphic part. This, however, contradicts Theorem 2.2.

### 4.2 Proof of Corollary 4.1

The corollary follows immediately from a stronger corollary which we state here.

**Corollary 4.1.** Suppose that  $F(\tau) \in H_{k_1}(\Gamma_1(N))$ . Then there does not exist a weakly holomorphic modular form g(z) of any weight  $k_2 \in \frac{1}{2}\mathbb{Z}$  on any congruence subgroup  $\Gamma_1(N')$  such that for every root of unity  $\zeta$  we have

$$\lim_{q \to \zeta} \left( F^+(\tau) - g(\tau) \right) = O(1).$$

*Proof.* As noted in the previous section,  $F^{-}(\tau)$  is O(1) as  $\tau$  approaches cusps, therefore this is equivalent to the statement that no such  $g(\tau)$  exists such that

$$\lim_{q \to \zeta} \left( F(\tau) - g(\tau) \right) = O(1).$$

Suppose such a  $g(\tau)$  existed. Following the argument in the previous section, if such a  $g(\tau)$  existed then both  $F(\tau)$  and  $g(\tau)$  must have the same principal parts at all cusps. Due to Theorem 2.2, at least one of these must be non constant. Without loss of generality, suppose there is a non-constant principal part at the cusp infinity. Consider the function  $h(\tau) := F(\tau) - g(\tau)$ . By hypothesis,  $h(\tau)$  has bounded radial limits as q approaches every root of unity. We note that  $F(\tau)$  and  $g(\tau)$  are modular on some common subgroup  $\Gamma_1(M)$ . If we take  $\gamma = \begin{pmatrix} a & b \\ c & d \end{pmatrix} \in \Gamma_1(M)$  with  $c \neq 0$ , then we have

$$h\left(\frac{a\tau+b}{c\tau+d}\right) = F\left(\frac{a\tau+b}{c\tau+d}\right) - g\left(\frac{a\tau+b}{c\tau+d}\right) = (c\tau+d)^{k_1}F(\tau) - (c\tau+d)^{k_2}g(\tau).$$
(4.1)

Letting  $z \to i\infty$ , we see that  $g(\tau)$  cannot cut out the exponential singularity of  $F(\tau)$  due to the difference between the weights, thus contradicting the assumption.

# 4.3 The f(q) example

Having proven that the poles of a mock theta function cannot mirror those of a weakly holomorphic modular form at every cusp, we return to Ramanujan's example of a subtler way in which a modular form may imitate a mock theta function at the cusps. Namely, Ramanujan conjectures for the mock theta function

$$f(q) = 1 + \frac{q}{(1+q)^2} + \frac{q^4}{(1+q)^2(1+q^2)^2} + \dots$$

and for the q-series b(q) (which is a weakly holomorphic modular form up to a  $q^{-\frac{1}{24}}$ )

$$b(q) := (1-q)(1-q^3)(1-q^5)\cdots(1-2q+2q^4-\ldots)$$

that as q approaches an even 2k order root of unity  $\zeta$ , b(q) is a "near miss" to f(q), in that half the time one must subtract, and half of the time one must add the values. More specifically, he conjectures that

$$f(q) - (-1)^k b(q) = O(1).$$

Folsom, Ono, and Rhoades show this conjecture and moreover prove the following:

**Theorem 4.2** (Theorem 1.1 of [121]). If  $\zeta$  is a primitive, even order 2k root of unity, then, as q approaches  $\zeta$  radially in the unit disk, we have that

$$\lim_{q \to \zeta} \left( f(q) - (-1)^k b(q) \right) = -4 \cdot \sum_{n=0}^{k-1} (1+\zeta)^2 (1+\zeta^2)^2 \cdots (1+\zeta^n)^2 \zeta^{n+1}.$$

# Chapter 5

# Weierstrass mock modular forms

# 5.1 Weierstrass Theory and the proof of Theorems 1.14, 1.15 and 1.16

Here we recall the essential features of the Weierstrass theory of elliptic curves. After recalling these facts, we then prove Theorems 1.14 and 1.15.

#### 5.1.1 Basic facts about Weierstrass theory

As noted in the introduction, the analytic parameterization  $\mathbb{C}/\Lambda_E \cong E$  of an elliptic curve is given by  $\mathfrak{z} \to P_{\mathfrak{z}} = (\wp(\Lambda_E; \mathfrak{z}), \wp'(\Lambda_E; \mathfrak{z}))$ . By evaluating the Weierstrass  $\wp$ -function at the Eichler integral given in (1.20), this analytic parameterization becomes the modular parameterization. The Eichler integral is not modular, however its obstruction to modularity is easily characterized. The map  $\Psi_E \colon \Gamma_0(N) \to \mathbb{C}$  given by

$$\Psi_E(\gamma) := \mathcal{E}_E(z) - \mathcal{E}_E(\gamma z) \tag{5.1}$$

is a homomorphism of groups. Its image in  $\mathbb{C}$  turns out to be the lattice  $\Lambda_E$ . Hence, since  $\wp(\Lambda_E; \mathfrak{z})$  is invariant on the lattice, the map  $\wp(\Lambda_E; \mathcal{E}_E(z))$  parameterizes E and is also a modular function.

Theorems 1.14 and 1.15 rely on a similar observation, but in this case involving the Weierstrass  $\zeta$ -function. Unlike the Weierstrass  $\wp$ -function, the  $\zeta$ -function itself is not lattice-invariant. However, Eisenstein [298] observed that it could be modified to become lattice-invariant but this modification necessarily sacrifices holomorphicity.

#### 5.1.2 Proofs of Theorems 1.14 and 1.15

We now prove Theorems 1.14 and 1.15.

Proof of Theorem 1.14. Eisenstein's modification to the  $\zeta$ -function is given by

$$\zeta(\Lambda_E; \mathfrak{z}) - S(\Lambda_E)\mathfrak{z} - \frac{\pi}{a(\Lambda_E)}\overline{\mathfrak{z}}.$$
(5.2)

Here S is as in (1.22) and  $a(\Lambda_E)$  is the area of a fundamental parallelogram for  $\Lambda_E$ .

Using the formula

$$a(\Lambda_E) = \frac{4\pi^2 ||F_E||^2}{\deg(\phi_E)},$$
(5.3)

we have that the function  $\mathfrak{Z}_E(\mathfrak{z})$  defined in (1.23) above is Eisenstein's corrected  $\zeta$ -function and is lattice-invariant. Formula (5.3) was first given by Zagier [307] for prime conductor and generalized by Cremona for general level [86]. Since  $\mathfrak{Z}_E(\mathfrak{z})$  is lattice-invariant,  $\widehat{\mathfrak{Z}}_E(z)$ , defined by (1.24), is modular.

Part (1) of Theorem 1.14 follows by noting that  $\mathfrak{Z}_E(\mathfrak{z})$  diverges precisely for  $\mathfrak{z} \in \Lambda_E$ . This divergence must result from a pole in the holomorphic part,  $\mathfrak{Z}_E^+(\mathfrak{z})$ .

In order to establish part (2), we consider the modular function  $\wp(\Lambda_E; \mathcal{E}_E(z))$ . We observe that  $\wp(\Lambda_E; \mathcal{E}_E(z))$  is meromorphic with poles precisely for those z such that  $\mathcal{E}_E(z) \in \Lambda_E$ . Therefore  $\wp(\Lambda_E; \mathcal{E}_E(z))$  may be decomposed into modular functions with algebraic coefficients, each with only a simple pole at one such z and possibly at cusps. These simple modular functions may be combined appropriately to construct the function  $M_E(z)$  to cancel the poles of  $\hat{\mathfrak{Z}}_E^+(z)$ .

The proof of (3) follows from straightforward calculations.  $\Box$ 

Using the theory of Atkin-Lehner involutions, we now prove Theorem 1.15.

Proof of Theorem 1.15. Recall that by classical theory of Atkin-Lehner, every newform of level  $N_E$  is an eigenform of the Atkin-Lehner involution

$$W_q = \begin{pmatrix} q^{\alpha}a & b \\ Nc & q^{\alpha}d \end{pmatrix},$$

for every prime power  $q||N_E$ , with eigenvalue  $\pm 1$ . We note that

$$\widehat{\mathfrak{Z}}_E(z)|_0 W_q = \mathfrak{Z}_E(\Lambda_E; \mathcal{E}_E(z)|_0 W_q).$$

It suffices to show  $\mathcal{E}_E(z) - \lambda_q \mathcal{E}_E(z) | W_q$  is equal to  $\Omega_q(F_E)$ . To this end note that

$$\begin{aligned} \mathcal{E}_{E}(z) - \lambda_{q} \mathcal{E}_{E}(z) | W_{q} &= -2\pi i \left[ \int_{z}^{i\infty} F_{E}(z) dz - \lambda_{q} \int_{W_{q}z}^{i\infty} F_{E}(z) dz \right] \\ &= -2\pi i \left[ \int_{z}^{i\infty} F_{E}(z) dz - \lambda_{q} \int_{z}^{W_{q}^{-1}i\infty} \frac{\det(W_{q})}{(Ncz + q^{\alpha}d)^{2}} F_{E}(W_{q}z) dz \right] \\ &= -2\pi i \left[ \int_{z}^{i\infty} F_{E}(z) dz + \lambda_{q}^{2} \int_{W_{q}^{-1}i\infty}^{z} F_{E}(z) dz \right] \\ &= -2\pi i \int_{W_{q}^{-1}i\infty}^{i\infty} F_{E}(z) dz. \end{aligned}$$
(5.4)

We note that if  $\Omega_q(F_E)$  is in the lattice, then we may ignore this term, and we see that  $\widehat{\mathfrak{Z}}_E(z)$  is an eigenfunction for the involution  $W_q$ . Otherwise,  $\widehat{\mathfrak{Z}}_E(z)|_0 W_q$  has a constant term equal to  $\mathfrak{Z}_E(\Omega_q(F_E))$ .

#### 5.1.3 Proof of Theorem 1.16

The proof of Theorem 1.16 is similar to recent work of Guerzhoy, Kent, and Ono [153]. We will need the following proposition.

**Proposition 5.1.** Suppose that R(z) is a meromorphic modular function on  $\Gamma_0(N)$  with  $\mathbb{Q}$ -rational coefficients. If  $p \nmid N$  is prime, then there is an A such that

$$\operatorname{ord}_p\left(q\frac{d}{dq}R|T(p^n)\right) \ge n-A.$$

Proof. For convenience, we let  $R(z) = \sum_{n \gg -\infty} a(n)q^n$ . We first show that the coefficients a(n) of R have bounded denominators. In other words, we have that  $A := \inf_n(\operatorname{ord}_p(a(n))) < \infty$ . Indeed, we can always multiply R with an appropriate power of (z) and a monic polynomial in j(z) with rational coefficients to obtain a cusp form of positive integer weight and rational coefficients. The resulting Fourier coefficients will have bounded denominators by Theorem 3.52 of [274]. One easily checks that dividing by the power of  $\Delta(z)$  and this polynomial in j(z) preserves the boundedness. The proposition now follows easily from

$$\left(q\frac{d}{dq}R\right)|T(p^n) = \sum_{m\gg\infty} \sum_{j=0}^{\min\{\operatorname{ord}_p(m),n\}} p^{n-j}ma(p^{n-2j}m)q^m.$$

*Remark.* Proposition 5.1 is analogous to Proposition 2.1 of [153] which concerns Atkin's U(p) operator.

Proof of Theorem 1.16. We first consider the case where E has CM. Suppose D < 0 is the discriminant of the imaginary quadratic field K. The nonzero coefficients of  $F_E(z)$  are supported on powers  $q^n$  with  $\chi_D(n) := \left(\frac{D}{n}\right) \neq -1$ . Let  $\varphi_D$  be the trivial character modulo |D|. We construct the modular function

$$\mathcal{Z}_E(z) = \frac{1}{2} \left( \widehat{\mathfrak{Z}}_E | \varphi_D + \widehat{\mathfrak{Z}}_E | \chi_D \right).$$
 (5.5)

Since the coefficients of the nonholomorphic part of  $\widehat{\mathfrak{Z}}_E(z)$  are supported on powers  $q^{-n}$  with  $\chi_D(-n) \neq 1$ , we see that the twisting action in the definition of  $\mathcal{Z}_E(z)$  kills the nonholomorphic part. Therefore,  $\mathcal{Z}_E(z)$  is a meromorphic modular function on  $\Gamma_0(ND^2)$  whose nonzero coefficients are supported on  $q^m$  where  $\chi_D(m) = 1$ , and are equal to the original coefficients of  $\widehat{\mathfrak{Z}}_E(z)$ .

We now aim to prove the following p-adic limits:

$$\lim_{n \to +\infty} \left[ q \frac{d}{dq} (\widehat{\mathfrak{Z}}_E(z)) \right] | T(p^n) = \lim_{n \to +\infty} \left[ q \frac{d}{dq} (\widehat{\mathfrak{Z}}_E(z) - \mathcal{Z}_E(z)) \right] | T(p^n) = 0.$$
(5.6)

By Proposition 5.1, the two limits are equal, and so it suffices to prove the vanishing of the second limit.

Since  $\chi_D(p^n) = 1$ , it follows that the coefficients of  $q^{p^n}$  (including  $q^1$ ) in  $\widehat{\mathfrak{Z}}_E^+(z) - \mathcal{Z}_E(z)$  all vanish. Therefore the coefficient of  $q^1$  for each n in the second limit of (5.6) is zero. Since the principal part of  $\widehat{\mathfrak{Z}}_E(z) - \mathcal{Z}_E(z)$  is  $q^{-1}$ , the principal parts in the second limit p-adically tend to 0 thanks to the definition of the Hecke operators  $T(p^n)$ .

Suppose that m > 1 is coprime to  $N_E$ . Then note that  $F_E$  is an eigenfunction for the Hecke operator T(m) with eigenvalue  $a_E(m)$ . Since the nonholomorphic part of  $\widehat{\mathfrak{Z}}_E(z)$  is the period integral of  $F_E(z)$ , it follows that  $Q_m(z) := m \widehat{\mathfrak{Z}}_E(z) |T(m) - a_E(m) \widehat{\mathfrak{Z}}_E(z) = m \widehat{\mathfrak{Z}}_E^+(z) |T(m) - a_E(m) \widehat{\mathfrak{Z}}_E^+(z)$  is a meromorphic modular function. Note that the functions  $q \frac{d}{dq} Q_m(z)$  have denominators that are bounded independently of m. This follows from the proof of Proposition 5.1 and the fact that (see Theorem 1.1 of [50])  $q \frac{d}{dq} \widehat{\mathfrak{Z}}_E(z)$ is a weight 2 meromorphic modular form. Since Hecke operators commute, we have

$$\left[q\frac{d}{dq}\widehat{\mathfrak{Z}}_{E}^{+}(z)\right]|T(p^{n})T(m) = \left[q\frac{d}{dq}(a_{E}(m)\widehat{\mathfrak{Z}}_{E}^{+}(z) + Q_{m}(z))\right]|T(p^{n}).$$

Modulo any fixed power of p, say  $p^t$ , Proposition 5.1 then implies that

$$\left[q\frac{d}{dq}\widehat{\mathfrak{Z}}_{E}^{+}(z)\right]|T(p^{n})T(m) \equiv a_{E}(m)\cdot\left[q\frac{d}{dq}\widehat{\mathfrak{Z}}_{E}^{+}(z)\right]|T(p^{n}) \left(\operatorname{mod} p^{t}\right),$$

for sufficiently large n. In other words, we have that  $\left[q\frac{d}{dq}\widehat{\mathfrak{Z}}_{E}^{+}(z)\right]|T(p^{n})$  is congruent to a Hecke eigenform for T(m) modulo  $p^{t}$  for sufficiently large n. By Proposition 5.1 again, we have that  $\left[q\frac{d}{dq}(\widehat{\mathfrak{Z}}_{E}^{+}(z) - \mathcal{Z}_{E}(z))\right]|T(p^{n})$  is an eigenform of T(m) modulo  $p^{t}$  for sufficiently large n. Obviously, this conclusion holds uniformly in n for all T(m) with  $gcd(m, N_{E}) = 1$ .

Generalizing this argument in the obvious way to incorporate Atkin's Uoperators (as in [153]), we conclude that these forms are eigenforms of all the Hecke operators. By the discussion above, combined with the fact that the constant terms vanish after applying  $q \frac{d}{dq}$ , these eigenforms are congruent to  $0+O(q^2) \pmod{p^t}$ . Such an eigenform must be identically  $0 \pmod{p^t}$ , thereby establishing (5.6).

To complete the proof in this case, we observe that  $p \nmid a_E(p^n)$  for any n. This follows from the recurrence relation on  $a_E(p^n)$  in n, combined with the fact that  $p \nmid a_E(p)$  since p is split in K. By (5.6) we have that

$$\lim_{n \to +\infty} \frac{\left[q \frac{d}{dq}(\widehat{\mathfrak{Z}}_E^+(z))\right] | T(p^n)}{a_E(p^n)} = 0.$$
(5.7)

The proof now follows from the identities

$$\widehat{\mathfrak{Z}}_{E}^{+}(z) = \zeta(\Lambda_{E}; \mathcal{E}_{E}(z)) - S(\Lambda_{E})\mathcal{E}_{E}(z) \text{ and } F_{E}(z) = q \frac{d}{dq} \mathcal{E}_{E}(z).$$

The proof for E without CM is nearly identical. We replace  $\widehat{\mathfrak{Z}}_{E}^{+}(z)$  by  $\widehat{\mathfrak{Z}}_{E}^{+}(z) + S(\Lambda_{E})\mathcal{E}_{E}(z)$ , which has  $\mathbb{Q}$ -rational coefficients. In (5.7) the limiting value of 0 is replaced by a constant multiple of  $F_{E}(z)$ .

### 5.2 Vector valued harmonic Maass forms

To ease exposition, the results in the introduction were stated using the classical language of half-integral weight modular forms. To treat the case of general levels and functional equations, it will be more convenient to work with vector-valued forms and certain Weil representations. Here we recall this framework, and we discuss important theta functions which will be required in the section to define the theta lift  $\mathcal{I}(\bullet; \tau)$ . We shall let  $q := e^{2\pi i \tau}$ . The modular parameter will always be clear in context. The need for multiple modular variables arises from the structure of the theta lift. As a rule of thumb,  $\tau$  shall be the modular variable for all the half-integral weight forms in the remainder of this paper.

For a positive integer N we consider the rational quadratic space of signature (1, 2) given by

$$V := \left\{ \lambda = \begin{pmatrix} \lambda_1 & \lambda_2 \\ \lambda_3 & -\lambda_1 \end{pmatrix}; \lambda_1, \lambda_2, \lambda_3 \in \mathbb{Q} \right\}$$

and the quadratic form  $Q(\lambda) := N \det(\lambda)$ . The associated bilinear form is  $(\lambda, \mu) = -N \operatorname{tr}(\lambda \mu)$  for  $\lambda, \mu \in V$ .

We let  $G = \operatorname{Spin}(V) \simeq SL_2$ , viewed as an algebraic group over  $\mathbb{Q}$  and write  $\overline{\Gamma}$  for its image in  $\operatorname{SO}(V) \simeq \operatorname{PSL}_2$ . By D we denote the associated symmetric space. It can be realized as the Grassmannian of lines in  $V(\mathbb{R})$  on which the quadratic form Q is positive definite,

$$D \simeq \{z \subset V(\mathbb{R}); \dim z = 1 \text{ and } Q|_z > 0\}.$$

Then the group  $SL_2(\mathbb{Q})$  acts on V by conjugation

$$g.\lambda := g\lambda g^{-1}$$

for  $\lambda \in V$  and  $g \in SL_2(\mathbb{Q})$ . In particular,  $G(\mathbb{Q}) \simeq SL_2(\mathbb{Q})$ .

We identify the symmetric space D with the upper-half of the complex plane  $\mathbb{H}$  in the usual way, and obtain an isomorphism between  $\mathbb{H}$  and D by

$$z \mapsto \mathbb{R}\lambda(z),$$

where, for z = x + iy, we pick as a generator for the associated positive line

$$\lambda(z) := \frac{1}{\sqrt{Ny}} \begin{pmatrix} -(z+\bar{z})/2 & z\bar{z} \\ -1 & (z+\bar{z})/2 \end{pmatrix}$$

The group G acts on  $\mathbb{H}$  by linear fractional transformations and the isomorphism above is G-equivariant. Note that  $Q(\lambda(z)) = 1$  and  $g.\lambda(z) = \lambda(gz)$  for  $g \in G(\mathbb{R})$ . Let  $(\lambda, \lambda)_z = (\lambda, \lambda(z))^2 - (\lambda, \lambda)$ . This is the minimal majorant of  $(\cdot, \cdot)$  associated with  $z \in D$ .

We can view  $\Gamma_0(N)$  as a discrete subgroup of Spin(V) and we write  $M = \Gamma_0(N) \setminus D$  for the attached locally symmetric space.

We identify the set of isotropic lines Iso(V) in  $V(\mathbb{Q})$  with  $P^1(\mathbb{Q}) = \mathbb{Q} \cup \{\infty\}$  via

$$\psi: P^1(\mathbb{Q}) \to \operatorname{Iso}(V), \quad \psi((\alpha:\beta)) = \operatorname{span}\left(\begin{pmatrix} \alpha\beta & \alpha^2\\ -\beta^2 & -\alpha\beta \end{pmatrix}\right)$$

The map  $\psi$  is a bijection and  $\psi(g(\alpha : \beta)) = g.\psi((\alpha : \beta))$ . Thus, the cusps of M (i.e. the  $\Gamma_0(N)$ -classes of  $P^1(\mathbb{Q})$ ) can be identified with the  $\Gamma_0(N)$ -classes of Iso(V).

If we set  $\ell_{\infty} := \psi(\infty)$ , then  $\ell_{\infty}$  is spanned by  $\lambda_{\infty} = \begin{pmatrix} 0 & 1 \\ 0 & 0 \end{pmatrix}$ . For  $\ell \in \operatorname{Iso}(V)$  we pick  $\sigma_{\ell} \in SL_2(\mathbb{Z})$  such that  $\sigma_{\ell}\ell_{\infty} = \ell$ .

Heegner points are given as follows. For  $\lambda \in V(\mathbb{Q})$  with  $Q(\lambda) > 0$  we let

$$D_{\lambda} = \operatorname{span}(\lambda) \in D.$$

For  $Q(\lambda) \leq 0$  we set  $D_{\lambda} = \emptyset$ . We denote the image of  $D_{\lambda}$  in M by  $Z(\lambda)$ .

#### 5.2.1 A lattice related to $\Gamma_0(\mathbf{N})$

We consider the lattice

$$L := \left\{ \begin{pmatrix} b & -a/N \\ c & -b \end{pmatrix}; \quad a, b, c \in \mathbb{Z} \right\}.$$

The dual lattice corresponding to the bilinear form  $(\cdot, \cdot)$  is given by

$$L' := \left\{ \begin{pmatrix} b/2N & -a/N \\ c & -b/2N \end{pmatrix}; \quad a, b, c \in \mathbb{Z} \right\}.$$

We identify the discriminant group  $L'/L =: \mathcal{D}$  with  $\mathbb{Z}/2N\mathbb{Z}$ , together with the  $\mathbb{Q}/\mathbb{Z}$  valued quadratic form  $x \mapsto -x^2/4N$ . The level of L is 4N.

For a fundamental discriminant  $\Delta \in \mathbb{Z}$  we will consider the rescaled lattice  $\Delta L$  together with the quadratic form  $Q_{\Delta}(\lambda) := \frac{Q(\lambda)}{|\Delta|}$ . The corresponding bilinear form is then given by  $(\cdot, \cdot)_{\Delta} = \frac{1}{|\Delta|}(\cdot, \cdot)$ . The dual lattice of  $\Delta L$  with respect to  $(\cdot, \cdot)_{\Delta}$  is equal to L'. We denote the discriminant group  $L'/\Delta L$  by  $\mathcal{D}(\Delta)$ .

For  $m \in \mathbb{Q}$  and  $h \in \mathcal{D}$ , we let

$$L_{m,h} = \{\lambda \in L+h; Q(\lambda) = m\}.$$

By reduction theory, if  $m \neq 0$  the group  $\Gamma_0(N)$  acts on  $L_{m,h}$  with finitely many orbits.

We will also consider the one-dimensional lattice  $K = \mathbb{Z} \begin{pmatrix} 1 & 0 \\ 0 & -1 \end{pmatrix} \subset L$ . We have  $L = K + \mathbb{Z}\ell + \mathbb{Z}\ell'$  where  $\ell$  and  $\ell'$  are the primitive isotropic vectors

$$\ell = \begin{pmatrix} 0 & 1/N \\ 0 & 0 \end{pmatrix}, \qquad \qquad \ell' = \begin{pmatrix} 0 & 0 \\ -1 & 0 \end{pmatrix}$$

Then  $K'/K \simeq L'/L$ .

# 5.2.2 The Weil representation and vector-valued automorphic forms

By  $\operatorname{Mp}_2(\mathbb{Z})$  we denote the integral metaplectic group. It consists of pairs  $(\gamma, \phi)$ , where  $\gamma = \begin{pmatrix} a & b \\ c & d \end{pmatrix} \in SL_2(\mathbb{Z})$  and  $\phi : \mathbb{H} \to \mathbb{C}$  is a holomorphic function with  $\phi^2(\tau) = c\tau + d$ . The group  $\widetilde{\Gamma} = \operatorname{Mp}_2(\mathbb{Z})$  is generated by  $S = (\begin{pmatrix} 0 & -1 \\ 1 & 0 \end{pmatrix}, \sqrt{\tau})$  and  $T = (\begin{pmatrix} 1 & 1 \\ 0 & 1 \end{pmatrix}, 1)$ . We let  $\widetilde{\Gamma}_{\infty} := \langle T \rangle \subset \widetilde{\Gamma}$ . We consider the Weil representation  $\rho_{\Delta}$  of  $\operatorname{Mp}_2(\mathbb{Z})$  corresponding to the discriminant group  $\mathcal{D}(\Delta)$  on the group ring  $\mathbb{C}[\mathcal{D}(\Delta)]$ , equipped with the standard scalar product  $\langle \cdot, \cdot \rangle$ , conjugate-linear in the second variable. We simply write  $\rho$  for  $\rho_1$ .

Let  $e(a) := e^{2\pi i a}$ . We write  $\mathfrak{e}_{\delta}$  for the standard basis element of  $\mathbb{C}[\mathcal{D}(\Delta)]$ corresponding to  $\delta \in \mathcal{D}(\Delta)$ . The action of  $\rho_{\Delta}$  on basis vectors of  $\mathbb{C}[\mathcal{D}(\Delta)]$  is given by the following formulas for the generators S and T of Mp<sub>2</sub>( $\mathbb{Z}$ )

$$\rho_{\Delta}(T)\mathbf{\mathfrak{e}}_{\delta} = e(Q_{\Delta}(\delta))\mathbf{\mathfrak{e}}_{\delta},$$

and

$$\rho_{\Delta}(S)\mathbf{e}_{\delta} = \frac{\sqrt{i}}{\sqrt{|\mathcal{D}(\Delta)|}} \sum_{\delta' \in \mathcal{D}(\Delta)} e(-(\delta', \delta)_{\Delta})\mathbf{e}_{\delta'}$$

Let  $k \in \frac{1}{2}\mathbb{Z}$ , and let  $A_{k,\rho_{\Delta}}$  be the vector space of functions  $f : \mathbb{H} \to \mathbb{C}[\mathcal{D}(\Delta)]$ , such that for  $(\gamma, \phi) \in \mathrm{Mp}_2(\mathbb{Z})$  we have

$$f(\gamma\tau) = \phi(\tau)^{2k} \rho_{\Delta}(\gamma, \phi) f(\tau).$$

A smooth function  $f \in A_{k,\rho_{\Delta}}$  is called a *harmonic (weak) Maass form of* weight k with respect to the representation  $\rho_{\Delta}$  if it satisfies in addition (see [53, Section 3]):

- 1.  $\Delta_k f = 0$ ,
- 2. the singularity at  $\infty$  is locally given by the pole of a meromorphic function.

Here we write  $\tau = u + iv$  with  $u, v \in \mathbb{R}$ , and

$$\Delta_k = -v^2 \left( \frac{\partial^2}{\partial u^2} + \frac{\partial^2}{\partial v^2} \right) + ikv \left( \frac{\partial}{\partial u} + i \frac{\partial}{\partial v} \right)$$
(5.8)

is the weight k Laplace operator. We denote the space of such functions by  $H_{k,\rho_\Delta}.$ 

By  $M_{k,\rho_{\Delta}}^{!} \subset H_{k,\rho_{\Delta}}$  we denote the subspace of weakly holomorphic modular forms. Recall that weakly holomorphic modular forms are meromorphic modular forms whose poles (if any) are supported at cusps. Similarly, we can define scalar-valued analogs of these spaces of automorphic forms. In this case, we require analogous conditions at all cusps of  $\Gamma_0(N)$  in (*ii*). We denote these spaces by  $H_k^+(N)$  and  $M_k^!(N)$ .

Note that the Fourier expansion of any harmonic Maass form uniquely decomposes into a holomorphic and a nonholomorphic part [53, Section 3]

$$f^{+}(\tau) = \sum_{h \in L'/L} \sum_{\substack{n \in \mathbb{Q} \\ n \gg -\infty}} c^{+}(n,h) q^{n} \mathfrak{e}_{h}$$
$$f^{-}(\tau) = \sum_{h \in L'/L} \sum_{n \in \mathbb{Q}} c^{-}(n,h) \Gamma(1-k,4\pi|n|v) q^{n} \mathfrak{e}_{h},$$

where  $\Gamma(a, x)$  denotes the incomplete  $\Gamma$ -function. The first summand is called the holomorphic part of f, the second one the nonholomorphic part.

We define a differential operator  $\xi_k$  by

$$\xi_k(f) := -2iv^k \overline{\frac{\partial}{\partial \bar{\tau}}f}.$$
(5.9)

We then have the following exact sequence [53, Corollary 3.8]

$$0 \longrightarrow M_{k,\rho_{\Delta}}^{!} \longrightarrow H_{k,\rho_{\Delta}} \xrightarrow{\xi_{k}} S_{2-k,\bar{\rho}_{\Delta}} \longrightarrow 0.$$

#### 5.2.3 Poincaré series and Whittaker functions

We recall some facts on Poincaré series with exponential growth at the cusps following Section 2.6 of [57].

We let  $k \in \frac{1}{2}\mathbb{Z}$ , and  $M_{\nu,\mu}(z)$  and  $W_{\nu,\mu}(z)$  denote the usual Whittaker functions (see p. 190 of [1]). For  $s \in \mathbb{C}$  and  $y \in \mathbb{R}_{>0}$  we put

$$\mathcal{M}_{s,k}(y) = y^{-k/2} M_{-\frac{k}{2},s-\frac{1}{2}}(y).$$

We let  $\Gamma_{\infty}$  be the subgroup of  $\Gamma_0(N)$  generated by  $\begin{pmatrix} 1 & 1 \\ 0 & 1 \end{pmatrix}$ . For  $k \in \mathbb{Z}, m \in \mathbb{N}$ ,  $z = x + iy \in \mathbb{H}$  and  $s \in \mathbb{C}$  with  $\Re(s) > 1$ , we define

$$F_m(z,s,k) = \frac{1}{2\Gamma(2s)} \sum_{\gamma \in \Gamma_\infty \setminus \Gamma_0(N)} \left[ \mathcal{M}_{s,k}(4\pi m y) e(-mx) \right] |_k \gamma.$$
(5.10)

This Poincaré series converges for  $\Re(s) > 1$ , and it is an eigenfunction of  $\Delta_k$ with eigenvalue  $s(1-s) + (k^2 - 2k)/4$ . Its specialization at  $s_0 = 1 - k/2$  is a harmonic Maass form [48, Proposition 1.10]. The principal part at the cusp  $\infty$  is given by  $q^{-m} + C$  for some constant  $C \in \mathbb{C}$ . The principal parts at the other cusps are constant.

We now define  $\mathbb{C}[L'/L]$ -valued analogs of these series. Let  $h \in L'/L$  and  $m \in \mathbb{Z} - Q(h)$  be positive. For  $k \in (\mathbb{Z} - \frac{1}{2})_{<1}$  we let

$$\mathcal{F}_{m,h}(\tau,s,k) = \frac{1}{2\Gamma(2s)} \sum_{\gamma \in \widetilde{\Gamma}_{\infty} \setminus \widetilde{\Gamma}} \left[ \mathcal{M}_{s,k}(4\pi m y) e(-mx) \mathfrak{e}_{h} \right]|_{k,\rho} \ \gamma.$$

The series  $\mathcal{F}_{m,h}(\tau, s, k)$  converges for  $\Re(s) > 1$  and it defines a harmonic Maass form of weight k for  $\widetilde{\Gamma}$  with representation  $\rho$ . The special value at s = 1 - k/2 is harmonic [48, Proposition 1.10]. For  $k \in \mathbb{Z} - \frac{1}{2}$  the principal part is given by  $q^{-m} \mathfrak{e}_h + q^{-m} \mathfrak{e}_{-h} + C$  for some constant  $C \in \mathbb{C}[L'/L]$ .

*Remark.* If we let (in the same setting as above)

$$\mathcal{F}_{m,h}(\tau,s,k) = \frac{1}{2\Gamma(2s)} \sum_{\gamma \in \widetilde{\Gamma}_{\infty} \setminus \widetilde{\Gamma}} \left[ \mathcal{M}_{s,k}(4\pi m y) e(-mx) \mathfrak{e}_{h} \right] |_{k,\bar{\rho}} \gamma,$$

then this has the same convergence properties. But for the special value at s = 1 - k/2, the principal part is given by  $q^{-m} \mathfrak{e}_h - q^{-m} \mathfrak{e}_{-h} + C$  for some constant  $C \in \mathbb{C}[L'/L]$ .

#### 5.2.4 Twisted theta series

We define a generalized genus character for  $\delta = \begin{pmatrix} b/2N & -a/N \\ c & -b/2N \end{pmatrix} \in L'$ . From now on let  $\Delta \in \mathbb{Z}$  be a fundamental discriminant and  $r \in \mathbb{Z}$  such that  $\Delta \equiv r^2 \pmod{4N}$ . Then

$$\chi_{\Delta}(\delta) = \chi_{\Delta}([a, b, Nc]) := \begin{cases} \left(\frac{\Delta}{n}\right), & \text{if } \Delta|b^2 - 4Nac \text{ and } (b^2 - 4Nac)/\Delta \text{ is a} \\ & \text{square mod } 4N \text{ and } \gcd(a, b, c, \Delta) = 1, \\ 0, & \text{otherwise.} \end{cases}$$

Here [a, b, Nc] is the integral binary quadratic form corresponding to  $\delta$ , and n is any integer prime to  $\Delta$  represented by [a, b, Nc].

The function  $\chi_{\Delta}$  is invariant under the action of  $\Gamma_0(N)$  and under the action of all Atkin-Lehner involutions. It can be computed by the following formula [148, Section I.2, Proposition 1]: If  $\Delta = \Delta_1 \Delta_2$  is a factorization of  $\Delta$  into discriminants and  $N = N_1 N_2$  is a factorization of N into positive factors such that  $(\Delta_1, N_1 a) = (\Delta_2, N_2 c) = 1$ , then

$$\chi_{\Delta}([a, b, Nc]) = \left(\frac{\Delta_1}{N_1 a}\right) \left(\frac{\Delta_2}{N_2 c}\right).$$

If no such factorizations of  $\Delta$  and N exist, we have  $\chi_{\Delta}([a, b, Nc]) = 0$ .

Since  $\chi_{\Delta}(\delta)$  depends only on  $\delta \in L'$  modulo  $\Delta L$ , we can view it as a function on the discriminant group  $\mathcal{D}(\Delta)$ .

We now let

$$\varphi_{\Delta}^{0}(\lambda, z) = p_{z}(\lambda)e^{-2\pi R(\lambda, z)/|\Delta|}, \qquad (5.11)$$

where  $p_z(\lambda) = (\lambda, \lambda(z))$  and  $R(\lambda, z) := \frac{1}{2}(\lambda, \lambda(z))^2 - (\lambda, \lambda)$ . This function was recently studied extensively by Hövel [166]. From now on, if  $\Delta = 1$ , we omit the index  $\Delta$  and simply write  $\varphi^0(\lambda, z)$ . Let  $\varphi(\lambda, \tau, z) = e^{2\pi i Q_\Delta(\lambda)\tau} \varphi^0_\Delta(\sqrt{v\lambda}, z)$ (for notational purposes we drop the dependence on *Delta*). By  $\pi$  we denote the canonical projection  $\pi : \mathcal{D}(\Delta) \to \mathcal{D}$ .

Moreover, we let  $\tilde{\rho} = \rho$ , if  $\Delta > 0$ , and  $\tilde{\rho} = \bar{\rho}$ , if  $\Delta < 0$ .

**Theorem 5.2.** The theta function

$$\Theta_{\Delta,r}(\tau,z,\varphi) := v^{1/2} \sum_{h \in \mathcal{D}} \sum_{\substack{\delta \in \mathcal{D}(\Delta) \\ \pi(\delta) = rh \\ Q_{\Delta}(\delta) \equiv \operatorname{sgn}(\Delta)Q(h)(\mathbb{Z})}} \chi_{\Delta}(\delta) \sum_{\lambda \in \Delta L + \delta} \varphi(\lambda,\tau,z) \mathfrak{e}_{h} \quad (5.12)$$

is a nonholomorphic  $\mathbb{C}[\mathcal{D}]$ -valued modular form of weight 1/2 for the representation  $\tilde{\rho}$  in the variable  $\tau$ . Furthermore, it is a nonholomorphic automorphic form of weight 0 for  $\Gamma_0(N)$  in the variable  $z \in D$ .

*Proof.* This follows from [166, Satz 2.8] and the results in [6].  $\Box$ 

We use the following representation for  $\Theta_{\Delta,r}(\tau, z, \varphi)$  as a Poincaré series using the lattice K. We let  $\epsilon = 1$ , when  $\Delta > 0$ , and  $\epsilon = i$ , when  $\Delta < 0$ . The following proposition can be found in [166, Satz 2.22].

**Proposition 5.3.** We have

$$\begin{split} \Theta_{\Delta,r}(\tau,z,\varphi) &= -\frac{Ny^2\bar{\epsilon}}{2i}\sum_{n=1}^{\infty}n\left(\frac{\Delta}{n}\right) \\ &\times \sum_{\gamma\in\widetilde{\Gamma}_{\infty}\setminus\widetilde{\Gamma}} \left[\frac{1}{v^{1/2}}e\left(-\frac{Nn^2y^2}{2i|\Delta|v}\right)\sum_{\lambda\in K'}e\left(\frac{\lambda^2}{2}|\Delta|\bar{\tau}-2nN\lambda x\right)\mathfrak{e}_{r\lambda}\right]|_{1/2,\widetilde{\rho_{K}}}\gamma. \end{split}$$

Now we define the theta kernel of the Shintani lift. Recall that for a lattice element  $\lambda \in L'/L$  we write  $\lambda = \begin{pmatrix} b/2N & -a/N \\ c & -b/2N \end{pmatrix}$ . Let

$$\varphi_{\rm Sh}(\lambda,\tau,z) = -\frac{cN\bar{z}^2 - b\bar{z} + a}{4Ny^2} e^{-2\pi v R(\lambda,z)/|\Delta|} e^{2\pi i Q_{\Delta}(\lambda)\tau}.$$

The Shintani theta function then transforms as follows.

**Theorem 5.4.** The theta function

$$\Theta_{\Delta,r}(\tau, z, \varphi_{Sh}) = v^{1/2} \sum_{h \in \mathcal{D}} \sum_{\substack{\delta \in \mathcal{D}(\Delta) \\ \pi(\delta) = rh \\ Q_{\Delta}(\delta) \equiv \operatorname{sgn}(\Delta)Q(h)(\mathbb{Z})}} \chi_{\Delta}(\delta) \sum_{\lambda \in \Delta L + \delta} \varphi_{Sh}(\lambda, \tau, z) \mathfrak{e}_{h}$$
(5.13)

is a nonholomorphic automorphic form of weight 2 for  $\Gamma_0(N)$  in the variable  $z \in D$ . Moreover,  $\overline{\Theta_{\Delta,r,h}(\tau, z, \varphi_{Sh})}$  is a nonholomorphic  $\mathbb{C}[\mathcal{D}]$ -valued modular form of weight 3/2 for the representation  $\overline{\widetilde{\rho}}$  in the variable  $\tau$ .

*Proof.* This follows from the results in [58, p. 142] and the results in [6].  $\Box$ 

We have the following relation between the two theta functions. This was already investigated in [53] and [37].

Lemma 5.5. We have

$$\xi_{1/2,\tau}\Theta_{\Delta,r}(\tau,z,\varphi) = 4i\sqrt{N}y^2\frac{\partial}{\partial z}\overline{\Theta_{\Delta,r}(\tau,z,\varphi_{Sh})}.$$

*Proof.* We first compute

$$\xi_{1/2,\tau}v^{1/2}\varphi(\lambda,\tau,z) = -v^{1/2}p_z(\lambda)e^{-2\pi vR(\lambda,z)/|\Delta|}e(-Q_\Delta(\lambda)\bar{\tau})\left(1 - 2\pi v\frac{R(\lambda,z)}{|\Delta|}\right)$$

For the derivative of complex conjugate of the Shintanti theta kernel we obtain

$$\begin{aligned} &-\frac{1}{4N}v^{1/2}e^{-2\pi vR(\lambda,z)/|\Delta|}e(-Q_{\Delta}(\lambda)\bar{\tau}) \\ &\times \left(\frac{\partial}{\partial z}y^{-2}(cNz^2-bz+a)+y^{-2}(cNz^2-bz+a)(-2\pi v)\frac{1}{|\Delta|}\frac{\partial}{\partial z}R(\lambda,z)\right) \\ &=\frac{i}{4\sqrt{N}y^2}v^{1/2}p_z(\lambda)e^{-2\pi vR(\lambda,z)/|\Delta|}e(-Q_{\Delta}(\lambda)\bar{\tau})\left(1-2\pi v\frac{R(\lambda,z)}{|\Delta|}\right),\end{aligned}$$

using that

$$\begin{aligned} &\frac{\partial}{\partial z}y^{-2}(cNz^2 - bz + a) = -i\sqrt{N}y^{-2}p_z(\lambda),\\ &\frac{\partial}{\partial z}R(\lambda, z) = -\frac{i}{2\sqrt{N}}y^{-2}p_z(\lambda)(cN\bar{z}^2 - b\bar{z} + a),\\ &y^{-2}(cNz^2 - bz + a)(cN\bar{z}^2 - b\bar{z} + a) = 2NR(\lambda, z). \end{aligned}$$

## 5.3 Theta lifts of harmonic Maass forms

Recall that  $\Delta$  is a fundamental discriminant and that  $r \in \mathbb{Z}$  is such that  $r^2 \equiv \Delta \pmod{4N}$ . Let F be a harmonic Maass form in  $H_0^+(N)$ . We define

the twisted theta lift of F as follows

$$\mathcal{I}_{\Delta,r}(\tau,F) = \int_M F(z)\Theta_{\Delta,r}(\tau,z,\varphi)d\mu(z).$$

**Theorem 5.6.** Let  $\Delta \neq 1$  and let F be a harmonic Maass form in  $H_0^+(N)$ with vanishing constant term at all cusps. Then  $\mathcal{I}_{\Delta,r}(\tau, F)$  is a harmonic Maass form of weight 1/2 transforming with respect to the representation  $\tilde{\rho}$ . Moreover, the theta lift is equivariant with respect to the action of O(L'/L).

To prove the theorem we establish a couple of results. Note that the transformation properties of the twisted theta function  $\Theta_{\Delta,r}(\tau, z, \varphi)$  directly imply that the lift transforms with representation  $\tilde{\rho}$ . The equivariance follows from [166, Proposition 2.7]. First we show that the lift is annihilated by the Laplace operator. Together with a result relating this theta lift to the Shintani lift, these results imply Theorem 5.6. We also compute the lift of Poincaré series and the constant function since this will be useful in Section 5.4. Further properties of this lift will be investigated in a forthcoming paper [5].

**Proposition 5.7.** Let F be a harmonic Maass form in  $H_0^+(N)$ . Then  $\mathcal{I}_{\Delta,r}(\tau, F)$  is well-defined and

$$\Delta_{1/2,\tau} \mathcal{I}_{\Delta,r}(\tau, F) = 0.$$

Proof. We first investigate the growth of the theta function  $\Theta_{\Delta,r}(\tau, z, \varphi) = \sum_{h \in L'/L} \theta_h(\tau, z, \varphi)$  in the cusps of M. For simplicity we let  $\Delta = N = 1$ . Then  $L = \mathbb{Z}^3$  and  $h = \begin{pmatrix} h' & 0 \\ 0 & h' \end{pmatrix}$  with h' = 0 or h' = 1/2. So we consider

$$\theta_h(\tau, z, \varphi) = \sum_{\substack{a, c \in \mathbb{Z} \\ b \in \mathbb{Z} + h'}} -\frac{v}{y} (c|z|^2 - bx + a) e^{-\frac{\pi v}{y} (c|z|^2 - bx + a)^2} e^{2\pi i \bar{\tau} (-b^2/4 + ac)}.$$

We apply Poisson summation on the sum over a. We consider the summands

as a function of a and compute the Fourier transform, i.e.

$$-\int_{-\infty}^{\infty} \frac{v}{y} (c|z|^2 - bx + a) e^{-\frac{\pi v}{y} (c|z|^2 - bx + a)^2} e^{2\pi i \bar{\tau} (-b^2/4 + ac)} e^{2\pi i wa} da$$
  
=  $-y e^{-\pi i \bar{\tau} b^2/2} e^{2\pi i (c\bar{\tau} + w)(bx - c|z|^2)} \int_{-\infty}^{\infty} t e^{-\pi t^2} e^{2\pi i t \frac{y}{\sqrt{v}} (c\bar{\tau} + w)} dt,$ 

where we set  $t = \frac{\sqrt{v}}{y}(c|z|^2 - bx + a)$ . Since the Fourier transform of  $xe^{-\pi x^2}$  is  $ixe^{-\pi x^2}$  this equals

$$-i\frac{y^2}{\sqrt{v}}e^{-\pi i\bar{\tau}b^2/2}e^{2\pi i(c\bar{\tau}+w)(bx-c|z|^2)}(c\bar{\tau}+w)e^{-\frac{\pi y^2}{v}(c\bar{\tau}+w)^2}$$
$$=-i\frac{y^2}{\sqrt{v}}(c\bar{\tau}+w)e^{-2\pi i\bar{\tau}(b/2-cx)^2}e^{2\pi i(bxw-cx^2w)}e^{-\frac{\pi y^2}{v}|c\tau+w|^2}$$

We obtain that

$$\theta_h(\tau, z, \varphi) = -\frac{y^2}{\sqrt{v}} \sum_{\substack{w, c \in \mathbb{Z} \\ b \in \mathbb{Z} + h'}} (c\bar{\tau} + w) e^{-2\pi i \bar{\tau} (b/2 - cx)^2} e^{2\pi i (bxw - cx^2w)} e^{-\frac{\pi y^2}{v} |c\tau + w|^2}.$$

If c and w are non-zero this decays exponentially, and if c = w = 0 it vanishes.

In general we obtain for  $h \in L'/L$  and at each cusp  $\ell$ 

$$\theta_h(\tau, \sigma_\ell z, \varphi) = O(e^{-Cy^2}), \quad \text{as } y \to \infty,$$

uniformly in x, for some constant C > 0.

Thus, the growth of  $\Theta_{\Delta,r}(\tau, z, \varphi)$  offsets the growth of F and the integral converges. By [166, Proposition 3.10] we have

$$\begin{split} \Delta_{1/2,\tau} \mathcal{I}_{\Delta,r}(\tau,F) &= \int_M F(z) \Delta_{1/2,\tau} \Theta_{\Delta,r}(\tau,z,\varphi) d\mu(z) \\ &= \frac{1}{4} \int_M F(z) \Delta_{0,z} \Theta_{\Delta,r}(\tau,z,\varphi) d\mu(z). \end{split}$$

By the rapid decay of the theta function we may move the Laplacian to F. Since  $F \in H_0^+(N)$  we have  $\Delta_{0,z}F = 0$ , which implies the vanishing of the integral. By  $\mathcal{I}_{\Delta,r}^{\mathrm{Sh}}(\tau, G)$  we denote the Shintani lifting of a cusp form G of weight 2 for  $\Gamma_0(N)$ . It is defined as

$$\mathcal{I}_{\Delta,r}^{\mathrm{Sh}}(\tau,G) = \int_{M} G(z) \overline{\Theta_{\Delta,r}(\tau,z,\varphi_{\mathrm{Sh}})} y^{2} d\mu(z)$$

We then have the following relation between the two theta lifts.

**Theorem 5.8.** Let  $F \in H_0^+(N)$  with vanishing constant term at all cusps. Then we have that

$$\xi_{1/2,\tau}\left(\mathcal{I}_{\Delta,r}(\tau,F)\right) = \frac{1}{2\sqrt{N}}\mathcal{I}_{\Delta,r}^{Sh}(\tau,\xi_{0,z}(F)).$$

*Proof.* By Stokes' theorem we have that

$$\begin{aligned} \mathcal{I}_{\Delta,r}^{\mathrm{Sh}}(\tau,\xi_{0,z}(F)) &= \int_{M} \xi_{0}(F(z)) \overline{\Theta_{\Delta,r}(\tau,z,\varphi_{\mathrm{Sh}})} y^{2} d\mu(z) \\ &= -\int_{M} \overline{F(z)} \xi_{2,z}(\Theta_{\Delta,r}(\tau,z,\varphi_{\mathrm{Sh}})) d\mu(z) + \lim_{t \to \infty} \int_{\partial \mathcal{F}_{t}} \overline{F(z)} \overline{\Theta_{\Delta,r}(\tau,z,\varphi_{\mathrm{Sh}})} d\bar{z}, \end{aligned}$$

where  $\mathcal{F}_t = \{z \in \mathbb{H} : \Im(z) \le t\}$  denotes the truncated fundamental domain. Lemma 5.5 implies that

$$-\int_{M} \overline{F(z)} \xi_{2,z}(\Theta_{\Delta,r}(\tau, z, \varphi_{\rm Sh})) d\mu(z)$$
  
$$= \frac{1}{2\sqrt{N}} \int_{M} \overline{F(z)} \xi_{1/2,\tau}(\Theta_{\Delta,r}(\tau, z, \varphi)) d\mu(z) = \frac{1}{2\sqrt{N}} \xi_{1/2,\tau}\left(\mathcal{I}_{\Delta,r}(\tau, F)\right)$$

It remains to show that

$$\lim_{t \to \infty} \int_{\partial \mathcal{F}_t} \overline{F(z)\Theta_{\Delta,r}(\tau, z, \varphi_{\rm Sh})} d\bar{z} = 0.$$

As in the proof of Proposition 5.7 we have to investigate the growth of the theta function in the cusps. We have (again,  $\Delta = N = 1$ ,  $L = \mathbb{Z}^3$ , and h' = 0, 1/2)

$$\Theta_{\Delta,r}(\tau, z, \varphi_{\mathrm{Sh}}) = \sum_{\substack{a,c \in \mathbb{Z} \\ b \in \mathbb{Z} + h'}} -\frac{c\bar{z}^2 - b\bar{z} + a}{4y^2} e^{-\frac{\pi v}{y^2}(c|z|^2 - bx + a)} e^{2\pi i\bar{\tau}(-b^2/4 + ac)},$$

and apply Poisson summation to the sum on a. Thus, we consider

$$\int_{-\infty}^{\infty} -\frac{c\bar{z}^2 - b\bar{z} + a}{4y^2} e^{-\frac{\pi v}{y^2}(c|z|^2 - bx + a)} e^{2\pi i\bar{\tau}(-b^2/4 + ac)} e^{2\pi iwa} da.$$

Proceeding as before, we obtain

$$\begin{aligned} \theta_h(\tau, z, \varphi_{\rm Sh}) &= -\frac{1}{4\sqrt{vy}} \sum_{\substack{w,c \in \mathbb{Z} \\ b \in \mathbb{Z} + h'}} e^{-2\pi i \bar{\tau}(b/2 - cx)^2} e^{2\pi i (bxw - cx^2w)} \\ &\times \left( c\bar{z}^2 + biy - c|z|^2 + i\frac{y^2}{v} (c\bar{\tau} + w) \right) e^{-\frac{\pi y^2}{v} |c\tau + w|^2}. \end{aligned}$$

If c and w are not both equal to 0 this vanishes in the limit as  $y \to \infty$ . In this case, the whole integral vanishes. But if c = w = 0 we have

$$-\frac{i}{4\sqrt{v}}\sum_{b\in\mathbb{Z}+h'}be^{\pi i\bar{\tau}b^2/2}.$$

Thus, we are left with (the complex conjugate of)

$$\int_{\partial \mathcal{F}_T} F(z) \Theta_{\Delta,r}(\tau, z, \varphi_{\mathrm{Sh}}) dz = \frac{i}{4\sqrt{v}} \sum_{b \in \mathbb{Z} + h'} b e^{\pi i \bar{\tau} b^2/2} \int_1^T \int_0^1 F(z) dx dy.$$

We see that

$$\lim_{T \to \infty} \int_{1}^{T} \int_{0}^{1} F(z) dx dy = 0$$

since the constant coefficient of F vanishes. Therefore,

$$\lim_{T \to \infty} \int_{\partial M_T} \overline{F(z)} \Theta_{\Delta,r}(\tau, z, \varphi_{\rm Sh}) d\bar{z} = 0.$$

Generalizing to arbitrary N, a similar result holds for the other cusps of M.

For a cusp form  $G = \sum_{n=1}^{\infty} b(n)q^n \in S_2^{\text{new}}(N)$  we let  $L(G, \Delta, s)$  be its twisted L-function

$$L(G, \Delta, s) = \sum_{n=1}^{\infty} \left(\frac{\Delta}{n}\right) b(n) n^{-s}.$$

The relation to the Shintani lifting directly implies

**Proposition 5.9.** Let  $F \in H_0^+(N)$  with vanishing constant term at all cusps and let  $\xi_{0,z}(F) = F_E \in S_2^{new}(N)$ . The lift  $\mathcal{I}_{\Delta,r}(\tau, F)$  is weakly holomorphic if and only if

$$L(F_E, \Delta, 1) = 0.$$

In particular, this happens if F is weakly holomorphic.

*Proof.* Clearly, the lift is weakly holomorphic if and only if the Shintani lifting of  $F_E$  vanishes. This is trivially the case when  $F_E = \xi_0(F) = 0$ , i.e. when F is weakly holomorphic. In the other case, the coefficients of the Shintani lifting are given by (in terms of Jacobi forms; for the definition of Jacobi forms and the cycle integral r see [148])

$$\mathcal{I}_{\Delta,r}^{\mathrm{Sh}}(\tau,\xi_{0,z}(F)) = \sum_{\substack{n,r_0 \in \mathbb{Z} \\ r_0^2 < 4nN}} r_{1,N,\Delta(r_0^2 - 4nN),rr_0,\Delta}(F_E) q^n \zeta^{r_0}.$$

Now by the Theorem and Corollary in Section II.4 in [148] we have

$$|r_{1,N,\Delta(r_0^2 - 4nN),rr_0,\Delta}(F_E)|^2 = \frac{1}{4\pi^2} |\Delta|^{1/2} |r_0^2 - 4nN|^{1/2} L(F_E,\Delta,1) L(F_E,r_0^2 - 4nN,1).$$

Since  $r_0$  and n vary this expression vanishes if and only if  $L(F_E, \Delta, 1)$  vanishes.

Proof of Theorem 5.6. Proposition 5.7 implies that an  $F \in H_0^+(N)$  with vanishing constant term at all cusps maps to a form of weight 1/2 transforming with representation  $\tilde{\rho}$  that is annihilated by the Laplace operator  $\Delta_{1/2,\tau}$ . Theorem 5.8 then implies, that the lift satisfies the correct growth conditions at all cusps.

#### 5.3.1 Fourier expansion of the holomorphic part

Now we turn to the computation of the Fourier coefficients of positive index of the holomorphic part of the theta lift. Let  $h \in L'/L$  and  $m \in \mathbb{Q}_{>0}$  with  $m \equiv \operatorname{sgn}(\Delta)Q(h)$  (Z). We define a twisted Heegner divisor on M by

$$Z_{\Delta,r}(m,h) = \sum_{\lambda \in \Gamma_0(N) \setminus L_{rh,m|\Delta|}} \frac{\chi_{\Delta}(\lambda)}{\left|\overline{\Gamma}_{\lambda}\right|} Z(\lambda).$$

Here  $\overline{\Gamma}_{\lambda}$  denotes the stabilizer of  $\lambda$  in  $\overline{\Gamma_0(N)}$ .

Let F be a harmonic Maass form of weight 0 in  $H_0^+(N)$ . Then the twisted modular trace function is defined as follows

$$\operatorname{tr}_{\Delta,r}(F;m,h) = \sum_{z \in Z_{\Delta,r}(m,h)} F(z) = \sum_{\lambda \in \Gamma \setminus L_{|\Delta|m},rh} \frac{\chi_{\Delta}(\lambda)}{|\bar{\Gamma}_{\lambda}|} f(D_{\lambda}).$$
(5.14)

Here we need to define a refined modular trace function. We let

$$L^+_{|\Delta|m,rh} = \left\{ \lambda = \begin{pmatrix} \frac{b}{2N} & -\frac{a}{N} \\ c & -\frac{b}{2N} \end{pmatrix} \in L_{|\Delta|m,rh} \, ; \, a \ge 0 \right\},$$

and similarly

$$L^{-}_{|\Delta|m,rh} = \left\{ \lambda = \begin{pmatrix} \frac{b}{2N} & -\frac{a}{N} \\ c & -\frac{b}{2N} \end{pmatrix} \in L_{|\Delta|m,rh}; \ -a > 0 \right\},$$

and define modular trace functions

$$\operatorname{tr}_{\Delta,r}^+(F;m,h) = \sum_{\lambda \in \Gamma \setminus L_{|\Delta|m,rh}^+} \frac{\chi_{\Delta}(\lambda)}{|\bar{\Gamma}_{\lambda}|} f(D_{\lambda})$$

and

$$\operatorname{tr}_{\Delta,r}^{-}(F;m,h) = \sum_{\lambda \in \Gamma \setminus L_{|\Delta|m,rh}^{-}} \frac{\operatorname{sgn}(\Delta)\chi_{\Delta}(\lambda)}{|\bar{\Gamma}_{\lambda}|} f(D_{\lambda}).$$

**Theorem 5.10.** Let F be a harmonic Maass form of weight 0 in  $H_0^+(N)$ , m > 0, and  $h \in L'/L$ . The coefficients of index (m, h) of the holomorphic part of the lift  $\mathcal{I}_{\Delta,r}(\tau, F)$  are given by

$$\frac{\sqrt{\Delta}}{2\sqrt{m}} \left( \operatorname{tr}_{\Delta,r}^+(F;m,h) - \operatorname{tr}_{\Delta,r}^-(F;m,h) \right).$$
 (5.15)

*Proof.* To ease notation we start proving the result when  $\Delta = 1$ . Using the arguments of the proof of Theorem 5.5 in [6] it is straightforward to later generalize to the case  $\Delta \neq 1$ .

We consider the Fourier expansion of  $\int_M F(z)\Theta(\tau, z, \varphi)d\mu(z)$ , namely

$$\sum_{h \in L'/L} \sum_{m \in \mathbb{Q}} \left( \sum_{\lambda \in L_{m,h}} \int_M F(z) v^{1/2} \varphi^0(\sqrt{v}\lambda, z) d\mu(z) \right) e^{2\pi i m \tau}.$$
 (5.16)

We denote the (m, h)-th coefficient of the holomorphic part of (5.16) by C(m, h). Using the usual unfolding argument implies that

$$\begin{split} C(m,h) &= \sum_{\lambda \in \Gamma \setminus L_{m,h}} \frac{1}{|\bar{\Gamma}_{\lambda}|} \int_{D} F(z) v^{1/2} \varphi^{0}(\sqrt{v}\lambda,z) d\mu(z) \\ &= \sum_{\lambda \in \Gamma \setminus L_{m,h}^{+}} \frac{1}{|\bar{\Gamma}_{\lambda}|} \int_{D} F(z) v^{1/2} \varphi^{0}(\sqrt{v}\lambda,z) d\mu(z) \\ &+ \sum_{\lambda \in \Gamma \setminus L_{m,h}^{-}} \frac{1}{|\bar{\Gamma}_{\lambda}|} \int_{D} F(z) v^{1/2} \varphi^{0}(\sqrt{v}\lambda,z) d\mu(z). \end{split}$$

Since  $\varphi^0(-\sqrt{v}\lambda, z) = -\varphi^0(\sqrt{v}\lambda, z)$  the latter summand equals

$$-\sum_{\lambda\in\Gamma\backslash L_{m,h}^{-}}\frac{1}{|\bar{\Gamma}_{-\lambda}|}\int_{D}F(z)v^{1/2}\varphi^{0}(-\sqrt{v}\lambda,z)d\mu(z).$$

As in [185] and [57] we rewrite the integral over D as an integral over  $G(\mathbb{R}) =$ SL<sub>2</sub>( $\mathbb{R}$ ). We normalize the Haar measure such that the maximal compact subgroup SO(2) has volume 1. We then have

$$\int_D F(z)\varphi^0(\sqrt{v}\lambda,z)d\mu(z) = \int_{G(\mathbb{R})} F(gi)\varphi^0(\pm\sqrt{v}\lambda,gi)dg, \quad \text{for } \lambda \in \Gamma \setminus L_{m,h}^{\pm}.$$

Note that in [185] it is assumed that  $SL_2(\mathbb{R})$  acts transitively on vectors of the same norm. This is not true. However,  $SL_2(\mathbb{R})$  acts transitively on vectors of the same norm satisfying a > 0. Therefore, there is a  $g_1 \in SL_2(\mathbb{R})$  such

that  $g_1^{-1} \cdot \lambda = \sqrt{m\lambda(i)}$  for  $\lambda \in L_{m,h}^+$ . Similarly, there is a  $g_1 \in \mathrm{SL}_2(\mathbb{R})$  such that  $g_1^{-1} \cdot (-\lambda) = \sqrt{m\lambda(i)}$  for  $\lambda \in L_{m,h}^-$ . So, we have

$$C(m,h) = \sum_{\lambda \in \Gamma \setminus L_{m,h}^+} \frac{1}{|\bar{\Gamma}_{\lambda}|} v^{1/2} \int_{G(\mathbb{R})} F(gg_1 i) \varphi^0 \left(\sqrt{v} \sqrt{m} g^{-1} \cdot \lambda(i), i\right) dg$$
$$- \sum_{\lambda \in \Gamma \setminus L_{m,h}^-} \frac{1}{|\bar{\Gamma}_{-\lambda}|} v^{1/2} \int_{G(\mathbb{R})} F(gg_1 i) \varphi^0 \left(\sqrt{v} \sqrt{m} g^{-1} \cdot \lambda(i), i\right) dg.$$

Using the Cartan decomposition of  $SL_2(\mathbb{R})$  we find proceeding as in [185] that

$$C(m,h) = \sum_{\lambda \in \Gamma \setminus L_{m,h}^+} \frac{1}{|\bar{\Gamma}_{\lambda}|} F(D_{\lambda}) v^{1/2} Y(\sqrt{mv}) - \sum_{\lambda \in \Gamma \setminus L_{m,h}^-} \frac{1}{|\bar{\Gamma}_{-\lambda}|} F(D_{-\lambda}) v^{1/2} Y(\sqrt{mv})$$

$$(5.17)$$

where

$$Y(t) = 4\pi \int_{1}^{\infty} \varphi^{0}(t\alpha(a)^{-1} \cdot \lambda(i), i) \frac{a^{2} - a^{-2}}{2} \frac{da}{a}.$$
 (5.18)

.

Here  $\alpha(a) = \begin{pmatrix} a & 0 \\ 0 & a^{-1} \end{pmatrix}$ . We have that

$$\varphi^0(t\alpha(a)^{-1}.\lambda(i),i) = t(a^2 + a^{-2})e^{-\pi t^2(a^2 - a^{-2})^2}$$

Substituting  $a = e^{r/2}$  we obtain that (5.18) equals

$$4\pi t \int_0^\infty \cosh(r)\sinh(r)e^{-4\pi t^2\sinh(r)^2}dr = \frac{1}{2t}$$

Thus, we have  $Y(\sqrt{mv}) = \frac{1}{2\sqrt{mv}}$  which implies that

$$C(m,h) = \frac{1}{2\sqrt{m}} \left( \sum_{\lambda \in \Gamma \setminus L_{m,h}^+} \frac{1}{|\bar{\Gamma}_{\lambda}|} F(D_{\lambda}) - \sum_{\lambda \in \Gamma \setminus L_{m,h}^-} \frac{1}{|\bar{\Gamma}_{\lambda}|} F(D_{\lambda}) \right),$$

since  $|\bar{\Gamma}_{\lambda}| = |\bar{\Gamma}_{-\lambda}|$  and  $D_{\lambda} = D_{-\lambda}$ .

Using the methods of [6] it is not hard to see that the (m, h)-th coefficient of the twisted lift is equal to

$$\frac{\sqrt{\Delta}}{2\sqrt{m}} \left( \sum_{\lambda \in \Gamma \setminus L_{m|\Delta|,rh}^+} \frac{\chi_{\Delta}(\lambda)}{|\bar{\Gamma}_{\lambda}|} F(D_{\lambda}) - \sum_{\lambda \in \Gamma \setminus L_{m|\Delta|,rh}^-} \frac{\chi_{\Delta}(-\lambda)}{|\bar{\Gamma}_{\lambda}|} F(D_{\lambda}) \right).$$

We have that  $\chi_{\Delta}(-\lambda) = \operatorname{sgn}(\Delta)\chi_{\Delta}(\lambda)$  which implies the result.

### 5.3.2 Lift of Poincaré series and constants

In this section we compute the lift of Poincaré series and the constant function in the case  $\Delta \neq 1$ . This will be useful for the computation of the principal part of the theta lift.

#### Theorem 5.11. We have

$$\mathcal{I}_{\Delta,r}(\tau, F_m(z, s, 0)) = \frac{2^{-s+1}i}{\Gamma(s/2)} \sqrt{\pi N|\Delta|} \bar{\epsilon} \sum_{n|m} \left(\frac{\Delta}{n}\right) \mathcal{F}_{\frac{m^2}{4Nn^2}|\Delta|, -\frac{m}{n}r}\left(\tau, \frac{s}{2} + \frac{1}{4}, \frac{1}{2}\right).$$

*Remark.* In particular, we have

$$\mathcal{I}_{\Delta,r}(\tau, F_m(z, 1, 0)) = i\bar{\epsilon}\sqrt{N|\Delta|} \sum_{n|m} \left(\frac{\Delta}{n}\right) \mathcal{F}_{\frac{m^2}{4Nn^2}|\Delta|, -\frac{m}{n}r}\left(\tau, \frac{3}{4}, \frac{1}{2}\right).$$

*Proof.* The proof follows the one in [47, Theorem 3.3] or [4, Theorem 4.3]. Using the definition of the Poincaré series (5.10) and an unfolding argument we obtain

$$\mathcal{I}_{\Delta,r}(\tau, F_m(z, s, 0)) = \frac{1}{\Gamma(2s)} \int_{\Gamma_{\infty} \setminus \mathbb{H}} \mathcal{M}_{s,0}(4\pi m y) e(-mx) \Theta_{\Delta,r}(\tau, z, \varphi) d\mu(z).$$

By Proposition 5.3 this equals

$$-\frac{\bar{\epsilon}N}{\Gamma(2s)2i}\sum_{n=1}^{\infty}\left(\frac{\Delta}{n}\right)n\sum_{\gamma\in\tilde{\Gamma}_{\infty}\setminus\tilde{\Gamma}}I(\tau,s,m,n)|_{1/2,\tilde{\rho}_{K}}\gamma,$$

where

$$I(\tau, s, m, n) = \int_{y=0}^{\infty} \int_{x=0}^{1} y^2 \mathcal{M}_{s,0}(4\pi my) e(-mx) \exp\left(-\frac{\pi n^2 N y^2}{|\Delta|v}\right) \\ \times v^{-1/2} \sum_{\lambda \in K'} e\left(|\Delta|Q(\lambda)\bar{\tau} - 2N\lambda nx\right) \mathfrak{e}_{r\lambda} \frac{dxdy}{y^2}.$$

Identifying  $K' = \mathbb{Z} \begin{pmatrix} 1/2N & 0 \\ 0 & -1/2N \end{pmatrix}$  we find that

$$\sum_{\lambda \in K'} e\left(|\Delta|Q(\lambda)\bar{\tau} - 2N\lambda nx\right) \mathfrak{e}_{r\lambda} = \sum_{b \in \mathbb{Z}} e\left(-|\Delta|\frac{b^2}{4N}\bar{\tau} - nbx\right) \mathfrak{e}_{rb}.$$

Inserting this in the formula for  $I(\tau, s, m, n)$ , and integrating over x, we see that  $I(\tau, s, m, n)$  vanishes whenever  $n \nmid m$  and the only summand occurs for b = -m/n, when  $n \mid m$ . Thus,  $I(\tau, s, m, n)$  equals

$$v^{-1/2}e\left(-|\Delta|\frac{m^2}{4Nn^2}\bar{\tau}\right) \cdot \int_{y=0}^{\infty} \mathcal{M}_{s,0}(4\pi my) \exp\left(-\frac{\pi n^2 Ny^2}{|\Delta|v}\right) dy \ \mathfrak{e}_{-rm/n}.$$
(5.19)

To evaluate the integral in (5.19) note that (see for example (13.6.3) in [1])

$$\mathcal{M}_{s,0}(4\pi my) = 2^{2s-1}\Gamma\left(s+\frac{1}{2}\right)\sqrt{4\pi my} \cdot I_{s-1/2}(2\pi my).$$

Substituting  $t = y^2$  yields

$$\int_{y=0}^{\infty} \mathcal{M}_{s,0}(4\pi my) \exp\left(-\frac{\pi n^2 N y^2}{|\Delta|v}\right) dy$$
  
=  $2^{2s-1} \Gamma\left(s + \frac{1}{2}\right) \int_{y=0}^{\infty} \sqrt{4\pi my} I_{s-1/2}(2\pi my) \exp\left(-\frac{\pi n^2 N y^2}{|\Delta|v}\right) dy$   
=  $2^{2s-1} \Gamma\left(s + \frac{1}{2}\right) \sqrt{m\pi} \int_{t=0}^{\infty} t^{-1/4} I_{s-1/2}(2\pi m t^{1/2}) \exp\left(-\frac{\pi n^2 N t}{|\Delta|v}\right) dt.$ 

The last integral is a Laplace transform and is computed in [116] (see (20) on p. 197). It equals

$$\frac{\Gamma\left(\frac{s}{2}+\frac{1}{2}\right)}{\Gamma\left(s+\frac{1}{2}\right)}(\pi m)^{-1}\left(\frac{\pi n^2 N}{|\Delta|v}\right)^{-1/4}\exp\left(\frac{\pi m^2|\Delta|v}{2n^2 N}\right)M_{-\frac{1}{4},\frac{s}{2}-\frac{1}{4}}\left(\frac{\pi m^2|\Delta|v}{n^2 N}\right).$$

Therefore, we have that  $I(\tau, s, m, n)$  equals

$$2^{2s-1}\Gamma\left(\frac{s}{2}+\frac{1}{2}\right)\sqrt{\frac{|\Delta|}{\pi N n^2}}e\left(-\frac{m^2|\Delta|u}{4n^2N}\right)\mathcal{M}_{s/2+1/4,1/2}\left(\frac{\pi m^2|\Delta|v}{n^2N}\right)\mathfrak{e}_{-rm/n}.$$

Putting everything together we obtain the following for the lift of  $F_m(z, s, 0)$ 

$$-\frac{2^{2s-2}\Gamma(s/2+1/2)\bar{\epsilon}}{\Gamma(2s)i}\sqrt{\frac{N|\Delta|}{\pi}}\sum_{n|m}\left(\frac{\Delta}{n}\right)$$

$$\times\sum_{\gamma\in\tilde{\Gamma}_{\infty}\backslash\tilde{\Gamma}}\left[e\left(-\frac{m^{2}|\Delta|u}{4Nn^{2}}\right)\mathcal{M}_{s/2+1/4,1/2}\left(\frac{\pi m^{2}|\Delta|v}{n^{2}N}\right)\mathfrak{e}_{-rm/n}\right]|_{1/2,\tilde{\rho}_{K}}\gamma$$

$$=-\frac{2^{-s+1}}{i\Gamma(s/2)}\sqrt{\pi N|\Delta|}\bar{\epsilon}\sum_{n|m}\left(\frac{\Delta}{n}\right)\mathcal{F}_{\frac{m^{2}}{4Nn^{2}}|\Delta|,-\frac{m}{n}r}\left(\tau,\frac{s}{2}+\frac{1}{4},\frac{1}{2}\right).$$

We define

$$\Theta_K(\tau) = \sum_{\lambda \in K'} e(Q(\lambda)\tau) \mathfrak{e}_{\lambda+K}.$$

**Theorem 5.12.** Let N = 1 and  $\Delta < 0$  (for  $\Delta > 0$  and N = 1 the lift vanishes),  $\epsilon_{\Delta}(n) = \left(\frac{\Delta}{n}\right)$  and  $L(\epsilon_{\Delta}, s)$  be the Dirichlet L-series associated with  $\epsilon_{\Delta}$ . We have

$$\mathcal{I}_{\Delta,r}(\tau,1) = \frac{\bar{\epsilon}\,i}{\pi} |\Delta| L\left(\epsilon_{\Delta},1\right) \Theta_{K}(\tau).$$

*Proof.* This result follows analogously to [55, Theorem 7.1, Corollary 7.2] and [6, Theorem 6.1]. We compute the lift of the nonholomorphic weight 0 Eisenstein series and then take residues at s = 1/2. Let  $z \in \mathbb{H}$ ,  $s \in \mathbb{C}$  and

$$\mathcal{E}_0(z,s) = \frac{1}{2} \zeta^*(2s+1) \sum_{\gamma \in \Gamma_\infty \setminus SL_2(\mathbb{Z})} (\Im(\gamma z))^{s+\frac{1}{2}},$$

where  $\zeta^*(s)$  is the completed Riemann Zeta function. The Eisenstein series  $\mathcal{E}_0(z,s)$  has a simple pole at  $s = \frac{1}{2}$  with residue  $\frac{1}{2}$ . Using the standard

unfolding trick we obtain

$$\mathcal{I}_{\Delta,r}(\tau,\mathcal{E}_0(z,s)) = \zeta^*(2s+1) \int_{\Gamma_\infty \setminus \mathbb{H}} \Theta_{\Delta,r}(\tau,z,\varphi) y^{s+\frac{1}{2}} d\mu(z).$$

By Proposition 5.3 we have that this equals

$$\begin{split} &-\zeta^*(2s+1)\frac{\bar{\epsilon}}{2i}\sum_{n\geq 1}n\left(\frac{\Delta}{n}\right)\sum_{\gamma\in\tilde{\Gamma}_{\infty}\backslash\tilde{\Gamma}}\phi(\tau)^{-1}\tilde{\rho}_K^{-1}(\gamma)\frac{1}{\Im(\gamma\tau)^{1/2}}\\ &\times\int_{y=0}^{\infty}y^{s+\frac{1}{2}}\exp\left(-\frac{\pi n^2y^2}{|\Delta|\Im(\gamma\tau)}\right)dy\\ &\times\int_{x=0}^{1}\sum_{\lambda\in K'}e\left(\frac{\lambda^2\bar{\tau}}{2|\Delta|}-2\lambda nx\right)\mathfrak{e}_{r\lambda}dx.\end{split}$$

The integral over x equals  $\mathfrak{e}_0$  and the one over y equals

$$\frac{1}{2}\Gamma\left(\frac{s}{2}+\frac{3}{4}\right)(|\Delta|\Im(\gamma\tau))^{\frac{s}{2}+\frac{3}{4}}\pi^{-\frac{s}{2}-\frac{3}{4}}n^{-s-\frac{3}{2}}.$$

Thus, we have

$$\mathcal{I}_{\Delta,r}(\tau,\mathcal{E}_0(z,s)) = -\zeta^*(2s+1)\frac{\overline{\epsilon}}{2i}\Gamma\left(\frac{s}{2}+\frac{3}{4}\right)|\Delta|^{\frac{s}{2}+\frac{3}{4}}\pi^{-\frac{s}{2}-\frac{3}{4}}$$
$$\times L\left(\epsilon_{\Delta},s+\frac{1}{2}\right) \frac{1}{2}\sum_{\gamma\in\widetilde{\Gamma}_{\infty}\setminus\widetilde{\Gamma}}(v^{\frac{1}{2}(s+\frac{1}{2})}\mathfrak{e}_0)|_{1/2,K}\gamma.$$

We now take residues at s = 1/2 on both sides. Note that the residue of the weight 1/2 Eisenstein series is given by (see [171, Proof of Proposition 5.14])

$$\operatorname{res}_{s=1/2}\left(\frac{1}{2}\sum_{\gamma\in\widetilde{\Gamma}_{\infty}\setminus\widetilde{\Gamma}}(v^{\frac{1}{2}(s+\frac{1}{2})}\mathfrak{e}_{0})|_{1/2,K}\gamma\right)=\frac{6}{\pi}\Theta_{K}(\tau).$$

We have  $\zeta^*(2) = \pi/6$  which concludes the proof of the theorem.

134

# 5.4 General version of Theorem 1.17 and its proof

Here we give the general version of Theorem 1.17, give its proof, and then conclude with some numerical examples.

We begin with some notation. Let L be the lattice of discriminant 2Ndefined in Section 5.2.1 and let  $\rho = \rho_1$  be as in Section 5.2.2. Let  $F_E \in S_2^{new}(\Gamma_0(N_E))$  be a normalized newform of weight 2 associated to the elliptic curve  $E/\mathbb{Q}$ . Let  $\epsilon \in \{\pm 1\}$  be the eigenvalue of the Fricke involution on  $F_G$ . If  $\epsilon = 1$ , we put  $\rho = \bar{\rho}$  and assume that  $\Delta$  is a negative fundamental disriminant. If  $\epsilon = -1$  we put  $\rho = \rho$  and assume that  $\Delta$  is a positive fundamental discriminant. There is a newform  $g_E \in S_{3/2,\rho}^{new}$  mapping to  $F_E$ under the Shimura correspondence. We may normalize  $g_E$  such that all its coefficients are contained in  $\mathbb{Q}$ .

Recall that

$$\widehat{\mathfrak{Z}}_E(z) = \zeta(\Lambda_E; \mathcal{E}_E(z)) - S(\Lambda_E)\mathcal{E}_E(z) - \frac{\deg(\phi_E)}{4\pi ||F_E||^2}\overline{\mathcal{E}_E(z)},$$

and  $M_E(z)$  is chosen such that  $\widehat{\mathfrak{Z}}_E(z) - M_E(z)$  is holomorphic on  $\mathbb{H}$ . By  $a_{\ell,\widehat{\mathfrak{Z}}_E}(0)$  and  $a_{\ell,M_E}(0)$  we denote the constant terms of these two functions at the cusp  $\ell$ .

We then let

$$\widehat{\mathfrak{Z}}_{E}^{*}(z) = \frac{1}{\sqrt{|\Delta|N}} \left( \widehat{\mathfrak{Z}}_{E}(z) - \sum_{\ell \in \Gamma \setminus \operatorname{Iso}(V)} a_{\ell,\widehat{\mathfrak{Z}}_{E}}(0) \right).$$

Analogously, we let

$$M_E^*(z) = \frac{1}{\sqrt{|\Delta|N}} \left( M_E(z) - \sum_{\ell \in \Gamma \setminus \operatorname{Iso}(V)} a_{\ell,M_E}(0) \right).$$

Then  $\widehat{\mathfrak{Z}}_{E}^{*}(z) - M_{E}^{*}(z)$  is a harmonic Maass form of weight 0.
By  $f_{E,\Delta,r} = f_E$  we denote the twisted theta lift of  $\widehat{\mathfrak{Z}}_E^*(z) - M_E^*(z)$  as in Section 5.3.

We begin with some notation. Let L be the lattice of discriminant 2Ndefined in Section 5.2.1 and let  $\rho = \rho_1$  be as in Section 5.2.2. Let  $k \in \frac{1}{2}\mathbb{Z} \setminus \mathbb{Z}$ . The space of vector-valued holomorphic modular forms  $M_{k,\bar{\rho}}$  is isomorphic to the space of skew holomorphic Jacobi forms  $J_{k+1/2,N}^{skew}$  of weight k + 1/2 and index N. Moreover,  $M_{k,\rho}$  is isomorphic to the space of holomorphic Jacobi forms  $J_{k+1/2,N}$ . The subspace  $S_{k,\bar{\rho}}^{new}$  of newforms of the cusp forms  $S_{k,\bar{\rho}}$  is isomorphic as a module over the Hecke algebra to the space of newforms  $S_{2k-1}^{new,+}(\Gamma_0(N))$  of weight 2k-1 for  $\Gamma_0(N)$  on which the Fricke involution acts by multiplication with  $(-1)^{k-1/2}$ . The isomorphism is given by the Shimura correspondence [273]. Similarly, the subspace  $S_{k,\rho}^{new}$  of newforms of  $S_{k,\rho}$  is isomorphic as a module over the Hecke algebra to the space of newforms  $S_{2k-1}^{new,-}(\Gamma_0(N))$  of weight 2k-1 for  $\Gamma_0(N)$  on which the Fricke involution acts by multiplication with  $(-1)^{k-1/2}$ . The isomorphism is given by the Shimura correspondence [273]. Similarly, the subspace  $S_{k,\rho}^{new}$  of newforms of  $S_{k,\rho}$  is isomorphic as a module over the Hecke algebra to the space of newforms  $S_{2k-1}^{new,-}(\Gamma_0(N))$  of weight 2k-1 for  $\Gamma_0(N)$  on which the Fricke involution acts by multiplication with  $(-1)^{k+1/2}$  [148]. Let  $\epsilon$  be the eigenvalue of the Fricke involution on G.

The Hecke *L*-series of any  $G \in S_{2k-1}^{new,\pm}(\Gamma_0(N))$  satisfies a functional equation under  $s \mapsto 2k - 1 - s$  with root number  $-\epsilon$ . If  $G \in S_{2k-1}^{new,\pm}(\Gamma_0(N))$  is a normalized newform (in particular a common eigenform of all Hecke operators), we denote by  $F_G$  the number field generated by the Hecke eigenvalues of *G*. It is well known that we may normalize the preimage of *G* under the Shimura correspondence such that all its Fourier coefficients are contained in  $F_G$ .

**Theorem 5.13.** Assume that  $E/\mathbb{Q}$  is an elliptic curve of square free conductor  $N_E$ , and suppose that  $F_E|_2W_{N_E} = \epsilon F_E$ . Denote the coefficients of  $f_E(\tau)$ by  $c_E^{\pm}(h, n)$ . Then the following are true:

(i) If  $d \neq 1$  is a fundamental discriminant and  $r \in \mathbb{Z}$  such that  $d \equiv$ 

 $r^2 \pmod{4N_E}$ , and  $\epsilon d < 0$ , then

$$L(E_d, 1) = 8\pi^2 ||F_E||^2 ||g_E||^2 \sqrt{\frac{|d|}{N_E}} \cdot c_E^-(\epsilon d, r)^2.$$

(ii) If  $d \neq 1$  is a fundamental discriminant and  $r \in \mathbb{Z}$  such that  $d \equiv r^2 \pmod{4N_E}$  and  $\epsilon d > 0$ , then

$$L'(E_d, 1) = 0 \quad \iff \quad c_E^+(\epsilon d, r) \in \overline{\mathbb{Q}} \quad \iff \quad c_E^+(\epsilon d, r) \in \mathbb{Q}.$$

*Remark.* In contrast to Bruinier and Ono in [47] we are able to relate the weight 1/2 form to the elliptic curve in a direct way.

*Proof.* To prove Theorem 5.13, we shall employ the results in Section 7 in [47]. It suffices to prove that  $f_E$  can be taken for f in Theorem 7.6 and 7.8 in [47]. Therefore, we need to prove that  $f_E$  has rational principal part and that  $\xi_{1/2}(f_E) \in \mathbb{R}g$ , where g is the preimage of  $F_E$  under the Shimura lift. (In the case we consider it suffices to require that  $\xi_{1/2}(f) \in \mathbb{R}g$  in [47, Theorem 7.6].)

We first prove that  $f_E$  has rational principal part at the cusp  $\infty$ . We write  $\widehat{\mathfrak{Z}}_E^*(z) - M_E^*(z)$  as a linear combination of Poincaré series and constants, i.e.

$$\begin{aligned} \widehat{\mathfrak{Z}}_{E}^{*}(z) - M_{E}^{*}(z) &= C + \frac{1}{\sqrt{|\Delta|N}} \sum_{m>0} a_{\widehat{\mathfrak{Z}}_{E}}(-m) F_{m}(z,1,0) \\ &+ \frac{1}{\sqrt{|\Delta|N}} \sum_{k>0} a_{M_{E}}(-k) F_{k}(z,1,0). \end{aligned}$$

Here C is a constant and the coefficients  $a_{\mathfrak{J}_E}(-m)$  and  $a_{M_E}(-k)$  are rational by construction.

Then, by Theorem 5.11 and Theorem 5.12 the coefficients of the principal part of  $f_E$  are rational. For the other cusps of  $\Gamma_0(N)$  this follows by the equivariance of the theta lift under O(L'/L) and the fact that we can identify O(L'/L) with the group generated by the Atkin-Lehner involutions. By construction we have

$$\xi_0\left(\widehat{\mathfrak{Z}}_E^*(z) - M_E^*(z)\right) = \frac{-\mathrm{deg}(\phi_E)}{\sqrt{|\Delta|N||F_E||^2}}F_E.$$

At the same time Theorem 5.8 implies that

$$\mathcal{I}_{\Delta,r}^{\mathrm{Sh}}\left(\frac{-\mathrm{deg}(\phi_E)}{\sqrt{|\Delta|N||F_E||^2}}F_E\right) = 2\sqrt{N}\xi_{1/2}(f_E)$$

Thus, we have that  $\xi_{1/2}(f_E) \in \mathbb{R}g$ .

### 5.5 Examples

Here we give examples which illustrate the results proved in this paper.

*Example* 5.14. For  $X_0(11)$ , we have a single isogeny class. The strong Weil curve

$$E: y^2 + y = x^3 - x^2 - 10x - 20,$$

has sign of the functional equation equal to +1 and the Mordell-Weil group  $E(\mathbb{Q})$  has rank 0. In terms of Dedekind's eta-function, we have that

$$F_E(z) = \eta^2(z)\eta^2(11z) = q - 2q^2 - q^3 + 2q^4 + q^5 + 2q^6 - 2q^7 - 2q^9 - 2q^{10} + q^{11} - \dots$$

We find that the corresponding mock modular form  $\widehat{\mathfrak{Z}}^+_E(z)$  is

$$\widehat{\mathfrak{Z}}_{E}^{+}(z) = q^{-1} + 1 + 0.9520...q + 1.5479...q^{2} + 0.3493...q^{3} + 1.9760...q^{4} - O(q^{5}).$$

The apparent transcendence of these coefficients arise from  $S(\Lambda_E) = 0.381246...$ . We find that  $\Omega_{11}(F_E) = 0.2538418...$  which is 1/5 of the real period of E. This 1/5 is related to the fact that the Mordell-Weil group has a cyclic torsion subgroup of order 5. A short calculation shows that the expansion of  $\mathfrak{Z}_E(z)$  at the cusp zero is given by

$$\widehat{\mathfrak{Z}}_{E}^{+}(z)|_{0}\begin{pmatrix} 0 & -1\\ 11 & 0 \end{pmatrix} = \widehat{\mathfrak{Z}}_{E}^{+}(z)|U(11) + \frac{12}{5}.$$

138

In particular, the constant term is 17/5.

We see that p = 5 is ordinary for  $X_0(11)$ . Here we illustrate Theorem 1.16. As a 5-adic expansion we have that

$$\mathfrak{S}_E(5) = 4 + 2 \cdot 5^2 + 4 \cdot 5^3 + \dots$$

which can be thought of as a 5-adic expansion of  $S(\Lambda_E)$  given above. It turns out that

$$\lim_{n \to +\infty} \frac{\left\lfloor q \frac{d}{dq} \zeta(\Lambda_E; \mathcal{E}_E(z)) \right\rfloor | T(5^n)}{a_E(5^n)} = \mathfrak{S}_E(5) F_E(z)$$

as a 5-adic limit. To illustrate this phenomenon, we let

$$T_n(E,z) := \frac{\left[q\frac{d}{dq}\zeta(\Lambda_E;\mathcal{E}_E(z))\right]|T(5^n)}{a_E(5^n)}.$$

We then have that

$$T_1(E,z) - 4F_E(z) = -5q^{-5} - \frac{50}{3}q - \frac{65}{3}q^2 + \dots \equiv 0 \pmod{5}$$

$$T_2(E,z) - (4+0\cdot 5)F_E(z) = \frac{25}{4}q^{-25} - \frac{25}{6}q + \frac{925}{3}q^2 - \dots \equiv 0 \pmod{5^2}$$

$$\vdots T_4(E,z) - (4+2\cdot 5^2 + 4\cdot 5^3)F_E(z) = -\frac{625}{11}q^{-625} + \frac{5^4 \cdot 61301717918}{33}q + \dots \equiv 0 \pmod{5^4}$$

*Example 5.15.* Here we illustrate Theorem 1.17 using the following numerical example computed by Strömberg [51]. We consider the elliptic curve 37a1 given by the Weierstrass model

$$E: y^2 + y = x^3 - x.$$

The sign of the functional equation of L(E, s) is -1, and  $E(\mathbb{Q})$  has rank 1. The *q*-expansion of  $F_E(z)$  begins with the terms

$$F_E(z) = q - 2q^2 - 3q^3 + 2q^4 - 2q^5 + 6q^6 - q^7 + 6q^9 + 4q^{10} - \dots \in S_2^{new}\left(\Gamma_0(37)\right).$$

Using Remark 3, we find that the corresponding mock modular form is

$$\widehat{\mathfrak{Z}}_{E}^{+}(z) = q^{-1} + 1 + 2.1132...q + 2.3867...q^{2} + 4.2201...q^{3} + 5.5566...q^{4} + 8.3547...q^{5} + O(q^{6}).$$
  
It turns out that the weight 1/2 harmonic Maass form  $f_{E}(z) = \mathcal{I}_{-3}(\tau, \widehat{\mathfrak{Z}}_{E}^{+}(z))$   
corresponds to the Poincaré series  $\mathcal{M}_{-3/148,21}$  (see Section 5.2.3)). Using  
Sage [277], Strömberg and Bruinier computed all values of  $L'(E_{d}, 1)$  for fun-  
damental discriminants  $d > 0$  such that  $\left(\frac{d}{37}\right) = 1$  and  $|d| \leq 15000$ . The  
following table illustrates Theorem 1.17.

d	$c^+(d)$	$L'(E_d,1)$	$\operatorname{rk}(E_d(\mathbb{Q}))$
1	$-0.2817617849\ldots$	$0.3059997738\ldots$	1
12	$-0.4885272382\dots$	4.2986147986	1
21	$-0.1727392572\ldots$	9.0023868003	1
28	$-0.6781939953\dots$	4.3272602496	1
33	$0.5663023201\ldots$	$3.6219567911\ldots$	1
:		÷	:
1489	9	0	3
:		:	
4393	66	0	3

Stephan Ehlen numerically confirmed that  $c^+(d) = \frac{1}{2\sqrt{d}} \left( \operatorname{tr}_{-3}^+(\widehat{\mathfrak{Z}}_E^+(z);d) - \operatorname{tr}_{-3}^-(\widehat{\mathfrak{Z}}_E^+(z);d) \right)$  using Sage [277].

*Example* 5.16. In [308] Zagier defines the generating functions for the twisted traces of the modular invariant. For coprime fundamental discriminants d < 0 and D > 1, he sets

$$f_d = q^{-d} + \sum_{D>0} \left( \frac{1}{\sqrt{D}} \sum_{Q \in \mathcal{Q}_{dD} \setminus \Gamma} \chi(Q) j(\alpha_Q) \right) q^D,$$

where  $\mathcal{Q}_{dD}$  are the quadratic forms of discriminant dD,  $\chi(Q) = \left(\frac{D}{p}\right)$ , where p is a prime represented by Q and  $\alpha_Q$  is the corresponding CM-point.

With  $d = -\Delta$  and D = m we rediscover a vector-valued version of his results. For example

$$\mathcal{I}_{-3}(\tau, j - 744) = f_3 = q^{-3} - 248q + 26752q^4 - 85995q^5 + 1707264q^8 - 4096248q^9 + \cdots$$

# Chapter 6

# SU(2)-Donaldson Invariants

## 6.1 Some relevant classical functions

Here we fix notation concerning theta functions, and we recall a few standard facts about Dedekind's eta-function and the *nearly* modular Eisenstein series  $E_2(\tau)$ . We use the following normalization for the Jacobi theta function

$$\vartheta_{ab}(v|\tau) = \sum_{n \in \mathbb{Z}} q^{\frac{(2n+a)^2}{8}} e^{\pi i \, (2n+a)(v+\frac{b}{2})},\tag{6.1}$$

where  $a, b \in \{0, 1\}$ ,  $v \in \mathbb{C}$ ,  $q = \exp(2\pi i \tau)$ ,  $\tau = x + iy \in \mathbb{H}$ , and  $\mathbb{H}$  is the complex upper half-plane. The relation to the standard Jacobi theta functions is summarized in the following table:

$\label{eq:phi_1} \boxed{\vartheta_1(v \tau) = \vartheta_{11}(v \tau)}$	$\vartheta_1(0 \tau) = 0$	$\vartheta_1'(0 \tau) = -2\pi\eta^3(\tau)$
$\vartheta_2(v \tau) = \vartheta_{10}(v \tau)$	$\vartheta_2(0 \tau) = \sum_{n \in \mathbb{Z}} q^{\frac{(2n+1)^2}{8}}$	$\vartheta_2'(0 \tau) = 0$
$\vartheta_3(v \tau) = \vartheta_{00}(v \tau)$	$\vartheta_3(0 \tau) = \sum_{n \in \mathbb{Z}} q^{\frac{n^2}{2}}$	$\vartheta_3'(0 \tau)=0$
$\vartheta_4(v \tau) = \vartheta_{01}(v \tau)$	$\vartheta_4(0 \tau) = \sum_{n \in \mathbb{Z}} (-1)^n q^{\frac{n^2}{2}}$	$\vartheta_4'(0 \tau) = 0$
		(6.1)

Here  $\eta(\tau)$  is the Dedekind eta-function with

$$\eta^{3}(\tau) = \sum_{n=0}^{\infty} (-1)^{n} (2n+1) q^{\frac{(2n+1)^{2}}{8}}.$$
 (6.3)

We will also use the notation  $\vartheta_j(\tau) = \vartheta_j(0|\tau)$  for j = 2, 3, 4, and

$$\vartheta_2(\tau) = 2\Theta_2\left(\frac{\tau}{8}\right), \quad \vartheta_3(\tau) = \Theta_3\left(\frac{\tau}{8}\right), \quad \vartheta_4(\tau) = \Theta_4\left(\frac{\tau}{8}\right).$$
 (6.4)

Also, we have that  $E_2(\tau)$  is the normalized *nearly modular* weight 2 Eisenstein series

$$E_2(\tau) := 1 - 24 \sum_{n=1}^{\infty} \sum_{d|n} d q^n .$$
(6.5)

## 6.2 SU(2)-Donaldson invariants on $\mathbb{CP}^2$

Here we recall a closed formula expression for these Donaldson invariants which is due to Göttsche and his collaborators [141, 143], and we recall the conjecture of Moore and Witten in this case. We then conclude this section with Theorem 6.5 which we shall use to prove Theorem 1.20.

The Donaldson invariants of a smooth, compact, oriented, simply connected Riemannian four-manifold (X, g) without boundary are defined by using intersection theory on the moduli space of anti-self-dual instantons for the gauge groups SU(2) and SO(3) [142]. Given a homology orientation some cohomology classes on the instanton moduli space can be associated to homology classes of X through the slant product and then evaluated on a fundamental class. We define

$$\mathbf{A}(X) := \operatorname{Sym}(H_0(X, \mathbb{Z}) \oplus H_2(X, \mathbb{Z})),$$

and we regard the Donaldson invariants as the functional

$$\mathcal{D}_{w_2(E)}^{X,g}: \mathbf{A}(X) \to \mathbb{Q} , \qquad (6.6)$$

where  $w_2(E) \in H^2(X, \mathbb{Z}_2)$  is the second Stiefel-Whitney class of the gauge bundles which are considered. Since X is simply connected, there is an integer class  $2\lambda_0 \in H^2(\mathbb{CP}^2, \mathbb{Z})$  that is not divisible by two and whose modtwo reduction is  $w_2(E)$ . Let  $\{s_i\}_{i=1,\dots,b_2}$  be a basis of the two-cycles of X. We introduce the formal sum  $S = \sum_{i=1}^{b_2} \kappa^i s_i$ , where  $\kappa^i$  are complex numbers. The generator of the zero-class of X will be denoted by  $\mathbf{x} \in H_0(X, \mathbb{Z})$ . The Donaldson-Witten generating function is

$$Z_{\rm DW}(p,\kappa) = \mathcal{D}_{w_2(E)}^{X,g}(e^{p\,\mathbf{x}+S}) , \qquad (6.7)$$

so that the Donaldson invariants are read off from the expansion of (6.7) as the coefficients of powers of p and  $\kappa = (\kappa^1, \ldots, \kappa^{b_2})$ .

In the case of the complex projective plane  $\mathbb{CP}^2$ , we have  $b_2 = b_2^+ = 1$ . The Fubini-Study metric g on  $\mathbb{CP}^2$  is Kähler with the Kähler form  $K = \frac{i}{2}g_{a\bar{b}} dz^a \wedge dz^{\bar{b}}$ . We denote the first Chern class of the dual of the hyperplane bundle over  $\mathbb{CP}^2$  by  $H = K/\pi$ , so that  $\int_{\mathbb{CP}^2} H^2 = 1$ ,  $c_1(\mathbb{CP}^2) = 3 H$ , and  $p_1(\mathbb{CP}^2) = 3 H^2$ . The Poincaré dual h of H is a generator of the rank-one homology group  $H_2(\mathbb{CP}^2,\mathbb{Z})$ . The SO(3)-bundles on four-dimensional manifolds are classified by the second Stiefel-Whitney class  $w_2(E) \in H^2(\mathbb{CP}^2,\mathbb{Z}_2)$ , and the first Pontrjagin class  $p_1(E) \in H^4(\mathbb{CP}^2,\mathbb{Z})$ , such that

$$p_1(E)[\mathbb{CP}^2] \equiv w_2^2(E)[\mathbb{CP}^2] \pmod{4}$$
.

Since  $\mathbb{CP}^2$  is simply connected, there is an integer class  $2\lambda_0 \in H^2(\mathbb{CP}^2,\mathbb{Z})$ whose mod-two reduction is  $w_2(E)$ . Then, there is a smooth complex twodimensional vector bundle  $\xi \to \mathbb{CP}^2$  with the Chern classes  $c_1(\xi) = 2\lambda_0$  and  $c_2(\xi)$ , such that  $c_1^2(\xi) - 4c_2(\xi) = p_1(E)$ . We denote by  $\mathfrak{M}(c_1, c_2)$  the moduli space of rank-two vector bundles  $\xi$  over  $\mathbb{CP}^2$  with Chern classes  $c_1, c_2$ . It is known that  $\mathfrak{M}(c_1, c_2)$  only depends on the discriminant  $c_1^2 - 4c_2$  with the discriminant being negative for stable bundles. The bundle  $\xi$  can be reduced to an SU(2)-bundle if and only if  $c_1(\xi) = 0$  and a SO(3)-bundle, which does not arise as the associated bundle for the adjoint representation of a SU(2)bundle, satisfies  $w_2(E) \neq 0$ .

From now on, we will restrict ourselves to  $w_2(E) = 0$ , i.e., the case of SU(2)-bundles and where  $c_1(\xi) = 0$  and  $c_2(\xi) = k \operatorname{H}^2$  with  $k \in \mathbb{N}$ . The moduli space of anti-selfdual irreducible SU(2)-connections with  $c_2(\xi)[\mathbb{CP}^2] = k$  modulo gauge transformations is then the smooth, projective variety  $\mathfrak{M}(0, k)$  of dimension  $2d_k = 8k - 6$  [280]. The generating function (6.7) can be described as follows

$$Z_{\rm DW}(p,\kappa) = \sum_{m,n\geq 0} \Phi_{m,n} \frac{p^m}{m!} \frac{\kappa^n}{n!} , \qquad (6.8)$$

where  $S = \kappa$  h. Here  $\Phi_{m,n}$  is the intersection number obtained by evaluating the top-dimensional cup product of the *m*th power of a universal four-form and the *n*th power of a two-form on the fundamental class of the Uhlenbeck compactification of  $\mathfrak{M}(0, k)$  such that 4m + 2n = 8k - 6 with  $k \in \mathbb{N}$ . Thus, for dimensional reasons we have  $\Phi_{m,n} = 0$  for  $2m + n \not\equiv 1 \pmod{4}$ 

#### 6.2.1 The work of Göttsche and his collaborators

The work of Göttsche and his collaborators [141,143] gives a closed expression for the SU(2) Donaldson invariants for the complex projective plane. Using the blowup formula for the Donaldson invariants, Göttsche [141] derived a closed formula expression for  $\Phi_{m,n}$  assuming the truth of the Kotschick-Morgan Conjecture. Recently, Göttsche, Nakajima, Hiraku, and Yoshioka [143] have unconditionally proved these formulas. His work was based on earlier work with Ellingsrud [115] and it extended the results previously obtained by Kotschick and Lisca [197] up to an overall sign convention. His work with Zagier [144] was an application of [141]. We state [141, Thm. 3.5, (1)] using the original sign convention of [115, 197]. We write the result in terms of the Jacobi theta-functions  $\vartheta_2, \vartheta_3, \vartheta_4$ . In this way, we obtain a closed formula expression for  $\Phi_{m,n}$ , which we shall later show equals the the Moore-Witten prediction based on the u-plane integral.

**Theorem 6.1** (Göttsche [141]). Assuming the notation and hypotheses above, then we have that the only non-vanishing coefficients in the generating function in (6.8) satisfy

$$\Phi_{m,2n+1} = \sum_{l=0}^{n} \sum_{j=0}^{l} \frac{(-1)^{n+j+1} 2^{2n-3l+4}}{3^{l}} \frac{(2n+1)!}{(2n-2l+1)! j! (l-j)!} \times \operatorname{Coeff}_{q^{0}} \left( \frac{\vartheta_{4}^{8}(\tau) \left[\vartheta_{2}^{4}(\tau) + \vartheta_{3}^{4}(\tau)\right]^{m+j}}{\left[\vartheta_{2}(\tau) \vartheta_{3}(\tau)\right]^{2m+2n+5}} E_{2}^{l-j}(\tau) K_{2(n-l)}(\tau) \right) ,$$

$$(6.9)$$

where  $m, n \in \mathbb{N}_0$ ,  $\operatorname{Coeff}_{q^0}$  is the constant term of a series expansion in  $q = \exp(2\pi i \tau)$ . The series  $K_t(\tau)$  is

$$K_t(\tau) := q^{\frac{1}{8}} \sum_{\beta=1}^{\infty} \sum_{\alpha=\beta}^{\infty} (-1)^{\alpha+\beta} (2\alpha+1) \beta^{t+1} q^{\frac{\alpha(\alpha+1)-\beta^2}{2}} .$$
 (6.10)

*Proof.* The following table summarizes the quantities used by Göttsche [141, Thm. 3.5, (1)] and in this article:

Göttsche	Present Paper	Göttsche	Present Paper
z	S	$\theta(\tau)$	$\vartheta_4( au)$
x	p	f( au)	$rac{1}{2\sqrt{i}}artheta_2( au)artheta_3( au)$
n	$2\beta+1,\;\beta\geq 0$	$\frac{\Delta^2(2\tau)}{\Delta(\tau)\Delta(4\tau)}$	$-16 \tfrac{\vartheta_4^8(\tau)}{\left[\vartheta_2(\tau)\vartheta_3(\tau)\right]^4}$
a	$2\alpha,\;\alpha\geq\beta+1$	$G_2(2\tau)$	$-\frac{1}{24} E_2(\tau)$
τ	$\frac{\tau-1}{2}$	$e_3(2\tau)$	$\frac{1}{12} \left[ \vartheta_2^4(\tau) + \vartheta_3^4(\tau) \right]$
q	$-q^{\frac{1}{2}}$	$\frac{-3ie_3(2\tau)}{f(\tau)^2}$	$\frac{\vartheta_2^4(\tau) + \vartheta_3^4(\tau)}{\left[\vartheta_2(\tau)\vartheta_3(\tau)\right]^2}$

We use

$$\begin{pmatrix} \frac{n}{2} \frac{\sqrt{i}}{f(\tau)} \end{pmatrix}^{2(n-l)} \left( -\frac{i}{2 f(\tau)^2} (2 G_2(2\tau) + e_3(2\tau)) \right)^l \\ = \frac{(-1)^{n+l}}{2^l 3^l} (2\beta + 1)^{2(n-l)} \frac{(-E_2(\tau) + [\vartheta_2^4(\tau) + \vartheta_3^4(\tau)])^l}{[\vartheta_2(\tau) \vartheta_3(\tau)]^{2n}}$$

An expansion of the exponential in [141, Thm. 3.5, (1)] then yields (6.9).  $\Box$ 

## 6.2.2 The *u*-plane integral and the work of Moore and Witten

Here we recall the theory of the u-plane and the work of Moore and Witten.

From now on we will assume that (X, g) is a smooth, compact, oriented, simply connected Riemannian four-manifold without boundary and  $b_2^+ = 1$ . The *u*-plane integral Z is a generating function in the variables p and  $\kappa$ whose coefficients are the integrals of certain modular forms over the fundamental domain of the group  $\Gamma_0(4)$ . It depends on the period point  $\omega$ , the lattice  $H_2(X,\mathbb{Z})$  together with the intersection form (.,.), the second Stiefel-Whitney classes of the gauge bundle  $w_2(E)$ , and the tangent bundle  $w_2(X)$ , whose integral liftings are denoted by  $2\lambda_0$  and  $w_2$  respectively. The *u*-plane integral is non-vanishing only for manifolds with  $b_2^+ = 1$ . The explicit form of Z for simply connected four-manifolds was first introduced in [241]. For the reader's convenience, we quickly review the explicit construction of the *u*-plane in this chapter. Our approach to the *u*-plane integral, as well as its normalization, closely follows the approach in [200, 216, 232].

We will denote the self-dual and anti-self-dual projections of any two-form  $\lambda \in H^2(X, \mathbb{Z}) + \lambda_0$  by  $\lambda_+ = (\lambda, \omega)\omega$  and  $\lambda_- = \lambda - \lambda_+$  respectively. We first

introduce the integral

$$\mathcal{G}(\rho) = \int_{\Gamma_0(4) \setminus \mathbb{H}}^{\text{reg}} \frac{dxdy}{y^{\frac{3}{2}}} \, \widehat{f}(p,\kappa) \, \bar{\Theta}(\xi) \, . \tag{6.11}$$

In this expression  $\widehat{f}(p,\kappa)$  is the almost holomorphic modular form given by

$$\widehat{f}(p,\kappa) = \frac{\sqrt{2}}{64\pi} \frac{\vartheta_4^{\sigma}}{h^3 \cdot f_2} e^{2p\,u + S^2 \widehat{T}},\tag{6.12}$$

where  $\sigma$  is the signature of X and  $S^2 = (S, S) = \sum_{i,j} \kappa^i \kappa^j (\mathbf{s}_i, \mathbf{s}_j)$ . Also,  $\overline{\Theta}$  is the Siegel-Narain theta function

$$\bar{\Theta}(\xi) = \exp\left[\frac{\pi}{2y} \left(\bar{\xi}_{+}^{2} - \bar{\xi}_{-}^{2}\right)\right] \times \sum_{\lambda \in H^{2} + \lambda_{0}} \exp\left[-i\pi\bar{\tau}(\lambda_{+})^{2} - i\pi\tau(\lambda_{-})^{2} - 2\pi i\left(\lambda,\bar{\xi}\right) + \pi i\left(\lambda,w_{2}\right)\right],$$
(6.13)

where  $\bar{\xi} = \bar{\xi}_+ + \bar{\xi}_-$ ,  $\bar{\xi}_+ = \rho y h \omega$ ,  $\bar{\xi}_- = S_-/(2\pi h)$ , and  $\rho \in \mathbb{R}$ . The Siegel-Narain theta function only depends on the lattice data  $(H^2(X), \omega, \lambda_0, w_2)$ . We have denoted the intersection form in two-cohomology by (., .), and we used Poincaré duality to convert cohomology classes into homology classes. In the above expressions, u, T, h, and  $f_2$  are the modular forms defined as follows:

$$u = \frac{\vartheta_2^4 + \vartheta_3^4}{2(\vartheta_2\vartheta_3)^2}, \qquad h = \frac{1}{2}\vartheta_2\vartheta_3,$$
  

$$T = -\frac{1}{24}\left(\frac{E_2}{h^2} - 8u\right), \qquad f_2 = \frac{\vartheta_2\vartheta_3}{2\vartheta_4^8}.$$
(6.14)

Note that T does not transform well under modular transformations, due to the presence of the second normalized Eisenstein series  $E_2 = E_2(\tau)$ . Therefore, in (6.12) we have used the related form  $\hat{T} = T + 1/(8\pi y h^2)$  which is not holomorphic but transforms well under modular transformations. We also define the related holomorphic function  $f(p, \kappa)$  as in (6.12), but with Tinstead of  $\hat{T}$ . The *u*-plane integral is defined to be

$$Z\left(X,\omega,\lambda_0,w_2\right) = \left[\left(S,\omega\right) + 2\frac{d}{d\rho}\right]\Big|_{\rho=0} \mathcal{G}(\rho) .$$
(6.15)

If there is no danger of confusion we suppress the arguments  $(X, \omega, \lambda_0, w_2)$  of Z.

#### Two remarks.

1) This regularized *u*-plane integral can be thought of as the regularized Peterson inner product of two half-integral weight modular forms on  $\Gamma_0(4)$ , where the regularization is obtained by integrating over the truncated fundamental domain for  $\Gamma_0(4)$  where neighborhoods of the cusps are removed. 2) Definition (6.15) agrees with the definition given in [216]. However, compared to the original definition in [241], a factor of exp  $[2\pi i(\lambda_0, \lambda_0) + \pi i(\lambda_0, w_2)]$ is missing. For the case considered in this article, this factor is equal to one.

The regularization procedure applied in the definition of the integral (6.11) was described in detail in [241]. It defines a way of extracting certain contributions for each boundary component near the cusps of  $\Gamma_0(4) \setminus \mathbb{H}$ . Since the cusps are located at  $\tau = \infty$ ,  $\tau = 0$ , and  $\tau = 2$ , we obtain  $Z_u$  as the sum of these contributions from the cusps:

$$Z_{\rm u} = Z_{\tau=0} + Z_{\tau=2} + Z_{\tau=\infty} . \tag{6.16}$$

We now apply the construction of the *u*-plane integral to  $X = \mathbb{CP}^2$ . We denote the integral lifting of  $w_2(E)$  by  $2\lambda_0 = a \operatorname{H} \in H^2(\mathbb{CP}^2, \mathbb{Z})$ , and the integral lifting of  $w_2(\mathbb{CP}^2)$  by  $w_2 = -b \operatorname{H} \in H^2(\mathbb{CP}^2, \mathbb{Z})$ . The following lemma then follows immediately from the definition:

**Lemma 6.2.** On  $X = \mathbb{CP}^2$  let  $\omega = H$  be the period point of the metric. Let  $2\lambda_0 = a H$  with  $a \in \{0, 1\}$  be an integral lifting of  $w_2(E)$ . For  $(X, \omega, \lambda_0, w_2 = -H)$ , the Siegel-Narain theta function is

$$\bar{\Theta} = \exp\left(\frac{\pi}{2y}\bar{\xi}_{+}^{2}\right)\overline{\vartheta_{a1}\left((\xi_{+},\mathrm{H})\big|\tau\right)},\tag{6.17}$$

where  $\bar{\xi} = \bar{\xi}_+ = \rho \, y \, h \, \omega$ .

It was shown in [241] that for  $\sigma = b_2^+ - b_2^- = 1$  and any value of a, we have

$$Z_{\tau=0} = Z_{\tau=2} = 0 , \qquad (6.18)$$

and so  $Z_{\rm u} = Z_{\tau=\infty}$ . We restrict ourselves to the case a = 0, i.e., the case of SU(2)-bundles on  $\mathbb{C}\mathrm{P}^2$ . The *u*-plane integral in (6.15) can be expanded as follows

$$Z_{\tau=\infty} = \sum_{m,n\in\mathbb{N}_0} \frac{p^m}{m!} \frac{\kappa^{2n+1}}{(2n+1)!} \mathcal{D}_{m,n},$$
(6.19)

where

$$D_{m,n} := -\frac{\sqrt{2}}{32\pi} \sum_{l=0}^{n} \int_{\Gamma_{0}(4)\backslash \mathbb{H}}^{\text{reg}} \frac{dx \, dy}{y^{\frac{3}{2}}} R_{mnl} \, \widehat{E}_{2}^{l} \, \overline{\vartheta_{01}(0|\tau)} \,. \tag{6.20}$$

For  $m, n \in \mathbb{N}_0$  and  $0 \leq l \leq n$  we have set

$$R_{mnl} := (-1)^{l+1} \frac{(2n+1)!}{l! (n-l)!} \frac{2^{m-3l-1}}{3^n} \frac{\vartheta_4 \cdot u^{m+n-l}}{h^{3+2l} \cdot f_2}, \qquad (6.21)$$

where u, h, and  $f_2$  were defined in (6.14). To evaluate the regularized uplane integral we introduce the non-holomorphic modular form  $Q_{ab}(\tau) = Q_{ab}^+(\tau) + Q_{ab}^-(\tau)$  of weight 3/2 such that

$$8\sqrt{2\pi} i \frac{d}{d\bar{\tau}} Q_{ab}(\tau) = y^{-\frac{3}{2}} \overline{\vartheta_{ab}(0|\tau)}, \qquad (6.22)$$

where a or b must be zero. These non-holomorphic modular forms were constructed by Zagier [306] and reviewed in [241]. The holomorphic parts of Zagier's weight 3/2 Maass-Eisenstein series, which first arose [161] in connection with intersection theory for certain Hilbert modular surfaces, are generating functions for Hurwitz class numbers. The holomorphic part of Zagier's weight 3/2 Maass-Eisenstein series is the generating function for Hurwitz class numbers. They have series expansions of the form

$$Q_{10}^{+}(\tau) = \frac{1}{q^{\frac{1}{8}}} \sum_{l>0} \mathcal{H}_{4l-1} q^{\frac{l}{2}} ,$$
  

$$Q_{00}^{+}(\tau) = \sum_{l\geq0} \mathcal{H}_{4l} q^{\frac{l}{2}} ,$$
(6.23)

where  $\mathcal{H}_{\alpha}$  are the Hurwitz class numbers. The first nonvanishing Hurwitz class numbers are as follows:

$\mathcal{H}_0$	$\mathcal{H}_3$	$\mathcal{H}_4$	$\mathcal{H}_7$	$\mathcal{H}_8$	$\mathcal{H}_{11}$	$\mathcal{H}_{12}$	
-1/12	1/3	1/2	1	1	1	4/3	

The non-holomorphic parts have series expansions of the form

$$Q_{10}^{-}(\tau) = \frac{1}{8\sqrt{2\pi}} \sum_{l=-\infty}^{\infty} \left(l + \frac{1}{2}\right) \cdot \Gamma\left(-\frac{1}{2}, 2\pi \left(l + \frac{1}{2}\right)^{2} y\right) q^{-\frac{(l+1/2)^{2}}{2}},$$

$$Q_{00}^{-}(\tau) = \frac{1}{8\sqrt{2\pi}} \sum_{l=-\infty}^{\infty} l \cdot \Gamma\left(-\frac{1}{2}, 2\pi l^{2} y\right) q^{-\frac{l^{2}}{2}},$$
(6.24)

where  $\Gamma(3/2, x)$  is the incomplete gamma function

$$\Gamma(\alpha, x) = \int_x^\infty e^{-t} t^{\alpha - 1} dt . \qquad (6.25)$$

The forms  $Q_{10}$  and  $Q_{00}$  combine to form a weight 3/2 form for the modular group. As explained in [241] the form  $Q_{01}(\tau) = Q_{00}(4\tau) - Q_{10}(4\tau) + \frac{1}{2}Q_{00}(\tau +$ 1) is modular for  $\Gamma_0(4)$  of weight 3/2. We write the holomorphic part as

$$Q_{01}^{+}(\tau) = \sum_{n \ge 0} \mathcal{R}_n \ q^{\frac{n}{2}} \ . \tag{6.26}$$

The first nonvanishing coefficients in the series expansion are as follows:

$\mathcal{R}_0$	$\mathcal{R}_1$	$\mathcal{R}_2$	$\mathcal{R}_3$	$\mathcal{R}_4$	
-1/8	-1/4	1/2	-1	5/4	

All non-holomorphic parts have an exponential decay since

$$\Gamma(\alpha, t) = t^{\alpha - 1} e^{-t} \left( 1 + O(t^{-1}) \right) \qquad (t \to \infty) .$$
 (6.27)

The following lemma was proved in [241, (9.18)]:

Lemma 6.3. The weakly holomorphic function

$$\mathcal{E}^{l}[Q_{01}] = \sum_{j=0}^{l} (-1)^{j} \binom{l}{j} \frac{\Gamma\left(\frac{3}{2}\right)}{\Gamma\left(\frac{3}{2}+j\right)} 2^{2j} 3^{j} E_{2}^{l-j}(\tau) \left(q \frac{d}{dq}\right)^{j} Q_{01}(\tau) \quad (6.28)$$

is modular for  $\Gamma_0(4)$  of weight 2l + 3/2 and satisfies

$$8\sqrt{2\pi} \, i \, \frac{d}{d\bar{\tau}} \, \mathcal{E}^l \left[ Q_{01} \right] = y^{-\frac{3}{2}} \, \widehat{E}_2^l(\tau) \, \overline{\vartheta_{01}(0|\tau)} \,. \tag{6.29}$$

#### The evaluation of the *u*-plane integral.

It was shown in [241] that the cusp contribution at  $\tau = \infty$  to the regularized *u*-plane integral can be evaluated as follows: in (6.31) we integrate by parts using the modular forms constructed in Lemma 6.3, i.e., we rewrite an integrand *f* as a total derivative using

$$dx \wedge dy \ \partial_{\bar{\tau}} f = \frac{1}{2} \, dx \wedge dy \ \left(\partial_x + i \, \partial_y\right) \, f = -\frac{i}{2} \, d\left(f \, dx + i \, f \, dy\right) \, .$$

We carry out the integral along the boundary  $x = \text{Re}(\tau) \in [0, 4]$  and  $y \gg 1$  fixed. This extracts the constant term coefficient. We then take the limit  $y \to \infty$ . Since all non-holomorphic parts have an exponential decay, the non-holomorphic dependence drops out. The following expression for the *u*-plane integral was obtained for the gauge group SU(2) in [241]. Additional information about the evaluation of the *u*-plane integral as well as the geometry of the Seiberg-Witten curve can be found in [225, 226].

**Theorem 6.4.** On  $X = \mathbb{CP}^2$ , let  $\omega = H$  be the period point of the metric. For  $(X, \omega, \lambda_0 = 0, w_2 = -H)$ , the u-plane integral in the variables  $p \ge H_0(X, \mathbb{Z})$ ,  $S = \kappa h \in H_2(X, \mathbb{Z})$  is

$$Z_{\rm u} = Z_{\tau=\infty} = \sum_{m,n\in\mathbb{N}_0} \frac{p^m}{m!} \frac{\kappa^{2n+1}}{(2n+1)!} \,\mathcal{D}_{m,n},\tag{6.30}$$

where

$$D_{m,n} = \sum_{l=0}^{n} \operatorname{Coeff}_{q^{0}} \left( R_{mnl} \ \mathcal{E}^{l}[Q_{01}^{+}(\tau)] \right)$$
(6.31)

and where  $R_{mnl}$  and  $\mathcal{E}^{l}[Q_{01}(\tau)]$  are defined in (6.21) and (6.28) respectively.

For concreteness, we list the first nonvanishing coefficients of the generating function in Theorem 6.4, i.e., if m + n = 2(k - 1) for some  $k \in \mathbb{N}$ .

k	m	n	$D_{m,n}$	$D_{m,n}$
1	0	0	$-\frac{3}{2}$	$-\frac{1}{2}\mathcal{R}_1+13\mathcal{R}_0$
2	0	2	1	$-2\mathcal{R}_2+7\mathcal{R}_1-30\mathcal{R}_0$
2	1	1	-1	$-rac{1}{4}\mathcal{R}_2+rac{1}{2}\mathcal{R}_1+6\mathcal{R}_0$
2	2	0	$-\frac{13}{8}$	$-rac{1}{32}\mathcal{R}_2 - rac{7}{16}\mathcal{R}_1 + rac{55}{4}\mathcal{R}_0$

#### 6.2.3 Criterion for proving Theorem 1.20

Here we combine the results of the previous two subsections to obtain a criterion for proving Theorem 1.20.

From a physics point of view, at a high energy scale, the SU(2)-Donaldson theory is described by the low energy effective field theory. Thus, the cuspidal contributions to the generating function of the low energy effective field theory should be equal to the generating function of the SU(2)-Donaldson theories The conjecture is equivalent to the assertion that the generating functions  $Z_{\rm DW}$  in (6.8) and  $Z_{\rm u}$  in (6.30) are equal. This amounts to proving that for all  $m, n \in \mathbb{N}_0$  we have

$$\Phi_{m,2n+1} = \mathcal{D}_{m,n} \ . \tag{6.32}$$

In particular, the coefficients in (6.32) vanish for  $m + n \equiv 1 \pmod{2}$ . We will prove (6.32) by proving:

**Theorem 6.5.** Theorem 1.20 is equivalent to the vanishing of constant terms,

for every pair of non-negative integers m and n, of the series

$$\sum_{l=0}^{n} \sum_{j=0}^{l} (-1)^{j+1} \frac{(2n+1)!}{(n-l)! \, j! \, (l-j)!} \frac{\vartheta_4^8(\tau) \, \left[\vartheta_2^4(\tau) + \vartheta_3^4(\tau)\right]^m}{\left[\vartheta_2(\tau) \, \vartheta_3(\tau)\right]^{2m+2n+4}} \, E_2^{l-j}(\tau) \\ \times \left[ \frac{(-1)^n \, 2^{2n-3l+4}}{3^l} \, \frac{(n-l)!}{(2n-2l+1)!} \, \frac{\left[\vartheta_2^4(\tau) + \vartheta_3^4(\tau)\right]^j}{\vartheta_2(\tau) \, \vartheta_3(\tau)} \, K_{2(n-l)}(\tau) \right. \\ \left. - \frac{(-1)^l \, 2^{2j-n+3}}{3^{n-j}} \, \frac{\Gamma\left(\frac{3}{2}\right)}{\Gamma\left(j+\frac{3}{2}\right)} \, \vartheta_4(\tau) \, \left[\vartheta_2^4(\tau) + \vartheta_3^4(\tau)\right]^{n-l} \, \left(q \frac{d}{dq}\right)^j Q_{01}^+(\tau) \right] \,,$$

$$(6.33)$$

where the series  $K_t(\tau)$  are defined in (6.10).

## 6.3 The proof of Theorem 1.20

Here we prove Theorem 1.20 by using the theory of non-holomorphic modular forms and meromorphic Jacobi forms to check the condition in Theorem 6.5. To this end, we recall the important q-series

$$K_t(\tau) := q^{\frac{1}{8}} \sum_{\beta=1}^{\infty} \sum_{\alpha=\beta}^{\infty} (-1)^{\alpha+\beta} (2\alpha+1) \beta^{t+1} q^{\frac{\alpha(\alpha+1)-\beta^2}{2}}$$
(6.34)

from (6.10). In the following section we relate  $K_{2t}(\tau)$  to derivatives of important power series.

#### 6.3.1 *q*-series identities

Here we begin with the following elementary identity.

**Proposition 6.6.** Let  $\rho = e^{2\pi i u}$  and  $\omega = e^{2\pi i v}$ , and let  $D_z = \frac{1}{2\pi i} \frac{d}{dz}$ , where z is one of u, v, or  $\tau$ . Then we have that

$$K_{2t}(8\tau) = 2^{-2t-1} \left. D_u^{2t+1} D_v \sum_{n \in \mathbb{Z}} \frac{(-1)^n \omega^{2n+1} q^{(2n+1)^2}}{1 - \rho^2 \omega^2 q^{8n+4}} \right|_{u=v=0}$$

*Proof.* By rearranging terms, it is not difficult to see that the summation on the right (prior to taking derivatives) is equal to

$$\sum_{n\geq 0} \frac{(-1)^n \omega^{2n+1} q^{(2n+1)^2}}{1 - \rho^2 \omega^2 q^{4(2n+1)}} - \rho^{-2} \omega^{-2} q^{4(2n+1)} \frac{(-1)^n \omega^{-(2n+1)} q^{(2n+1)^2}}{1 - \rho^{-2} \omega^{-2} q^{4(2n+1)}}$$
$$= \sum_{n\geq 0} (-1)^n \omega^{2n+1} q^{(2n+1)^2}$$
$$+ \sum_{n\geq 0} \sum_{m\geq 1} (-1)^n \left( \omega^{2n+1+2m} \rho^{2m} + \omega^{-(2n+1+2m)} \rho^{-2m} \right) q^{(2n+1)^2 + 4m(2n+1)}.$$
(6.35)

We then set  $\alpha = n + m$ , and  $\beta = m$ . After applying the derivatives, evaluating at u = v = 0, and factoring out the powers of 2, this becomes

$$\sum_{\beta=1}^{\infty} \sum_{\alpha=\beta}^{\infty} (-1)^{\alpha+\beta} (2\alpha+1) \beta^{2t+1} q^{4\alpha^2+4\alpha-4\beta^2+1} = K_{2t}(8\tau).$$
(6.36)

The summation in the right hand side of the equation in Proposition 6.6 is in the form of an Appell-Lerch function. In the next section, we show how to write this in terms of Zwegers's  $\mu$ -function, from which we can infer its modularity properties.

#### 6.3.2 Work of Zwegers

In his Ph.D. thesis on mock theta functions [311], Zwegers constructs weight 1/2 harmonic weak Maass forms by making use of the transformation properties of functions which were investigated earlier by Appell and Lerch. Here we briefly recall some of his results.

For  $\tau$  in  $\mathbb{H}$  and  $u, v \in \mathbb{C} \setminus (\mathbb{Z}\tau + \mathbb{Z})$ , Zwegers defines the function

$$\mu(u, v; \tau) := \frac{\rho^{1/2}}{\theta(v; \tau)} \cdot \sum_{n \in \mathbb{Z}} \frac{(-\omega)^n q^{n(n+1)/2}}{1 - \rho q^n}, \tag{6.37}$$

where  $\rho = e^{2\pi i u}$  and  $\omega = e^{2\pi i v}$  as above, and

$$\theta(v;\tau) := \sum_{\nu \in \mathbb{Z} + \frac{1}{2}} (-1)^{\nu - \frac{1}{2}} \omega^{\nu} q^{\nu^2/2}.$$
(6.38)

Zwegers's (see Section 1.3 of [311]) proves that  $\mu(u, v, \tau)$  satisfies the following important properties.

Lemma 6.7. Assuming the notation above, we have that

(1)  $\mu(u,v;\tau) = \mu(v,u,\tau),$ 

(2) 
$$\mu(u+1,v,\tau) = -\mu(u,v;\tau),$$

(3) 
$$\rho^{-1}\omega q^{-\frac{1}{2}}\mu(u+\tau,v;\tau) = -\mu(u,v;\tau) + \rho^{-\frac{1}{2}}\omega^{\frac{1}{2}}q^{-\frac{1}{8}},$$

(4) 
$$\mu(u,v;\tau+1) = \zeta_8^{-1}\mu(u,v;\tau) \quad (\zeta_N := e^{2\pi i/N})$$

(5) 
$$(\tau/i)^{-\frac{1}{2}}e^{\pi i(u-v)^2/\tau}\mu(\frac{u}{\tau},\frac{v}{\tau};-\frac{1}{\tau}) = -\mu(u,v;\tau) + \frac{1}{2}h(u-v;\tau),$$

where

$$h(z;\tau) := \int_{-\infty}^{\infty} \frac{e^{\pi i x^2 \tau - 2\pi x z} dx}{\cosh \pi x}$$

*Remark.* The integral  $h(z, \tau)$  is known as a *Mordell integral*.

Lemma 6.7 shows that  $\mu(u, v; \tau)$  is nearly a weight 1/2 Jacobi form, where  $\tau$  is the modular variable. Zwegers then uses  $\mu$  to construct weight 1/2 harmonic weak Maass forms. He achieves this by modifying  $\mu$  to obtain a function  $\hat{\mu}$  which he then uses as a building block for such Maass forms. To make this precise, for  $\tau \in \mathbb{H}$  and  $u \in \mathbb{C}$ , let

$$c := \operatorname{Im}(u) / \operatorname{Im}(\tau),$$

and let

$$R(u;\tau) := \sum_{\nu \in \mathbb{Z} + \frac{1}{2}} (-1)^{\nu - \frac{1}{2}} \left\{ \operatorname{sgn}(\nu) - E\left((v+c)\sqrt{2\operatorname{Im}(\tau)}\right) \right\} e^{-2\pi i\nu u} q^{-\nu^2/2},$$
(6.39)

where E(z) is the odd function

$$E(z) := 2 \int_0^z e^{-\pi u^2} du.$$

Using  $\mu$  and R, we let

$$\widehat{\mu}(u,v;\tau) := \mu(u,v;\tau) - \frac{1}{2}R(u-v;\tau).$$
(6.40)

Zwegers's construction of weight 1/2 harmonic weak Maass forms depends on the following theorem (see Section 1.4 of [311]).

Theorem 6.8. Assuming the notation above, we have that (1)  $\hat{\mu}(u, v; \tau) = \hat{\mu}(v, u, \tau),$ 

(2)  $\widehat{\mu}(u+1,v,\tau) = \rho^{-1}\omega q^{-\frac{1}{2}}\mu(u+\tau,v;\tau) = -\widehat{\mu}(u,v;\tau),$ 

(3) 
$$\zeta_8^{-1}\widehat{\mu}(u,v;\tau+1) = -(\tau/i)^{-\frac{1}{2}}e^{\pi i(u-v)^2/\tau}\widehat{\mu}(\frac{u}{\tau},\frac{v}{\tau};-\frac{1}{\tau}) = \widehat{\mu}(u,v;\tau)$$

(4) 
$$\widehat{\mu}\left(\frac{u}{c\tau+d},\frac{v}{c\tau+d};\frac{a\tau+b}{c\tau+d}\right) = \chi(A)^{-3}(c\tau+d)^{\frac{1}{2}}e^{-\pi i c(u-v)^2/(c\tau+d)}\cdot\widehat{\mu}(u,v;\tau),$$

where 
$$A = \begin{pmatrix} a & b \\ c & d \end{pmatrix}$$
, and  $\chi(A) := \eta(A\tau) / \left( (c\tau + d)^{\frac{1}{2}} \eta(\tau) \right)$ .

Theorem 6.8 gives the modular transformation properties for  $\hat{\mu}$ . In the following section we will write  $K_{2t}$  in terms of  $\mu$ , and we then use its properties to complete  $K_0(\tau)$  as a nonholomorphic modular form on  $\Gamma_0(8)$ .

#### 6.3.3 Modularity Properties of $K_0(\tau)$

We begin with the following proposition.

Proposition 6.9. We have that

$$K_{2t}(8\tau) = 2^{-2t-1} D_u^{2t+1} D_v \left( \rho^{-1} q^{-1} \mu (2u + 2v + 4\tau, 2v; 8\tau) \theta(2v; 8\tau) \right) \Big|_{u=v=0}$$

Moreover,  $\frac{K_0(8\tau)}{\eta^3(8\tau)}$  is the holomorphic part of a weight 3/2 weak Maass which is modular on  $\Gamma_0(8)$ , and whose non-holomorphic part is the period integral of  $\Theta_4(\tau)$ .

*Proof.* The first statement follows directly from Proposition 6.6 and the definition of the  $\mu$  function defined in (6.37).

To prove the remainder of the proposition, let

$$\widehat{K}_{0}(\tau) = 2^{-1} D_{u} D_{v} \left( \rho^{-1} q^{-1} \widehat{\mu} (2u + 2v + 4\tau, 2v; 8\tau) \theta(2v; 8\tau) \right) \Big|_{u=v=0}, \quad (6.41)$$

so that the holomorphic part of  $\widehat{K_0}(\tau)$  is  $K_0(8\tau)$ . Suppose  $A = \begin{pmatrix} a & b \\ 8c & d \end{pmatrix} \in$ 

 $\Gamma_0(8)$ . Then we note  $\overline{A} = \begin{pmatrix} a & 8b \\ c & d \end{pmatrix} \in SL_2(\mathbb{Z})$ . Using the transformation laws for  $\hat{\mu}$  found in Lemma 6.8, we have

$$e^{2\pi i \frac{-u-a\tau-b}{8c\tau+d}} \widehat{\mu} \left( \frac{2u+2v+4(a\tau+b)}{8c\tau+d}, \frac{2v}{8c\tau+b}; \frac{a8\tau+8b}{c8\tau+d} \right)$$
$$= \chi(\overline{A})^{-3} (-1)^{\frac{a-1}{2}} (8c\tau+d)^{1/2} e^{(2\pi i) \frac{-1}{4} \frac{8cu^2}{8c\tau+d}} \cdot \widehat{\mu} (2u+2v+4\tau, 2v, 8\tau),$$
(6.42)

which is obtained by substituting  $u \to \frac{u}{8c\tau+d}$ ,  $v \to \frac{v}{8c\tau+d}$ , and  $\tau \to \frac{a\tau+b}{8c\tau+d}$  into the expression on the right hand side of (6.41), before taking derivatives. With a little more algebra, we find that

$$\widehat{K}_0\left(\frac{a\tau+b}{8c\tau+d}\right) = \chi(\overline{A})^{-3}(-1)^{\frac{a-1}{2}}(8c\tau+d)^3\widehat{K}_0(\tau).$$
(6.43)

Therefore  $\widehat{K}_0(\tau)$  is modular on  $\Gamma_0(8)$  with weight 3/2. The non-holomorphic part of  $\widehat{K}_0(\tau)$  is

$$\frac{-1}{4} \left. D_u D_v R(2u+4\tau;8\tau) \theta(2v;8\tau) \right|_{u=v=0} = \frac{-1}{4} \eta^3(8\tau) \left. D_u R(2u+4\tau;8\tau) \right|_{u=0}.$$
(6.44)

After factoring out  $\eta^3(8\tau)$ , a straightforward calculation gives us that

$$\frac{\partial}{\partial \bar{\tau}} \frac{-1}{4} D_u R(2u+4\tau; 8\tau)|_{u=0} = \frac{-1}{32\pi y^{\frac{3}{2}}} \overline{\Theta_4(\tau)}.$$
(6.45)

#### 6.3.4 The proof of Theorem 1.20

Thanks to Theorem 6.5, it suffices to prove that the differences between certain q-series have vanishing constant term. We shall derive these conclusions by using differential operators, using methods very similar to those found in Section 8.1 of [227]. For brevity, we describe the n = 0 cases in detail, and then provide general remarks which are required to justify the remaining cases.

By (6.26) and Proposition 6.9, we have

$$8Q_{01}^{+}(8q) = -1 - 2q^{4} + 4q^{8} - 8q^{12} + 10q^{16} + \dots, \qquad (6.46)$$

and

$$8\frac{\widehat{K}_0(\tau)}{\eta^3(8\tau)} = 24q^4 + 80q^8 + 240q^{12} + 528q^{16}\dots$$
(6.47)

Comparing (6.22) (with  $8\tau$  substituted for  $\tau$ ) and (6.45), we see that both of these are the holomorphic parts of weight 3/2 harmonic weak Maass forms with equal non-holomorphic parts. Therefore, it follows that

$$8\frac{\widehat{K}_{0}(\tau)}{\eta^{3}(8\tau)} - 8Q_{01}(8\tau) = 1 + 26q^{4} + 76q^{8} + 248q^{12} + 518q^{16} + \dots$$
 (6.48)

is a modular form. A short calculation shows that

$$\frac{\widehat{K_0}(\tau)}{\eta^3(8\tau)} - Q_{01}(8\tau) = \frac{E^*(4\tau)}{8\Theta_4(\tau)}$$
(6.49)

where  $E^*(\tau)$  is the weight 2 Eisenstein series

$$E^*(\tau) := -E_2(\tau) + 2E_2(2\tau) = 1 + 24\sum_{n=1}^{\infty} \sigma_{\text{odd}}(n)q^n, \qquad (6.50)$$

and  $\sigma_{\rm odd}(n)$  denotes the sum of the positive odd divisors of n. Noting that

$$\eta^3(8\tau) = \Theta_2(\tau)\Theta_3(\tau)\Theta_4(\tau), \qquad (6.51)$$

where  $\Theta_1$ ,  $\Theta_2$ , and  $\Theta_3$  are defined in (6.4), we can rewrite this as

$$\frac{8\widehat{K}_0(\tau)}{\Theta_2(\tau)\Theta_3(\tau)} - 8\Theta_4(\tau)Q_{01}(8\tau) = E^*(4\tau).$$
(6.52)

For n = 0, Theorem 1.20 is equivalent to the claim, for every  $m \ge 0$ , that the constant term vanishes in the expression

$$\frac{\Theta_4(\tau)^8 (16\Theta_2(\tau)^4 + \Theta_3(\tau)^4)^m E^*(4\tau)}{\Theta_2(\tau)^{2m+4}\Theta_3(\tau)^{2m+4}}.$$
(6.53)

In order to verify this claim, we will find if helpful to define

$$Z(q) := \frac{E^*(4\tau)}{\Theta_2(\tau)^2 \Theta_3(\tau)^2}.$$
(6.54)

which has the derivative

$$q\frac{d}{dq}Z(q) = \frac{-2 \Theta_4(\tau)^8}{\Theta_2(\tau)^2 \Theta_3(\tau)^2}.$$
(6.55)

Here Z(q) is the same as  $\widehat{Z}_0(q)$  defined in Section 8.1 of [227]. We also note that

$$16\Theta_2(\tau)^4 + \Theta_3(\tau)^4 = 1 + 24q^4 + 24q^2 + \dots = E^*(4\tau).$$

Using this notation, (6.53) becomes

$$\frac{-1}{2(m+2)} \; q \frac{d}{dq} Z(q)^{m+2},$$

which has a vanishing constant term.

In fact, for each  $m, n \ge 0$ , we find a similar phenomenon. For every non-negative k, define

$$\mathcal{G}_{\ell}(q) := \sum_{j=0}^{\ell} {\binom{\ell}{j}} \frac{(-12)^{j} E_{2}(8\tau)^{\ell-j} \Gamma(\frac{3}{2})}{\left(\Theta_{2}(\tau)\Theta_{3}(\tau)\right)^{2\ell+2} 8^{j} \Gamma\left(\frac{3}{2}+j\right)} \\
\times \left[\frac{(-4)^{j} 8 K_{2j}(8\tau)}{\Theta_{2}(\tau)\Theta_{3}(\tau)} - 8\Theta_{4}(\tau) \left(q\frac{d}{dq}\right)^{j} Q_{01}^{+}(8\tau)\right].$$
(6.56)

Using this notation, the criterion given in Theorem 6.5 is equivalent to the claim that the constant coefficient of

$$\left(q\frac{d}{dq}Z(q)\right)Z(q)^m\sum_{\ell=0}^n \binom{n}{\ell}\left(-Z(q)\right)^{n-\ell}\mathcal{G}_\ell(\tau)$$
(6.57)

is zero for each non-negative m and n. It suffices to show that  $\mathcal{G}_{\ell}(q)$  is a polynomial in Z(q). We define  $M_0^*(\Gamma_0(8))$  to be the space of modular functions on  $\Gamma_0(8)$  which are holomorphic away from infinity, and is a subspace of  $\mathbb{C}((q^2))$ . One can easily verify that  $M_0^*(\Gamma_0(8))$  is precisely the set of polynomials in Z(q). In order to show that  $\mathcal{G}_{\ell}(\tau)$  is in  $M_0^*(\Gamma_0(8))$ , we first show that a similar function,  $\mathcal{H}_{\ell}(q)$  is in  $M_0^*(\Gamma_0(8))$ . We define the function

$$\mathcal{H}_{\ell}(q) := \frac{\Theta_{4}(\tau)}{\left(\Theta_{2}(\tau)\Theta_{3}(\tau)\right)^{2\ell+2}} \sum_{j=0}^{\ell} {\binom{\ell}{j}} \frac{\Gamma(\frac{3}{2})(-12)^{j} E_{2}(8\tau)^{\ell-j}}{\Gamma\left(\frac{3}{2}+j\right) 8^{j}} \left(q\frac{d}{dq}\right)^{j} \frac{E^{*}(8\tau)}{\Theta_{4}(\tau)}.$$
(6.58)

We can observe that  $\mathcal{H}_{\ell}(q)$  is modular on  $\Gamma_0(8)$  with weight 0 by comparing the summation to the expression  $\mathcal{E}^{\ell}\left[\frac{E^*(8\tau)}{\Theta_4(\tau)}\right]$ , where the bracket operator

$$\mathcal{E}^{\ell}[f] := \sum_{j=0}^{\ell} {\binom{\ell}{j}} \frac{\Gamma\left(\frac{3}{2}\right)}{\Gamma\left(\frac{3}{2}+j\right)} (-12)^{j} E_{2}^{\ell-j}(\tau) \left(q\frac{d}{dq}\right)^{j} f(\tau)$$

is defined as in equation (9.18) of [241] (See also [77]). This is the bracket operator used in Lemma 6.3 and, as noted, preserves modularity, but changes the weight from  $\frac{3}{2}$  to  $\frac{3}{2} + 2\ell$ . A calculation shows that  $(\Theta_2(\tau)\Theta_3(\tau))^{-2}$  and  $\Theta_4(\tau)^{-1}$  are holomorphic away from infinity, which, combined with the fact that  $\Theta_2(\tau)\Theta_3(\tau) \in \mathbb{Z}[[q^2]]$ , shows that  $\mathcal{H}_{\ell}(\tau)$  is in  $M_0^*(\Gamma_0(8))$ . Hence it suffices to show that  $\mathcal{G}_{\ell}(q) - \mathcal{H}_{\ell}(q)$  is in  $M_0^*(\Gamma_0(8))$  as well. From (6.49), we see that

$$\mathcal{G}_{\ell}(q) - \mathcal{H}_{\ell}(q) = \sum_{j=0}^{\ell} {\ell \choose j} \frac{\Gamma(\frac{3}{2})(-12)^{j} E_{2}(8\tau)^{\ell-j} \Theta_{4}(\tau)}{(\Theta_{2}(\tau)\Theta_{3}(\tau))^{2\ell+2} \Gamma(\frac{3}{2}+j) 8^{j}} \left[ \frac{(-4)^{j} 8K_{2j}(8\tau)}{\eta^{3}(8\tau)} - \left(q \frac{d}{dq}\right)^{j} \frac{8K_{0}(8\tau)}{\eta^{3}(8\tau)} \right].$$
(6.59)

Using Theorem 6.9, this can be written as

$$\begin{aligned} \mathcal{G}_{\ell}(q) - \mathcal{H}_{\ell}(q) \\ &= \frac{\Theta_{4}(\tau)}{\left(\Theta_{2}(\tau)\Theta_{3}(\tau)\right)^{2\ell+2}} \sum_{j=0}^{\ell} \binom{\ell}{j} \frac{\Gamma(\frac{3}{2})(-12)^{j}E_{2}(8\tau)^{\ell-j}}{\Gamma\left(\frac{3}{2}+j\right)8^{j-1}} \\ &\times \left[ (-1)^{j}D_{u}^{2j+1}D_{v} - D_{\tau}^{j}D_{u}D_{v} \right] \frac{\rho^{-1}q^{-1}\mu(2u+2v+4\tau,2v;8\tau)\theta(2v;4\tau)}{2\eta^{3}(8\tau)} \Big|_{u=v=0} \end{aligned}$$

$$(6.60)$$

Paying particular attention to the derivatives of the  $\mu$ -function above, we use the transformation laws for  $\mu$  found in Lemma 6.7, and observe that the Mordel integrals that arise as obstructions to the modular transformation of (6.60) cancel directly. Therefore an argument similar to the proof that the bracket operator preserves modularity suffices to show that  $\mathcal{G}_{\ell}(q) - \mathcal{H}_{\ell}(q)$  is modular with respect to  $\Gamma_0(8)$ . Some simple accounting shows that  $\mathcal{G}_{\ell}(q) - \mathcal{H}_{\ell}(q)$  is in  $\mathcal{H}_{\ell}(q)$  is supported on even exponents of q, and hence  $\mathcal{G}_{\ell}(q) - \mathcal{H}_{\ell}(q)$  is in  $M_0^*(\Gamma_0(8))$ . This completes the proof.

## 6.3.5 Examples

In the table below, we give the polynomial  $P_n(x)$  such that the expression in the statement of Theorem 6.5 can be written as

# Chapter 7

# Moonshine

We briefly return to a discussion of the history of classical moonshine, building up to the moonshine towers and the proofs of Theorem 7.10 and Corollary 7.11 which generalize Theorem 1.27 and Corollary 1.28 form the introduction.

# 7.1 Vertex operators and the proof of classical moonshine

In order to prove Thompson's conjecture, Frenkel-Lepowsky-Meurman generalized the homogeneous realization of the basic representation of an affine Lie algebra  $\hat{\mathfrak{g}}$  due, independently, to Frenkel-Kac [124] and Segal [267], in such a way that *Leech's lattice*  $\Lambda$  [201, 202]—the unique [78] even self-dual positive-definite lattice of rank 24 with no roots—could take on the role played by the root lattice of  $\mathfrak{g}$  in the Lie algebra case. In particular, their construction came equipped with rich algebraic structure, furnished by vertex operators, which had appeared first in the physics literature in the late 1960's.

We refer to [124], and also the introduction to [126] for accounts of the role played by vertex operators in physics (up to 1988) along with a detailed description of their application to the representation theory of affine Lie algebras. The first application of vertex operators to affine Lie algebra representations was obtained by Lepowsky–Wilson in [210].

Borcherds described a powerful axiomatic formalism for vertex operators in [29]. In particular, he introduced the notion of a vertex algebra, which can be regarded as similar to a commutative associative algebra, except that multiplications depend upon formal variables  $z_i$ , and can be singular, in a certain formal sense, along the canonical divisors  $\{z_i = 0\}, \{z_i = z_j\}$ (cf. [32, 123]).

The appearance of affine Lie algebras above, as a conceptual ingredient for the Frenkel-Lepowsky-Meurman construction of  $V^{\ddagger}$  hints at an analogy between complex Lie groups and the monster. Borcherds' vertex algebra theory makes this concrete, for Borcherds showed [29] that both in the case of the basic representation of an affine Lie algebra, and in the case of the moonshine module  $V^{\ddagger}$ , the vertex operators defined by Frenkel-Kac, Segal, and Frenkel-Lepowsky-Meurmann, extend naturally to vertex algebra structures.

In all of these examples the Virasoro algebra,  $\mathcal{V} = \bigoplus_n \mathrm{CL}(n) \oplus \mathrm{Cc}$ , being the unique universal central extension of the Lie algebra  $\mathrm{C}[t, t^{-1}] \frac{\mathrm{d}}{\mathrm{d}t}$  of polynomial vector fields on the circle,

$$[L(m), L(n)] = (m-n)L(m+n) + \frac{m^3 - m}{12}\delta_{m+n,0}\mathbf{c}, \quad [L(m), \mathbf{c}] = 0, \quad (7.1)$$

acts naturally on the underlying vector space. (See [181] for a detailed analysis of  $\mathcal{V}$ . The generator L(m) lies above the vector field  $-t^{m+1}\frac{\mathrm{d}}{\mathrm{d}t}$ .) This Virasoro structure, which has powerful applications, was axiomatized in [126], with the introduction of the notion of a vertex operator algebra. If V is a vertex operator algebra and the central element  $\mathbf{c}$  of the Virasoro algebra acts as c times the identity on V, for some  $c \in \mathbf{C}$ , then V is said to have central charge c.

For the basic representation of an affine Lie algebra  $\hat{\mathfrak{g}}$ , the group of vertex

operator algebra automorphisms—i.e. those vertex algebra automorphisms that commute with the Virasoro action—is the adjoint complex Lie group associated to  $\mathfrak{g}$ . For the moonshine module  $V^{\natural}$ , it was shown by Frenkel– Lepowsky–Meurman in [126], that the group of vertex operator algebra automorphisms is precisely the monster.

Theorem 7.1 (Frenkel-Lepowsky-Meurman). The moonshine module  $V^{\natural} = \bigoplus_{n=-1}^{\infty} V_n^{\natural}$  is a vertex operator algebra of central charge 24 whose graded dimension is given by  $J(\tau)$ , and whose automorphism group is M.

Vertex operator algebras are of relevance to physics, for we now recognize them as "chiral halves" of two-dimensional conformal field theories (cf. [130, 131]). From this point of view, the construction of  $V^{\natural}$  by Frenkel–Lepowsky– Meurman constitutes one of the first examples of an orbifold conformal field theory (cf. [94–96]). In the case of  $V^{\natural}$ , the underlying geometric orbifold is the quotient

$$\left(\mathbb{R}^{24}/\Lambda\right)/(\mathbb{Z}/2\mathbb{Z}),\tag{7.2}$$

of the 24-dimensional torus  $\Lambda \otimes_{\mathbb{Z}} \mathbb{R}/\Lambda \simeq \mathbb{R}^{24}/\Lambda$  by the Kummer involution  $x \mapsto -x$ , where  $\Lambda$  denotes the Leech lattice. So in a certain sense,  $V^{\ddagger}$  furnishes a "24-dimensional" construction of  $\mathbb{M}$ . We refer to [123,126,182,208] for excellent introductions to vertex algebra, and vertex operator algebra theory.

Affine Lie algebras are special cases of Kac–Moody algebras, first considered by Kac [177] and Moody [239, 240], independently. Roughly speaking, a Kac–Moody algebra is "built" from copies of  $\mathfrak{sl}_2$ , in such a way that most examples are infinite-dimensional, but much of the finite-dimensional theory carries through (cf. [183]). Borcherds generalized this further, allowing also copies of the three-dimensional Heisenberg Lie algebra to serve as building blocks, and thus arrived [28] at the notion of generalized Kac–Moody algebra, or Borcherds-Kac-Moody (BKM) algebra, which has subsequently found many applications in mathematics and mathematical physics (cf. [164, 260]).

One of the most powerful such applications occurred in moonshine, when Borcherds introduced a particular example—the monster Lie algebra  $\mathfrak{m}$  and used it to prove [30] the moonshine conjectures of Conway–Norton. His method entailed using monster-equivariant versions of the denominator identity for  $\mathfrak{m}$  to verify that the coefficients of the McKay–Thompson series  $T_g$ , defined by (1.37) according to the Frenkel–Lepowsky–Meurman construction of  $V^{\natural}$ , satisfy the replication formulas conjectured by Conway–Norton in [81]. This powerful result reduced the proof of the moonshine conjectures to a small, finite number of identities, that he could easily check by hand.

Theorem 7.2 (Borcherds). Let  $V^{\natural}$  be the moonshine module vertex operator algebra constructed by Frenkel-Lepowsky-Meurman, whose automorphism group is  $\mathbb{M}$ . If  $T_g$  is defined by (1.37) for  $g \in \mathbb{M}$ , and if  $\Gamma_g$  is the discrete subgroup of  $SL_2(\mathbb{R})$  specified by Conway-Norton in [81], then  $T_g$  is the unique normalized principal modulus for  $\Gamma_g$ .

Recall that an even self-dual lattice of signature (m, n) exists if and only if  $m - n = 0 \pmod{8}$  (cf. e.g. [83]). Such a lattice is unique up to isomorphism if mn > 0, and is typically denoted  $II_{m,n}$ . In the case that m = n = 1 we may take

$$II_{1,1} := \mathbb{Z}e + \mathbb{Z}f,\tag{7.3}$$

where e and f are isotropic,  $\langle e, e \rangle = \langle f, f \rangle = 0$ , and  $\langle e, f \rangle = 1$ . Then  $me + nf \in H_{1,1}$  has square-length 2mn. Note that  $H_{25,1}$  and  $\Lambda \oplus H_{1,1}$  are isomorphic, for  $\Lambda$  the Leech lattice, since both lattices are even and self-dual, with signature (25, 1).

In physical terms the monster Lie algebra  $\mathfrak{m}$  is ("about half" of) the space

of "physical states" of a bosonic string moving in the quotient

$$\left(\mathbb{R}^{24}/\Lambda \oplus \mathbb{R}^{1,1}/II_{1,1}\right)/(\mathbb{Z}/2\mathbb{Z})$$
(7.4)

of the 26-dimensional torus  $H_{25,1} \otimes_{\mathbb{Z}} \mathbb{R}/H_{25,1} \simeq \mathbb{R}^{24}/\Lambda \oplus \mathbb{R}^{1,1}/H_{1,1}$  by the Kummer involution  $x \mapsto -x$ . The monster Lie algebra  $\mathfrak{m}$  is constructed in a functorial way from  $V^{\natural}$  (cf. [66]), inherits an action by the monster from  $V^{\natural}$ , and admits a monster-invariant grading by  $H_{1,1}$ .

The denominator identity for a Kac-Moody algebra  $\mathfrak{g}$  equates a product indexed by the positive roots of  $\mathfrak{g}$  with a sum indexed by the Weyl group of  $\mathfrak{g}$ . A BKM algebra also admits a denominator identity, which for the case of the monster Lie algebra  $\mathfrak{m}$  is the beautiful Koike-Norton-Zagier formula

$$p^{-1} \prod_{\substack{m,n \in \mathbb{Z} \\ m > 0}} (1 - p^m q^n)^{c(mn)} = J(\sigma) - J(\tau),$$
(7.5)

where  $\sigma \in \mathbb{H}$  and  $p = e^{2\pi i\sigma}$  (and c(n) is the coefficient of  $q^n$  in  $J(\tau)$ , cf. (1.33)). Since the right hand side of (7.5) implies that the left hand side has no terms  $p^m q^n$  with  $mn \neq 0$ , this identity imposes many non-trivial polynomial relations upon the coefficients of  $J(\tau)$ . Among these is

$$c(4n+2) = c(2n+2) + \sum_{k=1}^{n} c(k)c(2n-k+1),$$
(7.6)

which was first found by Mahler [221] by a different method, along with similar expressions for c(4n), c(4n+1), and c(4n+3), which are also entailed in (7.5). Taken together these relations allow us to compute the coefficients of  $J(\tau)$  recursively, given just the values

$$c(1) = 196884,$$

$$c(2) = 21493760,$$

$$c(3) = 864299970,$$

$$c(5) = 333202640600.$$
(7.7)

To recover the replication formulas of [81, 245] we require to replace J with  $T_g$ , and  $c(n) = \dim(V_n^{\natural})$  with  $\operatorname{tr}(g|V_n^{\natural})$  in (7.5), and for this we require a categorification of the denominator identity, whereby the positive integers c(mn) are replaced with M-modules of dimension c(mn).

A categorification of the denominator formula for a finite-dimensional simple complex Lie algebra was obtained by Kostant [196], following an observation of Bott [34]. This was generalized to Kac-Moody algebras by Garland-Lepowsky [137], and generalized further to BKM algebras by Borcherds in [30]. In its most compact form, it is the identity of virtual vector spaces

$$\bigwedge_{-1}(\mathfrak{e}) = H(\mathfrak{e}),\tag{7.8}$$

where  $\mathfrak{e}$  is the sub Lie algebra of a BKM algebra corresponding to its positive roots (cf. [174, 175, 183]).

In (7.8) we understand  $\bigwedge_{-1}(\mathfrak{e})$  to be the specialization of the formal series

$$\bigwedge_t(\mathfrak{e}) := \sum_{k \ge 0} \wedge^k(\mathfrak{e}) t^k \tag{7.9}$$

to t = -1, where  $\wedge^k(\mathfrak{e})$  is the k-th exterior power of  $\mathfrak{e}$ , and we write

$$H(\mathfrak{e}) := \sum_{k \ge 0} (-1)^k H_k(\mathfrak{e}) \tag{7.10}$$

for the alternating sum of the Lie algebra homology groups of  $\mathfrak{e}$ .

In the case of the monster Lie algebra  $\mathfrak{m}$ , the spaces  $\wedge^k(\mathfrak{e})$  and  $H_k(\mathfrak{e})$  are graded by  $H_{1,1}$ , and acted on naturally by the monster. If we use the variables p and q to keep track of the  $H_{1,1}$ -gradings, then the equality of (7.8) holds in the ring  $R(\mathbb{M})[[p,q]][q^{-1}]$  of formal power series in p and q (allowing finitely many negative powers of q), with coefficients in the (integral) representation ring of  $\mathbb{M}$ . More precisely, (7.8) becomes

$$\left(\sum_{\substack{m,n\in\mathbb{Z}\\m>0}} V_{mn}^{\natural} p^m q^n\right) = \sum_{m\in\mathbb{Z}} V_m^{\natural} p^{m+1} - \sum_{n\in\mathbb{Z}} V_n^{\natural} p q^n,$$
(7.11)

which returns (7.5), once we replace  $V_k^{\natural}$  everywhere with  $\dim(V_k^{\natural}) = c(k)$ , and divide both sides by p. More generally, replacing  $V_k^{\natural}$  with  $\operatorname{tr}(g|V_k^{\natural})$  for  $g \in \mathbb{M}$ , the identity (7.11) implies

$$p^{-1} \exp\left(-\sum_{k>0} \sum_{\substack{m,n \in \mathbb{Z} \\ m>0}} \frac{1}{k} \operatorname{tr}(g^k | V_{mn}^{\natural}) p^{mk} q^{nk}\right) = T_g(\sigma) - T_g(\tau)$$
(7.12)

(cf. [30], and also [173]), which, in turn, implies the replication formulas formulated in [81, 245]. Taking g = e in (7.12) we recover (7.5), so (7.12) furnishes a natural, monster-indexed family of analogues of the identity (7.5).

## 7.2 Modularity

Despite the power of the BKM algebra theory developed by Borcherds, and despite some conceptual improvements (cf. [87, 175, 176]) upon Borcherds' original proof of the moonshine conjectures, a conceptual explanation for the principal modulus property of monstrous moonshine is yet to be established. Indeed, there are generalizations and analogs of the notion of replicability which hold for generic modular functions and forms (for example, see [49]), not just those modular functions which are principal moduli.

Zhu explained the modularity of the graded dimension  $\sum_n \dim(V_n^{\natural})q^n$  of  $V^{\natural}$  in [310], by proving that this is typical for vertex operator algebras satisfying quite general hypotheses, and Dong-Li-Mason extended Zhu's work in [100], obtaining modular invariance results for graded trace functions arising from the action of a finite group of automorphisms.

To prepare for a statement of the results of Zhu and Dong-Li-Mason, we mention that the module theory for vertex operators algebras includes so-called *twisted modules*, associated to finite order automorphisms. If g is a finite order automorphism of V, then V is called *g*-rational in case every g-twisted V-module is a direct sum of simple g-twisted V-modules. Dong-Li-Mason proved [99] that a g-rational vertex operator algebra has finitely many simple g-twisted modules up to isomorphism. So in particular, a rational vertex operator algebra has finitely many simple (untwisted) modules.

Theorem 7.3 (Zhu, Dong-Li-Mason). Let V be rational  $C_2$ -cofinite vertex operator algebra. Then the generating functions  $\sum_n \dim(M_n^i)q^n$ , of the graded dimensions of its simple modules  $M^i = \bigoplus_n M_n^i$ , span a finite-dimensional representation of  $SL_2(\mathbb{Z})$ . More generally, if G is a finite subgroup of Aut(V)and V is g-rational for every  $g \in G$ , then the graded trace functions  $\sum_n tr(\tilde{h}|M_n)q^n$ , attached to the triples  $(g, \tilde{h}, M)$ , where  $g, h \in G$  commute, M is a simple h-stable g-twisted module for V, and  $\tilde{h}$  is a lift of h to GL(M), span a finitedimensional representation of  $SL_2(\mathbb{Z})$ .

We refer to the Introduction of [100] (see also §2 of [101]) for a discussion of *h*-stable twisted modules, and the relevant notion of *lift*. Note that any two lifts for *h* differ only up to multiplication by a non-zero scalar, so  $\sum_{n} \operatorname{tr}(\tilde{h}|M_{n})q^{n}$  is uniquely defined by (g, h, M), up to a non-zero scalar.

In the case of  $V^{\natural}$ , there is a unique simple g-twisted module  $V_g^{\natural} = \bigoplus_n (V_g^{\natural})_n$  for every  $g \in \mathbb{M} = \operatorname{Aut}(V^{\natural})$  (cf. Theorem 1.2 of [99]), and  $V_g^{\natural}$  is necessarily h-stable for any  $h \in \mathbb{M}$  that commutes with g. Therefore, Theorem 7.3 suggests that the functions

$$T_{(g,\tilde{h})}(\tau) := \sum_{n} \operatorname{tr}(\tilde{h}|(V_g^{\natural})_n) q^n, \qquad (7.13)$$

associated to pairs (g, h) of commuting elements of  $\mathbb{M}$ , may be of interest.

Indeed, this was anticipated a decade earlier by Norton, following observations of Conway–Norton [81] and Queen [252], which associated principal moduli to elements of groups that appear as centralizers of cyclic subgroups in the monster. Norton formulated his *generalized moonshine* conjectures in [246] (cf. also [247], and the Appendix to [233]).
Conjecture 7.4 (Generalized Moonshine: Norton). There is an assignment of holomorphic functions  $T_{(g,\tilde{h})} : \mathbb{H} \to \mathbb{C}$  to every pair (g,h) of commuting elements in the monster, such that the following are true:

- 1. For every  $x \in \mathbb{M}$  we have  $T_{(x^{-1}gx,x^{-1}\tilde{h}x)} = T_{(g,\tilde{h})}$ .
- 2. For every  $\gamma \in SL_2(\mathbb{Z})$  we have that  $T_{(g,\tilde{h})\gamma}(\tau)$  is a scalar multiple of  $T_{(g,\tilde{h})}(\gamma\tau)$ .
- 3. The coefficient functions  $\tilde{h} \mapsto \operatorname{tr}(\tilde{h}|(V_g^{\natural})_n)$ , for fixed g and n, define characters of a projective representation of the centralizer of g in  $\mathbb{M}$ ,
- 4. We have that  $T_{(g,\tilde{h})}$  is either constant or a generator for the function field of a genus zero group  $\Gamma_{(g,h)} < SL_2(\mathbb{R})$ .
- 5. We have that  $T_{(q,\tilde{h})}$  is a scalar multiple of J if and only if g = h = e.

*Remark.* In Conjecture 7.4 (2) above, the right-action of  $SL_2(\mathbb{Z})$  on commuting pairs of elements of the monster is given by

$$(g,h)\gamma := (g^a h^c, g^b h^d) \tag{7.14}$$

for  $\gamma = \begin{pmatrix} a & b \\ c & d \end{pmatrix}$ . The (slightly ambiguous)  $T_{(g,\tilde{h})\gamma}$  denotes the graded trace of a lift of  $g^b h^d$  to  $GL(V_{g^a h^c}^{\natural})$ . Norton's generalized moonshine conjectures reduce to the original Conway–Norton moonshine conjectures of [81] when g = e.

Conjecture 7.4 is yet to be proven in full, but has been established for a number of special cases. Theorem 7.3 was used by Dong-Li-Mason in [100], following an observation of Tuite (cf. [98], and [286-288] for broader context), to prove Norton's conjecture for the case that g and h generate a cyclic subgroup of  $\mathbb{M}$ , and this approach, via twisted modules for  $V^{\natural}$ , has been extended by Ivanov-Tuite in [169, 170]. Höhn obtained a generalization of Borcherds' method by using a particular twisted module for  $V^{\natural}$  to construct a BKM algebra adapted to the case that g is in the class named 2A in [80] the smaller of the two conjugacy class of involutions in M—and in so doing established [162] generalized moonshine for the functions  $T_{(g,\tilde{h})}$  with  $g \in 2A$ . So far the most general results in generalized moonshine have been obtained by Carnahan [64–66]. (See [67] for a recent summary.)

Theorem 7.3 explains why the McKay–Thompson series  $T_g(\tau)$  of (1.37), and the  $T_{(g,\tilde{h})}(\tau)$  of (7.13) more generally, should be invariant under the actions of (finite index) subgroups of  $SL_2(\mathbb{Z})$ , but it does not explain the surprising predictive power of monstrous moonshine. That is, it does not explain why the full invariance groups  $\Gamma_g$  of the  $T_g$  should be so large that they admit normalized principal moduli, nor does it explain why the  $T_g$ should actually be these normalized principal moduli.

A program to establish a conceptual foundation for the principal modulus property of monstrous moonshine, via *Rademacher sums* and *threedimensional gravity*, was initiated in [110] by Duncan and Frenkel.

## 7.3 Rademacher Sums

To explain the conjectural connection between gravity and moonshine, we first recall some history. The roots of the approach of [110] extend back almost a hundred years, to Einstein's theory of general relativity, formulated in 1915, and the introduction of the circle method in analytic number theory, by Hardy–Ramanujan [158]. At the same time that pre-war efforts to quantize Einstein's theory of gravity were gaining steam (see [276] for a review), the circle method was being refined and developed, by Hardy–Littlewood (cf. [157]), and Rademacher [254], among others. (See [290] for a detailed account of what is now known as the *Hardy–Littlewood circle method*.) Despite being contemporaneous, these works were unrelated in science until this century: as we will explain presently, Rademacher's analysis led to a

Poincaré series-like expression—the prototypical Rademacher sum—for the elliptic modular invariant  $J(\tau)$ . It was suggested first in [253] (see also [231]) that this kind of expression might be useful for the computation of partition functions in quantum gravity.

Rademacher "perfected" the circle method introduced by Hardy–Ramanujan, and he obtained an exact convergent series expression for the combinatorial partition function p(n). In 1938 he generalized this work [255] and obtained such exact formulas for the Fourier coefficients of general modular functions. For the elliptic modular invariant  $J(\tau) = \sum_{n} c(n)q^{n}$  (cf. (1.33)), Rademacher's formula (which was obtained earlier by Petersson [251], via a different method) may be written as

$$c(n) = 4\pi^2 \sum_{c>0} \sum_{\substack{0 < a < c \\ (a,c)=1}} \frac{e^{-2\pi i \frac{a}{c}} e^{2\pi i n \frac{d}{c}}}{c^2} \sum_{k \ge 0} \frac{(4\pi^2)^k}{c^{2k}} \frac{1}{(k+1)!} \frac{n^k}{k!},$$
(7.15)

where d, in each summand, is a multiplicative inverse for a modulo c, and (a, c) is the greatest common divisor of a and c. Having established the formula (7.15), Rademacher sought to reverse the process, and use it to derive the modular invariance of  $J(\tau)$ . That is, he set out to prove directly that  $J_0(\tau+1) = J_0(-1/\tau) = J_0(\tau)$ , when  $J_0(\tau)$  is defined by setting  $J_0(\tau) = q^{-1} + \sum_{n>0} c(n)q^n$ , with c(n) defined by (7.15).

Rademacher achieved this goal in [256], by reorganizing the summation

$$\sum_{n>0} c(n)q^n = 4\pi^2 \sum_{n>0} \sum_{c>0} \sum_{\substack{0 < a < c \\ (a,c) = 1}} \frac{e^{-2\pi i \frac{a}{c}} e^{2\pi i n(\tau + \frac{a}{c})}}{c^2} \sum_{k\ge 0} \frac{(4\pi^2)^k}{c^{2k}} \frac{1}{(k+1)!} \frac{n^k}{k!}$$
(7.16)

into a Poincaré series-like expression for J. More precisely, Rademacher proved that

$$J(\tau) + 12 = e^{-2\pi i \tau} + \lim_{K \to \infty} \sum_{\substack{0 < c < K \\ -K^2 < d < K^2 \\ (c,d)=1}} e^{-2\pi i \frac{a\tau + b}{c\tau + d}} - e^{-2\pi i \frac{a}{c}},$$
(7.17)

where  $a, b \in \mathbb{Z}$  are chosen arbitrarily, in each summand, so that ad - bc = 1. We call the right hand side of (7.17) the first *Rademacher sum*.

Rademacher's expression (7.17) for the elliptic modular invariant J is to be compared to the formal sum

$$\sum_{\substack{c,d\in\mathbb{Z}\\(c,d)=1}} e^{-2\pi i m \frac{a\tau+b}{c\tau+d}},\tag{7.18}$$

for *m* a positive integer, which we may regard as a (formal) Poincaré series of weight zero for  $SL_2(\mathbb{Z})$ . In particular, (7.18) is (formally) invariant for the action of  $SL_2(\mathbb{Z})$ , as we see by recognizing the matrices  $\begin{pmatrix} a & b \\ c & d \end{pmatrix}$  as representatives for the right coset space  $\Gamma_{\infty} \setminus \Gamma$ , where  $\Gamma = SL_2(\mathbb{Z})$  and  $\Gamma_{\infty}$  is defined in (1.38): for a fixed bottom row (c, d) of matrices in  $SL_2(\mathbb{Z})$ , any two choices for the top row (a, b) are related by left-multiplication by some element of  $\Gamma_{\infty}$ .

The formal sum (7.18) does not converge for any  $\tau \in \mathbb{H}$ , so a regularization procedure is required. Rademacher's sum (7.17) achieves this, for m = 1, by constraining the order of summation, and subtracting the limit as  $\Im(\tau) \to \infty$ of each summand  $e^{-2\pi i \frac{a\tau+b}{c\tau+d}}$ , whenever this limit makes sense. Rademacher's method has by now been generalized in various ways by a number of authors. The earliest generalizations are due to Knopp [189–192], and a very general negative weight version of the Rademacher construction was given by Niebur in [243]. We refer to [71] for a detailed review and further references. A nice account of the original approach of Rademacher appears in [188].

We note here that one of the main difficulties in establishing formulas like (7.17) is the demonstration of convergence. When the weight w of the Rademacher sum under consideration lies in the range  $0 \le w \le 2$ , then one requires non-trivial estimates on sums of Kloosterman sums, like

$$\sum_{c>0} \sum_{\substack{0 < a < c \\ (a,c)=1}} \frac{e^{-2\pi i m \frac{a}{c}} e^{2\pi i n \frac{d}{c}}}{c^2}$$
(7.19)

(for the case that w = 0 or w = 2). The demonstration of convergence generally becomes more delicate as w approaches 1.

In [110] the convergence of a weight zero Rademacher sum  $R_{\Gamma}^{(-m)}(\tau)$  is shown, for m a positive integer and  $\Gamma$  an arbitrary subgroup of  $SL_2(\mathbb{R})$  that is commensurable with  $SL_2(\mathbb{Z})$ . Assuming that  $\Gamma$  contains -I and has width one at infinity (cf. (1.38)), we have

$$R_{\Gamma}^{(-m)}(\tau) = e^{-2\pi i m \tau} + \lim_{K \to \infty} \sum_{(\Gamma_{\infty} \setminus \Gamma)_{< K}^{\times}} e^{-2\pi i m \frac{a\tau + b}{c\tau + d}} - e^{-2\pi i m \frac{a}{c}}, \qquad (7.20)$$

where the summation, for fixed K, is over non-trivial right cosets of  $\Gamma_{\infty}$  in  $\Gamma$  (cf. (1.38)), having representatives  $\begin{pmatrix} a & b \\ c & d \end{pmatrix}$  such that 0 < c < K and  $|d| < K^2$ .

The modular properties of the  $R_{\Gamma}^{(-m)}$  are also considered in [110], and it is at this point that the significance of Rademacher sums in monstrous moonshine appears. To state the relevant result we give the natural generalization (cf. §3.2 of [110]) of the Rademacher–Petersson formula (7.15) for c(n), which is

$$c_{\Gamma}(-m,n) = 4\pi^2 \lim_{K \to \infty} \sum_{(\Gamma_{\infty} \setminus \Gamma/\Gamma_{\infty})_{
(7.21)$$

where the summation, for fixed K, is over non-trivial double cosets of  $\Gamma_{\infty}$  in  $\Gamma$  (cf. (1.38)), having representatives  $\begin{pmatrix} a & b \\ c & d \end{pmatrix}$  such that 0 < c < K. Note that this formula simplifies for n = 0, to

$$c_{\Gamma}(-m,0) = 4\pi^2 m \lim_{K \to \infty} \sum_{(\Gamma_{\infty} \setminus \Gamma / \Gamma_{\infty})_{< K}^{\times}} \frac{e^{-2\pi i m \frac{a}{c}}}{c^2}.$$
 (7.22)

The value  $c_{\Gamma}(-1,0)$  is the *Rademacher constant* attached to  $\Gamma$ . (Cf. §6 of [245] and §5.1 of [110].)

A normalized Rademacher sum  $T_{\Gamma}^{(-m)}(\tau)$  is defined in §4.1 of [110] by introducing an extra complex variable and taking a limit. It is shown in §4.4 of [110] that

$$T_{\Gamma}^{(-m)}(\tau) = R_{\Gamma}^{(-m)}(\tau) - \frac{1}{2}c_{\Gamma}(-m,0)$$
(7.23)

for any group  $\Gamma < SL_2(\mathbb{R})$  that is commensurable with  $SL_2(\mathbb{Z})$ . If  $\Gamma$  has width one at infinity (cf. (1.38)), then also

$$T_{\Gamma}^{(-m)}(\tau) = q^{-m} + \sum_{n>0} c_{\Gamma}(-m, n)q^n, \qquad (7.24)$$

so in particular,  $T_{\Gamma}^{(-m)}(\tau) = q^{-m} + O(q)$  as  $\Im(\tau) \to \infty$ . The following theorem by Duncan and Frenkel summarizes the central role of Rademacher sums and the principal modulus property.

Theorem 7.5 (Duncan-Frenkel [110]). Let  $\Gamma$  be a subgroup of  $SL_2(\mathbb{R})$  that is commensurable with  $SL_2(\mathbb{Z})$ . Then the normalized Rademacher sum  $T_{\Gamma}^{(-m)}$ is  $\Gamma$ -invariant if and only if  $\Gamma$  has genus zero. Furthermore, if  $\Gamma$  has genus zero then  $T_{\Gamma}^{(-1)}$  is the normalized principal modulus for  $\Gamma$ .

In the case that the normalized Rademacher sum  $T_{\Gamma}^{(-1)}$  is not  $\Gamma$ -invariant,  $T_{\Gamma}^{(-m)}$  is an *abelian integral of the second kind* for  $\Gamma$ , in the sense that it has at most exponential growth at the cusps of  $\Gamma$ , and satisfies  $T_{\Gamma}^{(-m)}(\gamma \tau) = T_{\Gamma}^{(-m)}(\tau) + \omega(\gamma)$  for  $\gamma \in \Gamma$ , for some function  $\omega : \Gamma \to \mathbb{C}$  (depending on m).

Theorem 7.5 is used as a basis for the formulation of a characterization of the discrete groups  $\Gamma_g$  of monstrous moonshine in terms of Rademacher sums in §6.5 of [110], following earlier work [84] of Conway–McKay–Sebbar. It also facilitates a proof of the following result, which constitutes a uniform construction of the McKay–Thompson series of monstrous moonshine.

Theorem 7.6 (Duncan–Frenkel [110]). Let  $g \in \mathbb{M}$ . Then the McKay–Thompson series  $T_g$  coincides with the normalized Rademacher sum  $T_{\Gamma_g}^{(-1)}$ .

*Proof.* Theorem 7.2 states that  $T_g$  is a normalized principal modulus for  $\Gamma_g$ , and in particular, all the  $\Gamma_g$  have genus zero. Given this, it follows from

Theorem 7.5 that  $T_{\Gamma_g}^{(-1)}$  is also a normalized principal modulus for  $\Gamma_g$ . A normalized principal modulus is unique if it exists, so we conclude  $T_g = T_{\Gamma_g}^{(-1)}$  for all  $g \in \mathbb{M}$ , as we required to show.

Perhaps most importantly, Theorem 7.5 is an indication of how the principal modulus property of monstrous moonshine can be explained conceptually. For if we can develop a mathematical theory in which the underlying objects are graded with graded traces that are provably

- 1. modular invariant, for subgroups of  $SL_2(\mathbb{R})$  that are commensurable with  $SL_2(\mathbb{Z})$ , and
- 2. given explicitly by Rademacher sums, such as (7.20),

then these graded trace functions are necessarily normalized principal moduli, according to Theorem 7.5.

We are now led to ask: what kind of mathematical theory can support such results? As we have alluded to above, Rademacher sums have been related to quantum gravity by articles in the physics literature. Also, a possible connection between the monster and three-dimensional quantum gravity was discussed in [304]. This suggests the possibility that three-dimensional quantum gravity and moonshine are related via Rademacher sums, and was a strong motivation for the work [110]. In the next section we will give a brief review of quantum gravity, since it is an important area of physical inquiry which has played a role in the development of moonshine, but we must first warn the reader: problems have been identified with the existing conjectures that relate the monster to gravity, and the current status of this connection is uncertain.

## 7.4 Quantum Gravity

A positive solution to the conjecture that  $V^{\ddagger}$  is dual to chiral three-dimensional gravity at m = 1 may furnish a conceptual explanation for why the graded dimension of  $V^{\ddagger}$  is the normalized principal modulus for  $SL_2(\mathbb{Z})$ . For on the one hand, modular invariance is a consistency requirement of the physical theory—the genus one partition function function is really defined on the moduli space  $SL_2(\mathbb{Z}) \setminus \mathbb{H}$  of genus one curves, rather than on  $\mathbb{H}$ —and on the other hand, the genus one partition function of chiral three-dimensional gravity is given by a Rademacher sum, as explained by Manschot–Moore [231], following earlier work [253] by Dijkgraaf–Maldacena–Moore–Verlinde. (Cf. also [228–230].) So, as we discussed in §7.3, the genus one partition function must be the normalized principal modulus  $J(\tau)$  for  $SL_2(\mathbb{Z})$ , according to Theorem 7.5.

In the analysis of [230, 231], the genus one partition function of chiral three-dimensional gravity is a Rademacher sum (7.20), because it is obtained as a sum over three-dimensional hyperbolic structures on a solid torus with genus one boundary, and such structures are naturally parameterized by the coset space  $\Gamma_{\infty} \setminus SL_2(\mathbb{Z})$  (cf. (1.38)), as explained in [224] (see also §5.1 of [253]). The terms  $e^{-2\pi i m \frac{a\tau+b}{c\tau+d}}$  in (7.20) are obtained by evaluating  $e^{-I_{\text{TMG}}}$ , with  $\mu = \sqrt{-\Lambda} = 1/16Gm$ , on a solution with boundary curve  $C/(\mathbb{Z} + \tau \mathbb{Z})$ , and the subtraction of  $e^{-2\pi i m \frac{a}{c}}$  represents quantum corrections to the classical action.

In [110], the above conjecture is extended so as to encompass the principal modulus property for all elements of the monster, with a view to establishing a conceptual foundation for monstrous moonshine. More specifically, the first main conjecture of [110] states the following.

Conjecture 7.7 (Duncan–Frenkel). There exists a monster-indexed family of twisted chiral three-dimensional gravity theories, whose genus one partition

functions at

$$\mu = \sqrt{-\Lambda} = 1/16G \tag{7.25}$$

are given by  $T_{\Gamma_g}^{(-1)}(-1/\tau)$ , where  $T_{\Gamma_g}^{(-1)}(\tau)$  is the normalized Rademacher sum attached to  $\Gamma_g$ , satisfying (7.23).

From the point of view of vertex operator algebra theory,  $T_g(-1/\tau)$  which coincides with  $T_{\Gamma_g}^{(-1)}(-1/\tau)$  according to Theorems 7.2 and 7.5—is the graded dimension of the unique simple g-twisted  $V^{\natural}$ -module  $V_g^{\natural}$  (cf. §7.2). This non-trivial fact about the functions  $T_g(-1/\tau)$  is proven by Carnahan in Theorem 5.1.4 of [66].

Geometrically, the twists of the above conjecture are defined by imposing (generalized) spin structure conditions on solutions to the chiral gravity equations, and allowing orbifold solutions of certain kinds. See §7.1 of [110] for a more complete discussion. The corresponding sums over geometries are then indexed by coset spaces  $\Gamma_{\infty} \setminus \Gamma$ , for various groups  $\Gamma < SL_2(\mathbb{R})$ , commensurable with  $SL_2(\mathbb{Z})$ . According to Theorem 7.5, the genus one partition function corresponding to such a twist, expected to be a Rademacher sum on physical grounds, will only satisfy the basic physical consistency condition of  $\Gamma$ -invariance if  $\Gamma$  is a genus zero group. One may speculate that a finer analysis of physical consistency will lead to the list of conditions given in §6.5 of [110], which characterize the groups  $\Gamma_g$  for  $g \in \mathbb{M}$ , according to Theorem 6.5.1 of [110]. Thus the discrete groups  $\Gamma_g$  of monstrous moonshine may ultimately be recovered as those defining physically consistent twists of chiral three-dimensional gravity.

On the other hand, it is reasonable to expect that twisted chiral gravity theories are determined by symmetries of the underlying untwisted theory. Conceptually then, but still conjecturally, the monster group appears as the symmetry group of chiral three-dimensional gravity, for which the corresponding twists exist. The principal modulus property of monstrous moonshine may then be explained: as a consequence of Theorem 7.5, together with the statement that the genus one partition function of a twisted theory is  $T_{\Gamma}^{(-1)}(-1/\tau)$ , where  $T_{\Gamma}^{(-1)}(\tau)$  is the normalized Rademacher sum attached to the subgroup  $\Gamma < SL_2(\mathbb{R})$  that parameterizes the geometries of the twist.

For more background on the mathematics and physics of black holes we refer the reader to [88]. We refer to [62,63] for reviews that focus on the particular role of conformal field theory in understanding quantum gravity.

## 7.5 Moonshine Tower

An optimistic view on the relationship between moonshine and gravity is adopted in §7 of [110]. In particular, in §7.2 of [110] the consequences of Conjecture 7.7 for the *second quantization* of chiral three-dimensional gravity are explored. (We warn the reader that the notion of second quantized gravity is very speculative at this stage.)

Motivated in part by the results on second quantized string theory in [93], the existence of a tower of monster modules

$$V^{(-m)} = \bigoplus_{n=-m}^{\infty} V_n^{(-m)}, \qquad (7.26)$$

parameterized by positive integer values of m, is predicted in §7.2 of [110]. Moreover, it is suggested that the graded dimension of  $V^{(-m)}$  should be given by

$$J^{(-m)} := m\hat{T}(m)J, (7.27)$$

where  $\hat{T}(m)$  denotes the (order m) Hecke operator, acting on  $SL_2(\mathbb{Z})$ -invariant holomorphic functions on  $\mathbb{H}$  according to the rule

$$(\hat{T}(m)f)(\tau) := \frac{1}{m} \sum_{\substack{ad=m\\0\le b< d}} f\left(\frac{a\tau+b}{d}\right).$$
(7.28)

Standard calculations (cf. e.g. Chp.VII, §5 of [270]) determine that  $m\hat{T}(m)J$  is an  $SL_2(\mathbb{Z})$ -invariant holomorphic function on  $\mathbb{H}$ , whose Fourier coefficients

$$J^{(-m)}(\tau) = \sum_{n} c(-m, n)q^{n}$$
(7.29)

are expressed in terms of those of  $J(\tau) = \sum_{n=-1}^{\infty} c(n)q^n$ , by  $c(-m,n) = \delta_{-m,n}$ for  $n \leq 0$ , and

$$c(-m,n) = \sum_{\substack{k>0\\k|(m,n)}} \frac{m}{k} c(mn/k^2),$$
(7.30)

for n > 0, where (m, n) denotes the greatest common divisor of m and n. In particular,  $J^{(-m)}(\tau) = q^{-m} + O(q)$  as  $\Im(\tau) \to \infty$ . There is only one such  $SL_2(\mathbb{Z})$ -invariant holomorphic function on  $\mathbb{H}$ , so we have

$$J^{(-m)}(\tau) = \sum_{n=-m}^{\infty} \dim(V^{(-m)})q^n = T_{\Gamma}^{(-m)}(\tau)$$
(7.31)

according to (7.24) and Theorem 7.5, when  $\Gamma = SL_2(\mathbb{Z})$ . So the graded dimension of  $V^{(-m)}$  is also a normalized Rademacher sum.

We would like to investigate the higher order analogues of the McKay– Thompson series  $T_g$  (cf. (1.37)), encoding the graded traces of monster elements on  $V^{(-m)}$ , but for this we must first determine the M-module structure on each homogeneous subspace  $V_n^{(-m)}$ .

A solution to this problem is entailed in Borcherds' proof [30] of the monstrous moonshine conjectures, and the identity (7.11), in particular. To explain this, recall the *Adams operation*  $\psi^k$  on virtual *G*-modules, defined, for  $k \ge 0$  and *G* a finite group, by requiring that

$$\operatorname{tr}(g|\psi^k(V)) = \operatorname{tr}(g^k|V) \tag{7.32}$$

for  $g \in G$ . (Cf. [16, 193] for more details on Adams operations.) Using the  $\psi^k$  we may equip  $V^{(-m)}$  with a virtual M-module structure (we will see momentarily that it is actually an M-module, cf. Proposition 7.9) by defining  $V_{-m}^{(-m)} := C$  to be the one-dimensional trivial M-module,  $V_n^{(-m)} := 0$  for  $-m < n \leq 0$ , and

$$V_n^{(-m)} := \bigoplus_{\substack{k>0\\k\mid(m,n)}} \mathcal{C}^{m/k} \otimes \psi^k(V_{mn/k^2}^{\natural})$$
(7.33)

for n > 0, where  $C^{m/k}$  denotes the trivial M-module of dimension m/k. For convenience later on, we also define  $V^{(0)} = V_0^{(0)} := C$  to be the trivial, onedimensional M-module, regarded as graded, with grading concentrated in degree n = 0.

Evidently  $\psi^k$  preserves dimension, so the graded dimension of  $V^{(-m)}$ is still given by  $J^{(-m)}$ , according to (7.30). Define the order *m* McKay– Thompson series  $T_g^{(-m)}$ , for  $m \ge 0$  and  $g \in \mathbb{M}$ , by setting

$$T_g^{(-m)}(\tau) := q^{-m} + \sum_{n>0} \operatorname{tr}(g|V_n^{(-m)})q^n.$$
(7.34)

Then  $T_g^{(0)} = 1$  for all  $g \in \mathbb{M}$ , and  $T_g^{(-1)}$  is the original McKay–Thompson series  $T_g$ . More generally, we have the following result, which constructs the  $T_g^{(-m)}$  uniformly and explicitly as Rademacher sums.

Theorem 7.8. For m > 0 and  $g \in \mathbb{M}$  we have  $T_g^{(-m)}(\tau) = T_{\Gamma_g}^{(-m)}(\tau)$ , where  $\Gamma_g$ is the invariance group of  $T_g(\tau)$ , and  $T_{\Gamma}^{(-m)}$  denotes the normalized Rademacher sum of order m attached to  $\Gamma$ , as in (7.23). In particular,  $T_g^{(-m)}(\tau)$  is a monic integral polynomial of degree m in  $T_g(\tau)$ .

*Proof.* We will use Borcherds' identity (7.11). To begin, note that  $T_g^{(-m)}$  is given explicitly in terms of traces on  $V^{\natural}$  by

$$T_g^{(-m)}(\tau) = q^{-m} + \sum_{n>0} \sum_{k|(m,n)} \frac{m}{k} \operatorname{tr}(g|\psi^k(V_{mn/k^2}^{\natural}))q^n$$
(7.35)

according to (7.33) and (7.34). Recall that R(G) denotes the integral representation ring of a finite group G. Extend the  $\psi^k$  from R(G) to  $R(G)[[p,q]][q^{-1}]$ , by setting  $\psi^k(Mp^mq^n) = \psi^k(M)p^{km}q^{km}$  for  $M \in R(G)$ . Then it is a general property of the Adams operations (cf. §5.2 of [173]) that

$$\log \bigwedge_{-1} (X) = -\sum_{k>0} \frac{1}{k} \psi^k(X)$$
 (7.36)

in  $R(G)[[p,q]][q^{-1}] \otimes_{\mathbb{Z}} \mathbb{Q}$ , for  $X \in R(G)[[p,q]][q^{-1}]$ . So taking  $X = \sum_{\substack{m,n \in \mathbb{Z} \\ m>0}} V_{mn}^{\natural} p^m q^n$  we obtain

$$\log \bigwedge_{-1} \left( \sum_{\substack{m,n \in \mathbb{Z} \\ m>0}} V_{mn}^{\natural} p^m q^n \right) = -\sum_{\substack{k>0 \\ m>0}} \sum_{\substack{m,n \in \mathbb{Z} \\ m>0}} \frac{1}{k} \psi^k (V_{mn}^{\natural}) p^{km} q^{kn}$$

$$= -\sum_{\substack{m,n \in \mathbb{Z} \\ m>0}} \sum_{\substack{k \mid (m,n) \\ m>0}} \frac{1}{k} \psi^k (V_{mn/k^2}^{\natural}) p^m q^n$$
(7.37)

for the logarithm of the left hand side of (7.11). If we now define  $V^{(-m)}(q) := \sum_n V_n^{(-m)} q^n$ , an element of  $R(\mathbb{M})[[q]][q^{-1}]$ , then the generating series  $\sum_{m>0} p^m V^{(-m)}(q)$  is obtained when we apply  $-p\partial_p$  to (7.37), according to the definition (7.33) of the  $V_n^{(-m)}$  as elements of  $R(\mathbb{M})$ . So apply  $-p\partial_p \log(\cdot)$  to both sides of (7.11) to obtain the identity

$$\sum_{m>0} V^{(-m)}(q) p^m = -1 - (p\partial_p V^{\natural}(p)) \sum_{k\geq 0} V^{\natural}(q)^k V^{\natural}(p)^{-k-1}$$
(7.38)

in  $R(\mathbb{M})[[p,q]][q^{-1}]$ , where  $V^{\natural}(q) = V^{(-1)}(q) = q^{-1} + \sum_{n>0} V_n^{\natural} q^n$ . The right hand side of (7.38) really is a taylor series in p, for we use  $V^{\natural}(p)^{-1}$  as a short hand for  $\sum_{k\geq 0} (-1)^k p^{k+1} V_+^{\natural}(p)$ , where  $V_+^{\natural}(p) := \sum_{n>0} V_n^{\natural} p^n$  is the regular part of  $V^{\natural}(p)$ .

The McKay–Thompson series  $T_g^{(-m)}(\tau)$  is just the trace of g on  $V^{(-m)}(q)$ , so an application of  $\operatorname{tr}(g|\cdot)$  to (7.38) replaces  $V^{(-m)}(q)$  with  $T_g^{(-m)}(\tau)$ , and  $V^{\natural}(q)$  with  $T_g(\tau)$ , etc. and shows that  $T_g^{(-m)}$  is indeed a polynomial in  $T_g$ , of degree m since the leading term of  $T_g^{(-m)}$  is  $q^{-m}$  by definition. In particular,  $T_g^{(-m)}$  is a modular function for  $\Gamma_g$ , with no poles away from the infinite cusp. Since  $\Gamma_g$  has genus zero, such a function is uniquely determined (up to an additive constant) by the polar terms in its Fourier expansion. The McKay–Thompson series  $T_g^{(-m)}$  and the Rademacher sum  $T_{\Gamma_g}^{(-m)}$  both satisfy  $q^{-m} + O(q)$  as  $\Im(\tau) \to \infty$  (cf. (7.24)), and neither have poles away from the infinite cusp, so they must coincide. This completes the proof.

*Remark.* The identity obtained by taking the trace of  $g \in \mathbb{M}$  on (7.38) may be compactly rewritten

$$\sum_{m \ge 0} T_g^{(-m)}(\tau) p^m = \frac{p \partial_p T_g(\sigma)}{T_g(\tau) - T_g(\sigma)},$$
(7.39)

where  $p = e^{2\pi i\sigma}$  and  $T_g(\sigma) = \sum_m \operatorname{tr}(g|V_m^{\natural})p^m$ . This expression (7.39) is proven for some special cases by a different method in [21].

Recall that the monster group has 194 irreducible ordinary representations, up to equivalence. Let us denote these by  $M_i$ , for  $1 \le i \le 194$ , where the ordering is as in [80], so that the character of  $M_i$  is the function denoted  $\chi_i$  in [80]. Define  $\mathbf{m}_i(-m, n)$  to be the multiplicity of  $M_i$  in  $V_n^{(-m)}$ , so that

$$V_n^{(-m)} \approx \bigoplus_{i=1}^{194} M_i^{\oplus \mathbf{m}_i(-m,n)}$$
(7.40)

as M-modules, and  $c(-m,n) = \sum_{i=1}^{194} \mathbf{m}_i(-m,n)\chi_i(e).$ 

A priori, the M-modules  $V_n^{(-m)}$  may be virtual, meaning that some of the integers  $\mathbf{m}_i(-m, n)$  are negative.

Proposition 7.9. The  $V_n^{(-m)}$  are all (non-virtual) modules for the monster. In particular, the integers  $\mathbf{m}_i(-m,n)$  are all non-negative.

*Proof.* The claim follows from the modification of Borcherds' proof of Theorem 7.2 presented by Jurisich–Lepowsky–Wilson in [173]. In [173] a certain free Lie sub algebra  $\mathfrak{u}^-$  of the monster Lie algebra  $\mathfrak{m}$  is identified, for which the identity  $\Lambda(\mathfrak{u}^-) = H(\mathfrak{u}^-)$  (or rather, the logarithm of this) yields

$$\sum_{m,n>0} \sum_{k|(m,n)} \frac{1}{k} \psi^k (V_{mn/k^2}^{\natural}) p^m q^n = \sum_{k>0} \frac{1}{k} \left( \sum_{m,n>0} V_{m+n-1}^{\natural} p^m q^n \right)^k$$
(7.41)

in  $R(\mathbb{M})[[p,q]][q^{-1}] \otimes \mathbb{Q}$ . (Notice the different range of summation, compared to (7.11).) We apply  $p\partial_p$  to (7.41), and recall the definition (7.33) of  $V_n^{(-m)}$ to obtain

$$\sum_{m,n>0} V_n^{(-m)} p^m q^n = \sum_{k>0} \left( \sum_{m,n>0} m V_{m+n-1}^{\natural} p^m q^n \right) \left( \sum_{m,n>0} V_{m+n-1}^{\natural} p^m q^n \right)^{k-1}.$$
(7.42)

The coefficient of  $p^m q^n$  in the right hand side of (7.42) is evidently a nonnegative integer combination of the M-modules  $V_n^{\natural}$ , so the proof of the claim is complete.

In §7.6 we will determine the behavior of the multiplicity functions  $\mathbf{m}_i(-m, n)$ (cf. (7.40)) as  $n \to \infty$ . For applications to gravity a slightly different statistic is more relevant. Recall from §7.4 that it is the Virasoro highest weight vectors—i.e. those  $v \in V_n^{\natural}$  with L(k)v = 0 for k > 0—that represent black hole states in chiral three-dimensional gravity at m = 1. Such vectors generate highest weight modules for  $\mathcal{V}$ , the structure of which has been determined by Feigin–Fuchs in [118]. (See [15] for an alternative treatment.) Specializing to the case that the central element  $\mathbf{c}$  (cf. (7.1)) acts as c = 24m times the identity, for some positive integer m, we obtain from the results of [118] that the isomorphism type of an irreducible highest weight module for  $\mathcal{V}$  depends only on the L(0)-eigenvalue of a generating highest weight vector, v, and if L(0)v = hv for h a non-negative integer, then

$$\sum_{n} \dim(L(h,c)_{n})q^{n} = \begin{cases} q^{-m}(1-q)(q)_{\infty}^{-1} & \text{if } h = 0, \\ q^{h-m}(q)_{\infty}^{-1} & \text{if } h > 0, \end{cases}$$
(7.43)

where L(h, c) denotes the irreducible highest weight  $\mathcal{V}$ -module generated by v. We write  $L(h, c)_n$  for the subspace of L(h, c) with L(0)-eigenvalue h - m in (7.43), and

$$(q)_{\infty} := \prod_{n>0} (1-q^n).$$
 (7.44)

(See [156] for details of the calculation that returns (7.43) in the case that m = 1.)

*Remark.* We may now recognize the leading terms in (1.47) as exactly those of the graded dimension of the Virasoro module L(0, 24m).

It is known that  $V^{\natural}$  is a direct sum of highest weight modules for the Virasoro algebra (cf. e.g. [156]). Since the Virasoro and monster actions on  $V^{\natural}$  commute, we have an isomorphism

$$V^{\natural} \simeq L(0, 24) \otimes W_{-1}^{\natural} \oplus \bigoplus_{n>0} L(n+1, 24) \otimes W_n^{\natural}$$
(7.45)

of modules for  $\mathcal{V} \times \mathbb{M}$ , where  $W_n^{\natural}$  denotes the subspace of  $V_n^{\natural}$  spanned by Virasoro highest weight vectors. To investigate how the black hole states in  $V^{\natural}$  are organized by the representation theory of the monster, we define non-negative integers  $\mathbf{n}_i(n)$  by requiring that

$$W_n^{\natural} \simeq \bigoplus_{i=1}^{194} M_i^{\oplus \mathbf{n}_i(n)},\tag{7.46}$$

for  $n \geq -1$ .

Evidently  $\mathbf{n}_i(n) \leq \mathbf{m}_i(-1, n)$  for all *i* and *n* since  $W_n^{\natural}$  is a subspace of  $V_n^{\natural}$ . To determine the precise relationship between the  $\mathbf{n}_i(n)$  and  $\mathbf{m}_i(-1, n)$ , define  $U_g(\tau)$  for  $g \in \mathbb{M}$  by setting

$$U_g(\tau) := \sum_{n=-1}^{\infty} \operatorname{tr}(g|W_n^{\natural}) q^n, \qquad (7.47)$$

so that  $U_g(\tau) = q^{-1} + \sum_{n>0} \sum_{i=1}^{194} \mathbf{n}_i(n)\chi_i(g)q^n$  (cf. (7.46)). Combining (7.43), (7.45) and (7.46), together with the definitions (1.37) of  $T_g$  and (7.47) of  $U_g$ , we obtain

$$T_g(\tau) = q^{-1} \frac{(1-q)}{(q)_{\infty}} + \sum_{n>0} q^n \frac{1}{(q)_{\infty}} \sum_{i=1}^{194} \mathbf{n}_i(n) \chi_i(g),$$
(7.48)

or equivalently,

$$U_g(\tau) = (q)_{\infty} T_g(\tau) + 1$$
(7.49)

for all  $g \in \mathbb{M}$ . (This computation also appears in [156].)

In §7.6 we will use (7.49) to determine the asymptotic behavior of the  $\mathbf{n}_i(n)$  (cf. Theorem 7.10), and thus the statistics of black hole states, at  $\ell = 16G$ , in the conjectural chiral three-dimensional gravity theory dual to  $V^{\natural}$ .

Remark. Note that we may easily construct modules for  $\mathcal{V} \times \mathbb{M}$  satisfying the extremal condition (1.47), for each positive integer m, by considering direct sums of the monster modules  $V^{(-m)}$  constructed in Proposition 7.9. A very slight generalization of the argument just given will then yield formulas for the graded traces of monster elements on the corresponding Virasoro highest weight spaces. Since it has been shown [134, 163] that such modules cannot admit vertex operator algebra structure, we do not pursue this here.

# 7.6 Monstrous Moonshine's Distributions

We now address the problem of determining exact formulas and asymptotic distributions of irreducible components. This work will rely heavily on the modularity of the underlying McKay–Thompson series (i.e. Theorems 1.23 and 7.8).

We prove formulas for the multiplicities  $\mathbf{m}_i(-m, n)$  and  $\mathbf{n}_i(n)$  which in turn imply the following asymptotics.

Theorem 7.10. If m is a positive integer and  $1 \le i \le 194$ , then as  $n \to +\infty$ we have

$$\mathbf{m}_{i}(-m,n) \sim \frac{\dim(\chi_{i})|m|^{1/4}}{\sqrt{2}|n|^{3/4}|\mathbb{M}|} \cdot e^{4\pi\sqrt{|mn|}}$$
$$\mathbf{n}_{i}(n) \sim \frac{\sqrt{12} \dim(\chi_{i})}{|24n+1|^{1/2}|\mathbb{M}|} \cdot e^{\frac{\pi}{6}\sqrt{23|24n+1|}}$$

These asymptotics immediately imply that the following limits are welldefined

$$\delta\left(\mathbf{m}_{i}(-m)\right) := \lim_{n \to +\infty} \frac{\mathbf{m}_{i}(-m,n)}{\sum_{i=1}^{194} \mathbf{m}_{i}(-m,n)}$$
$$\delta\left(\mathbf{n}_{i}\right) := \lim_{n \to +\infty} \frac{\mathbf{n}_{i}(n)}{\sum_{i=1}^{194} \mathbf{n}_{i}(n)}.$$
(7.50)

Corollary 7.11. In particular, we have that

$$\delta\left(\mathbf{m}_{i}(-m)\right) = \delta\left(\mathbf{n}_{i}\right) = \frac{\dim(\chi_{i})}{\sum_{j=1}^{194} \dim(\chi_{j})} = \frac{\dim(\chi_{i})}{5844076785304502808013602136}$$

#### 7.6.1 The modular groups in monstrous moonshine

To obtain exact formulas, we begin by recalling the modular groups which arise in monstrous moonshine. Suppose  $\Gamma_* < GL_2(\mathbb{R})$  is a discrete group which is commensurable with  $SL_2(\mathbb{Z})$ . If  $\Gamma_*$  defines a genus zero quotient of  $\mathbb{H}$ , then the field of modular functions which are invariant under  $\Gamma_*$  is generated by a single element, the principal modulus (cf. (1.39)). Theorem 7.2 implies that the  $T_g$  (defined by (1.37)) are principal moduli for certain groups  $\Gamma_g$ . We can describe these groups in terms of groups  $E_g$  which in turn may be described in terms of the congruence subgroups

$$\Gamma_0(N) := \left\{ \begin{pmatrix} a & b \\ c & d \end{pmatrix} \in SL_2(\mathbb{Z}) : c \equiv 0 \pmod{N} \right\},$$
(7.51)

and the Atkin–Lehner involutions  $W_e$  for  $\Gamma_0(N)$  given by

$$W_e := \begin{pmatrix} ae & b\\ cN & de \end{pmatrix}, \tag{7.52}$$

where e is an exact divisor of N (i.e. e|N, and (e, N/e) = 1), and a, b, c, and d are integers chosen so that  $W_e$  has determinant e.

Following Conway–Norton [81] and Conway–McKay–Sebbar [84], we denote the groups  $E_g$  by symbols of the form  $\Gamma_0(N|h) + e, f, \ldots$  (or simply  $N|h+e, f, \ldots$ ), where h divides (N, 24), and each of e, f, etc. exactly divide N/h. This symbol represents the group

$$\Gamma_0(N|h) + e, f, \dots := \begin{pmatrix} 1/h & 0\\ 0 & 1 \end{pmatrix} \langle \Gamma_0(N/h), W_e, W_f, \dots \rangle \begin{pmatrix} h & 0\\ 0 & 1 \end{pmatrix}$$

where  $W_e, W_f$ , etc. are representative of Atkin-Lehner involutions on  $\Gamma_0(N/h)$ . We use the notation  $\mathcal{W}_g := \{1, e, f, ...\}$  to denote this list of Atkin-Lehner involutions contained in  $E_g$ . We also note that  $\Gamma_0(N|h) + e, f, ...$  contains  $\Gamma_0(Nh)$ .

The groups  $E_g$  are eigengroups for the  $T_g$ , so that if  $\gamma \in E_g$ , then  $T_g(\gamma \tau) = \sigma_g(\gamma)T_g$ , where  $\sigma_g$  is a multiplicative group homomorphism from  $E_g$  to the group of *h*-th roots of unity. Conway and Norton [81] give the following values for  $\sigma_g$  evaluated on generators of  $N|h + e, f, \ldots$ .

Lemma 7.12 (Conway–Norton). Assuming the notation above, the following are true:

- (a)  $\sigma_g(\gamma) = 1$  if  $\gamma \in \Gamma_0(Nh)$
- (b)  $\sigma_g(\gamma) = 1$  if  $\gamma$  is an Atkin–Lehner involution of  $\Gamma_0(Nh)$  inside  $E_g$

(c) 
$$\sigma_g(\gamma) = e^{\frac{-2\pi i}{h}} \text{ if } \gamma = \begin{pmatrix} 1 & 1/h \\ 0 & 1 \end{pmatrix}$$

(d) 
$$\sigma_g(\gamma) = e^{-\lambda_g \frac{2\pi i}{h}} \text{ if } \gamma = \begin{pmatrix} 1 & 0 \\ N & 1 \end{pmatrix},$$

where  $\lambda_g$  in (d) is -1 if  $N/h \in \mathcal{W}_g$ , and +1 otherwise.

This information is sufficient to properly describe the modularity of the series  $T_g^{(-m)}(\tau)$  on  $E_g$ . In section 7.6.2, we give an explicit procedure for evaluating  $\sigma_g$ . The invariance group  $\Gamma_g$ , denoted by  $\Gamma_0(N||h) + e, f, \ldots$  (or by the symbol  $N||h+e, f, \ldots$ ), is defined as the kernel of  $\sigma_g$ . A complete list of the groups  $\Gamma_g$  can be found in the Appendix (§A.1) of this paper, or in table 2 of [81].

Theorems 1.23 and 7.8 are summarized by the following uniform statement.

Theorem 7.13. Let  $g \in \mathbb{M}$  and  $m \geq 1$ . Then  $T_g^{(-m)}$  is the unique weakly holomorphic modular form of weight zero for  $\Gamma_g$  that satisfies  $T_g^{(-m)} = q^{-m} + O(q)$ as  $\tau$  approaches the infinite cusp, and has no poles at any cusps inequivalent to the infinite one.

*Remark.* A *weakly holomorphic* modular form is a meromorphic modular form whose poles (if any) are supported at cusps.

# 7.6.2 Exact formulas for $T_g^{(-m)}$

Using Theorems 2.3 and 2.4, we can write exact formulas for the coefficients of the  $T_g^{(-m)}$  provided we know its principal parts at all cusps of  $\Gamma_0(Nh)$ . With this in mind, we now regard  $T_g$  as a modular function on  $\Gamma_0(Nh)$  with trivial multiplier. The location and orders of the poles were determined by Harada and Lang [155].

Lemma 7.14. [155, Lemma 7, 9] Suppose the  $\Gamma_g$  is given by the symbol  $N||h + e, f..., and let L = \begin{pmatrix} -\delta & \beta \\ \gamma & -\alpha \end{pmatrix} \in SL_2(\mathbb{Z}).$  Then  $T_g|_0L$  has a pole if and only if  $\left(\frac{\gamma}{(\gamma,h)}, \frac{N}{h}\right) = \frac{N}{eh}$ , for some  $e \in \mathcal{W}_g$  (Note, here we allow e = 1). The order of the pole is given by  $\frac{(h,\gamma)^2}{eh^2}$ .

Harada and Lang prove this lemma by showing that if u is an integer chosen such  $\frac{u\gamma-\alpha\cdot(h,\gamma)}{h}$  is integeral and divisible by e and  $U = \begin{pmatrix} \frac{e\cdot h}{(h,\gamma)} & \frac{u}{h} \\ 0 & \frac{(h,\gamma)}{h} \end{pmatrix}$ , then LU is an Atkin–Lehner involution  $W_e \in E_g$ . Therefore, we have that

$$T_g|_0 L = \sigma_g(LU)T_g\left(\frac{(h,\gamma)^2}{eh^2}\tau - \frac{u\cdot(h,\gamma)}{eh^2}\right).$$
(7.53)

Harada and Lang do not compute  $\sigma_g(LU)$ , however we will need these values in order to apply Theorem 2.4. Using Lemma 7.12, the following procedure allows us to compute  $\sigma_g(M)$  for any matrix  $M \in E_g$ .

Given a matrix  $M \in E_g$ , we may write M as  $M = \begin{pmatrix} ae & \frac{b}{h} \\ cN & de \end{pmatrix}$  with  $e \in \mathcal{W}_g$  and  $ade - bc\frac{N}{eh} = 1$ . We may also write  $h = h_e \cdot h_{\overline{e}}$ , where  $h_{\overline{e}}$  is the largest divisor of h co-prime to e. Since  $c\frac{N}{eh}$  is co-prime to both d and e, we may chose integers A, B, and  $C \pmod{h}$  such that:

•  $c_{eh}^{N}A + d$  is co-prime to  $h_{\overline{e}}$  but is divisible by  $h_{e}$ ,

• 
$$B \equiv -(eaA+b)(c\frac{N}{h}A+ed)^{-1} \pmod{h_{\overline{e}}}$$
 and  $Bc\frac{N}{eh}+b \equiv 0 \pmod{h_e}$ ,

• 
$$C \equiv -c(c\frac{N}{h}A + ed)^{-1} \pmod{h_{\overline{e}}}$$
, and  $C \equiv 0 \pmod{h_e}$ 

A calculation shows that  $\widehat{M} := \begin{pmatrix} 1 & \frac{B}{h} \\ 0 & 1 \end{pmatrix} M \begin{pmatrix} 1 & \frac{A}{h} \\ 0 & 1 \end{pmatrix} \begin{pmatrix} 1 & 0 \\ CN & 1 \end{pmatrix} \begin{pmatrix} h_e & 0 \\ 0 & h_e \end{pmatrix}$ is an Atkin–Lehner involution  $W_E$  for  $\Gamma_0(Nh)$  where  $E = e \cdot h_e^2$ . By Lemma 7.12, this implies  $\sigma_g(\widehat{M}) = 1$ , and therefore

$$\sigma_g(M) = exp\left(\frac{2\pi i}{h}(A + B + \lambda_g C)\right)$$

Combined with Lemma 7.14, this leads us to the following proposition.

Proposition 7.15. Given a matrix  $L = \begin{pmatrix} -\delta & \beta \\ \gamma & -\alpha \end{pmatrix} \in SL_2(\mathbb{Z})$ , let u and U be chosen as above, and define

$$\epsilon_g(L) := \sigma_g(LU) \cdot e^{2\pi i \frac{u \cdot (h,\gamma)}{eh^2}}.$$

Then by (7.53), we have that

$$T_g|_0 L = \epsilon_g(L)q^{-\frac{(h,\gamma)^2}{eh^2}} + O(q).$$

Using this notation, we are equipped to find exact formulas for the  $T_g^{(-m)}$ .

Theorem 7.16. Let  $g \in \mathbb{M}$ , with  $\Gamma_g = N|h + e, f, \ldots$ , and let  $\mathcal{S}_{Nh}$  and  $\mathcal{W}_g$  be as above. if m and n are positive integers, then there is a constant c for which

$$T_g^{(-m)}(\tau) = c + \sum_{e \in \mathcal{W}_g} \sum_{\substack{\frac{\alpha}{\gamma} \in \mathcal{S}_{Nh} \\ \left(\frac{\gamma}{(\gamma,h)},\frac{N}{h}\right) = \frac{N}{eh}}} \epsilon_g(L_\rho)^m \mathcal{P}_{\alpha/\gamma}^+(\tau,m,Nh,0).$$

The n-th coefficient of  $T_g^{(-m)}(\tau)$  is given by

$$\sum_{e \in \mathcal{W}_g} \sum_{\substack{\rho = \frac{\alpha}{\gamma} \in \mathcal{S}_{Nh} \\ \left(\frac{\gamma}{(\gamma,h)}, \frac{N}{h}\right) = \frac{N}{eh}}} \epsilon_g(L_\rho)^m 2\pi \left| \frac{-m}{n} \cdot \frac{(h,\gamma)^2}{eh^2} \right|^{\frac{1}{2}} \times \sum_{\substack{c > 0 \\ (c,Nh) = (\gamma,Nh)}} \frac{K_c(0,L,-m,n)}{c} \cdot I_1\left(\frac{4\pi}{c}\sqrt{\left|\frac{-mn \cdot (h,\gamma)^2}{eh^2}\right|}\right)$$

Proof. Every modular function is a harmonic Maass form. Therefore, the idea is to exhibit a linear combination of Maass-Poincaré series with exactly the same principal parts at all cusps as  $T_g^{(-m)}$ . By Lemma 2.2 and Theorem 2.4, this form equals  $T_g^{(-m)}$  up to an additive constant. Lemma 7.12 (c) implies that the coefficients  $c_g(n)$  of  $T_g$  are supported on the arithmetic progression  $n \equiv -1 \pmod{h}$ . The function  $T_g^{(-m)}$  is a polynomial in  $T_g$ , and as such must be the sum of powers of  $T_g$  each of which is congruent to  $m \pmod{h}$ . Therefore, if  $M \in \Gamma_g$ , then  $T^{(-m)}|_0 M = \sigma_g(M)^m T^{(-m)}$ . Given  $L \in SL_2(\mathbb{Z})$ , let U be a matrix as in (7.53) so that  $LU \in \Gamma_g$ . By applying Proposition 7.15, we find

$$T_g^{(-m)}|_0 L = \sigma_g(LU)^m T_g^{(-m)}|_0 U^{-1} = \epsilon_g(L)^m q^{-m\frac{(h,\gamma)^2}{eh^2}} + O(q).$$

Theorem 2.4, along with the observations that  $t_{\rho} = \frac{(h,\gamma)^2}{eh^2}$  and  $\kappa_{\rho} = 0$  for every  $\rho = \frac{\alpha}{\gamma} = L_{\rho}^{-1}\infty$ , implies the first part of the theorem. The formula for the coefficients follows by Theorem 2.3.

### 7.6.3 Exact formulas for $U_g$ up to a theta function

Following a similar process to that in the previous section, we may construct a series  $\hat{U}_g(\tau) = q^{-\frac{23}{24}} + O(q^{\frac{1}{24}})$  with principal parts matching those of  $\eta(\tau)T_g(\tau)$  at all cusps. Then according to (7.49), the difference  $q^{\frac{1}{24}}(U_g - 1) - \hat{U}_g$  is a weight  $\frac{1}{2}$  holomorphic modular form, which by a celebrated result [271] of Serre–Stark, is a finite linear combination of unary theta functions. This will not affect the asymptotics in Theorem 7.10. The functions  $T_g$  and  $\hat{U}_g$  differ primarily in their weight, and in that  $\hat{U}_g$  has a non-trivial multiplier  $\nu_{\eta} : M \to \frac{\eta(M\tau)}{(M:\tau)^{1/2}\eta(\tau)}$ . They also have slightly different orders of poles, which is accounted for by the fact that the multiplier  $\nu_{\eta}$  implies that  $\kappa_{\rho} = t_{\rho}/24$  at every cusp  $\rho$  for the  $\hat{U}_g$ , rather than 0 for the  $T_g$ . The proof of the following theorem is the same as that of Theorem 7.16, mutatis mutandis.

Theorem 7.17. Let  $g \in \mathbb{M}$ , with  $\Gamma_g = N|h+e, f, \ldots$ , and let  $\mathcal{S}_{Nh}$  and  $\mathcal{W}_g$  be as above. If m is a positive integer then

$$\widehat{U}_{g} = \sum_{e \in \mathcal{W}_{g}} \sum_{\substack{\rho = \frac{\alpha}{\gamma} \in \mathcal{S}_{Nh} \\ \left(\frac{\gamma}{(\gamma, h)}, \frac{N}{h}\right) = \frac{N}{eh}}} \epsilon_{g}(L_{\rho}) \mathcal{P}_{L_{\rho}}^{+}(\tau, 1, Nh, 1/2, \nu_{\eta}).$$

For n a non-negative integer, the coefficient of  $q^{n+\frac{1}{24}}$  in  $\widehat{U}_g$  is given by

$$\sum_{e \in \mathcal{W}_{g}} \sum_{\substack{\rho = \frac{\alpha}{\gamma} \in \mathcal{S}_{Nh} \\ \left(\frac{\gamma}{(\gamma,h)}, \frac{N}{h}\right) = \frac{N}{eh}}} \epsilon_{g}(L_{\rho}) \frac{1-i}{\sqrt{2}} 2\pi \left| \frac{-\frac{(h,\gamma)^{2}}{eh^{2}} + \frac{1}{24}}{n + \frac{1}{24}} \right|^{\frac{1}{4}} \times \sum_{\substack{c > 0 \\ (c,Nh) = (\gamma,Nh)}} \frac{K_{c}(\frac{1}{2}, L, \nu_{\eta}, -1, n)}{c} \cdot I_{\frac{1}{2}} \left( \frac{4\pi}{c} \sqrt{\left| -\frac{(h,\gamma)^{2}}{eh^{2}} + \frac{1}{24} \right|} \left| n + \frac{1}{24} \right| \right)}$$

This immediately admits the following corollary.

Corollary 7.18. Given the notation above, there is a weight  $\frac{1}{2}$  linear combination of theta functions  $h_g(\tau)$  for which the coefficient  $q^n$  in  $U_g(\tau) - q^{-\frac{1}{24}}h_g(\tau)$ coincides with the coefficient of  $q^{n+\frac{1}{24}}$  in  $\widehat{U}_g$ , given explicitly in Theorem 7.17.

#### 7.6.4 Proof of Theorem 7.10

*Proof of Theorem 7.10.* Following Harada and Lang [155], we begin by defining the functions

$$T_{\chi_i}^{(-m)}(\tau) := \frac{1}{|\mathbb{M}|} \sum_{g \in \mathbb{M}} \chi_i(g) T_g^{(-m)}(\tau).$$
(7.54)

The orthogonality of characters imply that for g and  $h \in \mathbb{M}$ ,

$$\sum_{i=1}^{194} \overline{\chi_i(g)} \chi_i(h) = \begin{cases} |C_{\mathbb{M}}(g)| & \text{if } g \text{ and } h \text{ are conjugate,} \\ 0 & \text{otherwise.} \end{cases}$$
(7.55)

Here  $|C_{\mathbb{M}}(g)|$  is the order of the centralizer of g in  $\mathbb{M}$ . Since the order of the centralizer times the order of the conjugacy class of an element is the order of the group, (7.55) and (7.54) together imply the inverse relation

$$T_g^{(-m)}(\tau) = \sum_{i=1}^{194} \overline{\chi_i(g)} T_{\chi_i}^{(-m)}(\tau).$$

In particular we have that  $T_e^{(-m)}(\tau) = \sum_{i=1}^{194} \dim(\chi_i) T_{\chi_i}^{(-m)}(\tau)$ , and therefore we can identify the  $\mathbf{m}_i(-m,n)$  as the Fourier coefficients of the  $T_{\chi_i}^{(-m)}(\tau) = \sum_{n=-m}^{\infty} \mathbf{m}_i(-m,n)q^n$ .

Using Theorem 7.16, we obtain exact formulas for the coefficients of  $T_{\chi_i}^{(-m)}(\tau)$ . Let  $g \in \mathbb{M}$  with  $\Gamma_g = N_g ||h_g + e_g, f_g, \ldots$  If m and n are pos-

itive integers, then the nth coefficient is given exactly by

$$\frac{1}{|\mathbb{M}|} \sum_{g \in \mathbb{M}} \chi_i(g) \sum_{e \in \mathcal{W}_g} \sum_{\substack{\frac{\alpha}{\gamma} \in \mathcal{S}_{N_g h_g} \\ \left(\frac{\gamma}{(\gamma, h_g)}, \frac{N_g}{h_g}\right) = \frac{N_g}{eh_g}}}{\sum_{\substack{c > 0 \\ (c, N_g h_g) = (\gamma, N_g h_g)}} \frac{K_c(2 - k, L, \nu, -m, n)}{c} \cdot I_1\left(\frac{4\pi}{c} \sqrt{\left|\frac{-mn \cdot (h_g, \gamma)^2}{eh_g^2}\right|}\right),$$

where  $\mathcal{S}_{N_gh_g}$  and  $\mathcal{W}_g$  are given as above.

Using the well-known asymptotics for the *I*-Bessel function

$$I_k(x) \sim \frac{e^x}{\sqrt{2\pi x}} \left( 1 - \frac{4k^2 - 1}{8x} + \dots \right)$$

we see that the formula for  $\mathbf{m}_i(-m,n)$  is dominated by the c = 1 term which appears only for g = e (so that  $N_e = h_e = 1$ ). This term yields the asymptotic

$$\mathbf{m}_{i}(-m,n) \sim \frac{\chi_{i}(\mathbf{e}) \cdot |m|^{1/4}}{\sqrt{2}n^{3/4}|\mathbb{M}|} \cdot e^{4\pi\sqrt{|mn|}}$$

as in the statement of the theorem.

The asymptotics for  $\mathbf{n}_i(n)$  follows similarly, using the formula

$$U_{\chi_i}(\tau) := \frac{1}{|\mathbb{M}|} \sum_{g \in \mathbb{M}} \chi_i(g) U_g^{(-m)}(\tau).$$

We note that the coefficients of the theta functions  $h_g(\tau)$  in Corollary 7.18 are bounded by constants and so do not affect the asymptotics. This yields

$$\mathbf{n}_{i}(n) \sim \frac{\sqrt{12} \chi_{i}(\mathbf{e})}{|24n+1|^{1/2}|\mathbb{M}|} \cdot e^{\frac{\pi}{6}\sqrt{23|24n+1|}}$$

as in the theorem.

196

#### 7.6.5 Examples of the exact formulas

We conclude with a few examples illustrating the exact formulas for the McKay–Thompson series. These formulas for the coefficients generally converge rapidly. However the rate of convergence is not uniform and often requires many more terms to converge to a given precision.

Example 7.19. We first consider the example g = e. In this case we have  $\Gamma_g = SL_2(\mathbb{Z})$ , which has only the cusp infinity. In this case Theorem 7.16 reduces to the well known expansion

$$T_{\rm e} = J(\tau) - 744 = q^{-1} + \sum_{n \ge 1} \frac{2\pi}{\sqrt{n}} \cdot \sum_{c > 0} \frac{K_c(\infty, -m, n)}{c} \cdot I_1\left(\frac{4\pi\sqrt{n}}{c}\right) q^n.$$

Table 7.1 below contains several approximations made by bounding the size of the c term in the summation.

Table	7.	1

	n = 1	n = 5	n = 10
$c \le 25$	196883.661	$333202640598.254\dots$	$22567393309593598.047\ldots$
$\leq 50$	196883.881	$333202640599.429\ldots$	$22567393309593598.660\ldots$
$\leq 75$	196883.840	333202640599.828	$22567393309593599.369\ldots$
$\leq 100$	196883.958	$333202640599.827\dots$	$22567393309593599.681\ldots$
$\infty$	196884	333202640600	22567393309593600

Example 7.20. The second example we consider is g in the conjugacy class 4B. In this case we have  $\Gamma_g = 4||2 + 2 \supset \Gamma_0(8)$ . The function  $T_g$  has a pole at each of the four cusps of  $\Gamma_0(8)$ :

1. The cusp  $\infty$  has e = 1, width t = 1, and coefficient  $\epsilon(L_{\infty}) = 1$ .

2. The cusp 0 has e = 2, width t = 8, and coefficient  $\epsilon(L_0) = 1$ .

- 3. The cusp 1/2 has e = 2, width t = 2, and coefficient  $\epsilon(L_{1/2}) = i$ .
- 4. The cusp 1/4 has e = 1, width t = 1, and  $\epsilon(L_{1/4}) = -1$ .

Table 7.2 below contains several approximations as in Table 7.1.

Tal	ole	7.	2

	n = 1	n = 5	n = 10
$c \le 25$	51.975	4760.372	0.107
$\leq 50$	52.003	4759.860	$0.117\ldots$
$\leq 75$	52.041	4760.066	0.092
$\leq 100$	51.894	4760.049	0.040
$\infty$	52	4760	0

# Chapter 8

# **Umbral Moonshine**

## 8.1 Proof of Theorem 1.32

We now prove the replicability formula as described in Theorem 1.32.

*Proof.* Fix a Niemeier lattice and its root system X, and let  $M = m^X$  denote its Coxeter number. Each  $H_{g,r}^X(\tau)$  is the holomorphic part of a weight  $\frac{1}{2}$ harmonic Maass form  $\widehat{H}_{g,r}^X(\tau)$ . To simplify the exposition in the following section, we will emphasize the case that the root system X is of pure A-type. If the root system X is of pure A-type, the shadow function  $S_{g,r}^X(\tau)$  is given by  $\widehat{\chi}_{g,r}^{X_A}S_{M,r}(\tau)$  (see §A.3.2), where

$$S_{M,r}(\tau) = \sum_{\substack{n \in \mathbb{Z} \\ m \equiv r \pmod{2M}}} n \ q^{\frac{n^2}{4M}},$$

and  $\hat{\chi}_{g,r}^{X_A} = \chi_g^{X_A}$  or  $\bar{\chi}_g^{X_A}$  depending on the parity of r is the twisted Euler character given in the appropriate table in §A.2.3, a character of  $G^X$ . (If Xis not of pure A-type, then the shadow function  $S_{g,r}^X(\tau)$  is a linear combination of similar functions as described in §A.3.2.)

Given X and g, the symbol  $n_g|h_g$  given in the corresponding table in §A.2.3 defines the modularity for the vector-valued function  $(\widehat{H}_{g,r}^X(\tau))$ . In particular, if the shadow  $(S_{g,r}^X(\tau))$  is nonzero, and if for  $\gamma \in \Gamma_0(n_g)$  we have that

$$(S_{g,r}^X(\tau))|_{3/2}\gamma = \sigma_{g,\gamma}(S_{g,r}^X(\tau))$$

then

$$(\widehat{H}_{g,r}^X(\tau))|_{1/2}\gamma = \overline{\sigma_{g,\gamma}}(\widehat{H}_{g,r}^X(\tau)).$$

Here, for  $\gamma \in \Gamma_0(n_g)$ , we have  $\sigma_{g,\gamma} = \nu_g(\gamma)\sigma_{e,\gamma}$  where  $\nu_g(\gamma)$  is a multiplier which is trivial on  $\Gamma_0(n_g h_g)$ . This identity holds even in the case that the shadow  $S_{q,r}^X$  vanishes.

The vector-valued function  $(H_{g,r}^X(\tau))$  has poles only at the infinite cusp of  $\Gamma_0(n_g)$ , and only at the component  $H_{g,r}^X(\tau)$  where r = 1 if X has pure A-type, or at components where  $r^2 \equiv 1 \pmod{2M}$  otherwise. These poles may only have order  $\frac{1}{4M}$ . This implies that the function  $(\widehat{H}_{g,r}^X(\tau)S_{g,r}^X(\tau))$  has no pole at any cusp, and is therefore a candidate for an application of Theorem 2.5.

The modular transformation of  $S_{M,r}(\tau)$  implies that

$$(\sigma_{e,S})^2 = (\sigma_{e,T})^{4M} = \mathbf{I}_{M-1}$$

where  $S = \begin{pmatrix} 0 & -1 \\ 1 & 0 \end{pmatrix}$ ,  $T = \begin{pmatrix} 1 & 1 \\ 0 & 1 \end{pmatrix}$ , and  $\mathbf{I}_{M-1}$  is the  $(M-1) \times (M-1)$  identity matrix. Therefore  $S_{M,r}^X(\tau)$ , viewed as a scalar-valued modular function, is modular on  $\Gamma(4M)$ , and so  $(\widehat{H}_{g,r}^X(\tau)S_{g,r}^X(\tau))$  is a weight 2 nonholomorphic scalar-valued modular form for the group  $\Gamma(4M) \cap \Gamma_0(n_g)$  with trivial multiplier.

Applying Theorem 2.5, we obtain a function  $F_{g,r}^X(\tau)$ —call it the holomorphic projection of  $\widehat{H_{g,r}^X}(\tau)S_{e,r}^X(\tau)$ —which is a weight 2 quasimodular form on  $\Gamma(4M) \cap \Gamma_0(n_g)$ . In the case that  $S_{g,r}^X(\tau)$  is zero, we substitute  $S_{e,r}^X(\tau)$  in its place to obtain a function  $\widetilde{F}_{g,r}^X(\tau) = H_{g,r}^X(\tau)S_{e,r}^X(\tau)$  which is a weight 2 holomorphic scalar-valued modular form for the group  $\Gamma(4M) \cap \Gamma_0(n_g)$  with multiplier  $\nu_g$  (or alternatively, modular for the group  $\Gamma(4M) \cap \Gamma_0(n_gh_g)$  with trivial multiplier).

The function  $F_{g,r}^X(\tau)$  may be determined explicitly as the sum of Eisenstein series and cusp forms on  $\Gamma(4M)\cap\Gamma_0(n_gh_g)$  using the standard arguments from the theory of holomorphic modular forms (i.e. the "first few" coefficients determine such a form). Therefore, we have the identity

$$F_{g,r}^X(\tau) = H_{g,r}^X(\tau) \cdot S_{g,r}^X(\tau) + D_{g,r}^X(\tau),$$

where the function  $D_{g,r}^X(\tau)$  is the correction term arising in Theorem 2.5. If X has pure A-type, then

$$D_{g,r}^{X}(\tau) = (\hat{\chi}_{g,r}^{X_{A}})^{2} \sum_{N=1}^{\infty} \sum_{\substack{m,n \in \mathbb{Z}_{+} \\ m^{2} - n^{2} = N}} \phi_{r}(m)\phi_{r}(n)(m-n)q^{\frac{N}{4M}},$$
(8.1)

where

$$\phi_r(\ell) = \begin{cases} \pm 1 & \text{if } \ell \equiv \pm r \pmod{2M} \\ 0 & \text{otherwise.} \end{cases}$$
  
Suppose  $H_{g,r}^X(\tau) = \sum_{n=0}^{\infty} A_{g,r}^X(n) q^{n-\frac{D}{4M}}$  where  $0 < D < 4M$  and  $D \equiv r^2 \pmod{4M}$ , and  $F_{g,r}^X(\tau) = \sum_{N=0}^{\infty} B_{g,r}^X(n) q^n$ . Then by Theorem 2.5, we find that

$$B_{g,r}^{X}(N) = \hat{\chi}_{g,r}^{X_{A}} \sum_{\substack{m \in \mathbb{Z} \\ m \equiv r \pmod{2M}}} m \cdot A_{g,r}^{X} \left( N + \frac{D - m^{2}}{4M} \right) + (\hat{\chi}_{g,r}^{X_{A}})^{2} \sum_{\substack{m,n \in \mathbb{Z}^{+} \\ m^{2} - n^{2} = N}} \phi_{r}(m)\phi_{r}(n)(m - n).$$
(8.2)

The function  $F_{g,r}^X(\tau)$  may be found by considering its first few coefficients as determined using the explicit prescriptions given in §A.3.4. It may also be found exactly as a sum of Eisenstein series and cusp forms in the following manner. The Eisenstein component is determined by the constant terms at cusps. Since  $D_{g,r}^X(\tau)$  (and the corresponding correction terms at other cusps) has no constant term, these are the same as the constant terms of  $\widehat{H_{g,r}^X}(\tau)S_{g,r}^X(\tau)$ , which are determined by the poles of  $\widehat{H_{g,r}^X}$ . The cuspidal component can be found by considering the order of vanishing of  $\widehat{H_{g,r}^X}(\tau)S_{g,r}^X(\tau)$  at cusps.

Once the  $B_{g,r}^X(N)$  are known, equation (8.2) provides a recursion relation which may be used to calculate the coefficients of  $H_{g,r}^X(\tau)$ . If the shadows  $S_{g,r}^X(\tau)$  are zero, then we may apply a similar procedure in order to determine  $\widetilde{F}_{g,r}^X(\tau)$ . For example, suppose  $\widetilde{F}_{g,r}^X(\tau) = \sum_{N=0}^{\infty} \widetilde{B}_{g,r}^X(n)q^n$ , and X has pure Atype. Then we find that the coefficients  $\widetilde{B}_{g,r}^X(N)$  satisfy

$$\widetilde{B}_{g,r}^X(N) = \widehat{\chi}_{g,r}^{X_A} \sum_{\substack{m \in \mathbb{Z} \\ m \equiv s \pmod{2M}}} m \cdot A_{g,r}^X \left( N + \frac{D - m^2}{4M} \right)$$
(8.3)

Proceeding in this way we obtain the claimed results.

# 8.2 Proof of Theorem 1.30

Here we prove Theorem 1.30. The idea is as follows. For each Niemeier root system X we begin with the vector-valued mock modular forms  $(H_g^X(\tau))$  for  $g \in G^X$ . We use their q-expansions to solve for the q-series whose coefficients are the alleged multiplicities of the irreducible components of the alleged infinite-dimensional  $G^X$ -module

$$\check{K}^X = \bigoplus_{\substack{r(\text{mod } 2m) \\ D = r^2(\text{mod } 4m)}} \bigoplus_{\substack{D \in \mathbb{Z}, \ D \le 0, \\ D = r^2(\text{mod } 4m)}} \check{K}^X_{r, -D/4m}.$$

These q-series turn out to be mock modular forms. The proof requires that we establish that these mock modular forms have non-negative integer coefficients.

Proof of Theorem 1.30. As in the previous section, we fix a root system X and set  $M := m^X$ , and we emphasize the case when X is of pure A-type.

The umbral moonshine conjecture asserts that

$$H_{g,r}^{X}(\tau) = \sum_{n=0}^{\infty} \sum_{\chi} m_{\chi,r}^{X}(n)\chi(g)q^{n-\frac{r^{2}}{4M}}$$

where the second sum is over the irreducible characters of  $G^X$ , and the  $m_{\chi,r}^X(n)$  are non-negative integers which are the multiplicities of the irreducible  $G^X$ -modules in the graded components of the alleged  $G^X$ -module  $\check{K}^X$ . Moreover, the umbral moonshine conjecture is true if and only if the coefficients of certain weight  $\frac{1}{2}$  mock modular forms are non-negative integers. Indeed, it turns out that the multiplicities  $m_{\chi,r}^X(n)$  are the Fourier coefficients of

$$H^X_{\chi,r}(\tau) := \frac{1}{|G^X|} \sum_g \overline{\chi(g)} H^X_{g,r}(\tau)$$
(8.4)

if and only if the conjecture is true. To see this, we recall the orthogonality of characters. We have that for irreducible characters  $\chi_i$  and  $\chi_j$ ,

$$\frac{1}{|G^X|} \sum_{g \in G^X} \overline{\chi_i(g)} \chi_j(g) = \begin{cases} 1 & \text{if } \chi_i = \chi_j, \\ 0 & \text{otherwise.} \end{cases}$$
(8.5)

We also have the relation for g and  $h \in G^X$ ,

$$\sum_{\chi} \overline{\chi_i(g)} \chi_i(h) = \begin{cases} |C_{G^X}(g)| & \text{if } g \text{ and } h \text{ are conjugate,} \\ 0 & \text{otherwise.} \end{cases}$$
(8.6)

Here  $|C_{G^X}(g)|$  is the order of the centralizer of g in  $G^X$ . Since the order of the centralizer times the order of the conjugacy class of an element is the order of the group, (8.4) and (8.6) together imply the inverse relation

$$H_{g,r}^X(\tau) = \sum_{\chi} \chi(g) H_{\chi,r}^X(\tau).$$

These lead to the key identity

$$H^X_{\chi,r}(\tau) = \sum_{n=0}^{\infty} m^X_{\chi,r}(n) q^{n - \frac{r^2}{4M}}.$$

Therefore, in order to prove the theorem it suffices to prove that the coefficients of the mock modular forms  $H_{\chi,r}^X(\tau)$  are all non-negative integers. For holomorphic modular forms, we may answer questions of this type by making use of Sturm's theorem [281] (see also Theorem 2.58 of [249]). This theorem provides a bound B associated to a space of modular forms such that a modular form  $f(\tau)$  is uniquely identified by it's first B coefficients. This bound reduces many questions about the Fourier coefficients of modular forms to finite calculations. In particular, because of the existence of integral bases, this bound B may be used to show that if the first B coefficients of the form  $f(\tau)$  are integral, then all coefficients of  $f(\tau)$  must be integral.

Sturm's Theorem relies on the finite dimensionality of certain spaces of modular forms, and so it can not be applied directly to spaces of mock modular forms. However, by making use of holomorphic projection we can adapt Sturm's theorem to this setting.

Let  $H^X_{\chi,r}(\tau)$  be defined as above. Recall that the transformation matrix for the vector-valued function  $\widehat{H^X_{g,r}}(\tau)$ ) is  $\overline{\sigma_{g,\gamma}}$ , the conjugate of the transformation matrix for  $(S^X_{e,r}(\tau))$  when  $\gamma \in \Gamma_0(n_g h_g)$ , and  $\sigma_{g,\gamma}$  is the identity for  $\gamma \in \Gamma(4M)$ . Therefore if

$$N_{\chi} := \operatorname{lcm}\{n_g h_g \mid g \in G, \chi(g) \neq 0\},\$$

then the scalar-valued functions  $\widehat{H^X}_{\chi,r}(\tau)$  are modular on  $\Gamma(4M) \cap \Gamma_0(N_\chi^X)$ . Suppose that  $H^X_{\chi,r}(\tau)$  has integral coefficients up to the Sturm bound for  $\Gamma(4M) \cap \Gamma_0(N_\chi^X)$ . Formulas for the shadow functions (cf. §A.3.2) show that the leading coefficient of  $S^X_{e,1}(\tau)$  is 1 and has integral coefficients. This implies that the function

$$A_{\chi,r}(\tau) := H^X_{\chi,r}(\tau) S^X_{e,1}(\tau)$$

also has integral coefficients up to the Sturm bound for  $\Gamma(4M) \cap \Gamma_0(N_{\chi}^X)$  and that every coefficient of  $A_{\chi,r}(\tau)$  is integral if and only if every coefficient of  $H_{\chi,r}^X$  is integral. The shadow of  $H_{\chi,r}^X(\tau)$  is given by

$$S_{\chi,r}^X(\tau) := \frac{1}{|G^X|} \sum_g \overline{\chi(g)} S_{g,r}^X(\tau).$$

If X is pure A-type, then  $S_{g,r}^X(\tau) = \chi_{g,r}^{X_A} S_{M,r}(\tau) = (\chi'(g) + \chi''(g)) S_{M,r}(\tau)$ for some irreducible characters  $\chi'$  and  $\chi''$ , according to §A.2.3 and §A.3.2. Therefore,

$$S_{\chi,r}^X(\tau) = \begin{cases} S_{M,r}(\tau) & \text{if } \chi = \chi' \text{ or } \chi'', \\ 0 & \text{otherwise.} \end{cases}$$

When X is not of pure A-type the shadow is some sum of such functions, but in every case has integer coefficients, and so, applying Theorem 2.5 to  $A_{\chi,r}(\tau)$ , we find that the holomorphic projection of this function has only integer coefficients if and only if  $A_{\chi,r}(\tau)$  has only integer coefficients. But the holomorphic projection is modular on  $\Gamma(4M) \cap \Gamma_0(N_{\chi}^X)$  and has integer coefficients up to the Sturm bound for  $\Gamma(4M) \cap \Gamma_0(N_{\chi}^X)$ . Therefore, in order to check that  $H_{\chi,r}^X(\tau)$  has only integer coefficients, it suffices to check up to the Sturm bound for  $\Gamma(4M) \cap \Gamma_0(N_{\chi})$ . These calculations were carried out using the sage mathematical software [277].

To complete the proof, it suffices to check that the multiplicities  $m_{\chi,r}^X(n)$ are non-negative. The proof of this claim follows easily by modifying step-bystep the argument in Gannon's proof of non negativity in the  $M_{24}$  case [136] (i.e.  $X = A_1^{24}$ ). Here we describe how this is done.

Conjectural expressions for the alleged McKay-Thompson series  $H_{g,r}^X(\tau)$ in terms of Rademacher sums and unary theta functions are given in §A.3.3. These expressions are known to hold in many cases, but in any case, the difference between  $H_{g,r}^X(\tau)$  and the corresponding Rademacher sum (cf. (A.27), (A.28)) is a unary theta function of bounded level, according to the Serre-Stark theorem [271] on modular forms of weight  $\frac{1}{2}$ . The unary theta functions have bounded coefficients, and so the non-negativity depends on the asymptotic growth of the coefficients of the Rademacher sums.

Exact formulas are known for all the coefficients of Rademacher sums because they are defined by averaging the special function  $r_{1/2}^{[\alpha]}(\gamma, \tau)$  (see (A.23)) over cosets of a specific modular group modulo  $\Gamma_{\infty}$ , the subgroup of translations. Therefore, Rademacher sums are standard Maass-Poincaré series, and as a result we have formulas for each of their coefficients as convergent infinite sums of Kloosterman-type sums weighted by values of the  $I_{1/2}$  modified Bessel function. (For example, see [45] for the general theory, and [73] for the specific case that  $X = A_1^{24}$ .) More importantly, this means also that the generating function for the multiplicities  $m_{\chi,r}^X(n)$  is a weight  $\frac{1}{2}$  harmonic Maass form, which in turn means that exact formulas (modulo the unary theta functions) are also available in similar terms. For positive integers n, this then means that (cf. Theorem 1.1 of [45])

$$m_{\chi,r}^{X}(n) = \sum_{\rho} \sum_{m<0} \frac{a_{\rho}^{X}(m)}{n^{\frac{1}{4}}} \sum_{c=1}^{\infty} \frac{K_{\rho}^{X}(m,n,c)}{c} \cdot \mathbb{I}^{X}\left(\frac{4\pi\sqrt{|nm|}}{c}\right), \qquad (8.7)$$

where the sums are over the cusps  $\rho$  of the group  $\Gamma_0(N_g^X)$ , and finitely many explicit negative rational numbers m. The constants  $a_{\rho}^X(m)$  are essentially the coefficients which describe the generating function in terms of Maass-Poincaré series. Here I is a suitable normalization and change of variable for the standard  $I_{1/2}$  modified Bessel-function.

The Kloosterman-type sums  $K^X_{\rho}(m, n, c)$  are well known to be related to Salié-type sums (for example see Proposition 5 of [194]). These Salié-type sums are of the form

$$S^{X}_{\rho}(m,n,c) = \sum_{\substack{x(\text{mod }c)\\x^{2} \equiv -D(m,n)(\text{mod }c)}} \epsilon^{X}_{\rho}(m,n) \cdot e\left(\frac{\beta^{X}x}{c}\right),$$

where  $\epsilon_{\rho}^{X}(m,n)$  is a root of unity, -D(m,n) is a discriminant of a positive definite binary quadratic form, and  $\beta^{X}$  is a nonzero positive rational number.

These Salié sums may then be estimated using the equidistribution of CM points with discriminant -D(m, n). This process was first introduced by Hooley [165], and it was first applied to the coefficients of weight  $\frac{1}{2}$  mock modular forms by Bringmann and Ono [41]. Gannon explains how to make effective the estimates for sums of this shape in §4 of [136], thereby reducing the proof of the  $M_{24}$  case of umbral moonshine to a finite calculation. In particular, in equations (4.6-4.10) of [136] Gannon shows how to bound coefficients of the form (8.7) in terms of the Selberg–Kloosterman zeta function, which is bounded in turn in his proof of Theorem 3 of [136]. We follow Gannon's proof *mutatis mutandis*. We find, for each root system, that the coefficients of each multiplicity generating function are positive beyond the 390th coefficient. Moreover, the coefficients exhibit subexponential growth. A finite computer calculation in sage has verified the non-negativity of the finitely many remaining coefficients.

*Remark.* It turns out that the estimates required for proving nonnegativity are the worst for the  $M_{24}$  case considered by Gannon.
# Appendix

# A.1 Monstrous Groups

The table below contains the symbols  $\Gamma_g = N ||h+e, f, \ldots$ , for each conjugacy class of the monster. Following [84], if h = 1, we omit the '||1' from the symbol. If  $\mathcal{W}_g = \{1\}$ , then we write N ||h, whereas if it contains every exact divisor of N/h, we write N ||h+.

$1\mathrm{A}$	1	12C	12  2+	21D	21 + 21
2A	2+	12D	12  3+	22A	22 +
2B	2	12E	12 + 3	22B	22 + 11
3A	3+	12F	12  2+6	23AB	23 +
3B	3	12G	12  2+2	24A	24  2+
$3\mathrm{C}$	3  3	12H	12 + 12	24B	24 +
4A	4+	12I	12	$24\mathrm{C}$	24 + 8
4B	4  2+	12J	12  6	24D	24  2+3
$4\mathrm{C}$	4	13A	13 +	$24\mathrm{E}$	24  6+
$4\mathrm{D}$	4  2	13B	13	24F	24  4+6
5A	5+	14A	14 +	24G	24  4+2
5B	5	14B	14 + 7	$24\mathrm{H}$	24  2+12
6A	6+	$14\mathrm{C}$	14 + 14	24I	24 + 24
6B	6 + 6	15A	15 +	24 J	24  12
$6\mathrm{C}$	6 + 3	15B	15 + 5	25A	25 +
6D	6 + 2	$15\mathrm{C}$	15 + 15	$26\mathrm{A}$	26 +
6E	6	$15\mathrm{D}$	15  3	26B	26 + 26
6F	6  3	16A	16  2+	27A	27 +
7A	7+	16B	16	27B	27 +
$7\mathrm{B}$	7	$16\mathrm{C}$	16 +	$28 \mathrm{A}$	28  2+
8A	8+	17A	17 +	28B	28 +
8B	8  2+	18A	18 + 2	$28\mathrm{C}$	28 + 7

34A	34 +	$46\mathrm{CD}$	46 +	68A	68  2+
35A	35 +	47AB	47+	69AB	69 +
35B	35 + 35	48A	48  2+	$70 \mathrm{A}$	70+
36A	36+	50A	50+	70B	70 + 10, 14, 35
36B	36 + 4	51A	51+	71AB	71 +
$36\mathrm{C}$	36  2+	52A	52  2+	78A	78 +
36D	36 + 36	52B	52  2+26	$78\mathrm{BC}$	78 + 6, 26, 39
38A	38 +	54A	54 +	84A	84  2+
39A	39+	55A	55 +	84B	84  2+6, 14, 21
39B	39  3+	56A	56 +	84C	84  3+
39CD	39 + 39	56BC	56  4+14	87AB	87+
40A	40  4+	57A	57  3+	88AB	88  2+
40B	40  2+	59AB	59 +	92AB	92 +
40CD	40  2+20	60A	60  2+	93AB	93  3+
41A	41+	60B	60+	94AB	94 +
42A	42 +	$60\mathrm{C}$	60+4, 15, 60	95AB	95 +
42B	42 + 6, 14, 21	60D	60 + 12, 15, 20	104AB	104  4+
42C	42  3+7	60E	60  2+5, 6, 30	105A	105 +
42D	42 + 3, 14, 42	60F	60  6+10	110A	110 +
44AB	44+	62AB	62 +	119AB	119 +
45A	45 +	66A	66+		
46AB	46 + 23	66B	66 + 6, 11, 66		

# A.2 The Umbral Groups

In this section we present the facts about the umbral groups that we have used in establishing the main results of this paper. We recall (from [68]) their construction in terms of Niemeier root systems in A.2.1, and we reproduce their character tables (appearing also in [68]) in A.2.2. Note that we use the abbreviations  $a_n := \sqrt{-n}$  and  $b_n := (-1 + \sqrt{-n})/2$  in the tables of §A.2.2.

The root system description of the umbral groups (cf. §A.2.1) gives rise to certain characters called *twisted Euler characters* which we recall (from [68]) in §A.2.3. The data appearing in §A.2.3 plays an important role in §A.3.2, where we use it to describe the shadows  $S_g^X$  of the umbral McKay-Thompson series  $H_q^X$  explicitly.

#### A.2.1 Construction

As mentioned in the introduction, there are exactly 24 self-dual even positivedefinite lattices of rank 24 up to isomorphism, according to the classification of Niemeier [244] (cf. also [82, 291]). Such a lattice L is determined up to isomorphism by its root system  $L_2 := \{\alpha \in L \mid \langle \alpha, \alpha \rangle = 2\}$ . The unique example without roots is the Leech lattice. We refer to the remaining 23 as the Niemeier lattices, and we call a root system X a Niemeier root system if it occurs as the root system of a Niemeier lattice.

The simple components of Niemeier root systems are root systems of ADE type, and it turns out that the simple components of a Niemeier root system X all have the same Coxeter number. Define  $m^X$  to be the Coxeter number of any simple component of X, and call this the *Coxeter number* of X.

For X a Niemeier root system write  $N^X$  for the corresponding Niemeier lattice. The *umbral group* attached to X is defined by setting

$$G^X := \operatorname{Aut}(N^X)/W^X \tag{A.1}$$

where  $W^X$  is the normal subgroup of  $\operatorname{Aut}(N^X)$  generated by reflections in root vectors.

Observe that  $G^X$  acts as permutations on the simple components of X. In general this action is not faithful, so define  $\overline{G}^X$  to be the quotient of  $G^X$  by its kernel. It turns out that the level of the mock modular form  $H_g^X$  attached to  $g \in G^X$  is given by the order, denoted  $n_g$ , of the image of g in  $\overline{G}^X$ . (Cf. §A.2.3 for the values  $n_g$ .)

The Niemeier root systems and their corresponding umbral groups are described in Table A.1. The root systems are given in terms of their simple components of ADE type. Here  $D_{10}E_7^2$ , for example, means the direct sum of one copy of the  $D_{10}$  root system and two copies of the  $E_7$  root system. The symbol  $\ell$  is called the *lambency* of X, and the Coxeter number  $m^X$  appears as the first summand of  $\ell$ .

In the descriptions of the umbral groups  $G^X$ , and their permutation group quotients  $\overline{G}^X$ , we write  $M_{24}$  and  $M_{12}$  for the sporadic simple groups of Mathieu which act quintuply transitively on 24 and 12 points, respectively. (Cf. e.g. [80].) We write  $GL_n(q)$  for the general linear group of a vector space of dimension n over a field with q elements, and  $SL_n(q)$  is the subgroup of linear transformations with determinant 1, &c. The symbols  $AGL_3(2)$  denote the affine general linear group, obtained by adjoining translations to  $GL_3(2)$ . We write  $Dih_n$  for the dihedral group of order 2n, and  $Sym_n$  denotes the symmetric group on n symbols. We use n as a shorthand for a cyclic group of order n.

We also use the notational convention of writing A.B to denote the middle term in a short exact sequence  $1 \rightarrow A \rightarrow A.B \rightarrow B \rightarrow 1$ . This introduces some ambiguity which is nonetheless easily navigated in practice. For example,  $2.M_{12}$  is the unique (up to isomorphism) double cover of  $M_{12}$  which is not  $2 \times M_{12}$ . The group  $AGL_3(2)$  naturally embeds in  $GL_4(2)$ , which in turn admits a unique (up to isomorphism) double cover  $2.GL_4(2)$  which is not a direct product. The group we denote  $2.AGL_3(2)$  is the preimage of  $AGL_3(2) < GL_4(2)$  in  $2.GL_4(2)$  under the natural projection.

X	$A_1^{24}$	$A_{2}^{12}$	$A_{3}^{8}$	$A_4^6$	$A_{5}^{4}D_{4}$	$A_6^4$	$A_{7}^{2}D_{5}^{2}$
«	2	3	4	5	6	7	8
$G^X$	$M_{24}$	$2.M_{12}$	$2.AGL_{3}(2)$	$GL_{2}(5)/2$	$GL_2(3)$	$SL_2(3)$	$Dih_4$
$\bar{G}^X$	$M_{24}$	$M_{12}$	$AGL_3(2)$	$PGL_2(5)$	$PGL_2(3)$	$PSL_2(3)$	$2^{2}$
X	$A_{8}^{3}$	$A_{9}^{2}D_{6}$	$A_{11}D_7E_6$	$A_{12}^2$	$A_{15}D_9$	$A_{17}E_{7}$	$A_{24}$
$\ll$	9	10	12	13	16	18	25
$G^X$	$Dih_6$	4	2	4	2	2	2
$\bar{G}^X$	$Sym_3$	2	1	2	1	1	1
·							
X	$D_4^6$	$D_6^4$	$D_{8}^{3}$	$D_{10}E_{7}^{2}$	$D_{12}^{2}$	$D_{16}E_{8}$	$D_{24}$
$\ll$	$6{+}3$	$10{+}5$	14 + 7	$18 \! + \! 9$	22 + 11	$30 {+} 15$	46 + 23
$G^X$	$3.Sym_6$	$Sym_4$	$Sym_3$	2	2	1	1
$\bar{G}^X$	$Sym_6$	$Sym_4$	$Sym_3$	2	2	1	1

Table A.1: The Umbral Groups

X	$E_{6}^{4}$	$E_{8}^{3}$
~	12 + 4	$30\!+\!6,\!10,\!15$
$G^X$	$GL_2(3)$	$Sym_3$
$\bar{G}^X$	$GL_2(3)$ $PGL_2(3)$	$Sym_3$

anes	
Unaracter	
A Z Z	

Table A.2: Character table of  $G^X \simeq M_{24}, X = A_1^{24}$ 

[g]	FS	1A	2A	2B	3A	3B 4	4A 4	4B 4(	4C 5A	A 6A	6B	7A	7B	8A 1	10A 1	11A	12A	12B	14A	14B	15A	15B	21A	21B	23A	23B
$[g^2]$		$1\mathrm{A}$	$1\mathrm{A}$	$1\mathrm{A}$	3A	3B 2	2A 2	A	2B 5A	A 3A	3B	7A	7B	4B	5A 1	11A	6A	6B	7A	7B	15A	15B	21A	21B	23A	23B
$[g^3]$		$1\mathrm{A}$	2A	2B	1A	1A 4	4A 4	4B 4(	4C 5A	A 2A	2B	7B	7A 8	8A 1	10A 1	11A	4A	4C	14B	14A	$5\mathrm{A}$	$5\mathrm{A}$	7B	7A	23A	23B
$[g^5]$			2A	2B	3A	$3B_{4}$	4A 4	4B 4(	4C 1A	A 6A	6B	7B	7A 8	8A	2B 1	11A	12A	12B	14B	14A	3A	3A	21B	21A	23B	23A
$[g^7]$			2A	2B	3A	$3B_{4}$	4A 4	4B 4(	4C $5A$	A 6A	6B	1A	1A	8A 1	10A 1	11A	12A	12B	2A	2A	15B	15A	3B	3B	23B	23A
$[g^{11}]$			2A	2B	3A	$3B_{4}$	4A 4	4B 4(	4C 5A	A 6A	6B	7A	7B	8A 1	10A	$1\mathrm{A}$	12A	12B	14A	14B	15B	15A	21A	21B	23B	23A
$[g^{23}]$		$1\mathrm{A}$	2A	2B	3A	$3B_{4}$	4A 4	4B 4(	4C 5A	A 6A	6B	7A	7B	8A 1	10A 1	11A	12A	12B	14A	14B	15A	15B	21A	21B	1A	$1\mathrm{A}$
$\chi_1$	+	1			1	-	Η	1		1 1	-	1	Η	Η		1	1	1	1	Η	1	1	1	1	1	Τ
$\chi_2$	+	23	4	-	Ŋ	- 		3 1	-	3 1	-	7	0	Η	-	1	Γ	-	0	0	0	0	-	Γ	0	0
$\chi_3$	0	45	-3	Ŋ	0	3 1	-3	Η	1	0 0	-1	$p_7$	$\frac{2q}{p_{1}}$	-1	0	1	0	1	$^{Lq-}$	$-\overline{b_7}$	0	0	$b_7$	$\overline{b_7}$	-	-
$\chi_4$	0	45	-3	Ŋ	0	3	-3	Η	1	0 (	-1	$\overline{b_7}$	$p_7$	-1	0	1	0	1	$-\overline{b_7}$	$-p_7$	0	0	$\overline{b_7}$	$b_7$	-1	-
$\chi_5$	0	231	7	-0	-3	- 0	-		ŝ	1	0	0	0	-1	Η	0	-1	0	0	0	$b_{15}$	$\overline{b_{15}}$	0	0	Η	Η
$\chi_6$	0	231	7	-0	-3	- 0	- 		ເ ເ	1	0	0	0	-1	Η	0	-1	0	0	0	$\overline{b_{15}}$	$b_{15}$	0	0	Η	Η
$\chi_7$	+	252	28	12	9	0	4	4	0	2 1	0	0	0	0	2	-1	Η	0	0	0	-1	-	0	0	-	-
$\chi_8$	+	253	13 -	-11	10		-3	μ	-	3 - 2	Τ	1	Η		-	0	0	1	-1	-1	0	0	Τ	Τ	0	0
$\chi_9$	+	483	35	က	0	0	က	က	ີ ຕ	2	0	0	0		-2	-	0	0	0	0	1	Τ	0	0	0	0
$\chi_{10}$	0	- 022	-14	10	Ŋ	2-	$\frac{2}{2}$	-2 -	5	0 1	Η	0	0	0	0	0	-1	Τ	0	0	0	0	0	0	$b_{23}$	$\overline{b_{23}}$
$\chi_{11}$	0	- 022	-14	10	ŋ	2-	$2^{-}$	-2 -	5	0 1	1	0	0	0	0	0	-1	Ļ	0	0	0	0	0	0	$\overline{b_{23}}$	$b_{23}$
$\chi_{12}$	0		-18 -	-10	0	3	<b>9</b>	$2^{-}$	5	0 (	-1	$b_7$	$\overline{b_7}$	0	0	0	0	μ	$p_7$	$\overline{b_7}$	0	0	$p_7$	$\overline{b_7}$	1	1
$\chi_{13}$	0		-18 -	-10	0	3	<b>9</b>	$2^{-}$	5	0 0	-1	$\overline{b_7}$	$b_7$	0	0	0	0	Ц	$\overline{b_7}$	$b_7$	0	0	$\overline{b_7}$	$b_7$	1	Ц
$\chi_{14}$	+	1035	27	35	0	<b>6</b>	r S	<del>,  </del>	3	0 0	<b>2</b>	-	$^{-1}$	μ	0	1	0	0	-	-	0	0	-	-1	0	0
$\chi_{15}$	0		-21	- 5	0	-3	က	3 1	1	0 (	1	$2b_7$	$2\overline{b_7}$	-	0	1	0	1	0	0	0	0	$-p_7$	$-\overline{b_7}$	0	0
$\chi_{16}$	0		-21	- 1 2	0	-3	ŝ	၊ အ	1	0 (	1	$2\overline{b_7}$	$2b_7$	-	0	1	0	1	0	0	0	0	$-\overline{b_7}$	$-p_7$	0	0
$\chi_{17}$	+	1265	49 -	-15	ഹ	$\infty$	2-	 		) 1	0	-2	-2	Η	0	0	-1	0	0	0	0	0	Η	Η	0	0
$\chi_{18}$	+		-21	11	16	7	3 1	-5 -		1 0	-	0	0	-1	Η	0	0	1	0	0	Η	Η	0	0	0	0
$\chi_{19}$	+	2024	$\infty$	24	-	$\infty$	$\infty$	0	0	1 - 1	0	1	Η	0	Ξ	0	-1	0	Η	Η	-1	1	1	Η	$213^{\circ}$	0
$\chi_{20}$	+	2277	21 -	-19	0	- 9	-3	- -	ີ ຕ	3 0	<b>2</b>	2	5	$-\frac{1}{2}$	μ	0	0	0	0	0	0	0	-1	-1	0	0
$\chi_{21}$	+	3312	48	16	0	-6	0	0	$\ddot{}$ 0	3 0	-2	1	μ	0	1	1	0	0	-1	-	0	0	Ļ	Η	0	0
$\chi_{22}$	+	3520	64	0	10	$\infty$	0	0	0	0 - 2	0	-1	$^{-1}$	0	0	0	0	0	1		0	0	-1	-1	-	-
$\chi_{23}$	+	5313	49	6	-15	0	  -	-3 -	 ന	3 1	0	0	0	-1	$^{-1}$	0	1	0	0	0	0	0	0	0	0	0
$\chi_{24}$	+	5544 -	-56	24	9	- 0	$\infty$	0	0 - 1	1 1	0	0	0	0		0		0	0	0	-1	-1	0	0	1	1

[g]	$\mathbf{F}_{\mathbf{S}}$	1A	2A 4A	A 2B	3 2C	C 3A	4 6A	4 3B	3 6B	3 4B	3 4C	3 5A	10A	12A	A 6C	0 6D	) 8A	<b>A</b> 8B	3 8C	C 8D	) 20A	A 20B	B 11A		22A 1	11B 2	22B
$[g^2]$		1A	1A 2A	4 1A	4 1A	4 3A	4 3A	4 3B	3 3B	3 2B	3 2B	3 5A	5A	6B	3 3A	4 3A	4B	3 4B	3 4C	C 4C	10A	A 10A	A 11	B 11	Ы	11A 1	11A
$[g^3]$		1A	2A 4A	$\Lambda 2B$	3 2C	C 1A	4 2A	4 1A	4 2A	4 4B	3 4C	35A	10A	4A	A 2B	3 2C	3 8A	A 8B	3 8C	0 8D	) 20A	A 20B		11A 22	22A 1	11B 2	22B
$[g^5]$		$1\mathrm{A}$	2A 4A	1.2B	3 2C	C 3A	4 6A	A 3B	3 6B	3 4B	3 4C	C 1A	2A	. 12A	4 6C	0.6D	) 8B	3 8A	∆ 8D	) 8C	0 4A	A 4A		11A 25	22A 1	11B 2	22B
$[g^{11}]$		$1\mathrm{A}$	2A 4A	A 2B	3 2C	C 3A	4 6A	4 3B	3 6B	3 4B	3 4C	5A	10A	. 12A	4 6C	0.00	) 8A	A 8B	3 8C	C 8D	) 20B	B 20A		1A 2	2A	1A	2A
$\chi_1$	+																							-			
$\chi_2$	+	11	11 - 1		- 			2	  -	 	1 0	3 1	-		1	0				-		 	1	0	0	0	0
$\chi_3$	+	11	11 - 1		- 	., 		2	  -		33 1	[ ]	-	, _,	1 0	0	<u> </u>	 		- 1		 	1	0	0	0	0
$\chi_4$	0	16	16 4	4 (	Ċ	;- 0	- - -	5		1	) (	) 1	-		1	0	)	) (	) (	0		 	1 $t$	$b_{11}$ <i>l</i>	$b_{11}$	$\overline{b_{11}}$	$\overline{b_{11}}$
$\chi_5$	0	16	$16 \ 4$	4 (	C	;- 0		2	, . ,	1 (	) (	) 1	-		1	0	)	) (	)	0		 	$1 \overline{b}$	$\overline{b_{11}}$ $\overline{b}$	$\overline{b_{11}}$	$b_{11}$	$b_{11}$
$\chi_6$	+	45	45 5	I	် က		_	0	ං ෆ	e S	, , 	0			1	) (				[-]	<b></b> 1	0	0				
$\chi_7$	+	54	54 6	) C	ç	9	0	0	0	; C	2	2 - 1		<u> </u>	) (	)	)	) (	)	0		<del>, _ 1</del>	ا ب	' 			-
$\chi_8$	+	55	55 - 5	2	2	2	,	<del>, _ ,</del>	,			1		_	 	. ,				[-]	<b></b> 1	0	0	0	0	0	0
$\chi_9$	+	55	55 - 5			 	 	<del>, _ ,</del>	,	•• —	3	1		_	[   	. ',   					<b></b> 1	0	0	0	0	0	0
$\chi_{10}$	+	55	55 - 5				,	H		 	1	3	J	_	[	. ,   		, , ,	 	[-]		0	0	0	0	0	0
$\chi_{11}$	+	99	66 6		2			ന	0	;- C	- 7	2	-	<u> </u>	0 - 1	. , 		0 (	)	0		<del>, _ 1</del>		0	0	0	0
$\chi_{12}$	+	66	99 - 1	-	 ന		0	0	ං ෆ	3	1	[-]	-		1	0		 	 	1		 	<del>,</del>	0	0	0	0
$\chi_{13}$	+	120	120 0	;- (	) X	× ×	 ന	0 ന	0	0	0 0	0		_	C	. ,	0	0 (	)	0		0	- 0	' 	 		
$\chi_{14}$	+	144	144 4	1 (	C	0	-	- - 0	ි ෆ	33	0	0 - 1	-		1 (	0	_	0 0	) (	0		  -					
$\chi_{15}$	+	176	176 - 4	1 (	0	<sup>7</sup> — 0	- 4	4 –	 	1	0 0	) 1	-		1 0	0	0	0 (	0	0		<del>,</del> 1	H	0	0	0	0
$\chi_{16}$	0	10	-10 0	;- (	2	5	   		5	2	0 0	0 (	0	_	0		$\mid a_2$	$\frac{1}{2}$	$\frac{1}{2}$ $a_2$	$\frac{1}{2}$		0	- 0	<del>, _ 1</del>	H	Ц Г	1
$\chi_{17}$	0	10	-10 0	;- (	2	5	  -		5	2	0 0	0	_		0		$\lfloor \overline{a_2} \rfloor$	$\frac{1}{2}$ $a_2$	$\frac{1}{a_2}$	$\frac{1}{2}$ $a_2$		0	- 0	<del>, _</del> 1	Н	Ц Г	Η
$\chi_{18}$	+	12	-12 (	~ (	- - -	4	് ന	- ന	0	0	0 0	)	1		0			0 0	0	0	_	0	0	, 1		Η	Γ
$\chi_{19}$	I	32	-32 (	) (	Ċ	·— 0	-4	4	2 - ,	2	0 0	0	-7	0	0 0	) (	)	0 (	0	0		0	- 0			Ξ	
$\chi_{20}$	0	44	-44 0	~ (	- -	4	, . ,	 	5 - - -	5	0	0 - 1		<u> </u>	0 ]			0 0	0	0	) a	$a_5 \overline{a}$	$\overline{a_5}$	0	214 0	0	0
$\chi_{21}$	0	44	-44 0	<sup>7</sup> (	4 -4	4	<del>, _ 1</del>		2 - 7	5	0	0 - 1	Ţ	<u> </u>	0		_	0 0	0	0	_	$\overline{a_5}$ a	$a_5$	0	0	0	0
$\chi_{22}$	0	110 -	-110 0	)   	- 9	9	I	2	2 -,	5	0	0 0	0		0 0	0	$a_2$	$\frac{1}{2}$	$\frac{1}{2}$	$\frac{1}{2}$ $a_2$		0	0	0	0	0	0
$\chi_{23}$	0	110 -	-110 0	)– (	- 9	9	5 - -	2	5   	5	0	0 0	_	_	0	)	$\overline{a_2}$	$\frac{1}{2}$ $a_2$	$a_2$	$\frac{2}{a_2}$		0	0	0	0	0	0
$\chi_{24}$	+	120 -	-120 C	; (	) X				0	0	0 0	0	_	_	0 - 1	. ,		0 0	_	0 0		0	- 0	<del>, _ 1</del>			<del>, _ 1</del>
$\chi_{25}$	0	160 -	-160 0		0	;- 0		7 7	5	5	0	0 0	0		0	) (	0	0 0	_	0 0		0	q - 0	$-b_{11}$ <i>b</i>	$b_{11} -$	$-\overline{b_{11}}$	$\overline{b_{11}}$

Table A.3: Character table of  $G^X \simeq 2.M_{12}, X = A_2^{12}$ 

[g]	$\mathbf{FS}$	1A	2A	2B	4A	4B	2C	3A	6A	6B	$6\mathrm{C}$	8A	$4\mathrm{C}$	7A	14A	7B	14B
$[g^2]$		1A	1A	1A	2A	2B	1A	3A	3A	3A	3A	4A	2C	7A	7A	$7\mathrm{B}$	7B
$[g^3]$		$1\mathrm{A}$	2A	2B	4A	$4\mathrm{B}$	2C	1A	2A	2B	2B	8A	$4\mathrm{C}$	$7\mathrm{B}$	14B	7A	14A
$[g^7]$		1A	2A	2B	4A	$4\mathrm{B}$	2C	3A	6A	6B	$6\mathrm{C}$	8A	$4\mathrm{C}$	1A	2A	1A	2A
$\chi_1$	+	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1
$\chi_2$	0	3	3	3	-1	-1	-1	0	0	0	0	1	1	$b_7$	$b_7$	$\overline{b_7}$	$\overline{b_7}$
$\chi_3$	0	3	3	3	-1	-1	-1	0	0	0	0	1	1	$\overline{b_7}$	$\overline{b_7}$	$b_7$	$b_7$
$\chi_4$	+	6	6	6	2	2	2	0	0	0	0	0	0	-1	-1	-1	-1
$\chi_5$	+	7	7	7	-1	-1	-1	1	1	1	1	-1	-1	0	0	0	0
$\chi_6$	+	8	8	8	0	0	0	-1	-1	-1	-1	0	0	1	1	1	1
$\chi_7$	+	7	7	-1	3	-1	-1	1	1	-1	-1	1	-1	0	0	0	0
$\chi_8$	+	7	7	-1	-1	-1	3	1	1	-1	-1	-1	1	0	0	0	0
$\chi_9$	+	14	14	-2	2	-2	2	-1	-1	1	1	0	0	0	0	0	0
$\chi_{10}$	+	21	21	-3	1	1	-3	0	0	0	0	-1	1	0	0	0	0
$\chi_{11}$	+	21	21	-3	-3	1	1	0	0	0	0	1	-1	0	0	0	0
$\chi_{12}$	+	8	-8	0	0	0	0	2	-2	0	0	0	0	1	-1	1	-1
$\chi_{13}$	0	8	-8	0	0	0	0	-1	1	$a_3$	$\overline{a_3}$	0	0	1	-1	1	-1
$\chi_{14}$	0	8	-8	0	0	0	0	-1	1	$\overline{a_3}$	$a_3$	0	0	1	-1	1	-1
$\chi_{15}$	0	24	-24	0	0	0	0	0	0	0	0	0	0	$\overline{b_7}$	$-\overline{b_7}$	$b_7$	$-b_{7}$
$\chi_{16}$	0	24	-24	0	0	0	0	0	0	0	0	0	0	$b_7$	$-b_{7}$	$\overline{b_7}$	$-\overline{b_7}$

Table A.4: Character table of  $G^X \simeq 2.AGL_3(2), X = A_3^8$ 

Table A.5: Character table of  $G^X \simeq GL_2(5)/2$ ,  $X = A_4^6$ 

[g]	FS	1A	2A	2B	$2\mathrm{C}$	3A	6A	$5\mathrm{A}$	10A	4A	$4\mathrm{B}$	$4\mathrm{C}$	4D	12A	12B
$[g^2]$		1A	1A	1A	1A	3A	3A	5A	5A	2A	2A	$2\mathrm{C}$	$2\mathrm{C}$	6A	6A
$[g^3]$		1A	2A	2B	2C	1A	2A	5A	10A	4B	4A	4D	$4\mathrm{C}$	4B	4A
$[g^5]$		1A	2A	2B	2C	3A	6A	1A	2A	4A	4B	$4\mathrm{C}$	4D	12A	12B
$\chi_1$	+	1	1	1	1	1	1	1	1	1	1	1	1	1	1
$\chi_2$	+	1	1	1	1	1	1	1	1	-1	-1	-1	-1	-1	-1
$\chi_3$	+	4	4	0	0	1	1	-1	-1	2	2	0	0	-1	-1
$\chi_4$	+	4	4	0	0	1	1	-1	-1	-2	-2	0	0	1	1
$\chi_5$	+	5	5	1	1	-1	-1	0	0	1	1	-1	-1	1	1
$\chi_6$	+	5	5	1	1	-1	-1	0	0	-1	-1	1	1	-1	-1
$\chi_7$	+	6	6	-2	-2	0	0	1	1	0	0	0	0	0	0
$\mathcal{V}_{\mathcal{O}}$		1	1	1	1	1	1	1	_1	0.	-01	<i>a</i> .	-a.	0.	-01

Table A.6: Character table of  $G^X \simeq GL_2(3), X \in \{A_5^4D_4, E_6^4\}$ 

[g]	FS	1A	2A	2B	4A	3A	6A	8A	8B
$[g^2]$		1A	$1\mathrm{A}$	$1\mathrm{A}$	2A	3A	3A	4A	4A
$[g^3]$		1A	2A	$2\mathrm{B}$	4A	1A	2A	8A	8B
$\chi_1$	+	1	1	1	1	1	1	1	1
$\chi_2$	+	1	1	-1	1	1	1	-1	-1
$\chi_3$	+	2	2	0	2	-1	-1	0	0
$\chi_4$	+	3	3	-1	-1	0	0	1	1
$\chi_5$	+	3	3	1	-1	0	0	-1	-1
$\chi_6$	0	2	-2	0	0	-1	1	$a_2$	$\overline{a_2}$
$\chi_7$	0	2	-2	0	0	-1	1	$\overline{a_2}$	$a_2$
$\chi_8$	+	4	-4	0	0	1	-1	0	0

_																		
	[g]	$\mathbf{FS}$	1A	3A	2A	6A	3B	$3\mathrm{C}$	4A	12A	5A	15A	15B	2B	$2\mathrm{C}$	4B	6B	$6\mathrm{C}$
	$[g^2]$		1A	3A	1A	3A	3B	$3\mathrm{C}$	2A	6A	5A	15A	15B	1A	1A	2A	3B	3C
	$[g^3]$		1A	1A	2A	2A	1A	1A	4A	4A	5A	5A	5A	2B	2C	4B	2B	2C
_	$[g^5]$		1A	3A	2A	6A	3B	$3\mathrm{C}$	4A	12A	1A	3A	3A	2B	2C	4B	6B	6C
	$\chi_1$	+	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1
	$\chi_2$	+	1	1	1	1	1	1	1	1	1	1	1	-1	-1	-1	-1	-1
	$\chi_3$	+	5	5	1	1	2	-1	-1	-1	0	0	0	3	-1	1	0	-1
	$\chi_4$	+	5	5	1	1	2	-1	-1	-1	0	0	0	-3	1	-1	0	1
	$\chi_5$	+	5	5	1	1	-1	2	-1	-1	0	0	0	-1	3	1	-1	0
	$\chi_6$	+	5	5	1	1	-1	2	-1	-1	0	0	0	1	-3	-1	1	0
	$\chi_7$	+	16	16	0	0	-2	-2	0	0	1	1	1	0	0	0	0	0
	$\chi_8$	+	9	9	1	1	0	0	1	1	-1	-1	-1	3	3	-1	0	0
	$\chi_9$	+	9	9	1	1	0	0	1	1	-1	-1	-1	-3	-3	1	0	0
	$\chi_{10}$	+	10	10	-2	-2	1	1	0	0	0	0	0	2	-2	0	-1	1
	$\chi_{11}$	+	10	10	-2	-2	1	1	0	0	0	0	0	-2	2	0	1	-1
	$\chi_{12}$	0	6	-3	-2	1	0	0	2	-1	1	$b_{15}$	$\overline{b_{15}}$	0	0	0	0	0
	$\chi_{13}$	0	6	-3	-2	1	0	0	2	-1	1	$\overline{b_{15}}$	$b_{15}$	0	0	0	0	0
	$\chi_{14}$	+	12	-6	4	-2	0	0	0	0	2	-1	-1	0	0	0	0	0
	$\chi_{15}$	+	18	-9	2	-1	0	0	2	-1	-2	1	1	0	0	0	0	0
	$\chi_{16}$	+	30	-15	-2	1	0	0	-2	1	0	0	0	0	0	0	0	0
_																		

Table A.7: Character table of  $G^X \simeq 3.Sym_6, X = D_4^6$ 

B 3A 3A 3B	34	4A	2A	1A	FS	[g]
	21				1	$\lfloor 9 \rfloor$
A 2A 1A 2A	31	2A	1A	1A		$[g^2]$
	17	4A	2A	1A		$[g^3]$
1 1 1 1		1	1	1	+	$\chi_1$
$b_3  \overline{b_3}  \overline{b_3}  b_3$	b	1	1	1	0	$\chi_2$
$\overline{b_3}$ $b_3$ $b_3$ $\overline{b_3}$	$\overline{b}$	1	1	1	0	$\chi_3$
0 0 0 0		-1	3	3	+	$\chi_4$
1 1 -1 1	_	0	-2	2	-	$\chi_5$
$\overline{b_3}$ $b_3$ $-b_3$ $\overline{b_3}$	$-\overline{b}$	0	-2	2	0	$\chi_6$
$b_3  \overline{b_3}  -\overline{b_3}  b_3$	-b	0	-2	2	0	$\chi_7$
$ \frac{1}{23}  b_3  b_3  b_3  \overline{b}_3  \overline{b}_3 $	$\overline{b}$ - $-\overline{b}$	$     \begin{array}{c}       1 \\       -1 \\       0 \\       0     \end{array} $	$1 \\ 3 \\ -2 \\ -2$	$\begin{vmatrix} 1\\ 3\\ 2\\ 2 \end{vmatrix}$	0 + - 0	$egin{array}{c} \chi_3 \ \chi_4 \ \chi_5 \ \chi_6 \end{array}$

Table A.8: Character table of  $G^X \simeq SL_2(3), X = A_6^4$ 

Table A.9: Character table of  $G^X \simeq Dih_4$ ,  $X = A_7^2 D_5^2$ 

[g]	$\mathbf{FS}$	1A	2A	2B	$2\mathrm{C}$	4A
$[g^2]$		1A	1A	1A	$1\mathrm{A}$	2A
$\chi_1$	+	1	1	1	1	1
$\chi_2$	+	1	1	-1	-1	1
$\chi_3$	+	1	1	-1	1	-1
$\chi_4$	+	1	1	1	-1	-1
$\chi_5$	+	2	-2	0	0	0

[g]	$\mathbf{FS}$	$1\mathrm{A}$	2A	2B	$2\mathrm{C}$	3A	6A
$[g^2]$		1A	$1\mathrm{A}$	1A	$1\mathrm{A}$	3A	3A
$[g^3]$		$1\mathrm{A}$	2A	2B	$2\mathrm{C}$	1A	2A
$\chi_1$	+	1	1	1	1	1	1
$\chi_2$	+	1	1	-1	-1	1	1
$\chi_3$	+	2	2	0	0	-1	-1
$\chi_4$	+	1	-1	-1	1	1	-1
$\chi_5$	+	1	-1	1	-1	1	-1
$\chi_6$	+	2	-2	0	0	-1	1

Table A.10: Character table of  $G^X \simeq Dih_6$ ,  $X = A_8^3$ 

Table A.11: Character table of  $G^X \simeq 4$ , for  $X \in \{A_9^2 D_6, A_{12}^2\}$ 

[g]	FS	1A	2A	4A	$4\mathrm{B}$
$[g^2]$		1A	1A	2A	2A
$\chi_1$	+	1	1	1	1
$\chi_2$	+	1	1	-1	-1
$\chi_3$	0	1	-1	$a_1$	$\overline{a_1}$
$\chi_4$	ο	1	-1	$\overline{a_1}$	$a_1$

[g]	$\mathbf{FS}$	1A	2A	3A	2B	4A
$[g^2]$		1A	1A	3A	$1\mathrm{A}$	2A
$[g^3]$		1A	2A	1A	2B	4A
$\chi_1$	+	1	1	1	1	1
$\chi_2$	+	1	1	1	-1	-1
$\chi_3$	+	2	2	-1	0	0
$\chi_4$	+	3	-1	0	1	-1
$\chi_5$	+	3	-1	0	-1	1

Table A.12: Character table of  $G^X \simeq PGL_2(3) \simeq Sym_4$ ,  $X = D_6^4$ 

Table A.13: Character table of  $G^X \simeq 2$ , for  $X \in \{A_{11}D_7E_6, A_{15}D_9, A_{17}E_7, A_{24}, D_{10}E_7^2, D_{12}^2\}$ 

[g]	FS	1A	2A
$[g^2]$		1A	$1\mathrm{A}$
$\chi_1$	+	1	1
$\chi_2$	+	1	-1

Table A.14: Character table of  $G^X \simeq Sym_3, X \in \{D_8^3, E_8^3\}$ 

[g]	$\mathbf{FS}$	1A	2A	3A
$[g^2]$		1A	$1\mathrm{A}$	3A
$[g^3]$		1A	2A	$1\mathrm{A}$
$\chi_1$	+	1	1	1
$\chi_2$	+	1	-1	1
$\chi_3$	+	2	0	-1

#### A.2.3 Twisted Euler Characters

In this section we reproduce certain characters—the *twisted Euler characters* which are attached to each group  $G^X$ , via its action on the root system X. (Their construction is described in detail in §2.4 of [68].)

To interpret the tables, write  $X_A$  for the (possibly empty) union of type A components of X, and interpret  $X_D$  and  $X_E$  similarly, so that if  $m = m^X$ Then  $X = A_{m-1}^d$  for some d, and  $X = X_A \cup X_D \cup X_E$ , for example. Then  $g \mapsto \bar{\chi}_g^{X_A}$  denotes the character of the permutation representation attached to the action of  $\bar{G}^X$  on the simple components of  $X_A$ . The characters  $g \mapsto \bar{\chi}_g^{X_D}$ and  $g \mapsto \bar{\chi}_g^{X_E}$  are defined similarly. The characters  $\chi_g^{X_A}, \chi_g^{X_D}, \chi_g^{X_E}$  and  $\tilde{\chi}_g^{X_D}$  incorporate outer automorphisms of simple root systems induced by the action  $G^X$  on X. We refer to §2.4 of [68] for full details of the construction. For the purposes of this work, it suffices to have the explicit descriptions in the tables in this section. The twisted Euler characters presented here will be used to specify the umbral shadow functions in §A.3.2.

The twisted Euler character tables also attach integers  $n_g$  and  $h_g$  to each  $g \in G^X$ . By definition,  $n_g$  is the order of the image of  $g \in G^X$  in  $\overline{G}^X$  (cf. §A.2.1). The integer  $h_g$  may be defined by setting  $h_g := N_g/n_g$  where  $N_g$  is the product of the shortest and longest cycle lengths appearing in the cycle shape attached to g by the action of  $G^X$  on a (suitable) set of simple roots for X.

[g]	1A	2A	2B	3A	3B	4A	4B	$4\mathrm{C}$	5A	6A	6B
$n_g h_g$	1 1	2 1	2 2	3 1	3 3	4 2	4 1	4 4	5 1	6 1	6 6
$\bar{\chi}_g^{X_A}$	24	8	0	6	0	0	4	0	4	2	0
[g]	7AB	8A	10A	11A	12A	12B	14AB	15AB	21AB	23AB	
$n_g   h_g$	7 1	8 1	10 2	11 1	12 2	12 12	14 1	15 1	21 3	23 1	
$\bar{\chi}_g^{X_A}$	3	2	0	2	0	0	1	1	0	1	

Table A.15: Twisted Euler characters at  $\ll = 2$ ,  $X = A_1^{24}$ 

Table A.16: Twisted Euler characters at  $\ll=3, X=A_2^{12}$ 

[g]	1A	2A	4A	2B	2C	3A	6A	3B	6B	4B	$4\mathrm{C}$	5A	10A	12A	60	C 6D	8AB	8CD	20AB	11AB	22AB
$n_g h_g$	1 1	1 4	2 8	2 1	2 2	3 1	3 4	3 33	3 12	4 2	4 1	5 1	5 4	6 24	6	16 2	2 8 4	8 1	10 8	11 1	11 4
$\bar{\chi}_g^{X_A}$	12	12	0	4	4	3	3	0	0	0	4	2	2	0	-	1 1	. 0	2	0	1	1
$\chi_g^{X_A}$	12 -	-12	0	4	-4	3	-3	0	0	0	0	2	-2	0	-	1 - 1	. 0	0	0	1	-1

Table A.17: Twisted Euler characters at  $\ll = 4$ ,  $X = A_3^8$ 

[g]	1A	2A	2B	4A	$4\mathrm{B}$	$2\mathrm{C}$	3A	6A	6BC	8A	$4\mathrm{C}$	7AB	14AB
$n_g   h_g$	1 1	1 2	2 2	2 4	4 4	2 1	3 1	3 2	6 2	4 8	4 1	7 1	7 2
$\bar{\chi}_g^{X_A}$	8	8	0	0	0	4	2	2	0	0	2	1	1
$\chi_g^{X_A}$	8	-8	0	0	0	0	2	-2	0	0	0	1	-1

[g]	1A	2A	2B	$2\mathrm{C}$	3A	6A	5A	10A	4AB	$4\mathrm{CD}$	12AB
$n_g   h_g$	1 1	1 4	2 2	2 1	3 3	3 12	5 1	5 4	2 8	4 1	6 24
$\bar{\chi}_g^{X_A}$	6	6	2	2	0	0	1	1	0	2	0
 $\chi_g^{X_A}$	6	-6	-2	2	0	0	1	-1	0	0	0

Table A.18: Twisted Euler characters at  $\ll = 5$ ,  $X = A_4^6$ 

Table A.19: Twisted Euler characters at  $\ll = 6$ ,  $X = A_5^4 D_4$ 

[g]	1A	2A	2B	4A	3A	6A	8AB
$n_g   h_g$	1 1	1 2	2 1	2 2	3 1	3 2	4 2
$\bar{\chi}_g^{X_A}$	4	4	2	0	1	1	0
$\chi_g^{X_A}$	4	-4	0	0	1	-1	0
$\bar{\chi}_g^{X_D}$	1	1	1	1	1	1	1
$\chi_g^{X_D}$	1	1	-1	1	1	1	-1
$\check{\chi}_g^{X_D}$	2	2	0	2	-1	-1	0

Table A.20: Twisted Euler characters at  $\ll = 6 + 3$ ,  $X = D_4^6$ 

[g]	1A	3A	2A	6A	3B	6C	4A	12A	5A	15AB	2B	2C	4B	6B	6C
$n_g h_g$	1 1	1 3	2 1	2 3	3 1	3 3	4 2	4 6	5 1	5 3	2 1	2 2	4 1	6 1	6 6
$\bar{\chi}_g^{X_D}$	6	6	2	2	3	0	0	0	1	1	4	0	2	1	0
$\chi_g^{X_D}$	6	6	2	2	3	0	0	0	1	1	-4	0	-2	-1	0
$\check{\chi}_g^{X_D}$	12	-6	4	-2	0	0	0	0	2	-1	0	0	0	0	0

[g]	1A	2A	4A	3AB	6AB
$n_g   h_g$	1 1	1 4	2 8	3 1	3 4
$\bar{\chi}_g^{X_A}$	4	4	0	1	1
$\chi_g^{X_A}$	4	-4	0	1	-1

Table A.21: Twisted Euler characters at  $\ll = 7$ ,  $X = A_6^4$ 

Table A.22: Twisted Euler characters at  $\ll = 8$ ,  $X = A_7^2 D_5^2$ 

[g]	1A	2A	2B	$2\mathrm{C}$	4A
$n_g   h_g$	1 1	1 2	2 1	2 1	2 4
$\bar{\chi}_g^{X_A}$	2	2	0	2	0
$\chi_g^{X_A}$	2	-2	0	0	0
$\bar{\chi}_g^{X_D}$	2	2	2	0	0
$\chi_g^{X_D}$	2	-2	0	0	0

Table A.23: Twisted Euler characters at  $\ll = 9$ ,  $X = A_8^3$ 

[g]	1A	2A	2B	$2\mathrm{C}$	3A	6A
$n_g   h_g$	1 1	1 4	2 1	2 2	3 3	3 12
$\bar{\chi}_g^{X_A}$	3	3	1	1	0	0
$\chi_g^{X_A}$	3	-3	1	-1	0	0

[g]	1A	2A	4AB
$n_g h_g$	1 1	1 2	2 2
$\bar{\chi}_g^{X_A}$	2	2	0
$\chi_g^{X_A}$	2	-2	0
$\bar{\chi}_g^{X_D}$	1	1	1
$\chi_g^{X_D}$	1	1	-1

Table A.24: Twisted Euler characters at  $\ll = 10, X = A_9^2 D_6$ 

Table A.25: Twisted Euler characters at  $\ll = 10 + 5$ ,  $X = D_6^4$ 

[g]	1A	2A	3A	2B	4A
$n_g   h_g$	1 1	2 2	3 1	2 1	4 4
$\bar{\chi}_g^{X_D}$	4	0	1	2	0
$\chi_g^{X_D}$	4	0	1	-2	0

Table A.26: Twisted Euler characters at  $\ll = 12$ ,  $X = A_{11}D_7E_6$ 

[g]	1A	2A
$n_g h_g$	1 1	1 2
$\bar{\chi}_g^{X_A}$	1	1
$\chi_g^{X_A}$	1	-1
$\bar{\chi}_g^{X_D}$	1	1
$\chi_g^{X_D}$	1	-1
$\bar{\chi}_g^{X_E}$	1	1
$\chi_g^{X_E}$	1	-1

[g]	1A	2A	2B	4A	3A	6A	8AB
$n_g h_g$	1 1	1 2	2 1	2 4	3 1	3 2	4 8
$\bar{\chi}_g^{X_E}$	4	4	2	0	1	1	0
$\chi_g^{X_E}$	4	-4	0	0	1	-1	0

Table A.27: Twisted Euler characters at  $\ll = 12 + 4$ ,  $X = E_6^4$ 

Table A.28: Twisted Euler characters at  $\ll = 13$ ,  $X = A_{12}^2$ 

[g]	1A	2A	4AB
$n_g h_g$	1 1	1 4	2 8
$\bar{\chi}_g^{X_A}$	2	2	0
$\chi_g^{X_A}$	2	-2	0

[g]	1A	2A	3A
$n_g   h_g$	1 1	2 1	3 3
$\bar{\chi}_g^{X_D}$	3	1	0
$\chi_g^{X_D}$	3	1	0

Table A.29: Twisted Euler characters at  $\ll = 14 + 7$ ,  $X = D_8^3$ 

Table A.30: Twisted Euler characters at  $\ll = 16$ ,  $X = A_{15}D_9$ 

[g]	1A	2A
$n_g   h_g$	1 1	1 2
$\bar{\chi}_g^{X_A}$	1	1
$\chi_g^{X_A}$	1	-1
$\bar{\chi}_g^{X_D}$	1	1
$\chi_g^{X_D}$	1	-1

Table A.31: Twisted Euler characters at  $\ll = 18$ ,  $X = A_{17}E_7$ 

[g]	1A	2A
$n_g   h_g$	1 1	1 2
$\bar{\chi}_g^{X_A}$	1	1
$\chi_g^{X_A}$	1	-1
$\bar{\chi}_g^{X_E}$	1	1

[g]	1A	2A
$n_g   h_g$	1 1	2 1
$\bar{\chi}_g^{X_D}$	1	1
$\chi_g^{X_D}$	1	-1
$\bar{\chi}_g^{X_E}$	2	0

Table A.32: Twisted Euler characters at  $\ll = 18 + 9$ ,  $X = D_{10}E_7^2$ 

Table A.33: Twisted Euler characters at  $\ll = 22 + 11$ ,  $X = D_{12}^2$ 

[g]	1A	2A
$n_g   h_g$	1 1	2 2
$\bar{\chi}_g^{X_D}$	2	0
$\chi_g^{X_D}$	2	0

Table A.34: Twisted Euler characters at  $\ll = 25, X = A_{24}$ 

[g]	1A	2A
$n_g   h_g$	1 1	1 4
$\bar{\chi}_g^{X_A}$	1	1
$\chi_g^{X_A}$	1	-1

Table A.35: Twisted Euler characters at  $\ll = 30 + 6, 10, 15, X = E_8^3$ 

[g]	1A	2A	3A
$n_g   h_g$	1 1	2 1	3 3
$\bar{\chi}_g^{X_E}$	3	1	0

## A.3 The Umbral McKay-Thompson Series

In this section we describe the umbral McKay-Thompson series in complete detail. In particular, we present explicit formulas for all the McKay-Thompson series attached to elements of the umbral groups by umbral moonshine in §A.3.4. Most of these expressions appeared first in [68,75], but some appear for the first time in this work.

In order to facilitate explicit formulations we recall certain standard functions in §A.3.1. We then, using the twisted Euler characters of §A.2.3, explicitly describe the shadow functions of umbral moonshine in §A.3.2. The Rademacher sum construction of the umbral McKay-Thompson series is described in §A.3.3.

### A.3.1 Special Functions

The Dedekind eta function is  $\eta(\tau) := q^{1/24} \prod_{n>0} (1-q^n)$ , where  $q = e^{2\pi i \tau}$ . Write  $\Lambda_M(\tau)$  for the function

$$\Lambda_M(\tau) := Mq \frac{\mathrm{d}}{\mathrm{d}q} \left( \log \frac{\mathbb{H}(M\tau)}{\mathbb{H}(\tau)} \right) = \frac{M(M-1)}{24} + M \sum_{k>0} \sum_{d|k} d\left(q^k - Mq^{Mk}\right),$$

which is a modular form of weight two for  $\Gamma_0(N)$  if M|N.

Define the Jacobi theta function  $\theta_1(\tau, z)$  by setting

$$\theta_1(\tau, z) := iq^{1/8}y^{-1/2} \sum_{n \in \mathbb{Z}} (-1)^n y^n q^{n(n-1)/2}.$$
 (A.2)

According to the Jacobi triple product identity we have

$$\theta_1(\tau, z) = -iq^{1/8}y^{1/2} \prod_{n>0} (1 - y^{-1}q^{n-1})(1 - yq^n)(1 - q^n).$$
(A.3)

The other Jacobi theta functions are

$${}_{2}(\tau, z) := q^{1/8} y^{1/2} \prod_{n>0} (1 + y^{-1} q^{n-1})(1 + yq^{n})(1 - q^{n}),$$
  

$${}_{3}(\tau, z) := \prod_{n>0} (1 + y^{-1} q^{n-1/2})(1 + yq^{n-1/2})(1 - q^{n}),$$
  

$${}_{4}(\tau, z) := \prod_{n>0} (1 - y^{-1} q^{n-1/2})(1 - yq^{n-1/2})(1 - q^{n}).$$
  
(A.4)

Define  $\Psi_{1,1}$  and  $\Psi_{1,-1/2}$  by setting

$$\Psi_{1,1}(\tau, z) := -i \frac{\theta_1(\tau, 2z)\eta(\tau)^3}{\theta_1(\tau, z)^2},$$

$$\Psi_{1,-1/2}(\tau, z) := -i \frac{\eta(\tau)^3}{\theta_1(\tau, z)}.$$
(A.5)

These are meromorphic Jacobi forms of weight one, with indexes 1 and -1/2, respectively. Here, the term meromorphic refers to the presence of simple poles in the functions  $z \mapsto \Psi_{1,*}(\tau, z)$ , for fixed  $\tau \in \mathbb{H}$ , at lattice points  $z \in \mathbb{Z}\tau + \mathbb{Z}$ .

The standard index m theta functions, for m a positive integer, are defined by

$$\theta_{m,r}(\tau, z) := \sum_{k \in \mathbb{Z}} y^{2mk+r} q^{(2mk+r)^2/4m},$$
(A.6)

where  $r \in \mathbb{Z}$ . Evidently,  $\theta_{m,r}$  only depends on  $r \mod 2m$ . Set  $S_{m,r}(\tau) := \frac{1}{2\pi i} \partial_z \theta_{m,r}(\tau, z)|_{z=0}$ , so that

$$S_{m,r}(\tau) = \sum_{k \in \mathbb{Z}} (2mk+r)q^{(2mk+r)^2/4m}.$$
 (A.7)

For a m a positive integer define

$$\mu_{m,0}(\tau,z) = \sum_{k \in \mathbb{Z}} y^{2km} q^{mk^2} \frac{yq^k + 1}{yq^k - 1} = \frac{y+1}{y-1} + O(q).$$
(A.8)

We recover  $\Psi_{1,1}$  upon specializing to m = 1. Observe that

$$\mu_{m,0}(\tau, z+1/2) = \sum_{k \in \mathbb{Z}} y^{2km} q^{mk^2} \frac{yq^k - 1}{yq^k + 1} = \frac{y-1}{y+1} + O(q).$$
(A.9)

Define the even and odd parts of  $\mu_{m,0}$  by setting

$$\mu_{m,0}^k(\tau,z) := \frac{1}{2}(\mu_{m,0}(\tau,z) + (-1)^k \mu_{m,0}(\tau,z+1/2))$$
(A.10)

for  $k \mod 2$ .

For  $m, r \in \mathbb{Z} + \frac{1}{2}$  with m > 0 define the half-integral index theta functions

$$\theta_{m,r}(\tau,z) := \sum_{k \in \mathbb{Z}} e(mk + r/2) y^{2mk+r} q^{(2mk+r)^2/4m},$$
(A.11)

and define also  $S_{m,r}(\tau) := \frac{1}{2\pi i} \partial_z \theta_{m,r}(\tau, z)|_{z=0}$ , so that

$$S_{m,r}(\tau) = \sum_{k \in \mathbb{Z}} e(mk + r/2)(2mk + r)q^{(2mk+r)^2/4m}.$$
 (A.12)

As in the integral index case,  $\theta_{m,r}$  depends only on  $r \mod 2m$ . We recover  $-\theta_1$  upon specializing  $\theta_{m,r}$  to m = r = 1/2.

For  $m \in \mathbb{Z} + 1/2$ , m > 0, define

$$\mu_{m,0}(\tau,z) := i \sum_{k \in \mathbb{Z}} (-1)^k y^{2mk+1/2} q^{mk^2+k/2} \frac{1}{1-yq^k} = \frac{-iy^{1/2}}{y-1} + O(q). \quad (A.13)$$

Given  $\alpha \in \mathbb{Q}$  write  $[\alpha]$  for the operator on q-series (in rational, possibility negative powers of q) that eliminates exponents not contained in  $\mathbb{Z} + \alpha$ , so that if  $f = \sum_{\beta \in \mathbb{Q}} c(\beta) q^{\beta}$  then

$$[\alpha]f := \sum_{n \in \mathbb{Z}} c(n+\alpha)q^{n+\alpha}$$
(A.14)

### A.3.2 Shadows

Let X be a Niemeier root system and let  $m = m^X$  be the Coxeter number of X. For  $g \in G^X$  we define the associated shadow function  $S_g^X = (S_{g,r}^X)$  by setting

$$S_g^X := S_g^{X_A} + S_g^{X_D} + S_g^{X_E}$$
(A.15)

where the  $S_g^{X_A}$ , &c., are defined in the following way, in terms of the twisted Euler characters  $\chi_g^{X_A}$ , &c. given in §A.2.3, and the unary theta series  $S_{m,r}$ (cf. (A.7)).

Note that if  $m = m^X$  then  $S_{g,r}^X = S_{g,r+2m}^X = -S_{g,-r}^X$  for all  $g \in G^X$ , so we need specify the  $S_{g,r}^{X_A}$ , &c., only for 0 < r < m.

If  $X_A = \emptyset$  then  $S_g^{X_A} := 0$ . Otherwise, we define  $S_{g,r}^{X_A}$  for 0 < r < m by setting

$$S_{g,r}^{X_A} := \begin{cases} \chi_g^{X_A} S_{m,r} & \text{if } r = 0 \mod 2, \\ \bar{\chi}_g^{X_A} S_{m,r} & \text{if } r = 1 \mod 2. \end{cases}$$
(A.16)

If  $X_D = \emptyset$  then  $S_g^{X_D} := 0$ . If  $X_D \neq \emptyset$  then *m* is even and  $m \ge 6$ . If m = 6 then set

$$S_{g,r}^{X_D} := \begin{cases} 0 & \text{if } r = 0 \mod 2, \\ \bar{\chi}_g^{X_D} S_{6,r} + \chi_g^{X_D} S_{6,6-r} & \text{if } r = 1,5 \mod 6, \\ \bar{\chi}_g^{X_D} S_{6,r} & \text{if } r = 3 \mod 6. \end{cases}$$
(A.17)

If m > 6 and  $m = 2 \mod 4$  then set

$$S_{g,r}^{X_D} := \begin{cases} 0 & \text{if } r = 0 \mod 2, \\ \bar{\chi}_g^{X_D} S_{m,r} + \chi_g^{X_D} S_{m,m-r} & \text{if } r = 1 \mod 2. \end{cases}$$
(A.18)

If m > 6 and  $m = 0 \mod 4$  then set

$$S_{g,r}^{X_D} := \begin{cases} \chi_g^{X_D} S_{m,m-r} & \text{if } r = 0 \mod 2, \\ \bar{\chi}_g^{X_D} S_{m,r} & \text{if } r = 1 \mod 2. \end{cases}$$
(A.19)

If  $X_E = \emptyset$  then  $S_g^{X_E} := 0$ . Otherwise, *m* is 12 or 18 or 30. In case m = 12 define  $S_{g,r}^{X_E}$  for 0 < r < 12 by setting

$$S_{g,r}^{X_E} = \begin{cases} \bar{\chi}_g^{X_E}(S_{12,1} + S_{12,7}) & \text{if } r \in \{1,7\}, \\ \bar{\chi}_g^{X_E}(S_{12,5} + S_{12,11}) & \text{if } r \in \{5,11\}, \\ \chi_g^{X_E}(S_{12,4} + S_{12,8}) & \text{if } r \in \{4,8\}, \\ 0 & \text{else.} \end{cases}$$
(A.20)

In case m = 18 define  $S_{g,r}^{X_E}$  for 0 < r < 18 by setting

$$S_{g,r}^{X_E} = \begin{cases} \bar{\chi}_g^{X_E}(S_{18,r} + S_{18,18-r}) & \text{if } r \in \{1, 5, 7, 11, 13, 17\}, \\ \bar{\chi}_g^{X_E}S_{18,9} & \text{if } r \in \{3, 15\}, \\ \bar{\chi}_g^{X_E}(S_{18,3} + S_{18,9} + S_{18,15}) & \text{if } r = 9, \\ 0 & \text{else.} \end{cases}$$
(A.21)

In case m = 30 define  $S_{g,r}^{X_E}$  for 0 < r < 30 by setting

$$S_{g,r}^{X_E} = \begin{cases} \bar{\chi}_g^{X_E}(S_{30,1} + S_{30,11} + S_{30,19} + S_{30,29}) & \text{if } r \in \{1, 11, 19, 29\}, \\ \bar{\chi}_g^{X_E}(S_{30,7} + S_{30,13} + S_{30,17} + S_{30,23}) & \text{if } r \in \{7, 13, 17, 23\}, \\ 0 & \text{else.} \end{cases}$$

## A.3.3 Rademacher Sums

Let  $\Gamma_{\infty}$  denote the subgroup of upper-triangular matrices in  $SL_2(\mathbb{Z})$ . Given  $\alpha \in \mathbb{R}$  and  $\gamma \in SL_2(\mathbb{Z})$ , define  $r_{1/2}^{[\alpha]}(\gamma, \tau) := 1$  if  $\gamma \in \Gamma_{\infty}$ . Otherwise, set

$$\mathbf{r}_{1/2}^{[\alpha]}(\gamma,\tau) := e(-\alpha(\gamma\tau - \gamma\infty)) \sum_{k \ge 0} \frac{(2\pi i\alpha(\gamma\tau - \gamma\infty))^{n+1/2}}{\Gamma(n+3/2)}, \qquad (A.23)$$

where  $e(x) := e^{2\pi i x}$ . Let *n* be a positive integer, and suppose that  $\nu$  is a multiplier system for vector-valued modular forms of weight 1/2 on  $\Gamma =$   $\Gamma_0(n)$ . Assume that  $\nu = (\nu_{ij})$  satisfies  $\nu_{11}(T) = e^{\pi i/2m}$ , for some basis  $\{\mathfrak{e}_i\}$ , for some positive integer m, where  $T = \begin{pmatrix} 1 & 1 \\ 0 & 1 \end{pmatrix}$ . To this data, attach the Rademacher sum

$$R_{\Gamma,\nu}(\tau) := \lim_{K \to \infty} \sum_{\gamma \in \Gamma_{\infty} \setminus \Gamma_{K,K^2}} \nu(\gamma) e\left(-\frac{\gamma\tau}{4m}\right) \mathfrak{e}_1 \, \mathbf{j}(\gamma,\tau)^{1/2} \, \mathbf{r}_{1/2}^{[-1/4m]}(\gamma,\tau), \quad (A.24)$$

where  $\Gamma_{K,K^2} := \{ \begin{pmatrix} a & b \\ c & d \end{pmatrix} \in \Gamma \mid 0 \leq c < K, |d| < K^2 \}$ , and  $j(\gamma, \tau) := (c\tau + d)^{-1}$  for  $\gamma = \begin{pmatrix} a & b \\ c & d \end{pmatrix}$ . (See [299] for a more general and detailed discussion of vector-valued Rademacher sums.)

For the special case that  $X = A_8^3$  we require 8-vector-valued functions  $\check{t}_g^{(9)} = (\check{t}_{g,r}^{(9)})$  for  $g \in G^X$  with order 3 or 6. For such g, define  $\check{t}_{g,r}^{(9)}$ , for 0 < r < 9, by setting

$$\check{t}_{3A,r}^{(9)}(\tau) := \begin{cases} 0, & \text{if } r \neq 0 \mod 3, \\ -\theta_{3,3}(\tau,0), & \text{if } r = 3, \\ \theta_{3,0}(\tau,0), & \text{if } r = 6, \end{cases}$$
(A.25)

in the case that g has order 3, and

$$\check{t}_{6A,r}^{(9)}(\tau) := \begin{cases} 0, & \text{if } r \neq 0 \mod 3, \\ -\theta_{3,3}(\tau,0), & \text{if } r = 3, \\ -\theta_{3,0}(\tau,0), & \text{if } r = 6, \end{cases}$$
(A.26)

when o(g) = 6. Here  $\theta_{m,r}(\tau, z)$  is as defined in (A.6).

The following conjecture is formulated in [69].

Conjecture A.1. Let X be a Niemeier root system and let  $g \in G^X$ . If  $X \neq A_8^3$ and  $g \in G^X$ , or if  $X = A_8^3$  and  $g \in G^X$  does not satisfy  $o(g) = 0 \mod 3$ , then we have

$$\check{H}_g^X(\tau) = -2R_{\Gamma_0(n_g),\check{\nu}_g^X}^X.$$
(A.27)

If  $X = A_8^3$  and  $g \in G^X$  satisfies  $o(g) = 0 \mod 3$  then

$$\check{H}_{g,r}^X(\tau) = -2R_{\Gamma_0(n_g),\check{\nu}_g^X}^X(\tau) + \check{t}_g^{(9)}(\tau).$$
(A.28)

Conjecture A.1 is a theorem in the case that  $X = A_1^{24}$ . This is the main result of [73]. A number of other cases of Conjecture A.1 are proved in [69].

#### A.3.4 Explicit Prescriptions

Here we give explicit expressions for all the umbral McKay-Thompson series  $H_g^X$ . Most of these appeared first in [68, 75]. The expressions in §§A.3.4, A.3.4, A.3.4, A.3.4, A.3.4 are taken from [108]. The expressions in §§A.3.4, A.3.4, A.3.4, A.3.4, A.3.4 are taken from [74]. The expressions for  $H_g^X$  with  $X = E_8^3$  appeared first in [107]. The expression for  $H_{2B,1}^{(6+3)}$  in §A.3.4, and the expressions for  $H_{4A,r}^{(12+4)}$  and  $H_{8AB,r}^{(12+4)}$  in §A.3.4, appear here for the first time.

The labels for conjugacy classes in  $G^X$  are as in §A.2.2.

 $\ell = 2, X = A_1^{24}$ 

We have  $G^{(2)} = G^X \simeq M_{24}$  and  $m^X = 2$ . So for  $g \in M_{24}$ , the associated umbral McKay-Thompson series  $H_g^{(2)} = (H_{g,r}^{(2)})$  is a 4-vector-valued function, with components indexed by  $r \in \mathbb{Z}/4\mathbb{Z}$ , satisfying  $H_{g,r}^{(2)} = -H_{g,-r}^{(2)}$ , and in particular,  $H_{g,r}^{(2)} = 0$  for  $r = 0 \mod 2$ . So it suffices to specify the  $H_{g,1}^{(2)}$ explicitly.

Define  $H_g^{(2)} = (H_{g,r}^{(2)})$  for g = e by requiring that

$$-2\Psi_{1,1}(\tau,z)\varphi_1^{(2)}(\tau,z) = -24\mu_{2,0}(\tau,z) + \sum_{r \mod 4} H_{e,r}^{(2)}(\tau)\theta_{2,r}(\tau,z), \quad (A.29)$$

where

$$\varphi_1^{(2)}(\tau, z) := 4 \left( \frac{2(\tau, z)^2}{2(\tau, 0)^2} + \frac{3(\tau, z)^2}{3(\tau, 0)^2} + \frac{4(\tau, z)^2}{4(\tau, 0)^2} \right).$$
(A.30)

More generally, for  $g \in G^{(2)}$  define

$$H_{g,1}^{(2)}(\tau) := \frac{\bar{\chi}_g^{(2)}}{24} H_{e,1}^{(2)}(\tau) - F_g^{(2)}(\tau) \frac{1}{S_{2,1}(\tau)}, \qquad (A.31)$$

where  $\bar{\chi}_g^{(2)}$  and  $F_g^{(2)}$  are as specified in Table A.36. Note that  $\bar{\chi}_g^{(2)} = \bar{\chi}_g^{X_A}$ , the latter appearing in Table A.15. Also,  $S_{2,1}(\tau) = \eta(\tau)^3$ .

The functions  $f_{23,a}$  and  $f_{23,b}$  in Table A.36 are cusp forms of weight two for  $\Gamma_0(23)$ , defined by

$$f_{23,a}(\tau) := \frac{\eta(\tau)^3 \eta(23\tau)^3}{\eta(2\tau)\eta(46\tau)} + 3\mathbb{H}(\tau)^2 \mathbb{H}(23\tau)^2 + 4\eta(\tau)\eta(2\tau)\eta(23\tau)\eta(46\tau) + 4\eta(2\tau)^2\eta(46\tau)^2$$
  
$$f_{23,b}(\tau) := \mathbb{H}(\tau)^2 \mathbb{H}(23\tau)^2.$$
  
(A.32)

Note that the definition of  $F_g^{(2)}$  appearing here for  $g \in 23A \cup 23B$  corrects errors in [72, 73].

$$\ell = 3, X = A_2^{12}$$

We have  $G^{(3)} = G^X \simeq 2.M_{12}$  and  $m^X = 3$ . So for  $g \in 2.M_{12}$ , the associated umbral McKay-Thompson series  $H_g^{(3)} = (H_{g,r}^{(3)})$  is a 6-vector-valued function, with components indexed by  $r \in \mathbb{Z}/6\mathbb{Z}$ , satisfying  $H_{g,r}^{(3)} = -H_{g,-r}^{(3)}$ , and in particular,  $H_{g,r}^{(3)} = 0$  for  $r = 0 \mod 3$ . So it suffices to specify the  $H_{g,1}^{(3)}$  and  $H_{g,2}^{(3)}$  explicitly.

Define  $H_g^{(3)} = (H_{g,r}^{(3)})$  for g = e by requiring that

$$-2\Psi_{1,1}(\tau,z)\varphi_1^{(3)}(\tau,z) = -12\mu_{3,0}(\tau,z) + \sum_{\substack{r \mod 6}} H_{e,r}^{(3)}(\tau)\theta_{3,r}(\tau,z), \quad (A.33)$$

where

$$\varphi_1^{(3)}(\tau,z) := 2\left(\frac{_3(\tau,z)^2}{_3(\tau,0)^2}\frac{_4(\tau,z)^2}{_4(\tau,0)^2} + \frac{_4(\tau,z)^2}{_4(\tau,0)^2}\frac{_2(\tau,z)^2}{_2(\tau,0)^2} + \frac{_2(\tau,z)^2}{_2(\tau,0)^2}\frac{_3(\tau,z)^2}{_3(\tau,0)^2}\right).$$
(A.34)

[g]	$\bar{\chi}_g^{(2)}$	$F_g^{(2)}( au)$
1A	24	0
2A	8	$16\Lambda_2(\tau)$
2B	0	$2\eta( au)^8\eta(2 au)^{-4}$
3A	6	$6\Lambda_3( au)$
3B	0	$2\eta( au)^6\eta(3 au)^{-2}$
4A	0	$2\eta(2\tau)^8\eta(4\tau)^{-4}$
4B	4	$4(-\Lambda_2(\tau) + \Lambda_4(\tau))$
4C	0	$2\eta(\tau)^4\eta(2\tau)^2\eta(4\tau)^{-2}$
5A	4	$2\Lambda_5( au)$
6A	2	$2(-\Lambda_2(\tau) - \Lambda_3(\tau) + \Lambda_6(\tau))$
6B	0	$2\eta(\tau)^2\eta(2\tau)^2\eta(3\tau)^2\eta(6\tau)^{-2}$
7AB	3	$\Lambda_7( au)$
8A	2	$-\Lambda_4( au) + \Lambda_8( au)$
10A	0	$2\eta(\tau)^{3}\eta(2\tau)\eta(5\tau)\eta(10\tau)^{-1}$
11A	2	$2(\Lambda_{11}(\tau) - 11\eta(\tau)^2\eta(11\tau)^2)/5$
12A	0	$\frac{2\eta(\tau)^{3}\eta(4\tau)^{2}\eta(6\tau)^{3}}{\eta(2\tau)\eta(3\tau)\eta(12\tau)^{2}}$
12B	0	$2\eta(\tau)^4\eta(4\tau)\eta(6\tau)\eta(2\tau)^{-1}\eta(12\tau)^{-1}$
14AB	1	$(-\Lambda_2(\tau) - \Lambda_7(\tau) + \Lambda_{14}(\tau))/3$
		$-14\eta(\tau)\eta(2\tau)\eta(7\tau)\eta(14\tau)/3$
15AB	1	$(-\Lambda_3(\tau) - \Lambda_5(\tau) + \Lambda_{15}(\tau))/4$
		$-15\eta(\tau)\eta(3\tau)\eta(5\tau)\eta(15\tau)/4$
21AB	0	$\frac{7\eta(\tau)^3\eta(7\tau)^3}{3\eta(3\tau)\eta(21\tau)} - \frac{\eta(\tau)^6}{3\eta(3\tau)^2}$
23AB	1	$(\Lambda_{23}(\tau) - 23f_{23,a}(\tau) - 69f_{23,b}(\tau))/11$

Table A.36: Character Values and Weight Two Forms for  $\ell=2, X=A_1^{24}$ 

More generally, for  $g \in G^{(3)}$  define

$$H_{g,1}^{(3)}(\tau) := \frac{\bar{\chi}_g^{(3)}}{12} H_{e,1}^{(3)}(\tau) + \frac{1}{2} \left( F_g^{(3)} + F_{zg}^{(3)} \right) \frac{1}{S_{3,1}(\tau)}, \tag{A.35}$$

$$H_{g,2}^{(3)}(\tau) := \frac{\chi_g^{(3)}}{12} H_{e,1}^{(3)}(\tau) + \frac{1}{2} \left( F_g^{(3)} - F_{zg}^{(3)} \right) \frac{1}{S_{3,2}(\tau)}, \tag{A.36}$$

where  $\chi_g^{(3)}$  and  $F_g^{(3)}$  are as specified in Table A.37, and z is the non-trivial central element of  $G^{(3)}$ . The action of  $g \mapsto zg$  on conjugacy classes can be read off Table A.37, for the horizontal lines indicate the sets  $[g] \cup [zg]$ .

Note the eta product identities,  $S_{3,1}(\tau) = \eta(2\tau)^5/\eta(4\tau)^2$ , and  $S_{3,2}(\tau) = 2\eta(\tau)^2\eta(4\tau)^2/\eta(2\tau)$ . Note also that  $\bar{\chi}_g^{(3)} = \bar{\chi}_g^{X_A}$  and  $\chi_g^{(3)} = \chi_g^{X_A}$ , the latter appearing in Table A.16.

The function  $f_{44}$  is the unique new cusp form of weight 2 for  $\Gamma_0(44)$ , normalized so that  $f_{44}(\tau) = q + O(q^3)$  as  $\Im(\tau) \to \infty$ . The coefficients  $c_g(d)$ and  $c'_q(d)$  for  $g \in 10A \cup 22A \cup 22B$  are given by

$$c_{10A}(2) = -5, c_{10A}(4) = -\frac{5}{3}, c_{10A}(5) = -\frac{2}{3}, c_{10A}(10) = 1, c_{10A}(20) = -\frac{1}{3},$$
(A.37)

$$c_{22AB}(2) = -\frac{11}{5}, c_{22AB}(4) = \frac{11}{5}, c_{22AB}(11) = -\frac{2}{15}, c_{22AB}(22) = \frac{1}{5}, c_{22AB}(44) = -\frac{1}{15},$$
(A.38)

$$c'_{22AB}(1) = 1, c'_{22AB}(2) = 4, c'_{22AB}(4) = 8.$$
 (A.39)

 $\ell = 4, X = A_3^8$ 

We have  $m^X = 4$ , so the umbral McKay-Thompson series  $H_g^{(4)} = (H_{g,r}^{(4)})$  associated to  $g \in G^{(4)}$  is an 8-vector-valued function, with components indexed by  $r \in \mathbb{Z}/8\mathbb{Z}$ .

[g]	$\bar{\chi}_{g}^{(3)}$	$\chi_g^{(3)}$	$F_g^{(3)}( au)$
$1\mathrm{A}$	12	12	0
2A	12	-12	0
4A	0	0	$-2\mathbb{H}(\tau)^4\mathbb{H}(2\tau)^2/\mathbb{H}(4\tau)^2$
$2\mathrm{B}$	4	4	$-16_{2}(\tau)$
$2\mathrm{C}$	4	-4	$16_2(\tau) - \frac{16}{3}_4(\tau)$
3A	3	3	$-6_3(\tau)$
6A	3	-3	$-9_2(\tau) - 2_3(\tau) + 3_4(\tau) + 3_6(\tau) - {}_{12}(\tau)$
	0	0	$8_3(\tau) - 2_9(\tau) + 2 \mathbb{H}^6(\tau) / \mathbb{H}^2(3\tau)$
6B	0	0	$-2\eta(\tau)^5\eta(3\tau)/\eta(2\tau)\eta(6\tau)$
4B	0	0	$-2\mathbb{H}(2\tau)^8/\mathbb{H}(4\tau)^4$
4C	4	0	$-8_4(\tau)/3$
5A	2	2	$-2_5(\tau)$
10A	2	-2	$\sum_{d 20} c_{10A}(d)_d(\tau) + \frac{20}{3}\eta(2\tau)^2\eta(10\tau)^2$
12A	0	0	$-2\mathbb{H}(\tau)\mathbb{H}(2\tau)^{5}\mathbb{H}(3\tau)/\mathbb{H}(4\tau)^{2}\mathbb{H}(6\tau)$
6C	1	1	$2(_{2}(\tau) + _{3}(\tau){6}(\tau))$
6D	1	-1	$-5_2(\tau) - 2_3(\tau) + \frac{5}{3}_4(\tau) + 3_6(\tau) - {}_{12}(\tau)$
8AB	0	0	$-2\mathbb{H}(2\tau)^4\mathbb{H}(4\tau)^2/\mathbb{H}(8\tau)^2$
8CD	2	0	$-2_2(\tau) + \frac{5}{34}(\tau) - {}_8(\tau)$
20AB	0	0	$-2\mathbb{H}(2\tau)^7\mathbb{H}(5\tau)/\mathbb{H}(\tau)\mathbb{H}(4\tau)^2\mathbb{H}(10\tau)$
11AB	1	1	$-\frac{2}{5} \frac{1}{11}(\tau) - \frac{33}{5} \eta(\tau)^2 \eta(11\tau)^2$
22AB	1	-1	$\sum_{d 44} c_g(d)_d(\tau) - \frac{11}{5} \sum_{d 4} c'_g(d) \eta(d\tau)^2 \eta(11d\tau)^2 + \frac{22}{3} f_{44}(\tau)$

Table A.37: Character Values and Weight Two Forms for  $\ell = 3, X = A_2^{12}$ 

Define  $H_g^{(4)} = (H_{g,r}^{(4)})$  for  $g \in G^{(4)}, g \notin 4C$ , by requiring that

$$\psi_{g}^{(4)}(\tau,z) = -\chi_{g}^{(4)}\mu_{4,0}^{0}(\tau,z) - \bar{\chi}_{g}^{(4)}\mu_{4,0}^{1}(\tau,z) + \sum_{r \mod 8} H_{g,r}^{(4)}(\tau)\theta_{4,r}(\tau,z),$$
(A.40)

where  $\chi_g^{(4)} := \chi_g^{X_A}$  and  $\bar{\chi}_g^{(4)} := \bar{\chi}_g^{X_A}$  (cf. Table A.17), and the  $\psi_g^{(4)}$  are meromorphic Jacobi forms of weight 1 and index 4 given explicitly in Table A.38.

Table A.38: Character Values and Meromorphic Jacobi Forms for  $\ell=4,$   $X=A_3^8$ 

[g]	$\chi_g^{(4)}$	$\bar{\chi}_g^{(4)}$	$\psi_g^{(4)}( au,z)$
1A	8	8	$2i_1(\tau,2z)^{3}{}_1(\tau,z)^{-4}\eta(\tau)^{3}$
2A	-8	8	$2i_1(\tau,2z)^3{}_2(\tau,z)^{-4}\eta(\tau)^3$
2B	0	0	$-2i_1(\tau,2z)^{3}{}_1(\tau,z)^{-2}{}_2(\tau,z)^{-2}\eta(\tau)^{3}$
4A	0	0	$-2i_1(\tau,2z)_2(\tau,2z)^2{}_2(2\tau,2z)^{-2}\eta(2\tau)^2\eta(\tau)^{-1}$
4B	0	0	$-2i_1(2\tau,2z)_3(2\tau,2z)^2_4(2\tau,2z)\eta(2\tau)^2\eta(\tau)^{-2}\eta(4\tau)^{-2}$
2C	0	4	$2i_1(\tau,2z)_2(\tau,2z)^2{}_1(\tau,z)^{-2}{}_2(\tau,z)^{-2}\eta(\tau)^3$
3A	2	2	$2i_1(3\tau, 6z)_1(\tau, z)^{-1}_1(3\tau, 3z)^{-1}\eta(\tau)^3$
6A	-2	2	$-2i_1(3\tau,6z)_2(\tau,z)^{-1}_2(3\tau,3z)^{-1}\eta(\tau)^3$
6BC	0	0	cf. (A.41)
8A	0	0	$-2i_1(\tau,2z)_2(2\tau,4z)_2(4\tau,4z)^{-1}\eta(\tau)\eta(4\tau)\eta(2\tau)^{-1}$
4C	0	2	$2i_1(\tau,2z)_2(2\tau,4z)_1(2\tau,2z)^{-2}\eta(2\tau)^7\eta(\tau)^{-3}\eta(4\tau)^{-2}$
7AB	1	1	cf. (A.41)
14AB	-1	1	cf. (A.41)

$$\psi_{6BC}^{(4)} := \left( {}_{1}(\tau, z + \frac{1}{3})_{1}(\tau, z + \frac{1}{6}) - {}_{1}(\tau, z - \frac{1}{3})_{1}(\tau, z - \frac{1}{6}) \right) \frac{-i_{1}(3\tau, 6z)}{{}_{1}(3\tau, 3z)_{2}(3\tau, 3z)} \eta(3\tau)$$
  

$$\psi_{7AB}^{(4)} := \left( \prod_{j=1}^{3} {}_{1}(\tau, 2z + \frac{j^{2}}{7})_{1}(\tau, z - \frac{j^{2}}{7}) + \prod_{j=1}^{3} {}_{1}(\tau, 2z - \frac{j^{2}}{7})_{1}(\tau, z + \frac{j^{2}}{7}) \right) \frac{-i_{1}(7\tau, 7z)}{{}_{1}(7\tau, 7z)} \frac{\eta(7\tau)}{\eta(\tau)^{4}}$$
  

$$\psi_{14AB}^{(4)} := \left( \prod_{j=1}^{3} {}_{1}(\tau, 2z + \frac{j^{2}}{7})_{2}(\tau, z - \frac{j^{2}}{7}) + \prod_{j=1}^{3} {}_{1}(\tau, 2z - \frac{j^{2}}{7})_{2}(\tau, z + \frac{j^{2}}{7}) \right) \frac{i_{2}(7\tau, 7z)}{{}_{2}(7\tau, 7z)} \frac{\eta(7\tau)}{\eta(\tau)^{4}}$$
  
(A.41)

For use later on, note that  $\psi_{1A}^{(4)} = -2\Psi_{1,1}\varphi_1^{(4)}$ , where

$$\varphi_1^{(4)}(\tau, z) := \frac{1(\tau, 2z)^2}{1(\tau, z)^2}.$$
(A.42)

 $\ell = 5, X = A_4^6$ 

We have  $m^X = 5$ , so the umbral McKay-Thompson series  $H_g^{(5)} = (H_{g,r}^{(5)})$  associated to  $g \in G^{(5)}$  is a 10-vector-valued function, with components indexed by  $r \in \mathbb{Z}/10\mathbb{Z}$ .

Define  $H_g^{(5)} = (H_{g,r}^{(5)})$  for  $g \in G^{(5)}$ ,  $g \notin 5A \cup 10A$ , by requiring that

$$\psi_{g}^{(5)}(\tau,z) = -\chi_{g}^{(5)}\mu_{5,0}^{0}(\tau,z) - \bar{\chi}_{g}^{(5)}\mu_{5,0}^{1}(\tau,z) + \sum_{r \mod 10} H_{g,r}^{(5)}(\tau)\theta_{5,r}(\tau,z),$$
(A.43)

where  $\chi_g^{(5)} := \chi_g^{X_A}$  and  $\bar{\chi}_g^{(5)} := \bar{\chi}_g^{X_A}$  (cf. Table A.18), and the  $\psi_g^{(5)}$  are meromorphic Jacobi forms of weight 1 and index 5 given explicitly in Table A.39.

[g]	$\chi_g^{(5)}$	$\bar{\chi}_{g}^{(5)}$	$\psi_g^{(5)}( au,z)$
1A	6	6	$2i_1(\tau,2z)_1(\tau,3z)_1(\tau,z)^{-3}\eta(\tau)^3$
2A	-6	6	$-2i_1(\tau,2z)_2(\tau,3z)_2(\tau,z)^{-3}\eta(\tau)^3$
$2\mathrm{B}$	-2	2	$-2i_1(\tau,2z)_1(\tau,3z)_1(\tau,z)^{-1}_2(\tau,z)^{-2}\eta(\tau)^3$
$2\mathrm{C}$	2	2	$2i_1(\tau,2z)_2(\tau,3z)_1(\tau,z)^{-2}_2(\tau,z)^{-1}\eta(\tau)^3$
3A	0	0	$-2i_1(\tau,2z)_1(\tau,3z)_1(3\tau,3z)^{-1}\eta(3\tau)$
6A	0	0	$-2i_1(\tau,2z)_2(\tau,3z)_2(3\tau,3z)^{-1}\eta(3\tau)$
4AB	0	0	cf. (A.44)
4CD	0	2	cf. (A.44)
12AB	0	0	cf. (A.44)

Table A.39: Character Values and Meromorphic Jacobi Forms for  $\ell=5,$   $X=A_4^6$ 

For  $g \in 5A$  use the formulas of §A.3.4 to define

$$H_{5A,r}^{(5)}(\tau) := H_{1A,r}^{(25)}(\tau/5) - H_{1A,10-r}^{(25)}(\tau/5) + H_{1A,10+r}^{(25)}(\tau/5) - H_{1A,20-r}^{(25)}(\tau/5) + H_{1A,20+r}^{(25)}(\tau/5).$$
(A.45)

For  $g \in 10A$  set  $H_{10A,r}^{(5)}(\tau) := -(-1)^r H_{5A,r}^{(5)}(\tau)$ .

For use later on we note that  $\psi_{1A}^{(5)} = -2\Psi_{1,1}\varphi_1^{(5)}$ , where

$$\varphi_1^{(5)}(\tau, z) := \frac{1(\tau, 3z)}{1(\tau, z)}.$$
(A.46)

 $\ell = 6, X = A_5^4 D_4$ 

We have  $m^X = 6$ , so the umbral McKay-Thompson series  $H_g^{(6)} = (H_{g,r}^{(6)})$  associated to  $g \in G^{(6)}$  is a 12-vector-valued function with components indexed by  $r \in \mathbb{Z}/12\mathbb{Z}$ . We have  $H_{g,r}^{(6)} = -H_{g,-r}^{(6)}$ , so it suffices to specify the  $H_{g,r}^{(6)}$  for  $r \in \{1, 2, 3, 4, 5\}$ .

To define  $H_g^{(6)} = (H_{g,r}^{(6)})$  for g = e, first define  $h(\tau) = (h_r(\tau))$  by requiring that

$$-2\Psi_{1,1}(\tau,z)\varphi_1^{(6)}(\tau,z) = -24\mu_{6,0}(\tau,z) + \sum_{r \mod 12} h_r(\tau)\theta_{6,r}(\tau,z), \quad (A.47)$$

where

$$\varphi_1^{(6)}(\tau, z) := \varphi_1^{(2)}(\tau, z)\varphi_1^{(5)}(\tau, z) - \varphi_1^{(3)}(\tau, z)\varphi_1^{(4)}(\tau, z).$$
(A.48)

(Cf. (A.30), (A.34), (A.42), (A.46).) Now define the  $H_{1A,r}^{(6)}$  by setting

$$H_{1A,1}^{(6)}(\tau) := \frac{1}{24} \left( 5h_1(\tau) + h_5(\tau) \right),$$
  

$$H_{1A,2}^{(6)}(\tau) := \frac{1}{6}h_2(\tau),$$
  

$$H_{1A,3}^{(6)}(\tau) := \frac{1}{4}h_3(\tau),$$
  

$$H_{1A,4}^{(6)}(\tau) := \frac{1}{6}h_4(\tau),$$
  

$$H_{1A,5}^{(6)}(\tau) := \frac{1}{24} \left( h_1(\tau) + 5h_5(\tau) \right).$$
  
(A.49)

Define  $H_{2A,r}^{(6)}$  by requiring

$$H_{2A,r}^{(6)}(\tau) := -(-1)^r H_{1A,r}^{(6)}(\tau).$$
(A.50)
For the remaining g, recall (A.14). The  $H_{g,r}^{(6)}$  for  $g \notin 1A \cup 2A$  are defined as follows for r = 2 and r = 4, noting that  $H_{g,4}^{(3)} = H_{g,-2}^{(3)} = -H_{g,2}^{(3)}$ .

$$H_{2B,r}^{(6)}(\tau) := \left[-\frac{r^2}{24}\right] H_{4C,r}^{(3)}(\tau/2)$$

$$H_{4A,r}^{(6)}(\tau) := \left[-\frac{r^2}{24}\right] H_{4B,r}^{(3)}(\tau/2)$$

$$H_{3A,r}^{(6)}(\tau) := \left[-\frac{r^2}{24}\right] H_{6C,r}^{(3)}(\tau/2)$$

$$H_{6A,r}^{(6)}(\tau) := \left[-\frac{r^2}{24}\right] H_{6D,r}^{(3)}(\tau/2)$$

$$H_{8AB,r}^{(6)}(\tau) := \left[-\frac{r^2}{24}\right] H_{8CD,r}^{(3)}(\tau/2)$$

For the  $H_{g,3}^{(6)}$  we define

$$H_{2B,3}^{(6)}(\tau), H_{4A,3}^{(6)}(\tau) := -[-\frac{9}{24}]H_{6A,1}^{(2)}(\tau/3),$$

$$H_{3A,3}^{(6)}(\tau), H_{6A,3}^{(6)}(\tau) := 0,$$

$$H_{8AB,3}^{(6)}(\tau) := -[-\frac{9}{24}]H_{12A,1}^{(2)}(\tau/3).$$
(A.52)

Noting that  $H_{g,5}^{(2)} = H_{g,1}^{(2)}$  and  $H_{g,5}^{(3)} = -H_{g,1}^{(3)}$ , the  $H_{g,1}^{(6)}$  and  $H_{g,5}^{(6)}$  are defined for  $o(g) \neq 0 \mod 3$  by setting

$$H_{2B,r}^{(6)}(\tau) := \left[-\frac{1}{24}\right] \frac{1}{2} \left( H_{6A,r}^{(2)}(\tau/3) + H_{4C,r}^{(3)}(\tau/2) \right)$$

$$H_{4A,r}^{(6)}(\tau) := \left[-\frac{1}{24}\right] \frac{1}{2} \left( H_{6A,r}^{(2)}(\tau/3) + H_{4B,r}^{(3)}(\tau/2) \right)$$

$$H_{8AB,r}^{(6)}(\tau) := \left[-\frac{1}{24}\right] \frac{1}{2} \left( H_{12A,r}^{(2)}(\tau/3) + H_{8CD,r}^{(3)}(\tau/2) \right)$$
(A.53)

It remains to specify the  $H_{g,r}^{(6)}$  when  $g \in 3A \cup 6A$  and r is 1 or 5. These cases are determined by using the formulas of §A.3.4 to set

$$H_{3A,1}^{(6)}(\tau), H_{6A,1}^{(6)}(\tau) := H_{1A,1}^{(18)}(3\tau) - H_{1A,11}^{(18)}(3\tau) + H_{1A,13}^{(18)}(3\tau), H_{3A,5}^{(6)}(\tau), H_{6A,5}^{(6)}(\tau) := H_{1A,5}^{(18)}(3\tau) - H_{1A,7}^{(18)}(3\tau) + H_{1A,17}^{(18)}(3\tau).$$
(A.54)

 $\ell = 6 + 3, X = D_4^6$ 

We have  $m^X = 6$ , so the umbral McKay-Thompson series  $H_g^{(6+3)} = (H_{g,r}^{(6+3)})$ associated to  $g \in G^{(6+3)}$  is a 12-vector-valued function with components indexed by  $r \in \mathbb{Z}/12\mathbb{Z}$ . In addition to the identity  $H_{g,r}^{(6+3)} = -H_{g,-r}^{(6+3)}$ , we have  $H_{g,r}^{(6+3)} = 0$  for  $r = 0 \mod 2$ . Thus it suffices to specify the  $H_{g,r}^{(6+3)}$  for  $r \in \{1,3,5\}$ .

Recall (A.14). For r = 1, define

$$\begin{split} H_{1A,1}^{(6+3)}(\tau), H_{3A,1}^{(6+3)}(\tau) &:= H_{1A,1}^{(6)}(\tau) + H_{1A,5}^{(6)}(\tau), \\ H_{2A,1}^{(6+3)}(\tau), H_{6A,1}^{(6+3)}(\tau) &:= H_{2B,1}^{(6)}(\tau) + H_{2B,5}^{(6)}(\tau), \\ H_{3B,1}^{(6+3)}(\tau) &:= H_{3A,1}^{(6)}(\tau) + H_{3A,5}^{(6)}(\tau), \\ H_{3C,1}^{(6+3)}(\tau) &:= -2\frac{\eta(\tau)^2}{\eta(3\tau)}, \\ H_{4A,1}^{(6+3)}(\tau), H_{12A,1}^{(6+3)}(\tau) &:= H_{8AB,1}^{(6)}(\tau) + H_{8AB,5}^{(6)}(\tau), \\ H_{5A,1}^{(6+3)}(\tau), H_{15A,1}^{(6+3)}(\tau) &:= [-\frac{1}{24}]H_{15AB,1}^{(2)}(\tau/3), \\ H_{2C,1}^{(6+3)}(\tau) &:= H_{4A,1}^{(6)}(\tau) - H_{8AB,5}^{(6)}(\tau), \\ H_{4B,1}^{(6+3)}(\tau) &:= H_{8AB,1}^{(6)}(\tau) - H_{8AB,5}^{(6)}(\tau), \\ H_{6B,1}^{(6+3)}(\tau) &:= H_{6A,1}^{(6)}(\tau) - H_{6A,5}^{(6)}(\tau), \\ H_{6B,1}^{(6+3)}(\tau) &:= H_{6A,1}^{(6)}(\tau) - H_{6A,5}^{(6)}(\tau), \\ H_{6B,1}^{(6+3)}(\tau) &:= -2\frac{\eta(2\tau)\eta(3\tau)}{\eta(6\tau)}. \end{split}$$

Then define  $H_{2B,1}^{(6+3)}$  by setting

$$H_{2B,1}^{(6+3)}(\tau) := 2H_{4B,1}^{(6+3)}(\tau) + 2\frac{\eta(\tau)^3}{\eta(2\tau)^2}.$$
 (A.56)

For r = 3 set

$$\begin{split} H_{1A,3}^{(6+3)}(\tau) &:= 2H_{1A,3}^{(6)}(\tau), \\ H_{3A,3}^{(6+3)}(\tau) &:= -H_{1A,3}^{(6)}(\tau), \\ H_{2A,3}^{(6+3)}(\tau) &:= 2H_{2B,3}^{(6)}(\tau), \\ H_{6A,3}^{(6+3)}(\tau) &:= -H_{2B,3}^{(6)}(\tau), \\ H_{4A,3}^{(6+3)}(\tau) &:= 2H_{8AB,3}^{(6)}(\tau), \\ H_{12A,3}^{(6+3)}(\tau) &:= -H_{8AB,3}^{(6)}(\tau), \\ H_{5A,3}^{(6+3)}(\tau) &:= -2[-\frac{9}{24}]H_{15AB,1}^{(2)}(\tau), \\ H_{15A,3}^{(6+3)}(\tau) &:= [-\frac{9}{24}]H_{15AB,1}^{(2)}(\tau), \end{split}$$

and

$$H_{3B,3}^{(6+3)}(\tau), H_{3C,3}^{(6+3)}(\tau), H_{2B,3}^{(6+3)}(\tau), H_{2C,3}^{(6+3)}(\tau), H_{4B,3}^{(6+3)}(\tau), H_{6B,3}^{(6+3)}(\tau), H_{6C,3}^{(6+3)}(\tau) := 0.$$
(A.58)

For r = 5 define  $H_{g,5}^{(6+3)}(\tau) := H_{g,1}^{(6+3)}(\tau)$  for  $[g] \in \{1A, 3A, 2A, 6A, 3B, 3C, 4A, 12A, 5A, 15AB$ and set  $H_{g,5}^{(6+3)}(\tau) := -H_{g,1}^{(6+3)}(\tau)$  for the remaining cases,  $[g] \in \{2B, 2C, 4B, 6B, 6C\}$ .

 $\ell = 7, X = A_6^4$ 

We have  $m^X = 7$ , so the umbral McKay-Thompson series  $H_g^{(7)} = (H_{g,r}^{(7)})$ associated to  $g \in G^{(7)} = G^X \simeq SL_2(3)$  is a 14-vector-valued function, with components indexed by  $r \in \mathbb{Z}/14\mathbb{Z}$ .

Define  $H_g^{(7)} = (H_{g,r}^{(7)})$  for  $g \in G^{(7)}$  by requiring that

$$\psi_{g}^{(7)}(\tau,z) = -\chi_{g}^{(7)}\mu_{7,0}^{0}(\tau,z) - \bar{\chi}_{g}^{(7)}\mu_{7,0}^{1}(\tau,z) + \sum_{r \mod 14} H_{g,r}^{(7)}(\tau)\theta_{7,r}(\tau,z),$$
(A.59)

where  $\chi_g^{(7)} := \chi_g^{X_A}$  and  $\bar{\chi}_g^{(7)} := \bar{\chi}_g^{X_A}$  (cf. Table A.21), and the  $\psi_g^{(7)}$  are meromorphic Jacobi forms of weight 1 and index 7 given explicitly in Table A.40.

Table A.40: Character Values and Meromorphic Jacobi Forms for  $\ell=7,$   $X=A_6^4$ 

[g]	$\chi_g^{(7)}$	$\bar{\chi}_{g}^{(7)}$	$\psi_g^{(7)}( au,z)$
$1\mathrm{A}$	4	4	$2i_1(\tau,4z)_1(\tau,z)^{-2}\eta(\tau)^3$
2A	-4	4	$-2i_1(\tau,4z)_2(\tau,z)^{-2}\eta(\tau)^3$
4A	0	0	$-2i_1(\tau,4z)_2(2\tau,2z)^{-1}\eta(2\tau)\eta(\tau)$
3A	1	1	cf. (A.60)
6A	-1	1	cf. (A.60)

$$\psi_{3A}^{(7)}(\tau,z) := -i \frac{i(\tau,4z+\frac{1}{3})_1(\tau,z-\frac{1}{3})+i(\tau,4z-\frac{1}{3})_1(\tau,z+\frac{1}{3})}{i(3\tau,3z)} \eta(3\tau)$$

$$\psi_{6A}^{(7)}(\tau,z) := -i \frac{i(\tau,4z+\frac{1}{3})_1(\tau,z-\frac{1}{6})-i(\tau,4z-\frac{1}{3})_1(\tau,z+\frac{1}{6})}{2(3\tau,3z)} \eta(3\tau)$$
(A.60)

For use later on we note that  $\psi_{1A}^{(7)} = -2\Psi_{1,1}\varphi_1^{(7)}$ , where

$$\varphi_1^{(7)}(\tau, z) := \frac{1(\tau, 4z)}{1(\tau, 2z)}.$$
(A.61)

 $\ell = 8, X = A_7^2 D_5^2$ 

We have  $m^X = 8$ , so the umbral McKay-Thompson series  $H_g^{(8)} = (H_{g,r}^{(8)})$  associated to  $g \in G^{(8)}$  is a 16-vector-valued function with components indexed by  $r \in \mathbb{Z}/16\mathbb{Z}$ . We have  $H_{g,r}^{(8)} = -H_{g,-r}^{(8)}$ , so it suffices to specify the  $H_{g,r}^{(8)}$  for  $r \in \{1, 2, 3, 4, 5, 6, 7\}$ .

To define  $H_g^{(8)} = (H_{g,r}^{(8)})$  for g = e, first define  $h(\tau) = (h_r(\tau))$  by requiring that

$$-2\Psi_{1,1}(\tau,z)\left(\varphi_1^{(8)}(\tau,z) + \frac{1}{2}\varphi_2^{(8)}(\tau,z)\right) = -24\mu_{8,0}(\tau,z) + \sum_{r \mod 16} h_r(\tau)\theta_{8,r}(\tau,z),$$
(A.62)

where

$$\varphi_1^{(8)}(\tau, z) := \varphi_1^{(3)}(\tau, z)\varphi_1^{(6)}(\tau, z) - 5\varphi_1^{(4)}(\tau, z)\varphi_1^{(5)}(\tau, z), 
\varphi_2^{(8)}(\tau, z) := \varphi_1^{(4)}(\tau, z)\varphi_1^{(5)}(\tau, z) - \varphi_1^{(8)}(\tau, z).$$
(A.63)

(Cf. (A.34), (A.42), (A.46), (A.48).) Now define the  $H_{1A,r}^{(8)}$  by setting

$$H_{1A,r}^{(8)}(\tau) := \frac{1}{6}h_r(\tau), \qquad (A.64)$$

for  $r \in \{1, 3, 4, 5, 7\}$ , and

$$H_{1A,2}^{(8)}(\tau), H_{1A,6}^{(8)}(\tau) := \frac{1}{12} \left( h_2(\tau) + h_6(\tau) \right).$$
 (A.65)

Define  $H_{2A,r}^{(8)}$  for  $1 \le r \le 7$  by requiring

$$H_{2A,r}^{(8)}(\tau) := -(-1)^r H_{1A,r}^{(8)}(\tau).$$
(A.66)

For the remaining g, recall (A.14). The  $H_{g,r}^{(8)}$  for  $g \in 2B \cup 2C \cup 4A$  are defined as follows for  $r \in \{1, 3, 5, 7\}$ , noting that  $H_{g,7}^{(4)} = H_{g,-1}^{(4)} = -H_{g,1}^{(4)}$ , &c.

$$H_{2BC,r}^{(8)}(\tau) := \left[-\frac{r^2}{32}\right] H_{4C,r}^{(4)}(\tau/2)$$

$$H_{4A,r}^{(8)}(\tau) := \left[-\frac{r^2}{32}\right] H_{4B,r}^{(4)}(\tau/2)$$
(A.67)

The  $H_{2BC,r}^{(8)}$  and  $H_{4A,r}^{(8)}$  vanish for  $r = 0 \mod 2$ .

 $\ell = 9, X = A_8^3$ 

We have  $m^X = 9$ , so for  $g \in G^{(9)}$  the associated umbral McKay-Thompson series  $H_g^{(9)} = (H_{g,r}^{(9)})$  is a 18-vector-valued function, with components indexed by  $r \in \mathbb{Z}/18\mathbb{Z}$ , satisfying  $H_{g,r}^{(9)} = -H_{g,-r}^{(9)}$ , and in particular,  $H_{g,r}^{(9)} = 0$  for  $r = 0 \mod 9$ . So it suffices to specify the  $H_{g,r}^{(9)}$  for  $r \in \{1, 2, 3, 4, 5, 6, 7, 8\}$ .

Define  $H_g^{(9)} = (H_{g,r}^{(9)})$  for g = e by requiring that

$$-\Psi_{1,1}(\tau,z)\varphi_1^{(9)}(\tau,z) = -3\mu_{9,0}(\tau,z) + \sum_{r \mod 18} H_{e,r}^{(9)}(\tau)\theta_{9,r}(\tau,z), \quad (A.68)$$

where

$$\varphi_1^{(9)}(\tau, z) := \varphi_1^{(3)}(\tau, z)\varphi_1^{(7)}(\tau, z) - \varphi_1^{(5)}(\tau, z)^2.$$
(A.69)

(Cf. (A.34), (A.46), (A.61).)

Recall (A.14). The  $H^{(9)}_{2B,r}$  are defined for  $r \in \{1, 2, 4, 5, 7, 8\}$  by setting

$$H_{2B,r}^{(9)}(\tau) := \left[-\frac{r^2}{36}\right] H_{6C,r}^{(3)}(\tau/3), \tag{A.70}$$

where we note that  $H_{g,4}^{(3)} = H_{g,-2}^{(3)} = -H_{g,2}^{(3)}$ , &c. We determine  $H_{2B,3}^{(9)}$  and  $H_{2B,6}^{(9)}$  by using §A.3.4 to set

$$H_{2B,r}^{(9)}(\tau) := H_{1A,r}^{(18)}(2\tau) - H_{1A,18-r}^{(18)}(2\tau)$$
(A.71)

for  $r \in \{3, 6\}$ .

The  $H_{3A,r}^{(9)}$  are defined by the explicit formulas

$$\begin{split} H_{3A,1}^{(9)}(\tau) &:= \left[-\frac{1}{36}\right] f_1^{(9)}(\tau/3), \\ H_{3A,2}^{(9)}(\tau) &:= \left[-\frac{4}{36}\right] f_2^{(9)}(\tau/3), \\ H_{3A,3}^{(9)}(\tau) &:= -\theta_{3,3}(\tau,0), \\ H_{3A,4}^{(9)}(\tau) &:= -\left[-\frac{16}{36}\right] f_2^{(9)}(\tau/3), \\ H_{3A,5}^{(9)}(\tau) &:= -\left[-\frac{25}{36}\right] f_1^{(9)}(\tau/3), \\ H_{3A,6}^{(9)}(\tau) &:= \theta_{3,0}(\tau,0), \\ H_{3A,7}^{(9)}(\tau) &:= \left[-\frac{13}{36}\right] f_1^{(9)}(\tau/3), \\ H_{3A,8}^{(9)}(\tau) &:= \left[-\frac{28}{36}\right] f_2^{(9)}(\tau/3), \\ \end{split}$$
(A.72)

where

$$f_{1}^{(9)}(\tau) := -2 \frac{\mathbb{H}(\tau)\mathbb{H}(12\tau)\mathbb{H}(18\tau)^{2}}{\mathbb{H}(6\tau)\mathbb{H}(9\tau)\mathbb{H}(36\tau)},$$

$$f_{2}^{(9)}(\tau) := \frac{\mathbb{H}(2\tau)^{6}\mathbb{H}(12\tau)\mathbb{H}(18\tau)^{2}}{\mathbb{H}(\tau)\mathbb{H}(4\tau)^{4}\mathbb{H}(6\tau)\mathbb{H}(9\tau)\mathbb{H}(36\tau)} - \frac{\mathbb{H}(\tau)\mathbb{H}(2\tau)\mathbb{H}(3\tau)^{2}}{\mathbb{H}(4\tau)^{2}\mathbb{H}(9\tau)}.$$
(A.73)

Finally, the  $H_{g,r}^{(9)}$  are determined for  $g \in 2A \cup 2C \cup 6A$  by setting

$$H_{2A,r}^{(9)}(\tau) := (-1)^{r+1} H_{1A,r}^{(9)}(\tau),$$
  

$$H_{2C,r}^{(9)}(\tau) := (-1)^{r+1} H_{2B,r}^{(9)}(\tau),$$
  

$$H_{6A,r}^{(9)}(\tau) := (-1)^{r+1} H_{3A,r}^{(9)}(\tau).$$
  
(A.74)

 $\ell = 10, X = A_9^2 D_6$ 

We have  $m^X = 10$ , so the umbral McKay-Thompson series  $H_g^{(10)} = (H_{g,r}^{(10)})$ associated to  $g \in G^{(10)}$  is a 20-vector-valued function with components indexed by  $r \in \mathbb{Z}/20\mathbb{Z}$ . We have  $H_{g,r}^{(10)} = -H_{g,-r}^{(10)}$ , so it suffices to specify the  $H_{g,r}^{(10)}$  for  $1 \leq r \leq 9$ .

To define  $H_g^{(10)} = (H_{g,r}^{(10)})$  for g = e, first define  $h(\tau) = (h_r(\tau))$  by requiring that

$$-6\Psi_{1,1}(\tau,z)\varphi_1^{(10)}(\tau,z) = -24\mu_{10,0}(\tau,z) + \sum_{r \mod 20} h_r(\tau)\theta_{10,r}(\tau,z), \quad (A.75)$$

where

$$\varphi_1^{(10)}(\tau, z) := 5\varphi_1^{(4)}(\tau, z)\varphi_1^{(7)}(\tau, z) - \varphi_1^{(5)}(\tau, z)\varphi_1^{(6)}(\tau, z).$$
(A.76)

(Cf. (A.42), (A.46), (A.48), (A.61).) Now define the  $H_{1A,r}^{(10)}$  for r odd by setting

$$H_{1A,1}^{(10)}(\tau) := \frac{1}{24} \left( 3h_1(\tau) + h_9(\tau) \right),$$
  

$$H_{1A,3}^{(10)}(\tau) := \frac{1}{24} \left( 3h_3(\tau) + h_7(\tau) \right),$$
  

$$H_{1A,5}^{(10)}(\tau) := \frac{1}{6} h_5(\tau),$$
  

$$H_{1A,3}^{(10)}(\tau) := \frac{1}{24} \left( h_3(\tau) + 3h_7(\tau) \right),$$
  

$$H_{1A,9}^{(10)}(\tau) := \frac{1}{24} \left( h_1(\tau) + 3h_9(\tau) \right).$$
  
(A.77)

For  $r = 0 \mod 2$  set

$$H_{1A,r}^{(10)}(\tau) := \frac{1}{12} h_r(\tau), \qquad (A.78)$$

and define  $H_{2A,r}^{(10)}$  for  $1 \le r \le 9$  by requiring

$$H_{2A,r}^{(10)}(\tau) := -(-1)^r H_{1A,r}^{(10)}(\tau).$$
(A.79)

It remains to specify  $H_{g,r}^{(10)}$  for  $g \in 4A \cup 4B$ . For  $r = 0 \mod 2$  set

$$H_{4AB,r}^{(10)}(\tau) := 0. \tag{A.80}$$

For r odd, recall (A.14), and define

$$H_{4A,r}^{(10)}(\tau) := \left[-\frac{r^2}{40}\right] \frac{1}{2} \left( H_{10A,r}^{(2)}(\tau/5) + H_{4CD,r}^{(5)}(\tau/2) \right).$$
(A.81)

 $\ell = 10 + 5, X = D_6^4$ 

We have  $m^X = 10$ , so the umbral McKay-Thompson series  $H_g^{(10+5)} = (H_{g,r}^{(10+5)})$ associated to  $g \in G^{(10+5)}$  is a 20-vector-valued function with components indexed by  $r \in \mathbb{Z}/20\mathbb{Z}$ . We have  $H_{g,r}^{(10+5)} = 0$  for  $r = 0 \mod 2$ , so it suffices to specify the  $H_{g,r}^{(10+5)}$  for r odd. Observing that  $H_{g,r}^{(10+5)} = -H_{g,-r}^{(10+5)}$  we may determine  $H_g^{(10+5)}$  by requiring that

$$\psi_{g}^{(5/2)}(\tau,z) = -2\chi_{g}^{(5/2)}i\mu_{5/2,0}(\tau,z) + \sum_{\substack{r \in \mathbb{Z}+1/2\\r \mod 5}} e(-r/2)H_{g,2r}^{(10+5)}(\tau)\theta_{5/2,r}(\tau,z),$$
(A.82)

where  $\chi_g^{(5/2)} := \bar{\chi}_g^{X_D}$  as in Table A.25, and the  $\psi_g^{(5/2)}$  are the meromorphic Jacobi forms of weight 1 and index 5/2 defined as follows.

 $\ell = 12, X = A_{11}D_7E_6$ 

We have  $m^X = 12$ , so the umbral McKay-Thompson series  $H_g^{(12)} = (H_{g,r}^{(12)})$ associated to  $g \in G^{(12)} \simeq \mathbb{Z}/2\mathbb{Z}$  is a 24-vector-valued function with components indexed by  $r \in \mathbb{Z}/24\mathbb{Z}$ . We have  $H_{g,r}^{(12)} = -H_{g,-r}^{(12)}$ , so it suffices to specify the  $H_{g,r}^{(12)}$  for  $1 \leq r \leq 11$ . Table A.41: Character Values and Meromorphic Jacobi Forms for  $\ell=10+5,$   $X=D_6^4$ 

[g]	$\bar{\chi}_g^{(5/2)}$	$\psi_g^{(5/2)}( au,z)$
$1\mathrm{A}$	4	$2i\theta_1(\tau,2z)^2\theta_1(\tau,z)^{-3}\eta(\tau)^3$
2A	0	$-2i\theta_1(\tau,2z)^2\theta_1(\tau,z)^{-1}\theta_2(\tau,z)^{-2}\eta(\tau)^3$
3A	1	$2i\theta_1(3\tau,6z)\theta_1(\tau,2z)^{-1}\theta_1(3\tau,3z)^{-1}\eta(\tau)^3$
2B	2	$2i\theta_1(\tau,2z)\theta_2(\tau,2z)\theta_1(\tau,z)^{-2}\theta_2(\tau,z)^{-1}\eta(\tau)^3$
4A	0	$-2i\theta_1(\tau,2z)\theta_2(\tau,2z)\theta_2(2\tau,2z)^{-1}\eta(\tau)\eta(2\tau)$

To define  $H_e^{(12)} = (H_{e,r}^{(12)})$ , first define  $h(\tau) = (h_r(\tau))$  by requiring that  $-2\Psi_{1,1}(\tau,z) \left(\varphi_1^{(12)}(\tau,z) + \varphi_2^{(12)}(\tau,z)\right) = -24\mu_{12,0}(\tau,z) + \sum_{r \mod 24} h_r(\tau)\theta_{12,r}(\tau,z),$ (A.83)

where  

$$\varphi_1^{(12)}(\tau, z) := 3\varphi_1^{(3)}(\tau, z)\varphi_1^{(10)}(\tau, z) - 8\varphi_1^{(4)}(\tau, z)\varphi_1^{(9)}(\tau, z) + \varphi_1^{(5)}(\tau, z)\varphi_1^{(8)}(\tau, z), 
\varphi_2^{(12)}(\tau, z) := 4\varphi_1^{(4)}(\tau, z)\varphi_1^{(9)}(\tau, z) - \varphi_1^{(5)}(\tau, z)\varphi_1^{(8)}(\tau, z) - \varphi_1^{(12)}(\tau, z).$$
(A.84)

(Cf. (A.34), (A.42), (A.46), (A.61), (A.63), (A.69), (A.76).) Now define the  $H_{1A,r}^{(12)}$  for  $r \neq 0 \mod 3$  by setting

$$H_{1A,1}^{(12)}(\tau) := \frac{1}{24} \left(3h_1(\tau) + h_7(\tau)\right),$$

$$H_{1A,2}^{(12)}(\tau), H_{1A,10}^{(12)}(\tau) := \frac{1}{24} \left(h_2(\tau) + h_{10}(\tau)\right),$$

$$H_{1A,4}^{(12)}(\tau), H_{1A,8}^{(12)}(\tau) := \frac{1}{12} \left(h_4(\tau) + h_8(\tau)\right),$$

$$H_{1A,5}^{(12)}(\tau) := \frac{1}{24} \left(3h_5(\tau) + h_{11}(\tau)\right),$$

$$H_{1A,7}^{(12)}(\tau) := \frac{1}{24} \left(h_1(\tau) + 3h_7(\tau)\right),$$

$$H_{1A,11}^{(12)}(\tau) := \frac{1}{24} \left(h_5(\tau) + 3h_{11}(\tau)\right).$$
(A.85)

For  $r = 0 \mod 3$  set

$$H_{1A,r}^{(12)}(\tau) := \frac{1}{12} h_r(\tau), \qquad (A.86)$$

and define  $H_{2A,r}^{(12)}$  by requiring

$$H_{2A,r}^{(12)}(\tau) := -(-1)^r H_{1A,r}^{(12)}(\tau).$$
(A.87)

 $\ell = 12 + 4, X = E_6^4$ 

We have  $m^X = 12$ , so the umbral McKay-Thompson series  $H_g^{(12+4)} = (H_{g,r}^{(12+4)})$ associated to  $g \in G^{(12+4)}$  is a 24-vector-valued function with components indexed by  $r \in \mathbb{Z}/24\mathbb{Z}$ . In addition to the identity  $H_{g,r}^{(12+4)} = -H_{g,-r}^{(12+4)}$ , we have  $H_{g,r}^{(12+4)} = 0$  for  $r \in \{2, 3, 6, 9, 10\}$ ,  $H_{g,1}^{(12+4)} = H_{g,7}^{(12+4)}$ ,  $H_{g,4}^{(12+4)} = H_{g,8}^{(12+4)}$ , and  $H_{g,5}^{(12+4)} = H_{g,11}^{(12+4)}$ . Thus it suffices to specify the  $H_{g,1}^{(12+4)}$ ,  $H_{g,4}^{(12+4)}$  and  $H_{g,5}^{(12+4)}$ .

Recall (A.14). Also, set  $S_1^{E_6}(\tau) := S_{12,1}(\tau) + S_{12,7}(\tau)$ , and  $S_5^{E_6}(\tau) := S_{12,5}(\tau) + S_{12,11}(\tau)$ . For r = 1 define

$$\begin{split} H_{1A,1}^{(12+4)}(\tau) &:= H_{1A,1}^{(12)}(\tau) + H_{1A,7}^{(12)}(\tau), \\ H_{2B,1}^{(12+4)}(\tau) &:= \left[-\frac{1}{48}\right] \left( H_{8AB,1}^{(6)}(\tau/2) - H_{8AB,5}^{(6)}(\tau/2) \right), \\ H_{4A,1}^{(12+4)}(\tau) &:= \frac{1}{S_{1}^{E_{6}}(\tau)^{2} - S_{5}^{E_{6}}(\tau)^{2}} \left( -2\frac{\eta(2\tau)^{8}}{\eta(\tau)^{4}} S_{1}^{E_{6}}(\tau) + 8\frac{\eta(\tau)^{4}\eta(4\tau)^{4}}{\eta(2\tau)^{4}} S_{5}^{E_{6}}(\tau) \right), \\ H_{3A,1}^{(12+4)}(\tau) &:= \left[-\frac{1}{48}\right] \left( H_{3A,1}^{(6)}(\tau/2) - H_{3A,5}^{(6)}(\tau/2) \right), \\ H_{8AB,1}^{(12+4)}(\tau) &:= \frac{1}{S_{1}^{E_{6}}(\tau)^{2} - S_{5}^{E_{6}}(\tau)^{2}} \left( -2F_{8AB,1}^{(12+4)}(\tau)S_{1}^{E_{6}}(\tau) + 12F_{8AB,5}^{(12+4)}(\tau/2)S_{5}^{E_{6}}(\tau) \right) \\ (A.88) \end{split}$$

In the expression for  $g \in 8AB$ , we write  $F_{8AB,1}^{(12+4)}$  for the unique modular form of weight 2 for  $\Gamma_0(32)$  such that

$$F_{8AB,1}^{(12+4)}(\tau) = 1 + 12q + 4q^2 - 24q^5 - 16q^6 - 8q^8 + O(q^9),$$
(A.89)

and we write  $F_{8AB,5}^{(12+4)}$  for the unique modular form of weight 2 for  $\Gamma_0(64)$  such that

$$F_{8AB,5}^{(12+4)}(\tau) = 3q + 4q^3 + 6q^5 - 8q^7 - 9q^9 + 12q^{11} - 18q^{13} - 24q^{15} + O(q^{17}).$$
(A.90)

For r = 4 define

$$H_{1A,4}^{(12+4)}(\tau) := H_{1A,4}^{(12)}(\tau) + H_{1A,8}^{(12)}(\tau),$$
  

$$H_{3A,4}^{(12+4)}(\tau) := H_{3A,2}^{(6)}(\tau/2) + H_{3A,4}^{(6)}(\tau/2),$$
(A.91)

and set  $H_{g,4}^{(12+4)}(\tau) := 0$  for  $g \in 2B \cup 4A \cup 8AB$ .

For r = 5 define

$$\begin{split} H_{1A,5}^{(12+4)}(\tau) &:= H_{1A,5}^{(12)}(\tau) + H_{1A,11}^{(12)}(\tau), \\ H_{2B,5}^{(12+4)}(\tau) &:= \left[-\frac{25}{48}\right] \left( H_{8AB,5}^{(6)}(\tau/2) - H_{8AB,1}^{(6)}(\tau/2) \right), \\ H_{4A,5}^{(12+4)}(\tau) &:= \frac{1}{S_{1}^{E_{6}}(\tau)^{2} - S_{5}^{E_{6}}(\tau)^{2}} \left( 2\frac{\eta(2\tau)^{8}}{\eta(\tau)^{4}} S_{5}^{E_{6}}(\tau) - 8\frac{\eta(\tau)^{4}\eta(4\tau)^{4}}{\eta(2\tau)^{4}} S_{1}^{E_{6}}(\tau) \right), \\ H_{3A,5}^{(12+4)}(\tau) &:= \left[-\frac{25}{48}\right] \left( H_{3A,5}^{(6)}(\tau/2) - H_{3A,1}^{(6)}(\tau/2) \right), \\ H_{8AB,5}^{(12+4)}(\tau) &:= \frac{1}{S_{1}^{E_{6}}(\tau)^{2} - S_{5}^{E_{6}}(\tau)^{2}} \left( 2F_{8AB,1}^{(12+4)}(\tau) S_{5}^{E_{6}}(\tau) - 12F_{8AB,5}^{(12+4)}(\tau/2) S_{1}^{E_{6}}(\tau) \right). \end{split}$$
(A.92)

Finally, define  $H_{g,r}^{(12+4)}$  for  $g \in 2A \cup 6A$  by setting

$$H_{2A,r}^{(12+4)}(\tau) := -(-1)^r H_{1A,r}^{(12+4)}(\tau),$$
  

$$H_{6A,r}^{(12+4)}(\tau) := -(-1)^r H_{3A,r}^{(12+4)}(\tau).$$
(A.93)

 $\ell = 13, X = A_{12}^2$ 

We have  $m^X = 13$ , so the umbral McKay-Thompson series  $H_g^{(13)} = (H_{g,r}^{(13)})$ associated to  $g \in G^{(13)} = G^X \simeq \mathbb{Z}/4\mathbb{Z}$  is a 26-vector-valued function, with components indexed by  $r \in \mathbb{Z}/26\mathbb{Z}$ . Define  $H_g^{(13)} = (H_{g,r}^{(13)})$  for  $g \in G^{(13)}$  by requiring that

$$\psi_{g}^{(13)}(\tau,z) = -\chi_{g}^{(13)}\mu_{13,0}^{0}(\tau,z) - \bar{\chi}_{g}^{(13)}\mu_{13,0}^{1}(\tau,z) + \sum_{r \mod 26} H_{g,r}^{(13)}(\tau)\theta_{13,r}(\tau,z),$$
(A.94)

where  $\chi_g^{(13)} := \chi_g^{X_A}$  and  $\bar{\chi}_g^{(13)} := \bar{\chi}_g^{X_A}$  (cf. Table A.28), and the  $\psi_g^{(13)}$  are meromorphic Jacobi forms of weight 1 and index 13 given explicitly in Table A.42.

Table A.42: Character Values and Meromorphic Jacobi Forms for  $\ell = 13$ ,  $X = A_{12}^2$ 

[g]	$\chi_g^{(13)}$	$\bar{\chi}_{g}^{(13)}$	$\psi_g^{(13)}( au,z)$
1A	2	2	$2i_1(\tau, 6z)_1(\tau, z)^{-1}_1(\tau, 3z)^{-1}\eta(\tau)^3$
2A	-2	2	$-2i_1(\tau,6z)_2(\tau,z)^{-1}_2(\tau,3z)^{-1}\eta(\tau)^3$
4A	0	0	cf. (A.95)

$$\psi_{4AB}^{(13)}(\tau,z) := -i_2(\tau,6z) \frac{1(\tau,z+\frac{1}{4})_1(\tau,3z+\frac{1}{4}) - 1(\tau,z-\frac{1}{4})_1(\tau,3z-\frac{1}{4})}{2(2\tau,2z)_2(2\tau,6z)} \frac{\eta(2\tau)^2}{\eta(\tau)}$$
(A.95)

For use later on we note that  $\psi_{1A}^{(13)} = -2\Psi_{1,1}\varphi_1^{(13)}$ , where

$$\varphi_1^{(13)}(\tau, z) := \frac{1(\tau, z)_1(\tau, 6z)}{1(\tau, 2z)_1(\tau, 3z)}.$$
(A.96)

 $\ell = 14 + 7, X = D_8^3$ 

We have  $m^X = 14$ , so the umbral McKay-Thompson series  $H_g^{(14+7)} = (H_{g,r}^{(14+7)})$ associated to  $g \in G^{(14+7)}$  is a 28-vector-valued function with components indexed by  $r \in \mathbb{Z}/28\mathbb{Z}$ . We have  $H_{g,r}^{(14+7)} = 0$  for  $r = 0 \mod 2$ , so it suffices to specify the  $H_{g,r}^{(14+7)}$  for r odd. Observing that  $H_{g,r}^{(14+7)} = -H_{g,-r}^{(14+7)}$  we may determine  $H_g^{(14+7)}$  by requiring that

$$\psi_{g}^{(7/2)}(\tau,z) = -2\bar{\chi}_{g}^{(7/2)}i\mu_{7/2,0}(\tau,z) + \sum_{\substack{r \in \mathbb{Z}+1/2\\r \mod 7}} e(-r/2)H_{g,2r}^{(14+7)}(\tau)\theta_{7/2,r}(\tau,z),$$
(A.97)

where  $\bar{\chi}_g^{(7/2)} := \bar{\chi}_g^{X_D}$  is the number of fixed points of  $g \in G^{(14+7)} \simeq S_3$ in the defining permutation representation on 3 points. The  $\psi_g^{(7/2)}$  are the meromorphic Jacobi forms of weight 1 and index 7/2 defined in Table A.43.

Table A.43: Character Values and Meromorphic Jacobi Forms for  $\ell=14+7,$   $X=D_8^3$ 

[g]	$\bar{\chi}_{g}^{(7/2)}$	$\psi_g^{(7/2)}( au,z)$
1A	3	$2i\theta_1(\tau,3z)\theta_1(\tau,z)^{-2}\eta(\tau)^3$
2A	1	$2i\theta_2(\tau, 3z)\theta_1(\tau, z)^{-1}\theta_2(\tau, z)^{-1}\eta(\tau)^3$
3A	0	$-2i_1(\tau,z)_1(\tau,3z)_1(3\tau,3z)^{-1}\eta(3\tau)$

 $\ell = 16, X = A_{15}D_9$ 

We have  $m^X = 16$ , so the umbral McKay-Thompson series  $H_g^{(16)} = (H_{g,r}^{(16)})$ associated to  $g \in G^{(16)} \simeq \mathbb{Z}/2\mathbb{Z}$  is a 32-vector-valued function with components indexed by  $r \in \mathbb{Z}/32\mathbb{Z}$ . We have  $H_{g,r}^{(16)} = -H_{g,-r}^{(16)}$ , so it suffices to specify the  $H_{g,r}^{(16)}$  for  $1 \leq r \leq 15$ .

To define  $H_g^{(16)} = (H_{g,r}^{(16)})$  for g = e, first define  $h(\tau) = (h_r(\tau))$  by requiring that

$$-6\Psi_{1,1}(\tau,z)\left(\varphi_1^{(16)}(\tau,z) + \frac{1}{2}\varphi_2^{(16)}(\tau,z)\right) = -24\mu_{16,0}(\tau,z) + \sum_{\substack{r \mod 32\\ \text{(A.98)}}} h_r(\tau)\theta_{16,r}(\tau,z),$$

where

$$\varphi_{1}^{(16)}(\tau,z) := 8\varphi_{1}^{(4)}(\tau,z)\varphi_{1}^{(13)}(\tau,z) - \varphi_{1}^{(5)}(\tau,z)\varphi_{1}^{(12)}(\tau,z) + \varphi_{1}^{(7)}(\tau,z)\varphi_{1}^{(10)}(\tau,z), 
\varphi_{2}^{(16)}(\tau,z) := 12\varphi_{1}^{(4)}(\tau,z)\varphi_{1}^{(13)}(\tau,z) - \varphi_{1}^{(5)}(\tau,z)\varphi_{1}^{(12)}(\tau,z) - 3\varphi_{1}^{(16)}(\tau,z). 
(A.99)$$

(Cf. (A.42), (A.46), (A.61), (A.76), (A.84), (A.96).) Now define the  $H_{1A,r}^{(16)}$  by setting

$$H_{1A,r}^{(16)}(\tau) := \frac{1}{12}h_r(\tau) \tag{A.100}$$

for r odd. For r even,  $2 \le r \le 14$ , use

$$H_{1A,r}^{(16)}(\tau) := \frac{1}{24} \left( h_r(\tau) + h_{16-r}(\tau) \right).$$
 (A.101)

Define  $H_{2A,r}^{(16)}$  by requiring

$$H_{2A,r}^{(16)}(\tau) := -(-1)^r H_{1A,r}^{(16)}(\tau).$$
(A.102)

 $\ell = 18, X = A_{17}E_7$ 

We have  $m^X = 18$ , so the umbral McKay-Thompson series  $H_g^{(18)} = (H_{g,r}^{(18)})$ associated to  $g \in G^{(18)} \simeq \mathbb{Z}/2\mathbb{Z}$  is a 36-vector-valued function with components indexed by  $r \in \mathbb{Z}/36\mathbb{Z}$ . We have  $H_{g,r}^{(18)} = -H_{g,-r}^{(18)}$ , so it suffices to specify the  $H_{g,r}^{(18)}$  for  $1 \leq r \leq 17$ .

To define  $H_g^{(18)} = (H_{g,r}^{(18)})$  for g = e, first define  $h(\tau) = (h_r(\tau))$  by requiring that

$$-24\Psi_{1,1}(\tau,z)\phi^{(18)}(\tau,z) = -24\mu_{18,0}(\tau,z) + \sum_{r \mod 36} h_r(\tau)\theta_{18,r}(\tau,z), \quad (A.103)$$

where

$$\phi^{(18)} := \frac{1}{12} \left( \varphi_1^{(18)} + \frac{1}{3} \varphi_3^{(18)} + 4 \frac{\frac{1}{12}}{\eta^{12}} \left( \varphi_1^{(12)} + 2\varphi_2^{(12)} + \frac{1}{3} \varphi_3^{(12)} \right) \right).$$
(A.104)

For the definition of  $\phi^{(18)}$  we require

$$\begin{split} \varphi_{2}^{(9)}(\tau,z) &:= \varphi_{1}^{(4)}(\tau,z)\varphi_{1}^{(6)}(\tau,z) - 4\varphi_{1}^{(5)}(\tau,z)^{2} - 4\varphi_{1}^{(9)}(\tau,z), \\ \varphi_{1}^{(11)}(\tau,z) &:= 3\varphi_{1}^{(5)}(\tau,z)\varphi_{1}^{(7)}(\tau,z) + 2\varphi_{1}^{(3)}(\tau,z)\varphi_{1}^{(9)}(\tau,z) - \varphi_{1}^{(4)}(\tau,z)\varphi_{1}^{(8)}(\tau,z), \\ \varphi_{3}^{(12)}(\tau,z) &:= \varphi_{1}^{(4)}(\tau,z)\varphi_{2}^{(9)}(\tau,z), \\ \varphi_{1}^{(12)}(\tau,z) &:= 3\varphi_{1}^{(5)}(\tau,z)\varphi_{1}^{(10)}(\tau,z) + \varphi_{1}^{(3)}(\tau,z)\varphi_{1}^{(12)}(\tau,z) - 4\varphi_{1}^{(4)}(\tau,z)\varphi_{1}^{(11)}(\tau,z), \\ \varphi_{1}^{(15)}(\tau,z) &:= \varphi_{1}^{(5)}(\tau,z)\varphi_{1}^{(11)}(\tau,z) + 6\varphi_{1}^{(3)}(\tau,z)\varphi_{1}^{(13)}(\tau,z) - \varphi_{1}^{(4)}(\tau,z)\varphi_{1}^{(12)}(\tau,z), \\ \varphi_{2}^{(15)}(\tau,z) &:= \varphi_{1}^{(4)}(\tau,z)\varphi_{1}^{(12)}(\tau,z) - 2\varphi_{1}^{(5)}(\tau,z)\varphi_{1}^{(11)}(\tau,z) - 2\varphi_{1}^{(15)}(\tau,z), \\ \varphi_{1}^{(18)}(\tau,z) &:= \varphi_{1}^{(5)}(\tau,z)\varphi_{1}^{(14)}(\tau,z) + 3\varphi_{1}^{(3)}(\tau,z)\varphi_{1}^{(16)}(\tau,z) - 4\varphi_{1}^{(4)}(\tau,z)\varphi_{1}^{(15)}(\tau,z), \\ \varphi_{3}^{(18)}(\tau,z) &:= \varphi_{1}^{(4)}(\tau,z)\varphi_{2}^{(15)}(\tau,z), \end{split}$$
(A.105)

in addition to the other  $\varphi_k^{(m)}$  that have appeared already. Now define the  $H_{1A,r}^{(18)}$  by setting

$$H_{1A,r}^{(18)}(\tau) := \frac{1}{24} h_r(\tau) \tag{A.106}$$

for r even. For r odd, use

$$\begin{aligned} H_{1A,1}^{(18)}(\tau) &:= \frac{1}{24} \left( 2h_1(\tau) + h_{17}(\tau) \right), \\ H_{1A,3}^{(18)}(\tau) &:= \frac{1}{24} \left( h_3(\tau) + h_9(\tau) \right), \\ H_{1A,5}^{(18)}(\tau) &:= \frac{1}{24} \left( 2h_5(\tau) + h_{13}(\tau) \right), \\ H_{1A,7}^{(18)}(\tau) &:= \frac{1}{24} \left( 2h_7(\tau) + h_{11}(\tau) \right), \\ H_{1A,9}^{(18)}(\tau) &:= \frac{1}{24} \left( h_3(\tau) + 2h_9(\tau) + h_{15}(\tau) \right), \end{aligned}$$
(A.107)  
$$\begin{aligned} H_{1A,11}^{(18)}(\tau) &:= \frac{1}{24} \left( h_7(\tau) + 2h_{11}(\tau) \right), \\ H_{1A,13}^{(18)}(\tau) &:= \frac{1}{24} \left( h_5(\tau) + 2h_{13}(\tau) \right), \\ H_{1A,15}^{(18)}(\tau) &:= \frac{1}{24} \left( h_{15}(\tau) + h_9(\tau) \right), \\ H_{1A,15}^{(18)}(\tau) &:= \frac{1}{24} \left( h_{15}(\tau) + h_9(\tau) \right). \end{aligned}$$

Define  $H_{2A,r}^{(18)}$  in the usual way for root systems with a type A component, by requiring

$$H_{2A,r}^{(18)}(\tau) := -(-1)^r H_{1A,r}^{(18)}(\tau).$$
(A.108)

 $\ell = 18 + 9, X = D_{10}E_7^2$ 

We have  $m^X = 18$ , so the umbral McKay-Thompson series  $H_g^{(18+9)} = (H_{g,r}^{(18+9)})$ associated to  $g \in G^{(18+9)} \simeq \mathbb{Z}/2\mathbb{Z}$  is a 36-vector-valued function with components indexed by  $r \in \mathbb{Z}/36\mathbb{Z}$ . We have  $H_{g,r}^{(18+9)} = -H_{g,-r}^{(18+9)}$ ,  $H_{g,r}^{(18+9)} = H_{g,18-r}^{(18+9)}$ for  $1 \leq r \leq 17$ , and  $H_{g,r}^{(18+9)} = 0$  for  $r = 0 \mod 2$ , so it suffices to specify the  $H_{g,r}^{(18+9)}$  for  $r \in \{1, 3, 5, 7, 9\}$ .

Define

$$H_{1A,r}^{(18+9)}(\tau) := H_{1A,r}^{(18)}(\tau) + H_{1A,18-r}^{(18)}(\tau),$$

$$H_{2A,r}^{(18+9)}(\tau) := H_{1A,r}^{(18)}(\tau) - H_{1A,18-r}^{(18)}(\tau),$$
(A.109)

for  $r \in \{1, 3, 5, 7, 9\}$ .

 $\ell = 22 + 11, X = D_{12}^2$ 

We have  $m^X = 22$ , so the umbral McKay-Thompson series  $H_g^{(22+11)} = (H_{g,r}^{(22+11)})$  associated to  $g \in G^{(22+11)} \simeq \mathbb{Z}/2\mathbb{Z}$  is a 44-vector-valued function with components indexed by  $r \in \mathbb{Z}/44\mathbb{Z}$ . We have  $H_{g,r}^{(22+11)} = -H_{g,-r}^{(22+11)}$  and  $H_{g,r}^{(22+11)} = 0$  for  $r = 0 \mod 2$ , so it suffices to specify the  $H_{g,r}^{(22+11)}$  for r odd. Observing that  $H_{g,r}^{(22+11)} = -H_{g,-r}^{(22+11)}$  we may determine  $H_g^{(22+11)}$  by requiring that

$$\psi_{g}^{(11/2)}(\tau,z) = -2\bar{\chi}_{g}^{(11/2)}i\mu_{11/2,0}(\tau,z) + \sum_{\substack{r \in \mathbb{Z}+1/2\\r \mod 11}} e(-r/2)H_{g,2r}^{(22+11)}(\tau)\theta_{11/2,r}(\tau,z)$$
(A.110)

where  $\bar{\chi}_{1A}^{(11/2)} := 2$ ,  $\bar{\chi}_{2A}^{(11/2)} := 0$ , and the  $\psi_g^{(11/2)}$  are the meromorphic Jacobi forms of weight 1 and index 11/2 defined as follows.

$$\psi_{1A}^{(11/2)}(\tau, z) := 2i \frac{\theta_1(\tau, 4z)}{\theta_1(\tau, z)\theta_1(\tau, 2z)} \eta(\tau)^3$$
  

$$\psi_{2A}^{(11/2)}(\tau, z) := -2i \frac{\theta_1(\tau, 4z)}{\theta_2(\tau, z)\theta_2(\tau, 2z)} \eta(\tau)^3$$
(A.111)

 $\ell = 25, X = A_{24}$ 

We have  $m^X = 25$ , so for  $g \in G^{(25)} \simeq \mathbb{Z}/2\mathbb{Z}$ , the associated umbral McKay-Thompson series  $H_g^{(25)} = (H_{g,r}^{(25)})$  is a 50-vector-valued function, with components indexed by  $r \in \mathbb{Z}/50\mathbb{Z}$ , satisfying  $H_{g,r}^{(25)} = -H_{g,-r}^{(25)}$ , and in particular,  $H_{g,r}^{(25)} = 0$  for  $r = 0 \mod 25$ . So it suffices to specify the  $H_{g,r}^{(25)}$  for  $1 \le r \le 24$ . Define  $H_g^{(25)} = (H_{g,r}^{(25)})$  for g = e by requiring that

$$-\Psi_{1,1}(\tau,z)\varphi_1^{(25)}(\tau,z) = -\mu_{25,0}(\tau,z) + \sum_{\substack{r \mod 50}} H_{e,r}^{(25)}(\tau)\theta_{25,r}(\tau,z), \quad (A.112)$$

where

$$\varphi_1^{(25)}(\tau, z) := \frac{1}{2} \varphi_1^{(5)}(\tau, z) \varphi_1^{(21)}(\tau, z) - \varphi_1^{(7)}(\tau, z) \varphi_1^{(19)}(\tau, z) + \frac{1}{2} \varphi_1^{(13)}(\tau, z)^2.$$
(A.113)

For the definition of  $\varphi_1^{(25)}$  we require

$$\begin{split} \varphi_{1}^{(17)}(\tau,z) &:= 4\varphi_{1}^{(5)}(\tau,z)\varphi_{1}^{(13)}(\tau,z) - \varphi_{1}^{(9)}(\tau,z)^{2}, \\ \varphi_{1}^{(19)}(\tau,z) &:= \varphi_{1}^{(4)}(\tau,z)\varphi_{1}^{(16)}(\tau,z) + 2\varphi_{1}^{(7)}(\tau,z)\varphi_{1}^{(13)}(\tau,z) - \varphi_{1}^{(5)}(\tau,z)\varphi_{1}^{(15)}(\tau,z), \\ \varphi_{1}^{(21)}(\tau,z) &:= \varphi_{1}^{(5)}(\tau,z)\varphi_{1}^{(17)}(\tau,z) - 2\varphi_{1}^{(9)}(\tau,z)\varphi_{1}^{(13)}(\tau,z), \end{split}$$

$$(A.114)$$

in addition to the other  $\varphi_k^{(m)}$  that have appeared already. Define  $H_{2A,r}^{(25)}$  in the usual way for root systems with a type A component, by requiring

$$H_{2A,r}^{(18)}(\tau) := -(-1)^r H_{1A,r}^{(18)}(\tau).$$
(A.115)

 $\ell = 30 + 15, \, X = D_{16}E_8$ 

We have  $m^X = 30$ , so the umbral McKay-Thompson series  $H_g^{(30+15)} = (H_{g,r}^{(30+15)})$  associated to  $g \in G^{(30+15)} = \{e\}$  is a 60-vector-valued function with components indexed by  $r \in \mathbb{Z}/60\mathbb{Z}$ . We have  $H_{e,r}^{(30+15)} = -H_{e,-r}^{(30+15)}$ ,  $H_{e,r}^{(30+15)} = H_{e,30-r}^{(30+15)}$  for  $1 \le r \le 29$ , and  $H_{e,r}^{(30+15)} = 0$  for  $r = 0 \mod 2$ , so it suffices to specify the  $H_{e,r}^{(30+15)}$  for  $r \in \{1, 3, 5, 7, 9, 11, 13, 15\}$ .

Define

$$\begin{split} H_{1A,1}^{(30+15)}(\tau) &:= \frac{1}{2} \left( H_{1A,1}^{(30+6,10,15)} + [-\frac{1}{120}] H_{3A,1}^{(10+5)}(\tau/3) \right), \\ H_{1A,3}^{(30+15)}(\tau) &:= [-\frac{9}{120}] H_{3A,3}^{(10+5)}(\tau/3), \\ H_{1A,5}^{(30+15)}(\tau) &:= [-\frac{25}{120}] H_{3A,5}^{(10+5)}(\tau/3), \\ H_{1A,7}^{(30+15)}(\tau) &:= \frac{1}{2} \left( H_{1A,7}^{(30+6,10,15)} + [-\frac{49}{120}] H_{3A,3}^{(10+5)}(\tau/3) \right), \\ H_{1A,11}^{(30+15)}(\tau) &:= \frac{1}{2} \left( H_{1A,1}^{(30+6,10,15)} - [-\frac{1}{120}] H_{3A,1}^{(10+5)}(\tau/3) \right), \\ H_{1A,13}^{(30+15)}(\tau) &:= \frac{1}{2} \left( H_{1A,7}^{(30+6,10,15)} - [-\frac{49}{120}] H_{3A,3}^{(10+5)}(\tau/3) \right), \\ H_{1A,13}^{(30+15)}(\tau) &:= -[-\frac{105}{120}] H_{3A,5}^{(10+5)}(\tau/3). \end{split}$$

 $\ell = 30 + 6, 10, 15, X = E_8^3$ 

We have  $m^X = 30$ , and  $G^{(30+6,10,15)} = G^X \simeq S_3$ . The umbral McKay-Thompson series  $H^{(30+6,10,15)}$  is a 60-vector-valued function with components indexed by  $r \in \mathbb{Z}/60\mathbb{Z}$ . We have

$$H_{g,r}^{(30+6,10,15)}(\tau) = \begin{cases} \pm H_{g,1}^{(30+6,10,15)} & \text{if } r = \pm 1, \pm 11, \pm 19, \pm 29 \mod 60, \\ \pm H_{g,7}^{(30+6,10,15)} & \text{if } r = \pm 7, \pm 13, \pm 17, \pm 27 \mod 60, \\ 0 & \text{else}, \end{cases}$$
(A.117)

so it suffices to specify the  $H_{g,r}^{(30+6,10,15)}$  for r = 1 and r = 7. These functions may be defined as follows.

$$\begin{split} H_{1A,1}^{(30+6,10,15)} &:= -2\frac{1}{\eta(\tau)^2} \left( \sum_{k,l,m\geq 0} + \sum_{k,l,m<0} \right) (-1)^{k+l+m} q^{(k^2+l^2+m^2)/2+2(kl+lm+mk)+(k+l+m)/2+3/40} \\ H_{2A,1}^{(30+6,10,15)} &:= -2\frac{1}{\eta(2\tau)} \left( \sum_{k,m\geq 0} - \sum_{k,m<0} \right) (-1)^{k+m} q^{3k^2+m^2/2+4km+(2k+m)/2+3/40} \\ H_{3A,1}^{(30+6,10,15)} &:= -2\frac{\eta(\tau)}{\eta(3\tau)} \sum_{k\in\mathbb{Z}} (-1)^k q^{15k^2/2+3k/2+3/40} \\ H_{1A,7}^{(30+6,10,15)} &= -2\frac{1}{\eta(\tau)^2} \left( \sum_{k,l,m\geq 0} + \sum_{k,l,m<0} \right) (-1)^{k+l+m} q^{(k^2+l^2+m^2)/2+2(kl+lm+mk)+3(k+l+m)/2+27/40} \\ H_{2A,7}^{(30+6,10,15)} &= 2\frac{1}{\eta(2\tau)} \left( \sum_{k,m\geq 0} - \sum_{k,m<0} \right) (-1)^{k+m} q^{3k^2+m^2/2+4km+3(2k+m)/2+27/40} \\ H_{3A,7}^{(30+6,10,15)} &= -2\frac{\eta(\tau)}{\eta(3\tau)} \sum_{k\in\mathbb{Z}} (-1)^k q^{15k^2/2+9k/2+27/40} \\ (A.118) \end{split}$$

 $\ell = 46 + 23, X = D_{24}$ 

We have  $m^X = 22$ , and  $G^{(46+23)} = \{e\}$ . The umbral McKay-Thompson series  $H_e^{(46+23)} = (H_{e,r}^{(46+23)})$  is a 92-vector-valued function with components

indexed by  $r \in \mathbb{Z}/92\mathbb{Z}$ . We have  $H_{e,r}^{(46+23)} = -H_{e,-r}^{(46+23)}$  and  $H_{e,r}^{(46+23)} = 0$  for  $r = 0 \mod 2$ , so it suffices to specify the  $H_{e,r}^{(46+23)}$  for r odd. Observing that  $H_{e,r}^{(46+23)} = -H_{e,-r}^{(46+23)}$  we may determine  $H_e^{(46+23)}$  by requiring that

$$\psi_e^{(23/2)}(\tau, z) = -2i\mu_{23/2,0}(\tau, z) + \sum_{\substack{r \in \mathbb{Z} + 1/2\\r \mod 23}} e(-r/2) H_{g,2r}^{(46+23)}(\tau) \theta_{23/2,r}(\tau, z),$$
(A.119)

where  $\psi_e^{(23/2)}$  is the meromorphic Jacobi forms of weight 1 and index 23/2 defined by setting

$$\psi_e^{(23/2)}(\tau, z) := 2i \frac{\theta_1(\tau, 6z)}{\theta_1(\tau, 2z)\theta_1(\tau, 3z)} \eta(\tau)^3.$$
(A.120)

## Bibliography

- Milton Abramowitz and Irene A. Stegun. Handbook of mathematical functions with formulas, graphs, and mathematical tables, volume 55 of National Bureau of Standards Applied Mathematics Series. For sale by the Superintendent of Documents, U.S. Government Printing Office, Washington, D.C., 1964.
- [2] Amod Agashe, Kenneth A. Ribet, and William A. Stein. The modular degree, congruence primes, and multiplicity one. In *Number theory*, analysis and geometry, pages 19–49. Springer, New York, 2012.
- [3] D. Alexander, C. Cummins, J. McKay, and C. Simons. Completely replicable functions. In *Groups, combinatorics & geometry (Durham, 1990)*, volume 165 of *London Math. Soc. Lecture Note Ser.*, pages 87–98. Cambridge Univ. Press, Cambridge, 1992.
- [4] C. Alfes. Formulas for the coefficients of half-integral weight harmonic maass forms. *Math. Zeitschrift, to appear.*
- [5] C. Alfes. Periods related to harmonic weak maass forms, in preparation.
- [6] Claudia Alfes and Stephan Ehlen. Twisted traces of CM values of weak Maass forms. J. Number Theory, 133(6):1827–1845, 2013.

- [7] George E. Andrews. An analytic generalization of the Rogers-Ramanujan identities for odd moduli. Proc. Nat. Acad. Sci. U.S.A., 71:4082–4085, 1974.
- [8] George E. Andrews. Problems and prospects for basic hypergeometric functions. In Theory and application of special functions (Proc. Advanced Sem., Math. Res. Center, Univ. Wisconsin, Madison, Wis., 1975), pages 191–224. Math. Res. Center, Univ. Wisconsin, Publ. No. 35. Academic Press, New York, 1975.
- [9] George E. Andrews. The theory of partitions. Addison-Wesley Publishing Co., Reading, Mass.-London-Amsterdam, 1976. Encyclopedia of Mathematics and its Applications, Vol. 2.
- [10] George E. Andrews, R. J. Baxter, and P. J. Forrester. Eight-vertex sos model and generalized Rogers-Ramanujan-type identities. J. Statist. Phys., 35(3-4):193-266, 1984.
- [11] George E. Andrews, Anne Schilling, and S. Ole Warnaar. An A<sub>2</sub> Bailey lemma and Rogers-Ramanujan-type identities. J. Amer. Math. Soc., 12(3):677-702, 1999.
- [12] Thomas Appelquist. Dimensional reduction in quantum gravity. In Problems in unification and supergravity (La Jolla, Calif., 1983), volume 116 of AIP Conf. Proc., pages 1–10. Amer. Inst. Phys., New York, 1984.
- [13] Michael Aschbacher. The status of the classification of the finite simple groups. Notices Amer. Math. Soc., 51(7):736-740, 2004.
- [14] R. Askey and Mourad E. H. Ismail. A generalization of ultraspherical polynomials. In *Studies in pure mathematics*, pages 55–78. Birkhäuser, Basel, 1983.

- [15] A. Astashkevich. On the structure of Verma modules over Virasoro and Neveu-Schwarz algebras. Comm. Math. Phys., 186(3):531–562, 1997.
- [16] M. F. Atiyah. K-theory. Advanced Book Classics. Addison-Wesley Publishing Company, Advanced Book Program, Redwood City, CA, second edition, 1989. Notes by D. W. Anderson.
- [17] Máximo Bañados, Marc Henneaux, Claudio Teitelboim, and Jorge Zanelli. Geometry of the 2+1 black hole. *Phys. Rev. D* (3), 48(4):1506–1525, 1993.
- [18] Máximo Bañados, Claudio Teitelboim, and Jorge Zanelli. Black hole in three-dimensional spacetime. *Phys. Rev. Lett.*, 69(13):1849–1851, 1992.
- [19] N. Bartlett and S. O. Warnaar. Hall-littlewood polynomials and characters of affine lie algebras. arXiv:1304.1602, 2013.
- [20] R. J. Baxter. Rogers-Ramanujan identities in the hard hexagon model.
   J. Statist. Phys., 26(3):427-452, 1981.
- [21] L. Beneish and H. Larson. Traces of singular values of Hauptmoduln. Int. J. Numb. Th., accepted for publication.
- [22] Alexander Berkovich and Barry M. McCoy. Rogers-Ramanujan identities: a century of progress from mathematics to physics. In Proceedings of the International Congress of Mathematicians, Vol. III (Berlin, 1998), number Extra Vol. III, pages 163–172 (electronic), 1998.
- [23] B. C. Berndt. Ramanujan, his lost notebook, its importance. In I. M. Vandoulakis and L. Dun, editors, *Navigating across Mathematical Cultures and Times.* to appear.

- [24] Bruce C. Berndt, Heng Huat Chan, and Liang-Cheng Zhang. Explicit evaluations of the Rogers-Ramanujan continued fraction. J. Reine Angew. Math., 480:141–159, 1996.
- [25] Bruce C. Berndt and Robert A. Rankin. Ramanujan, volume 9 of History of Mathematics. American Mathematical Society, Providence, RI; London Mathematical Society, London, 1995. Letters and commentary.
- [26] B. J. Birch and H. P. F. Swinnerton-Dyer. Notes on elliptic curves. I. J. Reine Angew. Math., 212:7–25, 1963.
- [27] B. J. Birch and H. P. F. Swinnerton-Dyer. Notes on elliptic curves. II. J. Reine Angew. Math., 218:79–108, 1965.
- [28] Richard Borcherds. Generalized Kac-Moody algebras. J. Algebra, 115(2):501–512, 1988.
- [29] Richard E. Borcherds. Vertex algebras, Kac-Moody algebras, and the Monster. Proc. Nat. Acad. Sci. U.S.A., 83(10):3068-3071, 1986.
- [30] Richard E. Borcherds. Monstrous moonshine and monstrous Lie superalgebras. Invent. Math., 109(2):405-444, 1992.
- [31] Richard E. Borcherds. Automorphic forms with singularities on Grassmannians. Invent. Math., 132(3):491-562, 1998.
- [32] Richard E. Borcherds. Quantum vertex algebras. In Taniguchi Conference on Mathematics Nara '98, volume 31 of Adv. Stud. Pure Math., pages 51-74. Math. Soc. Japan, Tokyo, 2001.
- [33] A. Borel, S. Chowla, C. S. Herz, K. Iwasawa, and J.-P. Serre. Seminar on complex multiplication. Seminar held at the Institute for Advanced Study, Princeton, N.J., 1957-58. Lecture Notes in Mathematics, No. 21. Springer-Verlag, Berlin-New York, 1966.

- [34] Raoul Bott. Homogeneous vector bundles. Ann. of Math. (2), 66:203–248, 1957.
- [35] D. M. Bressoud. On partitions, orthogonal polynomials and the expansion of certain infinite products. Proc. London Math. Soc. (3), 42(3):478-500, 1981.
- [36] David M. Bressoud. An analytic generalization of the Rogers-Ramanujan identities with interpretation. Quart. J. Math. Oxford Ser. (2), 31(124):385-399, 1980.
- [37] K. Bringmann, B. Kane, and M. Viazovska. Mock theta lifts and local maass forms. Mathematical Research Letters, accepted for publication.
- [38] Kathrin Bringmann and Amanda Folsom. On the asymptotic behavior of Kac-Wakimoto characters. Proc. Amer. Math. Soc., 141(5):1567– 1576, 2013.
- [39] Kathrin Bringmann and Amanda Folsom. Almost harmonic Maass forms and Kac-Wakimoto characters. J. Reine Angew. Math., 694:179– 202, 2014.
- [40] Kathrin Bringmann and Karl Mahlburg. Asymptotic formulas for coefficients of Kac-Wakimoto characters. Math. Proc. Cambridge Philos. Soc., 155(1):51-72, 2013.
- [41] Kathrin Bringmann and Ken Ono. The f(q) mock theta function conjecture and partition ranks. *Invent. Math.*, 165(2):243–266, 2006.
- [42] Kathrin Bringmann and Ken Ono. Arithmetic properties of coefficients of half-integral weight Maass-Poincaré series. *Math. Ann.*, 337(3):591– 612, 2007.

- [43] Kathrin Bringmann and Ken Ono. Some characters of Kac and Wakimoto and nonholomorphic modular functions. *Math. Ann.*, 345(3):547– 558, 2009.
- [44] Kathrin Bringmann and Ken Ono. Dyson's ranks and Maass forms. Ann. of Math. (2), 171(1):419–449, 2010.
- [45] Kathrin Bringmann and Ken Ono. Coefficients of harmonic Maass forms. In Partitions, q-series, and modular forms, volume 23 of Dev. Math., pages 23–38. Springer, New York, 2012.
- [46] Kathrin Bringmann, Ken Ono, and Robert C. Rhoades. Eulerian series as modular forms. J. Amer. Math. Soc., 21(4):1085–1104, 2008.
- [47] Jan Bruinier and Ken Ono. Heegner divisors, L-functions and harmonic weak Maass forms. Ann. of Math. (2), 172(3):2135–2181, 2010.
- [48] Jan H. Bruinier. Borcherds products on O(2, l) and Chern classes of Heegner divisors, volume 1780 of Lecture Notes in Mathematics. Springer-Verlag, Berlin, 2002.
- [49] Jan H. Bruinier, Winfried Kohnen, and Ken Ono. The arithmetic of the values of modular functions and the divisors of modular forms. *Compos. Math.*, 140(3):552–566, 2004.
- [50] Jan H. Bruinier, Ken Ono, and Robert C. Rhoades. Differential operators for harmonic weak Maass forms and the vanishing of Hecke eigenvalues. *Math. Ann.*, 342(3):673–693, 2008.
- [51] Jan H. Bruinier and Fredrik Strömberg. Computation of harmonic weak Maass forms. *Exp. Math.*, 21(2):117–131, 2012.
- [52] Jan Hendrik Bruinier. Harmonic Maass forms and periods. Math. Ann., 357(4):1363–1387, 2013.

- [53] Jan Hendrik Bruinier and Jens Funke. On two geometric theta lifts. Duke Math. J., 125(1):45–90, 2004.
- [54] Jan Hendrik Bruinier and Jens Funke. On two geometric theta lifts. Duke Math. J., 125(1):45–90, 2004.
- [55] Jan Hendrik Bruinier and Jens Funke. Traces of CM values of modular functions. J. Reine Angew. Math., 594:1–33, 2006.
- [56] Jan Hendrik Bruinier, Paul Jenkins, and Ken Ono. Hilbert class polynomials and traces of singular moduli. Math. Ann., 334(2):373–393, 2006.
- [57] Jan Hendrik Bruinier and Ken Ono. Algebraic formulas for the coefficients of half-integral weight harmonic weak Maass forms. Adv. Math., 246:198–219, 2013.
- [58] Jan Hendrik Bruinier, Gerard van der Geer, Günter Harder, and Don Zagier. The 1-2-3 of modular forms. Universitext. Springer-Verlag, Berlin, 2008. Lectures from the Summer School on Modular Forms and their Applications held in Nordfjordeid, June 2004, Edited by Kristian Ranestad.
- [59] Clemens Bruschek, Hussein Mourtada, and Jan Schepers. Arc spaces and the Rogers-Ramanujan identities. *Ramanujan J.*, 30(1):9–38, 2013.
- [60] Bryden Cais and Brian Conrad. Modular curves and Ramanujan's continued fraction. J. Reine Angew. Math., 597:27–104, 2006.
- [61] Luca Candelori. Harmonic weak Maass forms of integral weight: a geometric approach. Math. Ann., 360(1-2):489–517, 2014.
- [62] S. Carlip. Conformal field theory, (2 + 1)-dimensional gravity and the BTZ black hole. *Classical Quantum Gravity*, 22(12):R85–R123, 2005.

- [63] Steven Carlip. Effective conformal descriptions of black hole entropy. Entropy, 13(7):1355-1379, 2011.
- [64] Scott Carnahan. Generalized moonshine I: genus-zero functions. Algebra Number Theory, 4(6):649–679, 2010.
- [65] Scott Carnahan. Generalized moonshine, II: Borcherds products. Duke Math. J., 161(5):893–950, 2012.
- [66] Scott Carnahan. Generalized moonshine iv: Monstrous lie algebras. 2012.
- [67] Scott Carnahan. Monstrous Lie Algebras. RIMS Kôkyûroku, 1872, 2014.
- [68] M. C. N. Cheng, J. F. R. Duncan, and J. A. Harvey. Umbral Moonshine and the Niemeier Lattices. *Research in the Mathematical Sciences*, 1(3), 2014.
- [69] M. C. N. Cheng, J. F. R. Duncan, and J. A. Harvey. Weight One Jacobi Forms and Umbral Moonshine. 2015.
- [70] Miranda C. N. Cheng. K3 surfaces,  $\mathcal{N} = 4$  dyons and the Mathieu group  $M_{24}$ . Commun. Number Theory Phys., 4(4):623-657, 2010.
- [71] Miranda C. N. Cheng and John F. R. Duncan. Rademacher sums and rademacher series. In Winfried Kohnen and Rainer Weissauer, editors, *Conformal Field Theory, Automorphic Forms and Related Topics*, volume 8 of *Contributions in Mathematical and Computational Sciences*.
- [72] Miranda C. N. Cheng and John F. R. Duncan. The largest Mathieu group and (mock) automorphic forms. In *String-Math 2011*, volume 85 of *Proc. Sympos. Pure Math.*, pages 53–82. Amer. Math. Soc., Providence, RI, 2012.

- [73] Miranda C. N. Cheng and John F. R. Duncan. On Rademacher sums, the largest Mathieu group and the holographic modularity of moonshine. *Commun. Number Theory Phys.*, 6(3):697–758, 2012.
- [74] Miranda C. N. Cheng and John F. R. Duncan. Meromorphic Jacobi Forms of Half-Integral Index and Umbral Moonshine Modules. 2014.
- [75] Miranda C. N. Cheng, John F. R. Duncan, and Jeffrey A. Harvey. Umbral moonshine. Commun. Number Theory Phys., 8(2):101-242, 2014.
- [76] Ivan Cherednik and Boris Feigin. Rogers-Ramanujan type identities and Nil-DAHA. Adv. Math., 248:1050–1088, 2013.
- [77] YoungJu Choie and Min Ho Lee. Rankin-Cohen brackets on pseudodifferential operators. J. Math. Anal. Appl., 326(2):882–895, 2007.
- [78] J. H. Conway. A characterisation of Leech's lattice. Invent. Math., 7:137-142, 1969.
- [79] J. H. Conway. A simple construction for the Fischer-Griess monster group. Invent. Math., 79(3):513-540, 1985.
- [80] J. H. Conway, R. T. Curtis, S. P. Norton, R. A. Parker, and R. A. Wilson. Atlas of finite groups. Oxford University Press, Eynsham, 1985. Maximal subgroups and ordinary characters for simple groups, With computational assistance from J. G. Thackray.
- [81] J. H. Conway and S. P. Norton. Monstrous moonshine. Bull. London Math. Soc., 11(3):308-339, 1979.
- [82] J. H. Conway and N. J. A. Sloane. On the enumeration of lattices of determinant one. J. Number Theory, 15(1):83-94, 1982.

- [83] J. H. Conway and N. J. A. Sloane. Sphere packings, lattices and groups, volume 290 of Grundlehren der Mathematischen Wissenschaften [Fundamental Principles of Mathematical Sciences]. Springer-Verlag, New York, third edition, 1999. With additional contributions by E. Bannai, R. E. Borcherds, J. Leech, S. P. Norton, A. M. Odlyzko, R. A. Parker, L. Queen and B. B. Venkov.
- [84] John Conway, John McKay, and Abdellah Sebbar. On the discrete groups of Moonshine. Proc. Amer. Math. Soc., 132(8):2233-2240, 2004.
- [85] David A. Cox. Primes of the form  $x^2 + ny^2$ . A Wiley-Interscience Publication. John Wiley & Sons, Inc., New York, 1989. Fermat, class field theory and complex multiplication.
- [86] J. E. Cremona. Computing the degree of the modular parametrization of a modular elliptic curve. *Math. Comp.*, 64(211):1235–1250, 1995.
- [87] C. J. Cummins and T. Gannon. Modular equations and the genus zero property of moonshine functions. *Invent. Math.*, 129(3):413–443, 1997.
- [88] R. McNees D. Grumiller and Jakob Salzer. Black holes and thermodynamics - The first half century. 2014.
- [89] R. M. Damerell. L-functions of elliptic curves with complex multiplication. I. Acta Arith., 17:287–301, 1970.
- [90] R. M. Damerell. L-functions of elliptic curves with complex multiplication. II. Acta Arith., 19:311–317, 1971.
- [91] S. Deser and R. Jackiw. Erratum: "Topologically massive gauge theories" [Ann. Physics 140 (1982), no. 2, 372–411; MR0665601 (84j:81128)]
   by Deser, Jackiw and S. Templeton. Ann. Physics, 185(2):406, 1988.

- [92] S. Deser, R. Jackiw, and S. Templeton. Topologically massive gauge theories. Ann. Physics, 140(2):372-411, 1982.
- [93] Robbert Dijkgraaf, Gregory Moore, Erik Verlinde, and Herman Verlinde. Elliptic genera of symmetric products and second quantized strings. Comm. Math. Phys., 185(1):197–209, 1997.
- [94] L. Dixon, P. Ginsparg, and J. Harvey. Beauty and the beast: superconformal symmetry in a Monster module. *Comm. Math. Phys.*, 119(2):221-241, 1988.
- [95] L. Dixon, J. Harvey, C. Vafa, and E. Witten. Strings on orbifolds. *Nuclear Phys. B*, 261(4):678–686, 1985.
- [96] L. Dixon, J. Harvey, C. Vafa, and E. Witten. Strings on orbifolds. II. Nuclear Phys. B, 274(2):285–314, 1986.
- [97] S. K. Donaldson and P. B. Kronheimer. The geometry of fourmanifolds. Oxford Mathematical Monographs. The Clarendon Press, Oxford University Press, New York, 1990. Oxford Science Publications.
- [98] Chongying Dong, Haisheng Li, and Geoffrey Mason. Some twisted sectors for the Moonshine module. In *Moonshine*, the Monster, and related topics (South Hadley, MA, 1994), volume 193 of Contemp. Math., pages 25-43. Amer. Math. Soc., Providence, RI, 1996.
- [99] Chongying Dong, Haisheng Li, and Geoffrey Mason. Twisted representations of vertex operator algebras. Math. Ann., 310(3):571-600, 1998.
- [100] Chongying Dong, Haisheng Li, and Geoffrey Mason. Modularinvariance of trace functions in orbifold theory and generalized Moonshine. *Comm. Math. Phys.*, 214(1):1–56, 2000.

- [101] Chongying Dong and Geoffrey Mason. Nonabelian orbifolds and the boson-fermion correspondence. *Comm. Math. Phys.*, 163(3):523-559, 1994.
- [102] W. Duke. Continued fractions and modular functions. Bull. Amer. Math. Soc. (N.S.), 42(2):137–162, 2005.
- [103] W. Duke, Ö. Imamoğlu, and Á. Tóth. Cycle integrals of the *j*-function and mock modular forms. Ann. of Math. (2), 173(2):947–981, 2011.
- [104] W. Duke, Ö. Imamoğlu, and Á. Tóth. Real quadratic analogs of traces of singular moduli. Int. Math. Res. Not. IMRN, (13):3082–3094, 2011.
- [105] William Duke and Paul Jenkins. Integral traces of singular values of weak Maass forms. Algebra Number Theory, 2(5):573–593, 2008.
- [106] J. Duncan, M. J. Griffin, and K. Ono. Moonshine. arXiv:1411.6571.
- [107] J. Duncan and J. Harvey. The Umbral Mooonshine Module for the Unique Unimodular Niemeier Root System. arXiv:1412.8191.
- [108] J. Duncan and A. O'Desky. Super Vertex Algebras, Meromorphic Jacobi Forms, and Umbral Moonshine. 2015.
- [109] J. F. R. Duncan and K. Ono. The Jack Daniels Problem.
- [110] John F. R. Duncan and Igor B. Frenkel. Rademacher sums, moonshine and gravity. Commun. Number Theory Phys., 5(4):849–976, 2011.
- [111] Johan L. Dupont and Chih-Han Sah. Dilogarithm identities in conformal field theory and group homology. *Comm. Math. Phys.*, 161(2):265– 282, 1994.

- [112] Freeman J. Dyson. A walk through Ramanujan's garden. In Ramanujan revisited (Urbana-Champaign, Ill., 1987), pages 7–28. Academic Press, Boston, MA, 1988.
- [113] Tohru Eguchi and Kazuhiro Hikami. Note on twisted elliptic genus of K3 surface. Phys. Lett. B, 694(4-5):446-455, 2011.
- [114] Tohru Eguchi, Hirosi Ooguri, and Yuji Tachikawa. Notes on the K3 surface and the Mathieu group  $M_{24}$ . Exp. Math., 20(1):91–96, 2011.
- [115] Geir Ellingsrud and Lothar Göttsche. Wall-crossing formulas, the Bott residue formula and the Donaldson invariants of rational surfaces. Quart. J. Math. Oxford Ser. (2), 49(195):307-329, 1998.
- [116] A. Erdélyi, W. Magnus, F. Oberhettinger, and F. G. Tricomi. *Tables of integral transforms. Vol. I.* McGraw-Hill Book Company, Inc., New York-Toronto-London, 1954. Based, in part, on notes left by Harry Bateman.
- [117] John D. Fay. Fourier coefficients of the resolvent for a Fuchsian group.
   J. Reine Angew. Math., 293/294:143-203, 1977.
- B. L. Feĭgin and D. B. Fuchs. Verma modules over the Virasoro algebra. In Topology (Leningrad, 1982), volume 1060 of Lecture Notes in Math., pages 230-245. Springer, Berlin, 1984.
- [119] Boris Feigin, Omar Foda, and Trevor Welsh. Andrews-Gordon type identities from combinations of Virasoro characters. *Ramanujan J.*, 17(1):33-52, 2008.
- [120] Boris Feigin and Edward Frenkel. Coinvariants of nilpotent subalgebras of the Virasoro algebra and partition identities. In *I. M. Gel'fand*

Seminar, volume 16 of Adv. Soviet Math., pages 139–148. Amer. Math. Soc., Providence, RI, 1993.

- [121] A. Folsom, K. Ono, and R. C. Rhoades. Ramanujan's radial limits. In Proceedings of the Ramanujan 125 Conference.
- [122] David Ford, John McKay, and Simon Norton. More on replicable functions. Comm. Algebra, 22(13):5175–5193, 1994.
- [123] Edward Frenkel and David Ben-Zvi. Vertex algebras and algebraic curves, volume 88 of Mathematical Surveys and Monographs. American Mathematical Society, Providence, RI, second edition, 2004.
- [124] I. B. Frenkel and V. G. Kac. Basic representations of affine Lie algebras and dual resonance models. *Invent. Math.*, 62(1):23–66, 1980/81.
- [125] I. B. Frenkel, J. Lepowsky, and A. Meurman. A natural representation of the Fischer-Griess Monster with the modular function J as character. *Proc. Nat. Acad. Sci. U.S.A.*, 81(10, Phys. Sci.):3256–3260, 1984.
- [126] Igor Frenkel, James Lepowsky, and Arne Meurman. Vertex operator algebras and the Monster, volume 134 of Pure and Applied Mathematics. Academic Press, Inc., Boston, MA, 1988.
- [127] Igor B. Frenkel, James Lepowsky, and Arne Meurman. A moonshine module for the Monster. In Vertex operators in mathematics and physics (Berkeley, Calif., 1983), volume 3 of Math. Sci. Res. Inst. Publ., pages 231-273. Springer, New York, 1985.
- [128] Jason Fulman. The Rogers-Ramanujan identities, the finite general linear groups, and the Hall-Littlewood polynomials. Proc. Amer. Math. Soc., 128(1):17-25, 2000.

- [129] Jason Fulman. A probabilistic proof of the Rogers-Ramanujan identities. Bull. London Math. Soc., 33(4):397–407, 2001.
- [130] M. Gaberdiel. An introduction to conformal field theory. *Rept. Prog. Phys.*, 53:607–667, 2000.
- [131] M. Gaberdiel. 2D conformal field theory and vertex operator algebras. 2005.
- [132] Matthias R. Gaberdiel, Stefan Hohenegger, and Roberto Volpato. Mathieu twining characters for K3. J. High Energy Phys., (9):058, 20, 2010.
- [133] Matthias R. Gaberdiel, Christoph A. Keller, and Roberto Volpato. Genus two partition functions of chiral conformal field theories. Commun. Number Theory Phys., 4(2):295–363, 2010.
- [134] Davide Gaiotto. Monster symmetry and extremal CFTs. J. High Energy Phys., (11):149, front matter + 9, 2012.
- [135] Davide Gaiotto and Xi Yin. Genus two partition functions of extremal conformal field theories. J. High Energy Phys., (8):029, 13 pp. (electronic), 2007.
- [136] Terry Gannon. Much ado about Mathieu. arXiv:1211.5531.
- [137] Howard Garland and James Lepowsky. Lie algebra homology and the Macdonald-Kac formulas. *Invent. Math.*, 34(1):37–76, 1976.
- [138] Kristina Garrett, Mourad E. H. Ismail, and Dennis Stanton. Variants of the Rogers-Ramanujan identities. Adv. in Appl. Math., 23(3):274–299, 1999.

- [139] Basil Gordon. A combinatorial generalization of the Rogers-Ramanujan identities. Amer. J. Math., 83:393–399, 1961.
- [140] Eugene Gorsky, Alexei Oblomkov, and Jacob Rasmussen. On stable Khovanov homology of torus knots. *Exp. Math.*, 22(3):265–281, 2013.
- [141] Lothar Göttsche. Modular forms and Donaldson invariants for 4manifolds with  $b_{+} = 1$ . J. Amer. Math. Soc., 9(3):827-843, 1996.
- [142] Lothar Göttsche. Donaldson invariants in algebraic geometry. In School on Algebraic Geometry (Trieste, 1999), volume 1 of ICTP Lect. Notes, pages 101–134. Abdus Salam Int. Cent. Theoret. Phys., Trieste, 2000.
- [143] Lothar Göttsche, Hiraku Nakajima, and Kōta Yoshioka. Instanton counting and Donaldson invariants. J. Differential Geom., 80(3):343– 390, 2008.
- [144] Lothar Göttsche and Don Zagier. Jacobi forms and the structure of Donaldson invariants for 4-manifolds with  $b_{+} = 1$ . Selecta Math. (N.S.), 4(1):69-115, 1998.
- [145] Robert L. Griess, Jr. The structure of the "monster" simple group. In Proceedings of the Conference on Finite Groups (Univ. Utah, Park City, Utah, 1975), pages 113–118. Academic Press, New York, 1976.
- [146] Robert L. Griess, Jr. A construction of F<sub>1</sub> as automorphisms of a 196,883-dimensional algebra. Proc. Nat. Acad. Sci. U.S.A., 78(2, part 1):686-691, 1981.
- [147] Robert L. Griess, Jr. The friendly giant. Invent. Math., 69(1):1-102, 1982.
- [148] B. Gross, W. Kohnen, and D. Zagier. Heegner points and derivatives of L-series. II. Math. Ann., 278(1-4):497–562, 1987.
- [149] Benedict H. Gross and Don B. Zagier. Heegner points and derivatives of L-series. Invent. Math., 84(2):225–320, 1986.
- [150] S. S. Gubser, I. R. Klebanov, and A. M. Polyakov. Gauge theory correlators from non-critical string theory. *Phys. Lett. B*, 428(1-2):105– 114, 1998.
- [151] P. Guerzhoy. Zagier's adele.
- [152] P. Guerzhoy. A mixed mock modular solution of the Kaneko-Zagier equation. Ramanujan J., 36(1-2):149–164, 2015.
- [153] Pavel Guerzhoy, Zachary A. Kent, and Ken Ono. p-adic coupling of mock modular forms and shadows. Proc. Natl. Acad. Sci. USA, 107(14):6169-6174, 2010.
- [154] R. A. Gustafson. The Macdonald identities for affine root systems of classical type and hypergeometric series very-well-poised on semisimple Lie algebras. In *Ramanujan International Symposium on Analysis* (*Pune, 1987*), pages 185–224. Macmillan of India, New Delhi, 1989.
- [155] Koichiro Harada and Mong Lung Lang. The McKay-Thompson series associated with the irreducible characters of the Monster. In *Moon-shine, the Monster, and related topics (South Hadley, MA, 1994)*, volume 193 of *Contemp. Math.*, pages 93–111. Amer. Math. Soc., Providence, RI, 1996.
- [156] Koichiro Harada and Mong Lung Lang. Modular forms associated with the Monster module. In *The Monster and Lie algebras (Columbus, OH,* 1996), volume 7 of *Ohio State Univ. Math. Res. Inst. Publ.*, pages 59– 83. de Gruyter, Berlin, 1998.

- [157] G. H. Hardy. Collected papers of G. H. Hardy (Including Joint papers with J. E. Littlewood and others). Vol. I. Edited by a committee appointed by the London Mathematical Society. Clarendon Press, Oxford, 1966.
- [158] G. H. Hardy and S. Ramanujan. Asymptotic formulæ in combinatory analysis [Proc. London Math. Soc. (2) 17 (1918), 75–115]. In *Collected* papers of Srinivasa Ramanujan, pages 276–309. AMS Chelsea Publ., Providence, RI, 2000.
- [159] Kazuhiro Hikami. Volume conjecture and asymptotic expansion of qseries. Experiment. Math., 12(3):319–337, 2003.
- [160] Kazuhiro Hikami. q-series and L-functions related to half-derivatives of the Andrews-Gordon identity. Ramanujan J., 11(2):175–197, 2006.
- [161] F. Hirzebruch and D. Zagier. Intersection numbers of curves on Hilbert modular surfaces and modular forms of Nebentypus. *Invent. Math.*, 36:57–113, 1976.
- [162] G. Hoehn. Generalized Moonshine for the Baby Monster. 2003.
- [163] G. Hoehn. Self-dual vertex operator superalgebras of large minimal weight. arXiv:0801.1822v1, January 2008.
- [164] S. Hohenegger. Automorphic products, generalized kac-moody algebras and string amplitudes. J. Phys. Conf. Ser., 346, 2012.
- [165] Christopher Hooley. On the number of divisors of a quadratic polynomial. Acta Math., 110:97–114, 1963.
- [166] M. Hövel. Mock theta functions. PhD thesis, Automorphe Formen mit Singularitäten auf dem hyperbolischen Raum, 2012.

- [167] James E. Humphreys. Introduction to Lie algebras and representation theory. Springer-Verlag, New York-Berlin, 1972. Graduate Texts in Mathematics, Vol. 9.
- [168] Ozlem Imamoğlu, Martin Raum, and Olav K. Richter. Holomorphic projections and Ramanujan's mock theta functions. Proc. Natl. Acad. Sci. USA, 111(11):3961–3967, 2014.
- [169] Rossen Ivanov and Michael Tuite. Rational generalised moonshine from abelian orbifoldings of the moonshine module. Nuclear Phys. B, 635(3):435-472, 2002.
- [170] Rossen Ivanov and Michael Tuite. Some irrational generalised moonshine from orbifolds. Nuclear Phys. B, 635(3):473-491, 2002.
- [171] J. Funke J. H. Bruinier and Ö. Imamoğlu. Regularized theta liftings and periods of modular functions. J. Reine Angew. Math., accepted for publication.
- [172] Ho Yun Jung, Ja Kyung Koo, and Dong Hwa Shin. On some arithmetic properties of Siegel functions (II). Forum Math., 26(1):25–57, 2014.
- [173] E. Jurisich, J. Lepowsky, and R. L. Wilson. Realizations of the Monster Lie algebra. Selecta Math. (N.S.), 1(1):129–161, 1995.
- [174] Elizabeth Jurisich. An exposition of generalized Kac-Moody algebras. In Lie algebras and their representations (Seoul, 1995), volume 194 of Contemp. Math., pages 121–159. Amer. Math. Soc., Providence, RI, 1996.
- [175] Elizabeth Jurisich. Generalized Kac-Moody Lie algebras, free Lie algebras and the structure of the Monster Lie algebra. J. Pure Appl. Algebra, 126(1-3):233-266, 1998.

- [176] Elizabeth Jurisich. Borcherd's proof of the Conway-Norton conjecture. In Moonshine: the first quarter century and beyond, volume 372 of London Math. Soc. Lecture Note Ser., pages 219–235. Cambridge Univ. Press, Cambridge, 2010.
- [177] V. G. Kac. Simple irreducible graded Lie algebras of finite growth. Izv. Akad. Nauk SSSR Ser. Mat., 32:1323–1367, 1968.
- [178] V. G. Kac. Infinite-dimensional Lie algebras, and the Dedekind  $\eta$ -function. Funkcional. Anal. i Priložen., 8(1):77–78, 1974.
- [179] V. G. Kac. Infinite-dimensional algebras, Dedekind's η-function, classical Möbius function and the very strange formula. Adv. in Math., 30(2):85–136, 1978.
- [180] V. G. Kac. An elucidation of: "Infinite-dimensional algebras, Dedekind's η-function, classical Möbius function and the very strange formula". E<sub>8</sub><sup>(1)</sup> and the cube root of the modular invariant j. Adv. in Math., 35(3):264-273, 1980.
- [181] V. G. Kac and A. K. Raina. Bombay lectures on highest weight representations of infinite-dimensional Lie algebras, volume 2 of Advanced Series in Mathematical Physics. World Scientific Publishing Co., Inc., Teaneck, NJ, 1987.
- [182] Victor Kac. Vertex algebras for beginners, volume 10 of University Lecture Series. American Mathematical Society, Providence, RI, second edition, 1998.
- [183] Victor G. Kac. Infinite-dimensional Lie algebras. Cambridge University Press, Cambridge, third edition, 1990.

- [184] Victor G. Kac and Dale H. Peterson. Infinite-dimensional Lie algebras, theta functions and modular forms. Adv. in Math., 53(2):125-264, 1984.
- [185] Svetlana Katok and Peter Sarnak. Heegner points, cycles and Maass forms. Israel J. Math., 84(1-2):193-227, 1993.
- [186] Rinat Kedem, Barry M. McCoy, and Ezer Melzer. The sums of Rogers, Schur and Ramanujan and the Bose-Fermi correspondence in (1 + 1)dimensional quantum field theory. In *Recent progress in statistical mechanics and quantum field theory (Los Angeles, CA, 1994)*, pages 195–219. World Sci. Publ., River Edge, NJ, 1995.
- [187] Anatol N. Kirillov. New combinatorial formula for modified Hall-Littlewood polynomials. In q-series from a contemporary perspective (South Hadley, MA, 1998), volume 254 of Contemp. Math., pages 283– 333. Amer. Math. Soc., Providence, RI, 2000.
- [188] Marvin I. Knopp. Rademacher on J(τ), Poincaré series of nonpositive weights and the Eichler cohomology. Notices Amer. Math. Soc., 37(4):385–393, 1990.
- [189] Marvin Isadore Knopp. Construction of a class of modular functions and forms. *Pacific J. Math.*, 11:275–293, 1961.
- [190] Marvin Isadore Knopp. Construction of a class of modular functions and forms. II. Pacific J. Math., 11:661–678, 1961.
- [191] Marvin Isadore Knopp. Construction of automorphic forms on Hgroups and supplementary Fourier series. Trans. Amer. Math. Soc., 103:168–188, 1962.
- [192] Marvin Isadore Knopp. On abelian integrals of the second kind and modular functions. Amer. J. Math., 84:615–628, 1962.

- [193] Donald Knutson. λ-rings and the representation theory of the symmetric group. Lecture Notes in Mathematics, Vol. 308. Springer-Verlag, Berlin-New York, 1973.
- [194] Winfried Kohnen. Fourier coefficients of modular forms of half-integral weight. Math. Ann., 271(2):237–268, 1985.
- [195] V. A. Kolyvagin. Finiteness of E(Q) and SH(E, Q) for a subclass of Weil curves. Izv. Akad. Nauk SSSR Ser. Mat., 52(3):522-540, 670-671, 1988.
- [196] Bertram Kostant. Lie algebra cohomology and the generalized Borel-Weil theorem. Ann. of Math. (2), 74:329–387, 1961.
- [197] D. Kotschick and P. Lisca. Instanton invariants of CP<sup>2</sup> via topology. Math. Ann., 303(2):345–371, 1995.
- [198] C. Krattenthaler. Advanced determinant calculus. Sém. Lothar. Combin., 42:Art. B42q, 67 pp. (electronic), 1999. The Andrews Festschrift (Maratea, 1998).
- [199] Daniel S. Kubert and Serge Lang. Modular units, volume 244 of Grundlehren der Mathematischen Wissenschaften [Fundamental Principles of Mathematical Science]. Springer-Verlag, New York-Berlin, 1981.
- [200] J. M. F. Labastida and Carlos Lozano. Duality in twisted  $\mathcal{N} = 4$  supersymmetric gauge theories in four dimensions. Nuclear Phys. B, 537(1-3):203-242, 1999.
- [201] John Leech. Some sphere packings in higher space. Canad. J. Math., 16:657–682, 1964.

- [202] John Leech. Notes on sphere packings. Canad. J. Math., 19:251–267, 1967.
- [203] Jeffrey S. Leon and Charles C. Sims. The existence and uniqueness of a simple group generated by {3,4}-transpositions. Bull. Amer. Math. Soc., 83(5):1039-1040, 1977.
- [204] J. Lepowsky. Generalized Verma modules, loop space cohomology and MacDonald-type identities. Ann. Sci. École Norm. Sup. (4), 12(2):169– 234, 1979.
- [205] J. Lepowsky. Affine Lie algebras and combinatorial identities. In Lie algebras and related topics (New Brunswick, N.J., 1981), volume 933 of Lecture Notes in Math., pages 130–156. Springer, Berlin-New York, 1982.
- [206] J. Lepowsky and S. Milne. Lie algebraic approaches to classical partition identities. Adv. in Math., 29(1):15–59, 1978.
- [207] J. Lepowsky and S. Milne. Lie algebras and classical partition identities. Proc. Nat. Acad. Sci. U.S.A., 75(2):578–579, 1978.
- [208] James Lepowsky and Haisheng Li. Introduction to vertex operator algebras and their representations, volume 227 of Progress in Mathematics. Birkhäuser Boston, Inc., Boston, MA, 2004.
- [209] James Lepowsky and Mirko Primc. Structure of the standard modules for the affine Lie algebra  $A_1^{(1)}$ , volume 46 of Contemporary Mathematics. American Mathematical Society, Providence, RI, 1985.
- [210] James Lepowsky and Robert Lee Wilson. Construction of the affine Lie algebra  $A_1(1)$ . Comm. Math. Phys., 62(1):43–53, 1978.

- [211] James Lepowsky and Robert Lee Wilson. A new family of algebras underlying the Rogers-Ramanujan identities and generalizations. Proc. Nat. Acad. Sci. U.S.A., 78(12, part 1):7254–7258, 1981.
- [212] James Lepowsky and Robert Lee Wilson. The Rogers-Ramanujan identities: Lie theoretic interpretation and proof. Proc. Nat. Acad. Sci. U.S.A., 78(2, part 1):699-701, 1981.
- [213] James Lepowsky and Robert Lee Wilson. A Lie theoretic interpretation and proof of the Rogers-Ramanujan identities. Adv. in Math., 45(1):21– 72, 1982.
- [214] James Lepowsky and Robert Lee Wilson. The structure of standard modules. I. Universal algebras and the Rogers-Ramanujan identities. *Invent. Math.*, 77(2):199–290, 1984.
- [215] Wei Li, Wei Song, and Andrew Strominger. Chiral gravity in three dimensions. J. High Energy Phys., (4):082, 15, 2008.
- [216] C. Lozano and M. Marino. Donaldson invariants on product ruled surfaces and two-dimensional gauge theories. *Commun. Math. Phys.*, 220:231–261, 2001.
- [217] S. Hohenegger M. Gaberdiel and R. Volpato. Mathieu moonshine in the elliptic genus of k3. JHEP, 1010:062, 2010.
- [218] I. G. Macdonald. Affine root systems and Dedekind's η-function. Invent. Math., 15:91–143, 1972.
- [219] I. G. Macdonald. Symmetric functions and Hall polynomials. Oxford Mathematical Monographs. The Clarendon Press, Oxford University Press, New York, second edition, 1995. With contributions by A. Zelevinsky, Oxford Science Publications.

- [220] Percy A. MacMahon. Combinatory analysis. Vol. I, II (bound in one volume). Dover Phoenix Editions. Dover Publications, Inc., Mineola, NY, 2004. Reprint of it An introduction to combinatory analysis (1920) and it Combinatory analysis. Vol. I, II (1915, 1916).
- [221] K. Mahler. On a class of non-linear functional equations connected with modular functions. J. Austral. Math. Soc. Ser. A, 22(1):65–118, 1976.
- [222] J. Maldacena. The gauge gravity duality. In Proceedings of Xth Quark Confinement and the Hadron Spectrum (Confinement X).
- [223] Juan Maldacena. The large N limit of superconformal field theories and supergravity. Adv. Theor. Math. Phys., 2(2):231–252, 1998.
- [224] Juan Maldacena and Andrew Strominger. AdS<sub>3</sub> black holes and a stringy exclusion principle. J. High Energy Phys., (12):Paper 5, 24, 1998.
- [225] Andreas Malmendier. Expressions for the generating function of the Donaldson invariants for CP(2). ProQuest LLC, Ann Arbor, MI, 2007. Thesis (Ph.D.)-Massachusetts Institute of Technology.
- [226] Andreas Malmendier. The signature of the Seiberg-Witten surface. In Surveys in differential geometry. Volume XV. Perspectives in mathematics and physics, volume 15 of Surv. Differ. Geom., pages 255-277. Int. Press, Somerville, MA, 2011.
- [227] Andreas Malmendier and Ken Ono. SO(3)-Donaldson invariants of  $\mathbb{CP}^2$  and mock theta functions. *Geom. Topol.*, 16(3):1767–1833, 2012.
- [228] Alexander Maloney, Wei Song, and Andrew Strominger. Chiral gravity, log gravity, and extremal CFT. Phys. Rev. D, 81(6):064007, 19, 2010.

- [229] Alexander Maloney and Edward Witten. Quantum gravity partition functions in three dimensions. J. High Energy Phys., (2):029, 58, 2010.
- [230] Jan Manschot. AdS<sub>3</sub> partition functions reconstructed. J. High Energy Phys., (10):103, 7, 2007.
- [231] Jan Manschot and Gregory W. Moore. A modern fareytail. Commun. Number Theory Phys., 4(1):103–159, 2010.
- [232] Marcos Mariño and Gregory Moore. Integrating over the Coulomb branch in N = 2 gauge theory. Nuclear Phys. B Proc. Suppl., 68:336– 347, 1998. Strings '97 (Amsterdam, 1997).
- [233] Geoffrey Mason. Finite groups and modular functions. In The Arcata Conference on Representations of Finite Groups (Arcata, Calif., 1986), volume 47 of Proc. Sympos. Pure Math., pages 181–210. Amer. Math. Soc., Providence, RI, 1987. With an appendix by S. P. Norton.
- [234] Henry McKean and Victor Moll. *Elliptic curves*. Cambridge University Press, Cambridge, 1997. Function theory, geometry, arithmetic.
- [235] M. Mertens. Eichler-selberg type identities for mixed mock modular forms. arXiv:1404.5491.
- [236] Arne Meurman and Mirko Primc. Vertex operator algebras and representations of affine Lie algebras. Acta Appl. Math., 44(1-2):207-215, 1996. Representations of Lie groups, Lie algebras and their quantum analogues.
- [237] Alison Miller and Aaron Pixton. Arithmetic traces of non-holomorphic modular invariants. Int. J. Number Theory, 6(1):69-87, 2010.
- [238] Stephen C. Milne. The  $C_l$  Rogers-Selberg identity. SIAM J. Math. Anal., 25(2):571–595, 1994.

- [239] R. V. Moody. Lie algebras associated with generalized Cartan matrices. Bull. Amer. Math. Soc., 73:217–221, 1967.
- [240] Robert V. Moody. A new class of Lie algebras. J. Algebra, 10:211–230, 1968.
- [241] Gregory Moore and Edward Witten. Integration over the u-plane in Donaldson theory. Adv. Theor. Math. Phys., 1(2):298–387, 1997.
- [242] Douglas Niebur. A class of nonanalytic automorphic functions. Nagoya Math. J., 52:133-145, 1973.
- [243] Douglas Niebur. Construction of automorphic forms and integrals. Trans. Amer. Math. Soc., 191:373–385, 1974.
- [244] Hans-Volker Niemeier. Definite quadratische Formen der Dimension 24 und Diskriminante 1. J. Number Theory, 5:142–178, 1973.
- [245] S. P. Norton. More on moonshine. In Computational group theory (Durham, 1982), pages 185–193. Academic Press, London, 1984.
- [246] S. P. Norton. Generalized Moonshine. Proc. Symp. Pure Math, 47:208– 209, 1987.
- [247] Simon Norton. From Moonshine to the Monster. In Proceedings on Moonshine and related topics (Montréal, QC, 1999), volume 30 of CRM Proc. Lecture Notes, pages 163–171. Amer. Math. Soc., Providence, RI, 2001.
- [248] Andrew P. Ogg. Automorphismes de courbes modulaires. In Séminaire Delange-PisotPoitou (16e année: 1974/75), Théorie des nombres, Fasc. 1, Exp. No. 7, page 8. Secrétariat Mathématique, Paris, 1975.

- [249] Ken Ono. The web of modularity: arithmetic of the coefficients of modular forms and q-series, volume 102 of CBMS Regional Conference Series in Mathematics. Published for the Conference Board of the Mathematical Sciences, Washington, DC; by the American Mathematical Society, Providence, RI, 2004.
- [250] Ken Ono. Unearthing the visions of a master: harmonic Maass forms and number theory. In *Current developments in mathematics*, 2008, pages 347–454. Int. Press, Somerville, MA, 2009.
- [251] Hans Petersson. Über die Entwicklungskoeffizienten der automorphen Formen. Acta Math., 58(1):169–215, 1932.
- [252] Larissa Queen. Modular functions arising from some finite groups. Math. Comp., 37(156):547-580, 1981.
- [253] G. Moore R. Dijkgraaf, J. Maldacena and E. Verlinde. A black hole farey tail. 2007.
- [254] Hans Rademacher. On the Partition Function p(n). Proc. London Math. Soc., S2-43(4):241.
- [255] Hans Rademacher. The Fourier Coefficients of the Modular Invariant  $J(\tau)$ . Amer. J. Math., 60(2):501-512, 1938.
- [256] Hans Rademacher. The Fourier Series and the Functional Equation of the Absolute Modular Invariant J(τ). Amer. J. Math., 61(1):237-248, 1939.
- [257] K. G. Ramanathan. On Ramanujan's continued fraction. Acta Arith., 43(3):209–226, 1984.

- [258] Srinivasa Ramanujan. The lost notebook and other unpublished papers.
  Springer-Verlag, Berlin; Narosa Publishing House, New Delhi, 1988.
  With an introduction by George E. Andrews.
- [259] Robert A. Rankin. Modular forms and functions. Cambridge University Press, Cambridge-New York-Melbourne, 1977.
- [260] Urmie Ray. Generalized Kac-Moody algebras and some related topics. Bull. Amer. Math. Soc. (N.S.), 38(1):1–42 (electronic), 2001.
- [261] William F. Reynolds. Thompson's characterization of characters and sets of primes. J. Algebra, 156(1):237-243, 1993.
- [262] Bruce Richmond and George Szekeres. Some formulas related to dilogarithms, the zeta function and the Andrews-Gordon identities. J. Austral. Math. Soc. Ser. A, 31(3):362–373, 1981.
- [263] L. J. Rogers. Second Memoir on the Expansion of certain Infinite Products. Proc. London Math. Soc., S1-25(1):318.
- [264] L. J. Rogers. Proof of certain identities in combinatory analysis [Proc. Cambridge Philos. Soc. 19 (1919), 214-216]. In *Collected papers of Srinivasa Ramanujan*, pages 214-215. AMS Chelsea Publ., Providence, RI, 2000.
- [265] R. Jackiw S. Deser and S. Templeton. Three-Dimensional Massive Gauge Theories. *Phys. Rev. Lett.*, 48:975–978, 1982.
- [266] I. Schur. Ein Beitrag zur Hilbertschen Theorie der vollstetigen quadratischen Formen. Math. Z., 12(1):287–297, 1922.
- [267] Graeme Segal. Unitary representations of some infinite-dimensional groups. Comm. Math. Phys., 80(3):301-342, 1981.

- [268] N. Seiberg and E. Witten. Electric-magnetic duality, monopole condensation, and confinement in N = 2 supersymmetric Yang-Mills theory. Nuclear Phys. B, 426(1):19–52, 1994.
- [269] A. Selberg. über einige arithmetische identitäten. Avh. Norske Vid.-Akad. Oslo. I. Mat.-Naturv. Kl., 8:1-23, 1936.
- [270] J.-P. Serre. A course in arithmetic. Springer-Verlag, New York-Heidelberg, 1973. Translated from the French, Graduate Texts in Mathematics, No. 7.
- [271] J.-P. Serre and H. M. Stark. Modular forms of weight 1/2. In Modular functions of one variable, VI (Proc. Second Internat. Conf., Univ. Bonn, Bonn, 1976), pages 27–67. Lecture Notes in Math., Vol. 627. Springer, Berlin, 1977.
- [272] Goro Shimura. Introduction to the arithmetic theory of automorphic functions. Publications of the Mathematical Society of Japan, No. 11. Iwanami Shoten, Publishers, Tokyo; Princeton University Press, Princeton, N.J., 1971. Kanô Memorial Lectures, No. 1.
- [273] Goro Shimura. On modular forms of half integral weight. Ann. of Math. (2), 97:440-481, 1973.
- [274] Goro Shimura. Introduction to the arithmetic theory of automorphic functions, volume 11 of Publications of the Mathematical Society of Japan. Princeton University Press, Princeton, NJ, 1994. Reprint of the 1971 original, Kanô Memorial Lectures, 1.
- [275] Stephen D. Smith. On the head characters of the Monster simple group. In *Finite groups—coming of age (Montreal, Que., 1982)*, volume 45 of *Contemp. Math.*, pages 303–313. Amer. Math. Soc., Providence, RI, 1985.

- [276] John Stachel. The early history of quantum gravity (1916–1940). In Black holes, gravitational radiation and the universe, volume 100 of Fund. Theories Phys., pages 525–534. Kluwer Acad. Publ., Dordrecht, 1999.
- [277] W. A. Stein et al. Sage Mathematics Software (Version 6.2.beta1). The Sage Development Team, 2014. http://www.sagemath.org.
- [278] John R. Stembridge. Hall-Littlewood functions, plane partitions, and the Rogers-Ramanujan identities. Trans. Amer. Math. Soc., 319(2):469–498, 1990.
- [279] Andrew Strominger. A simple proof of the chiral gravity conjecture. In Geometry and analysis. No. 2, volume 18 of Adv. Lect. Math. (ALM), pages 329–333. Int. Press, Somerville, MA, 2011.
- [280] Stein Arild Strømme. Ample divisors on fine moduli spaces on the projective plane. Math. Z., 187(3):405–423, 1984.
- [281] Jacob Sturm. On the congruence of modular forms. In Number theory (New York, 1984–1985), volume 1240 of Lecture Notes in Math., pages 275–280. Springer, Berlin, 1987.
- [282] Leonard Susskind. The world as a hologram. J. Math. Phys., 36(11):6377–6396, 1995.
- [283] J. G. Thompson. Finite groups and modular functions. Bull. London Math. Soc., 11(3):347–351, 1979.
- [284] J. G. Thompson. Some numerology between the Fischer-Griess Monster and the elliptic modular function. Bull. London Math. Soc., 11(3):352– 353, 1979.

- [285] J. Tits. On R. Griess' "friendly giant". Invent. Math., 78(3):491–499, 1984.
- [286] Michael P. Tuite. Monstrous Moonshine from orbifolds. Comm. Math. Phys., 146(2):277–309, 1992.
- [287] Michael P. Tuite. Generalized moonshine from abelian orbifold constructions. Contemp. Maths., 193:353–368, 1995.
- [288] Michael P. Tuite. On the relationship between Monstrous Moonshine and the uniqueness of the Moonshine module. *Comm. Math. Phys.*, 166(3):495–532, 1995.
- [289] J. B. Tunnell. A classical Diophantine problem and modular forms of weight 3/2. Invent. Math., 72(2):323-334, 1983.
- [290] R. C. Vaughan. The Hardy-Littlewood method, volume 125 of Cambridge Tracts in Mathematics. Cambridge University Press, Cambridge, second edition, 1997.
- [291] B. B. Venkov. On the classification of integral even unimodular 24dimensional quadratic forms. *Trudy Mat. Inst. Steklov.*, 148:65–76, 273, 1978. Algebra, number theory and their applications.
- [292] S. Ole Warnaar. Hall-Littlewood functions and the  $A_2$  Rogers-Ramanujan identities. Adv. Math., 200(2):403–434, 2006.
- [293] S. Ole Warnaar and Wadim Zudilin. Dedekind's  $\eta$ -function and Rogers-Ramanujan identities. Bull. Lond. Math. Soc., 44(1):1–11, 2012.
- [294] G. N. Watson. The Final Problem : An Account of the Mock Theta Functions. J. London Math. Soc., S1-11(1):55.

- [295] G. N. Watson. A New Proof of the Rogers-Ramanujan Identities. J. London Math. Soc., S1-4(1):4.
- [296] G. N. Watson. Theorems Stated by Ramanujan (VII): Theorems on Continued Fractions. J. London Math. Soc., S1-4(1):39.
- [297] G. N. Watson. Theorems Stated by Ramanujan (XI). J. London Math. Soc., S1-6(1):59.
- [298] André Weil. Elliptic functions according to Eisenstein and Kronecker. Springer-Verlag, Berlin-New York, 1976. Ergebnisse der Mathematik und ihrer Grenzgebiete, Band 88.
- [299] D. Whalen. Vector-valued rademacher sums and automorphic integrals. arXiv:1406.0571.
- [300] E. Witten. On S-duality in abelian gauge theory. Selecta Math. (N.S.), 1(2):383-410, 1995.
- [301] Edward Witten. Topological quantum field theory. Comm. Math. Phys., 117(3):353-386, 1988.
- [302] Edward Witten. Monopoles and four-manifolds. Math. Res. Lett., 1(6):769-796, 1994.
- [303] Edward Witten. Anti de Sitter space and holography. Adv. Theor. Math. Phys., 2(2):253-291, 1998.
- [304] Edward Witten. Three-Dimensional Gravity Revisited. 2007.
- [305] Xi Yin. Partition functions of three-dimensional pure gravity. Commun. Number Theory Phys., 2(2):285-324, 2008.
- [306] Don Zagier. Nombres de classes et formes modulaires de poids 3/2. C.
  R. Acad. Sci. Paris Sér. A-B, 281(21):Ai, A883-A886, 1975.

- [307] Don Zagier. Modular parametrizations of elliptic curves. Canad. Math. Bull., 28(3):372–384, 1985.
- [308] Don Zagier. Traces of singular moduli. In Motives, polylogarithms and Hodge theory, Part I (Irvine, CA, 1998), volume 3 of Int. Press Lect. Ser., pages 211–244. Int. Press, Somerville, MA, 2002.
- [309] Don Zagier. Ramanujan's mock theta functions and their applications (after Zwegers and Ono-Bringmann). Astérisque, (326):Exp. No. 986, vii–viii, 143–164 (2010), 2009. Séminaire Bourbaki. Vol. 2007/2008.
- [310] Yongchang Zhu. Modular invariance of characters of vertex operator algebras. J. Amer. Math. Soc., 9(1):237–302, 1996.
- [311] S. Zwegers. Mock theta functions. PhD thesis, Universiteit Utrecht, 2002. (Advisor: D. Zagier).
- [312] S. P. Zwegers. Mock θ-functions and real analytic modular forms. In q-series with applications to combinatorics, number theory, and physics (Urbana, IL, 2000), volume 291 of Contemp. Math., pages 269–277. Amer. Math. Soc., Providence, RI, 2001.